# UNIVERSITÉ PARIS XI

U.E.R. MATHÉMATIQUE

91405 ORSAY FRANCE

N° 64

ON THE THEORY AND NUMERICAL ANALYSIS OF THE NAVIER-STOKES EQUATIONS

by

Roger TEMAM

June 1973

# ON THE THEORY AND NUMERICAL ANALYSIS OF THE NAVIER-STOKES EQUATIONS

Ъy

Roger Temam

Lecture Note #9

June 1, 1973

University of Maryland College Park, Maryland

Department of Mathematics

#### INTRODUCTION

These lecture notes develop some results related to the theory and numerical analysis of the Navier-Stokes equations of viscous incompressible fluids.

We successively consider the linearized stationary case (Chapter I), the nonlinear stationary case (Chapter II), and the full nonlinear evolution case (Chapter III).

In Chapter I, after a short presentation of existence and uniqueness results, we describe the approximation of the Stokes problem by several finite difference and finite element methods. This is an opportunity to introduce various methods of approximation of the divergence free vector functions which are also essential tools for the numerical aspects of the problems studied in Chapters II and III.

In Chapter II, we introduce compactness tools in both the continuous and discrete cases. Otherwise this chapter is just an extension to the nonlinear case of the results obtained for the linear case in the preceding chapter.

Chapter III deals with the full nonlinear evolution case. We first describe some typical results of the present state of the mathematical theory of the Navier-Stokes equations (existence and uniqueness results). We then give a brief introduction to the numerical aspects, combining the discretization in the space variables discussed in Chapter I with the usual discretization techniques in the time variable. The stability and convergence of the schemes are studied by energy methods. The numerical analysis of the full Navier-Stokes equations will be more fully developed in an extended version of this course to appear elsewhere.

The material of these Notes was taught at the University of Maryland during the first semester of 1972-73 as part of a special year on the Navier-Stokes and nonlinear partial differential equations.<sup>(\*)</sup>

The author is very grateful to his colleagues of the Department of

(\*) Preparation of these Notes was supported by the National Science Foundation under grant GU 2061.

i

Mathematics and of the Institute of Fluid Dynamics and Applied Mathematics who encouraged him to write these Notes, either directly or indirectly by attending his course and by their great interest.

Direct contributions to the preparation of the manuscript are due to Annett Williamson and Professors J. Osborn, J. Sather, and P. Wolfe. I wish to thank them for correcting the manuscript of many english mistakes and for their interesting comments and suggestions which helped improve the manuscript.

Finally, I thank Mrs. Debbie Curran for her excellent typing.

# CONTENTS

Page

Chapter	I.	The S	Stationary	Stokes	Equations.
---------	----	-------	------------	--------	------------

§1.	Some	e function spaces.	
	1.1	Notation.	1
	1.2	A density theorem.	4
	1.3	A trace theorem.	8
	1.4	Characterization of the space H.	12
	1.5	Characterization of the space V.	18
§2.	Exis	stence and uniqueness for the Stokes equations.	23
	2.1	Variational formulation of the problem.	23
	2.2	The projection theorem.	25
	2.3	The unbounded case.	31
	2.4	Regularity results.	33
§3.	Discretization of the Stokes equations (I).		38
	3.1	Approximation of a normed space.	38
	3.2	A general convergence theorem.	42
	3.3	Approximation by finite differences.	46
§4.	Disc	cretization of the Stokes equations (II).	63
	4.1	Preliminary results.	63
	4.2	Finite elements of degree 2 (n = 2).	70
	4.3	Finite elements of degree 3 (n = 3).	83
	4.4	An internal approximation of V.	88
	4.5	Nonconforming finite elements.	95
§5.	Numerical algorithms.		
	5.1	Uzawa algorithm.	116
	5.2	Arrow-Hurwicz algorithm.	120
	5.3	Discrete form of these algorithms.	122
§6.	Slightly compressible fluids.		126
	6.1	Convergence of u to u.	126
	6.2	Asymptotic development of $u_{\epsilon}$ .	129
	6.3	Numerical algorithms.	132

		Page
hapter I	II. Stationary Navier-Stokes Equations.	
	Introduction.	135
§1.	1. Existence and uniqueness theorems.	
	<ul> <li>1.1 Sobolev inequalities and compactness theorems.</li> <li>1.2 The homogeneous Navier-Stokes equations.</li> <li>1.3 The homogeneous Navier-Stokes equations (continued).</li> </ul>	136 138
	The unbounded case. 1.4 The inhomogeneous Navier-Stokes equations.	147 151
§2.	Discrete inequalities and compactness theorems.	160
	<ul><li>2.1 Discrete Sobolev inequalities.</li><li>2.2 A discrete compactness theorem.</li></ul>	160 166
§3.	§3. Approximation of the stationary Navier-Stokes equations.	
	<ul><li>3.1 A general convergence theorem.</li><li>3.2 Applications.</li><li>3.3 Numerical algorithms.</li></ul>	172 178 187
hapter ]	III. The Evolution Navier-Stokes Equations.	
	Introduction.	
§1.	The linear case.	194
	<ul> <li>1.1 Notations.</li> <li>1.2 The existence and uniqueness theorem</li> <li>1.3 Proof of the existence in Theorem 1.1</li> <li>1.4 Proof of the continuity and uniqueness.</li> <li>1.5 Miscellaneous remarks</li> </ul>	194 198 200 205 208
§2.	Compactness theorems.	214
	<ul> <li>2.1 A preliminary result.</li> <li>2.2 A compactness theorem in Banach spaces.</li> <li>2.3 A compactness theorem involving fractional derivatives.</li> </ul>	214 215 217
§3.	Existence and uniqueness theorems ( $n \leq 4$ ).	223
	3.1 An existence theorem in $\mathbb{R}^n$ $(n \le 4)$ . 3.2 Proof of Theorem 3.1. 3.3 Regularity and uniqueness $(n = 2)$ . 3.4 About regularity and uniqueness $(n = 3)$ . 3.5 More regular solutions.	223 227 233 237 240

		Page
§4.	Alternate proof of existence by semi-discretization.	249
	<ul> <li>4.1 Statement of the problem.</li> <li>4.2 The approximate solutions.</li> <li>4.3 A priori estimates.</li> <li>4.4 Passage to the limit.</li> </ul>	249 252 254 256
§5.	Discretization of the Navier-Stokes equations: General stability and convergence theorems.	259
	<ul> <li>5.1 Description of the approximation schemes.</li> <li>5.2 Stability of the Schemes 5.1 and 5.2.</li> <li>5.3 Stability of Scheme 5.3.</li> <li>5.4 Stability of Scheme 5.4.</li> <li>5.5 A complementary estimate for Scheme 5.2.</li> <li>5.6 Other a priori estimates.</li> <li>5.7 Convergence of the numerical schemes.</li> </ul>	260 264 267 270 274 275 277
	5./ Convergence of the numerical schemes.	277

#### CHAPTER I

#### THE STATIONARY STOKES EQUATIONS

#### INTRODUCTION

In this chapter we study the stationary Stokes equations; that is, the stationary linearized form of the Navier-Stokes equations. The study of the Stokes equations is useful by itself; it also gives us an opportunity to introduce several tools necessary for a treatment of the full Navier-Stokes equations.

In Section 1 we consider some function spaces (spaces of divergence free vector functions with  $L^2$ -components). In Section 2 we give the variational formulation of the Stokes equations and prove existence and uniqueness of solutions by the projection theorem. In Sections 3 and 4 we recall a few definitions and results on the approximation of a normed space and of a variational linear equation (Section 3). We propose then several types of approximation of a certain fundamental space V of divergence free vector functions; this includes an approximation by the finite differences method (Section 3), and by conforming and non-conforming finite element methods (Section 4). In Section 5 we discuss some algorithms of approximation for the Stokes equations and the corresponding discretized equations. The purpose of these algorithms is to overcome the difficulty caused by the condition div u = 0. As it will be shown, this difficulty, sometimes, is not merely solved by discretization.

Finally in Section 6 we study the linearized equations of slightly compressible fluids and their asymptotic convergence to the linear equations of incompressible fluids (i.e., Stokes equations).

#### §1. SOME FUNCTION SPACES

In this section we introduce and study some fundamental function spaces. The results are important for what follows, but the methods used in this section will not reappear so that the reader can skim through the proofs and retain only the general notations described in Sec. 1.1 and the results summarized in Remark 1.6.

1.1 Notation

In euclidian space  $\mathbb{R}^n$  we write  $e_1 = (1,0,\dots,0)$ ,  $e_2 = (0,1,0,\dots,0)$ ,  $\cdots$ ,  $e_n = (0,\dots,0,1)$ , the canonical basis, and  $\mathbf{x} = (\mathbf{x}_1,\dots,\mathbf{x}_n)$ ,  $\mathbf{y} = (\mathbf{y}_1,\dots,\mathbf{y}_n)$ ,  $\mathbf{z} = (\mathbf{z}_1,\dots,\mathbf{z}_n)$ ,  $\cdots$ , will denote points of the space.

 $z = (z_1, \dots, z_n), \dots,$  will denote points of the space. The differentiation operator  $\frac{\partial}{\partial x_i}$   $(1 \le i \le n)$ , will be written  $D_i$  and if  $j = (j_1, \dots, j_n)$  is a multi-index,  $D^j$  will be the differentiation operator

(1.1) 
$$D^{j} = D_{1}^{j_{1}} \cdots D_{n}^{j_{n}} = \frac{\partial^{j_{1}}}{\partial x_{1}^{j_{1}} \cdots \partial x_{n}^{j_{n}}}$$

where

$$(1.2) \qquad \qquad \left[j\right] = j_1 + \cdots + j_n$$

When  $j_i = 0$  for some i,  $D_i^{j_i}$  is the identity operator; in particular if [j] = 0,  $D^j$  is the identity. The Set  $\Omega$ 

Let  $\Omega$  be an open set of  $\mathbb{R}^n$  with boundary  $\Gamma$ . In general we will need some smoothness property for  $\Omega$ . Sometimes we will assume that  $\Omega$  is smooth in the following sense:

(1.3) The boundary  $\Gamma$  is a (n-1)-dimensional manifold of class  $\mathcal{C}^{\mathbf{r}}$  (r  $\geq 1$ which must be specified) and  $\Omega$  is locally located on one side of  $\Gamma$ . We will say that a set  $\Omega$  satisfying (1.3) is of class  $\mathcal{C}^{\mathbf{r}}$ . However this hypothesis is too strong for practical situations (such as a flow in a square) and all the main results will be proved under a weaker condition:

(1.4) The boundary of  $\Omega$  is locally Lipschitzian.

This means that in a neighborhood of any point  $x \in \Gamma$ ,  $\Gamma$  admits a representation as a hypersurface  $y_n = O(y_1, \dots, y_{n-1})$  where O is a lipschitzian function, and  $(y_1, \dots, y_n)$  are rectangular coordinates in  $\mathbb{R}^n$  in a basis that may be different from the canonical basis  $e_1, \dots, e_n$ .

Of course if  $\Omega$  is of class  $\mathcal{C}^1$ , then  $\Omega$  is locally lipschitzian.

It is useful for the sequel of this section, to notice that a set r satisfying (1.4) is "locally star-shaped" This means that each point  $x_j \in \Gamma$ , has an open neighborhood  $\mathcal{O}_j$ , such that  $\mathcal{O}'_j = \Omega \cap \mathcal{O}'_j$  is star-shaped with respect to one of its points. According to (1.4)



respect to one of its points. According to (1.4) we can moreover suppose that the boundary of  $\theta_1^r$  is lipschitzian.

If  $\Gamma$  is bounded, it can be covered by a finite family of such sets  $\vartheta_j$ ,  $j \in J$ ; if  $\Gamma$  is not bounded, the family  $(\vartheta_j)_{j \in J}$  can be chosen to be locally finite.

It is understood that  $\Omega$  will always satisfy (1.4), unless we mention explicitly that  $\Omega$  is any open set in  $\mathbb{R}^n$  or that some other smoothness property is required.  $L^p$  and Sobolev Spaces

Let  $\Omega$  be any open set in  $\mathbb{R}^n$ . We denote by  $L^p(\Omega)$ ,  $1 \leq p \leq +\infty$ (resp.  $L^{\infty}(\Omega)$ ) the space of real functions defined on  $\Omega$  with the  $p^{\text{th}}$  power absolutely integrable

2

(resp. essentially bounded real functions) for the Lebesgue measure  $dx = dx_1 \cdots dx_n$ . This is a Banach space with the norm

(1.5) 
$$\|u\|_{L^{P}(\Omega)} = \{\int_{\Omega} |u(x)|^{p} dx\}^{\frac{1}{p}}$$

(resp.

$$\|u\|_{L^{\infty}(\Omega)} = \operatorname{ess. sup} |u(x)|$$
.

For p = 2,  $L^{2}(\Omega)$  is a Hilbert space with the scalar product

(1.6) 
$$(u,v) = \int_{\Omega} u(x)v(x)dx$$

The Sobolev space  $W^{m,p}(\Omega)$  is the space of functions in  $L^{p}(\Omega)$  with derivatives of order less than or equal to m in  $L^{p}(\Omega)$  (m = integer,  $1 \le p \le +\infty$ ). This is a Banach space with the norm

(1.7) 
$$\|u\|_{W^{m,p}(\Omega)} = \{ \sum_{\substack{j \leq m \\ L^{p}(\Omega)}} \|D^{j}u\|^{p} \}^{\overline{p}}.$$

When p = 2,  $W^{m,2}(\Omega) = H^{m}(\Omega)$  is a Hilbert space with the scalar product

(1.8) 
$$((u,v)) = \sum_{\substack{H^m(\Omega) \\ H^m(\Omega)}} (D^j u, D^j v).$$

Let  $\mathcal{P}(\Omega)$  (resp.  $\mathcal{P}(\overline{\Omega})$ ) be the space of  $\mathcal{C}^{\infty}$  functions with compact support contained in  $\Omega$  (resp.  $\overline{\Omega}$ ). The closure of  $\mathcal{P}(\Omega)$  in  $W^{m,p}(\Omega)$  is denoted by  $W_0^{m,p}(\Omega)$  ( $H_0^m(\Omega)$ ) when p = 2).

We will recall when needed the classical properties of these spaces such as the density or trace theorems (assuming regularity properties for  $\Omega$ ).

We will often be concerned with n-dimensional vector functions with components in one of these spaces. We will use the notation

$$L^{p}(\Omega) = \{L^{p}(\Omega)\}^{n}, \quad W^{m,p}(\Omega) = \{W^{m,p}(\Omega)\}^{n}$$
$$H^{m}(\Omega) = \{H^{m}(\Omega)\}^{n}, \quad \mathfrak{D}(\Omega) = \{\mathfrak{D}(\Omega)\}^{n},$$

and we suppose that these product spaces are equipped with the usual product norm or an equivalent norm (except  $\mathfrak{D}(\Omega)$  and  $\mathfrak{P}(\Omega)$  which are not normed spaces.)

The following spaces will appear very frequently

$$L^{2}(\Omega)$$
,  $L^{2}(\Omega)$ ,  $H^{1}_{0}(\Omega)$ ,  $H^{1}_{0}(\Omega)$ .

The scalar product and the norm are denoted by  $(\cdot, \cdot)$  and  $|\cdot|$  on  $L^2(\Omega)$  or  $L^2(\Omega)$ 

(resp. [., ] and [.] on  $H_0^1(\Omega)$  or  $H_0^1(\Omega)$ ).

We recall that if  $\Omega$  is <u>bounded in some direction</u><sup>(1)</sup> then the Poincaré inequality holds:

(1.9) 
$$\| u \|_{L^{2}(\Omega)} \leq c(\Omega) \| D u \|_{L^{2}(\Omega)}, \quad \forall u \in H^{1}_{0}(\Omega),$$

where D is the derivative in that direction and  $c(\Omega)$  is a constant depending only on  $\Omega$ . In this case the norm [•] on  $H_0^1(\Omega)$  (resp.  $H_0^1(\Omega)$ ) is equivalent to the norm:

(1.10) 
$$||u|| = (\sum_{i=1}^{n} |D_{i}u|^{2})^{\frac{1}{2}}$$

The space  $H_0^1(\Omega)$  (resp.  $H_0^1(\Omega)$ ) is also a Hilbert space with the associated scalar product

(1.11) 
$$((u,v)) = \sum_{i=1}^{n} (D_{i}u, D_{i}v).$$

This scalar product and this norm are denoted by  $((\cdot, \cdot))$  and  $\|\cdot\|$  on  $H_0^1(\Omega)$ and  $H_0^1(\Omega)$  ( $\Omega$  bounded in some direction).

Let  $\mathcal V$  be the space (without topology)

(1.12) 
$$\mathcal{N} = \{ u \in \mathfrak{D}(\Omega), \text{ div } u = 0 \}.$$

The closure of  $\mathcal{V}$  in  $\mathbb{L}^2(\Omega)$  and in  $\mathbb{H}^1_0(\Omega)$  are two basic spaces in the study of the Navier-Stokes equations; we denote them by H and V. The results of this section will allow us to give a characterization of V and H.

1.2 A Density Theorem

Let  $E(\Omega)$  be the following auxiliary space:

$$E(\Omega) = \{ u \in L^2(\Omega), div u \in L^2(\Omega) \}.$$

This is a Hilbert space when equipped with the scalar product

(1.13) 
$$((u,v))_{E(\Omega)} = (u,v) + (div u, div v).$$

It is clear that (1.13) is a scalar product on  $E(\Omega)$ ; it is easy to see that  $E(\Omega)$  is

<sup>(1)</sup> i.e.,  $\Omega$  is included in a slab whose boundary is two hyperplanes which are orthogonal to this direction.

complete for the associated norm

$$\|u\|_{E(\Omega)} = \{((u,u))_{E(\Omega)}\}^{\frac{1}{2}}(1).$$

Our goal is to prove a trace theorem: for  $u \in E(\Omega)$  one can define the value on  $\Gamma$  of the normal component  $u \cdot v$ , v = the unit vector normal to the boundary. The method for that is a classical method of Lions-Magenes [1]. We first start by proving the

# Theorem 1.1.

Let  $\Omega$  be a locally lipschitzian open set in  $\mathbb{R}^n$ . Then the set of vector functions in  $\mathscr{D}(\overline{\Omega})$  is dense in  $E(\Omega)$ .

#### Proof.

Let u be some element of  $E(\Omega)$ . We have to prove that u is limit in  $E(\Omega)$  of vector functions of  $\mathscr{D}(\overline{\Omega})$ .

i) We first approximate u by functions of  $E(\Omega)$  with compact support in  $\Omega$ . Let  $\phi \in \mathscr{D}(\mathbb{R}^n)$ ,  $0 \le \phi \le 1$ ,  $\phi = 1$  for  $|x| \le 1$ , and  $\phi = 0$  for  $|x| \ge 2$ . For a > 0 let  $\phi_a$  be the restriction to  $\Omega$  of the function  $x \mapsto \phi(\frac{x}{a})$ . It is easy to check that  $\phi_a u \in E(\Omega)$  and that  $\phi_a u$  converges to u in this space as  $a \longrightarrow \infty$ .

The functions with compact support are a dense subspace of  $E(\Omega)$  and we can suppose that u has compact support.

ii) Let us consider first the case  $\Omega = \mathbb{R}^n$ ; hence  $u \in E(\mathbb{R}^n)$  and u has compact support.

The result is then proved by regularization. Let  $\rho \in \mathscr{D}(\mathbb{R}^n)$  be a smooth  $\mathfrak{C}^{\infty}$  function with compact support, such that  $\rho \geq 0$ ,  $\int_{\mathbb{R}^n} \rho(\mathbf{x}) d\mathbf{x} = 1$ . For  $\varepsilon \in (0,1]$ , let

 $\rho_{\varepsilon}$  denote the function  $x \mapsto \frac{1}{\varepsilon^n} \rho(\frac{x}{\varepsilon})$ . When  $\varepsilon \to 0$ ,  $\rho_{\varepsilon}$  converges in the distribution sense to the Dirac distribution and it is classical that

. . .

(1.14) 
$$\rho_{\varepsilon} * v \rightarrow v \quad \text{in } L^{2}(\mathbb{R}^{n}), \quad \forall v \in L^{2}(\mathbb{R}^{n}). \quad (2)$$

(1) For if u is a Cauchy sequence in  $E(\Omega)$ , then u is also a Cauchy sequence in  $L^2(\Omega)$ and also div u is a Cauchy sequence in  $L^2(\Omega)$ ; u converges to some limit u in  $L^2(\Omega)$ , and div u converges to some limit g in  $L^2(\Omega)$ , necessarily g =div u, and so  $u \in E(\Omega)$  and u converges to u in  $E(\Omega)$ . (2) \* is the convolution operator  $\int$ 

$$f(\mathbf{f} * \mathbf{g})(\mathbf{x}) = \int_{\mathbb{R}^n} f(\mathbf{x} - \mathbf{y}) g(\mathbf{y}) d\mathbf{y}$$

If  $f \in L^1(\mathbb{R}^n)$ ,  $g \in L^p(\mathbb{R}^n)$ ,  $1 \le p \le \infty$ , then  $f \ast g$  makes sense and belongs to  $L^p(\mathbb{R}^n)$ .

Now  $\rho_{\varepsilon} \star u$  belongs to  $\mathscr{D}(\mathbb{R}^n)$  since this function has a compact support  $(C(\text{support } \rho_{\varepsilon}) + \text{ support } u)$  and components which are  $\mathcal{C}^{\infty}$ . According to (1.14)  $\rho_{\varepsilon} \star u$  converges to u in  $\mathbb{L}^2(\mathbb{R}^n)$  as  $\varepsilon \longrightarrow 0$ , and

$$\operatorname{div}(\rho_{E} * u) = \rho_{E} * \operatorname{div} u$$
 converges to  $\operatorname{div} u$  in  $\mathbb{L}^{2}(\mathbb{R}^{n})$ ,

as  $\varepsilon \rightarrow 0$ . Hence u is the limit in  $E(\mathbb{R}^n)$  of functions of  $\mathcal{D}(\mathbb{R}^n)$ .

iii) For the general case,  $\Omega \neq \mathbb{R}^n$ , we use the remark after (1.4):  $\Omega$  is locally star-shaped. The sets  $\Omega, (\mathcal{O}_j)_{j \in J}$ , form an open covering of  $\overline{\Omega}$ . Let us consider a partition of unity subordinated to this covering

(1.15) 
$$1 = \phi + \sum_{j \in J} \phi_j, \ \phi \in \mathcal{D}(\Omega), \ \phi_j \in \mathcal{D}(\mathcal{O}_j).$$

We can write

$$u = \phi u + \sum_{j \in J} \phi_{j} u,$$

the sum  $\sum_{i=1}^{n}$  is actually finite since the support of u is compact.

Since the function  $\phi u$  has compact support in  $\Omega$  it can be shown as in ii) that  $\phi u$  is limit in  $E(\Omega)$  of functions belonging to  $\mathcal{D}(\Omega)$  (the function  $\phi u$  extended by 0 outside  $\Omega$  belongs to  $E(\mathbb{R}^n)$  and for  $\varepsilon$  sufficiently small,  $\rho_{\varepsilon}^*(\phi u)$  has compact support in  $\Omega$ ).

Let us consider now one of the functions  $u_j = \phi_j u$  not identically equal to zero. The set  $\mathcal{O}'_j = \mathcal{O}' \cap \Omega$  is star-shaped with respect to one of its points; after a translation in  $\mathbb{R}^n$  we can suppose this point is 0. Let  $\sigma_{\lambda}$ ,  $\lambda \neq 0$ , be the linear (homothetic) transformation  $x \mapsto \lambda x$ . It is clear that

$$\begin{cases} \partial'_{j} \subset \overline{\partial'}_{j} \subset \sigma_{\lambda} \partial'_{j} & \text{for } \lambda > 1 \\ \\ \sigma_{\lambda} \partial'_{j} \subset \overline{\sigma_{\lambda} \partial'}_{j} \subset \partial'_{j} & \text{for } 0 < \lambda < 1. \end{cases}$$

Let  $\sigma_{\lambda} \circ v$  denote the function  $x \mapsto v(\sigma_{\lambda}(x))$ ; because of Proposition 1.1 below, the restriction to  $\vartheta'_{j}$  of the function  $\sigma_{\lambda} \circ u_{j}$ ,  $\lambda > 1$ , converges to  $u_{j}$  in  $E(\vartheta'_{j})$ (or  $E(\Omega)$ ) as  $\lambda \mapsto 1$ . But if  $\psi_{j} \in \vartheta(\sigma_{\lambda}(\vartheta'_{j}))$  and  $\psi_{j} = 1$  on  $\vartheta'_{j}$ , the function  $\psi_{j}(\sigma_{\lambda} \circ u)$  clearly belongs to  $E(\mathbb{R}^{n})$ . Hence we must only approximate in place of the function  $u_{j}$ , a function  $v_{j} \in E(\Omega)$  which is the restriction to  $\Omega$  of a function  $w_{j} \in E(\mathbb{R}^{n})$ with compared support (take  $w_{j} = \psi_{j}(\sigma_{\lambda} \circ u)$ ). The result follows then from point ii). It remains only to prove the Proposition 1.1 giving some results needed before, and some results we will need later.

Proposition 1.1.

Let  $\mathcal{O}$  be an open set which is star-shaped with respect to 0.

(i) If  $p \in \mathcal{D}'(\mathcal{O})$  is a distribution in  $\mathcal{O}$ , then a distribution  $\sigma_{\lambda} \circ p$  can be defined in  $\mathcal{D}'(\sigma_{\lambda}\mathcal{O})$  by

(1.16) 
$$\langle \sigma_{\lambda} \circ p, \phi \rangle = \frac{1}{\lambda^{n}} \langle p, \sigma_{\underline{\lambda}} \circ \phi \rangle, \quad \forall \phi \in \mathcal{D}(\sigma_{\lambda} \mathcal{O}) (\lambda > 0).$$

The derivatives of  $\sigma_{\lambda} \circ p$  are related to the derivatives of p by the formula

(1.17) 
$$D_{i}(\sigma_{\lambda} \circ p) = \lambda \sigma_{\lambda} \circ (D_{i}p), \quad 1 \leq i \leq n.$$

If  $\lambda > 1$ ,  $\lambda \longrightarrow 1$ , the restriction to  $\mathcal{O}$  of  $\sigma_{\lambda} \circ p$  converges in the distribution sense to p.

(ii) If  $p \in L^{\alpha}(\mathcal{O})$ ,  $1 \leq \alpha < +\infty$ , then  $\sigma_{\lambda} \circ p \in L^{\alpha}(\sigma_{\lambda}\mathcal{O})$ . For  $\lambda > 1$ ,  $\lambda \longrightarrow 1$ , the restriction to  $\mathcal{O}$  of  $\sigma_{\lambda} \circ p$  converges to p in  $L^{\alpha}(\mathcal{O})$ .

Proof.

It is clear that the mapping  $\phi \mapsto \frac{1}{\lambda^n} < p, \sigma \circ \phi >$  is linear and continuous on  $\mathscr{D}(\sigma_{\lambda} \vartheta)$  and hence defines a distribution which is denoted by  $\sigma_{\lambda} \circ p$ .

The formula (1.17) is easy,

$$\langle D_{i}(\sigma_{\lambda} \circ p), \phi \rangle = - \langle (c_{\lambda} \circ p), D_{i} \phi \rangle$$

$$= -\frac{1}{\lambda^{n}} \langle p, \sigma_{1} \circ (D_{i} \phi) \rangle$$

$$= -\frac{1}{\lambda^{n-1}} \langle p, D_{i}(\sigma_{1} \circ \phi) \rangle$$

$$= +\frac{1}{\lambda^{n-1}} \langle D_{i}p, \sigma_{1} \circ \phi \rangle$$

$$= \lambda \langle \sigma_{\lambda} \circ D_{i}p, \phi \rangle$$

When  $\lambda > 1$ ,  $\lambda \rightarrow 1$ , the functions  $\sigma_1 \circ \phi$  have compact support in  $\mathcal{O}$  for  $\lambda - 1$  small enough, and converge to  $\phi$  in  $\mathcal{D}(\mathcal{O}) \xrightarrow{\lambda} as \lambda \rightarrow 1$ .

(ii) It is clear that

$$\int_{\sigma_{\lambda} \mathcal{O}} |(\sigma_{\lambda} p)(y)|^{\alpha} dy = \lambda^{n} \int_{\mathcal{O}} |p(x)|^{\alpha} dx$$

and

$$\|\sigma_{\lambda} \circ p\|_{L^{\alpha}(\sigma_{\lambda} \theta)} = \|p\|_{L^{\alpha}(\theta)}$$

It is then sufficient to prove that  $\sigma_{\lambda} \circ p$  restricted to  $\mathscr{O}$  converge to p, for the p's belonging to some dense subspace of  $L^{\alpha}(\Omega)$ . But  $\mathscr{P}(\mathscr{O})$  is dense in  $L^{\alpha}(\mathscr{O})$ , and the result is obvious if  $p \in \mathscr{P}(\mathscr{O})$ .

#### 1.3. A Trace Theorem

We suppose here that  $\Omega$  is an open bounded set of class  $\mathcal{C}^2$ . It is known that there exists a linear continuous operator  $\gamma_0 \in \mathfrak{L}(\mathrm{H}^1(\Omega), \mathrm{L}^2(\Gamma))$  (the trace operator), such that  $\gamma_0 u$  = the restriction of u to  $\Gamma$  for every function  $u \in \mathrm{H}^1(\Omega)$  which is twice continuously differentiable in  $\overline{\Omega}$ . The space  $\mathrm{H}_0^1(\Omega)$  is equal to the kernel of  $\gamma_0$ . The image space  $\gamma_0(\mathrm{H}^1(\Omega))$  is a dense subspace of  $\mathrm{L}^2(\Gamma)$  denoted  $\mathrm{H}^{\frac{1}{2}}(\Gamma)$ ; the space  $\mathrm{H}^{\frac{1}{2}}(\Gamma)$  can be equipped with the norm carried from  $\mathrm{H}^1(\Gamma)$  by  $\gamma_0$ . There exists moreover a linear continuous operator  $\ell_{\Omega} \in \mathfrak{L}(\mathrm{H}^{\frac{1}{2}}(\Gamma), \mathrm{H}^1(\Omega))$  (which is called a lifting operator), such that  $\gamma_0 \circ \ell_{\Omega}$  = the identity operator in  $\mathrm{H}^{\frac{1}{2}}(\Gamma)$ . All these results are given in Lions [1], Lions-Magenes [1].

We want to prove a similar result for the vector functions in  $E(\Omega)$ .

Let  $H^{-\frac{1}{2}}(\Gamma)$  be the dual space of  $H^{\frac{1}{2}}(\Gamma)$ ; since  $H^{\frac{1}{2}}(\Gamma) \subset L^{2}(\Gamma)$  with a stronger topology,  $L^{2}(\Gamma)$  is contained in  $H^{-\frac{1}{2}}(\Gamma)$  with a stronger topology. We have the following trace theorem (which means that one can define  $u \cdot v|_{\Gamma}$  when  $u \in E$ ):

# Theorem 1.2.

Let  $\Omega$  be an open bounded set class  $\mathcal{C}^2$ . Then there exists a linear continuous operator  $\gamma_{\mathcal{V}} \in \mathcal{S}(E(\Omega), \mathbb{H}^{-2}(\Gamma))$ , such that

(1.18)  $\gamma_{\nu} u = \underline{the restriction of } u \cdot v \underline{to } \Gamma$ , for every  $u \in \mathfrak{D}(\overline{\Omega})$ .

The following generalized Stokes formula is true, for each  $u \in E(\Omega)$  and  $w \in H^{1}(\Omega)$ 

(1.19) (u, grad w) + (div u, w) = 
$$\langle \gamma_{u} u, \gamma_{0} w \rangle$$
.

Proof.

Let  $\phi \in H^{\frac{1}{2}}(\Gamma)$  and let  $w \in H^{1}(\Omega)$  with  $\gamma_{0}w = \phi$ . For  $u \in E(\Omega)$ , let us set

$$X_{u}(\phi) = \int_{\Omega} [\operatorname{div} u(x)w(x) + u(x) \operatorname{grad} w(x)] dx$$
$$= (\operatorname{div} u,w) + (u, \operatorname{grad} w).$$

Lemma 1.1.

$$X_{u}(\phi)$$
 is independent of the choice of w, as long as  $w \in H^{1}(\Omega)$  and  $\gamma_{0}w = \phi$ .

Proof.

Let  $w_1$  and  $w_2$  belong to  $H^1(\Omega)$ , with

$$\gamma_0 w_1 = \gamma_0 w_2 = \phi,$$

and let  $w = w_1 - w_2$ .

One has to prove that

$$(div u, w_1) + (u, grad w_1) = (div u, w_2) + (u, grad w_2),$$

that is to say

(1.20) 
$$(\operatorname{div} u, w) + (u, \operatorname{grad} w) = 0.$$

But since  $w \in H^1(\Omega)$  and  $\gamma_0 w = 0$ , w belongs to  $H^1_0(\Omega)$  and is the limit in  $H^1(\Omega)$  of smooth functions with compact support:  $w = \lim w_m$ ,  $w_m \in \mathcal{D}(\Omega)$ . It is obvious that

$$(\text{div u, } w_{m}) + (u, \text{ grad } w_{m}) = 0, \forall w_{m} \in \mathscr{D}(\Omega)$$

and (1.20) follows as  $m \rightarrow \infty$ .

Let us take now,  $w = \ell_0 \phi$  (see above). Then by the Schwarz inequality

$$|\mathbf{X}_{\mathbf{u}}(\phi)| \leq \|\mathbf{u}\|_{\mathbf{E}(\Omega)} \|\mathbf{w}\|_{\mathbf{H}^{1}(\Omega)},$$

and since  $l_{\Omega} \in \mathcal{J}(\overline{H^{2}}(\Gamma), H^{1}(\Omega))$ 

(1.21) 
$$|X_{u}(\phi)| \leq c_{0} ||u||_{E(\Omega)} ||\phi||_{H^{\frac{1}{2}}(\Gamma)}$$

where  $c_0 =$  the norm of the linear operator  $\ell_{\Omega}$ .

Therefore the mapping  $\phi \mapsto X_u(\phi)$  is a linear continuous mapping from  $H^2(\Gamma)$ into R. Thus there exists  $g = g(u) \in H^{-\frac{1}{2}}(\Gamma)$  such that

(1.22) 
$$X_{11}(\phi) = \langle g, \phi \rangle.$$

It is clear that the mapping  $u \mapsto g(u)$  is linear and, by (1.21),

(1.23) 
$$\|g\|_{H^{-\frac{1}{2}}(\Gamma)} \leq c_{0} \|u\|_{E(\Omega)};$$

this proves that the mapping  $u \mapsto g(u) = \gamma_{v}u$  is continuous from  $E(\Omega)$  into  $H^{-\frac{1}{2}}(\Gamma)$ . The last point is to prove (1.18), since (1.19) follows directly from the definition of  $\gamma_{v}u$ .

9

 $\frac{\text{Lemma}}{\text{If } u \in \mathscr{D}(\overline{\Omega}), \text{ then}}$ 

$$\gamma_{v} \cdot u = \underline{the restriction of } u \cdot v \underline{on } \Gamma.$$

Proof.

For such a smooth u and for any  $w \in \mathcal{D}(\overline{\Omega})$  (or if u and w are twice continuously differentiable in  $\overline{\Omega}$ ),

$$\begin{split} \mathbf{X}_{\mathbf{u}}(\boldsymbol{\gamma}_{0}\mathbf{w}) &= \int_{\Omega} \operatorname{div}(\mathbf{u} \ \mathbf{w}) \, \mathrm{dx} \\ &= (\text{by the Stokes formula}) \\ &= \int_{W} (\mathbf{u} \cdot \boldsymbol{\nu}) \, \mathrm{d}\Gamma = \int_{\Gamma} (\mathbf{u} \cdot \boldsymbol{\nu}) \, (\boldsymbol{\gamma}_{0}\mathbf{w}) \, \mathrm{d}\Gamma. \\ &= < \mathbf{u} \cdot \boldsymbol{\nu}, \ \boldsymbol{\gamma}_{0}\mathbf{w} >. \end{split}$$

Since for these functions w, the traces  $\gamma_0 w$  form a dense subset of  $H^{\frac{1}{2}}(\Gamma)$ , the formula

$$X_{11}(\phi) = \langle u.v, \phi \rangle$$

is also true by continuity for every  $\phi \in H^{\frac{1}{2}}(\Gamma)$ . By comparison with (1.22), we get,  $\gamma_{v} u = u \cdot v|_{\Gamma}$ .

#### Remark 1.1.

Theorem 1.1 is not explicitly used in the proof of Theorem 1.2, but the density theorem combined with Lemma 1.2 shows that the operator  $\gamma_{V}$  is unique since its value on a dense subset is known.

Remark 1.2.

The operator  $\gamma_{\mathcal{V}}$  actually maps  $E(\Omega)$  onto  $H^{-\frac{1}{2}}(\Gamma)$ . Let  $\phi$  be given in  $H^{-\frac{1}{2}}(\Gamma)$ , such that

$$<\phi,1>=0.$$

Then the Neumann problem

(1.24) 
$$\begin{cases} -\Delta p = 0 \quad \text{in } \Omega \\ \frac{\partial p}{\partial \nu} = \phi \quad \text{on } \Gamma \end{cases}$$

has a generalized solution  $p = p(\phi) \in H^1(\Omega)$  which is unique up to an additive

constant (See Lions-Magenes [1]). For one of these solutions p let

u = grad p.

It is clear that  $u \in E(\Omega)$  and  $\gamma_{\nu} u = \phi$ . In addition it is clear that there exists a vector function  $u_0$  with components in  $\mathcal{C}^1(\overline{\Omega})$  such that  $\gamma_{\nu} \cdot u_0 = 1$ . Then for any  $\psi$  in  $H^{\frac{1}{2}}(\Gamma)$ , writing

(1.25) 
$$\psi = \phi + \frac{\langle \psi, 1 \rangle}{\operatorname{mes} \Gamma}, \quad \phi = \psi - \frac{\langle \psi, 1 \rangle}{\operatorname{mes} \Gamma},$$

one can define a  $u = u(\psi)$  such that  $\gamma_{u} = \psi$  by setting

(1.26) 
$$\mathbf{u} = \operatorname{grad} \mathbf{p}(\psi) + \frac{\langle \psi, 1 \rangle}{\operatorname{mes} \Gamma} \mathbf{u}_0$$

Moreover the mapping  $\psi \mapsto u(\psi)$  is a linear continuous mapping from  $H^{\frac{1}{2}}(\Gamma)$  into  $E(\Omega)$  (i.e., a <u>lifting</u> operator as  $\ell_{\Omega}$ ).

Let  $E_0(\Omega)$  be the closure of  $\mathfrak{D}(\Omega)$  in  $E(\Omega)$ . We have <u>Theorem 1.3</u>. <u>The kernel of</u>  $\gamma_{\mathcal{V}}$  is equal to  $E_0(\Omega)$ . Proof.

If  $u \in E_0(\Omega)$ , then by the definition of this space, there exists a sequence of functions  $u_m \in \mathfrak{D}(\Omega)$  which converges to u in  $E(\Omega)$  as  $m \to \infty$ . Theorem 1.2 implies that  $\gamma_{\mathcal{V}} u_m = 0$  and hence,  $\gamma_{\mathcal{V}} \cdot u = \lim_{m \to \infty} \gamma_{\mathcal{V}} \cdot u_m = 0$ .

Conversely let us prove that if  $u \in E(\Omega)$  and  $\gamma_{v} \cdot u = 0$ , then u is the limit in  $E(\Omega)$  of vector functions in  $\mathcal{D}(\Omega)^{n}$ .

Let  $\Phi$  be any function in  $\mathscr{D}(\mathbb{R}^n)$ , and  $\phi$  the restriction of  $\Phi$  to  $\Omega$ . Since  $\gamma_{ij} \cdot u = 0$ , we have  $\langle \gamma_{ij} \cdot u, \gamma_0 \phi \rangle = 0$  which means

$$\int_{\Omega} [\operatorname{div} u \cdot \phi + u \cdot \operatorname{grad} \phi] dx = 0.$$

Hence

$$\int_{\mathbb{R}^n} [\widetilde{\operatorname{div} u} \cdot \Phi + \widetilde{u} \cdot \operatorname{grad} \Phi] dx = 0, \forall \Phi \in \mathscr{D}(\mathbb{R}^n)$$

and so

$$(1.27) div \tilde{u} = div u,$$

where  $\tilde{v}$  denotes the function equal to v in  $\Omega$  and to 0 in  $(\Omega$ . This implies that  $\tilde{u} \in E(\mathbb{R}^n)$ .

Following exactly the steps in proving Theorem 1.1 (in particular points i) and ii)) it is possible to reduce the general case to the case where the function u has support included in one of the sets  $\mathcal{O}_i \cap \overline{\Omega}$ .

For such a function u, we remark that  $\tilde{u} \in E(\mathbb{R}^n)$  and that  $\sigma_{\lambda} \circ \tilde{u}$  has a compact support in  $\mathcal{O}'_j$ , for  $0 < \lambda < 1$  ( $\mathcal{O}'_j$  is supposed star-shaped with respect to 0). According to Proposition 1.1,  $\sigma_{\lambda} \circ \tilde{u}$  restricted to  $\mathcal{O}'_j$  (or  $\Omega$ ) converges to u in  $E(\mathcal{O}'_j)$  (or  $E(\Omega)$ ) as  $\lambda \longrightarrow 1$ . The problem is then reduced to that of approximating by elements of  $\mathcal{D}(\mathbb{R}^n)$  a function u with compact support in  $\Omega$ ; this is obvious by regularization (as in point ii) in the proof of Theorem 1.1).

#### <u>Remark</u> 1.3.

If the set  $\Omega$  is unbounded or if its boundary is not smooth, some partial results remain true: for example, if  $u \in E(\Omega)$ , we can define  $\lambda_{\mathcal{V}} u$  on each bounded part  $\Gamma_0$  of  $\Gamma$  of class  $\mathcal{C}^2$ , and  $\gamma_{\mathcal{V}} u \in H^1(\Gamma_0)$ . If  $\Omega$  is smooth but unbounded or if its boundary is the union of a finite number of bounded (n-1) dimensional manifolds of class  $\mathcal{C}^2$ , then  $\gamma_{\mathcal{V}} u$  is defined, in this way, on all  $\Gamma$ . Nevertheless the generalized Stokes formula (1.19) is not available.

The results will be more precise if we know more about the trace of functions in  $H^1(\Omega)$ . Let us suppose that the following results are available:

- there exists  $\gamma_0 \in \mathcal{S}(H^1(\Omega), L^2(\Gamma))$  such that  $\gamma_0 u=u|_{\Gamma}$  for every  $u \in \mathfrak{D}(\Omega)$ . We denote  $\gamma_0(H^1(\Gamma))$  by  $\mathfrak{R}^2(\Gamma)$ .

- there exists a lifting operator  $l_{\Omega} \in \mathcal{L}(\mathcal{H}^{\frac{1}{2}}(\Gamma), \mathcal{H}^{1}(\Omega))$ , such that  $\gamma_{0} \circ l_{\Omega} =$  the identity;  $\mathcal{H}^{\frac{1}{2}}(\Gamma)$  is equipped with the norm carried by  $\gamma_{0}$ .

-  $\Omega$  is a locally lipschitzian set.

Then all the preceding results can be extended to this case. Theorems 1.1 and 1.3 are true. The proof of Theorem 1.2 leads to a definition of  $\gamma_{v}$  as an element of  $\mathcal{H}^{\frac{1}{2}}(\Gamma)$  = the dual space of  $\mathcal{H}^{\frac{1}{2}}(\Gamma)$ . The generalized Stokes formula (1.19) is valid.

# 1.4 Characterization of the Space H.

We recall the notation introduced at the end of Section 1.1:

$$\mathcal{V} = \{ u \in \mathfrak{D}(\Omega), \text{ div } u = 0 \},$$
  
H = the closure of  $\mathcal{V}$  in  $\mathbb{L}^2(\Omega),$   
V = the closure of  $\mathcal{V}$  in  $\mathbb{H}^1_0(\Omega).$ 

We can give now the following characterization of H and H<sup> $\perp$ </sup> (the orthogonal complement of L<sup>2</sup>( $\Omega$ )).

Theorem 1.4.

#### Proof of (1.28).

i) Let u belong to the space in the right-hand member of (1.28). Then for each  $v \in \mathcal{V}$ ,

$$(1.30) (u,v) = (grad p, v) = -(p, div v) = 0.$$

Hence  $u \in H^{\perp}$ .

ii) We will prove conversely that  $H^{\perp}$  is contained in the space in the righthand member of (1.28), but this inclusion is less easy to show.

Let u be an element of  $\mathbb{H}^{\perp}$ . We write first that u is orthogonal to some particular elements of  $\mathbb{V}$ : for every i,j,  $1 \leq i, j \leq n$ , for each  $\psi \in \mathcal{P}(\Omega)$ , let  $v \in \mathbb{V}$  be defined by

(1.31) 
$$v_i = D_j \psi, v_j = -D_i \psi, v_\alpha = 0$$
 if  $\alpha \neq i, \alpha \neq j$ .

Writing that (u,v) = 0, we obtain

$$(\mathbf{u},\mathbf{v}) = (\mathbf{u}_{\mathbf{i}},\mathbf{D}_{\mathbf{j}}\psi) - (\mathbf{u}_{\mathbf{j}},\mathbf{D}_{\mathbf{i}}\psi) = 0, \forall \psi \in \mathfrak{D}(\Omega),$$

which means that

$$(1.32) D_{i}u_{i} = D_{i}u_{i}$$

in the distribution sense. If  $u_1, \cdots u_n$ , were smooth functions, it would follow from a classical result that (1.32) implies that u is the gradient of some function g on each simply connected open subset of  $\Omega$ . The same result is true for distributions by the duality theorem of de Rham [1]; u is the gradient of some distribution p on every simply connected open subset of  $\Omega$ .

Of course if  $\Omega$  is simply connected, u = grad p in all  $\Omega$ , and it follows from a result of Deny-Lions [1] (see Proposition 1.2 in Section 1.5) that  $p \in L^2(\Omega)$ ; (1.28) is proved in this case.

iii) When  $\Omega$  is not simply connected, let us make some smooth cuts in  $\Omega$ ,  $\Omega = \Sigma \cup \dot{\Omega}$ , where  $\dot{\Omega}$  is open, simply connected, and locally lipschitzian;  $\Sigma$  is a  $\dot{C}^2$ , (n-1)dimensional manifold and  $\dot{\Omega} \cap \Sigma = \phi$ .<sup>(1)</sup>

Applying point ii) u = grad p in  $\hat{\Omega}$ , where p belongs to  $L^2(\hat{\Omega})$ . Hence



(1) We assume that we can make such smooth cuts, but this is a very weak hypothesis on  $\Omega$ ; in practice it is always satisfied.  $p \in H^1(\hat{\Omega})$  and we can define the traces of p on both sides of  $\Sigma$ , but these traces may be <u>different</u>. If we prove that p has the same trace on both sides of  $\Sigma$ , then p will be a single valued function and u = grad p in all  $\Omega$ .

Let  $\gamma_{+}p$  and  $\gamma_{-}p$  be the traces of p on the two sides  $\Sigma_{+}$  and  $\Sigma_{-}$  of  $\Sigma_{-}$ . In order to show that  $\gamma_{+}p = \gamma_{-}p$ , it is sufficient to show that

(1.33) 
$$\int_{\Sigma} \phi(\gamma_{+}p - \gamma_{-}p) d\Gamma = 0,$$

for every smooth function  $\phi$  defined on  $\Sigma$ , with support contained in a "small" ball centered on  $\Sigma$ .

By Lemma 1.3 which will follow, for each function  $\phi$  there exists w $\in$ H, div w = 0, with compact support in  $\Omega$  such that

(1.34) 
$$\gamma_{v_{+}} w = \phi$$
 (1)

We write now that

$$0 = (u,w) = \int_{\Omega} u \cdot w \, dx$$
  
=  $\int_{\Omega} \operatorname{grad} p \cdot w \, dx = (by \text{ Stokes formula (1.19)})$   
=  $\langle \gamma_{\mathcal{V}} w, \gamma_{0} p \rangle_{\partial \Omega} - \int_{\Omega} p \cdot \operatorname{div} w \, dx$  <sup>(2)</sup>  
=  $\langle \gamma_{\mathcal{V}} w, \gamma_{+} p - \gamma_{-} p \rangle_{\Sigma} = \int_{\Sigma} (\gamma_{+} p - \gamma_{-} p) \phi d\Gamma$ ,

and (1.33) follows.

Lemma 1.3.

Let  $\phi$  be a smooth function on  $\Sigma$ , with compact support in  $\Sigma_0 \subset \overline{\Sigma}_0 \subset \Sigma$ . There exists  $w \in H$ , with compact support in  $\Omega$  such that div w = 0, and  $\gamma_{v_{\perp}} w = \phi$  on  $\Sigma$ .

<sup>(1)</sup> If  $w \in H$ , div w = 0, then  $w \in E(\Omega)$  and we can define  $\gamma_{V_{+}} w =$  the trace of  $w \cdot v_{+}$ on  $\Sigma$ ; here  $v_{+}$  stands for the unit normal to  $\Sigma$  pointing in the direction of  $\Sigma_{-}$ . <sup>(2)</sup> Since w has a compact support in  $\Omega$ , an extended form of (1.19) is valid without needing any regularity property for  $\partial\Omega$  (but  $\Sigma$  is of class  $\mathcal{C}^{2}$ ). Proof.

Let  $\mathcal{O}$  be an open tube in  $\Omega$  with a smooth lateral boundary and cross sections  $\Sigma_0^+$  and  $\Sigma_0^-$ ; we suppose that the lateral boundary of  $\mathcal{O}$  matches  $\Sigma_0^+$  and  $\Sigma_0^-$  smoothly so that the whole boundary of  $\mathcal{O}$  is of class  $\mathcal{C}^2$ .



Let g be the solution of the following Neumann problem in  $\, \vartheta \,$ 

(1.35) 
$$\begin{cases} \Delta g = 0 \quad \text{in } \theta' \\ \frac{\partial g}{\partial v} = 0 \quad \text{on } \partial \theta - \Sigma_0 \\ \frac{\partial g}{\partial \gamma_+} = \phi \quad \text{on } \Sigma_0^+, \quad \frac{\partial g}{\partial v_-} = -\phi \quad \text{on } \Sigma_0^-. \end{cases}$$

Since the necessary condition  $\int_{\partial \mathcal{O}} \frac{\partial g}{\partial v} d\Gamma = 0$  is satisfied, the problem (1.35) has an infinity of solutions, any two of them differing by a constant. Let then w be the vector defined almost everywhere in  $\Omega$  (or  $\Omega$ ) by

(1.36) 
$$w = \begin{cases} \operatorname{grad} g & \operatorname{in} \mathcal{O} \\ 0 & \operatorname{in} \Omega - \mathcal{O}. \end{cases}$$

We will prove that w satisfies all the conditions.

It is clear that  $w \in \mathbb{L}^2(\Omega)$  and  $w \in \mathbb{E}(\dot{\Omega})$ ; let us show that  $w \in \mathbb{E}(\Omega)$  and moreover

$$(1.37) \qquad \text{div } w = 0 \quad \text{in } \Omega.$$

Let  $\psi$  be any test function in  $\mathscr{P}(\Omega)$ ; by definition,

$$\langle \text{div } w, \psi \rangle = - (w, \text{grad } \psi)$$

$$= - \int_{\mathcal{O}} \mathbf{w} \cdot \operatorname{grad} \psi \, d\mathbf{x} = (by \ (1.19))$$
$$= -\langle \gamma_{v} \mathbf{w}, \psi \rangle_{\partial \mathcal{O}} + \int_{\mathcal{O}} \operatorname{div} \mathbf{w} \cdot \psi \, d\mathbf{x}$$

The last integral equals 0 since div  $w = \Delta g = 0$  in  $\Omega$ ;  $\gamma_{V}w = \frac{\partial g}{\partial V}$  vanishes on the lateral boundary of  $\mathcal{O}$  and with (1.35), the sum of the integrals on  $\Sigma_{0}^{+}$  and  $\Sigma_{0}^{-}$ vanishes. There remains

$$< \operatorname{div} w, \psi > = 0, \forall \psi \in \mathcal{D}(\Omega),$$

and (1.37) follows.

In order to prove that w is the limit in  $\mathbb{L}^{2}(\Omega)$  of functions in  $\mathbb{V}$ , let us consider a regularization of w,  $w_{\varepsilon} = \rho_{\varepsilon} * w$ , where as usual  $\rho_{\varepsilon}(x) = \frac{1}{\varepsilon^{n}} \rho(\frac{x}{\varepsilon})$ ,  $\rho \in \mathcal{D}(\mathbb{R}^{n})$ ,  $0 \leq \rho \leq 1$ ,  $\rho(x) = 0$  for  $|x| \geq 1$ ,  $\int_{\mathbb{R}} \rho(x) dx = 1$ . For  $\varepsilon$  less than the distance between  $\mathcal{O}$  and  $\Gamma$ ,  $w_{\varepsilon}$  has compact support in  $\Omega$  and thus  $w_{\varepsilon} \in \mathcal{D}(\Omega)$ ; we have also

div 
$$w_{\rho} = \operatorname{div}(\rho_{\rho} * w) = \rho_{\rho} * \operatorname{div} w = 0$$

so that  $w \in \mathcal{V}$ .

It is clear that, for  $\varepsilon \longrightarrow 0$ ,  $w_{\varepsilon}$  converges to w in  $\mathbb{L}^{2}(\Omega)$ , thus  $w \in \mathbb{H}$ .

## Proof of (1.29).

Let H<sup>•</sup> be the space in the right-hand member of (1.29) and let us prove that  $H \subset H^{\bullet}$ . If  $u \in H$ , then  $u = \lim_{m \to \infty} u_m$ ,  $u_m \in \mathcal{V}$ ; this convergence in  $\mathbb{L}^2(\Omega)$  implies convergence in the distribution sense; then since differentiation is a continuous operator in the distribution space and div  $u_m = 0$ , we see that div u = 0. Since div  $u_m = \operatorname{div} u = 0$ ,  $u_m$  and u belong to  $E(\Omega)$  and

$$\left\| \mathbf{u} - \mathbf{u}_{\mathbf{m}} \right\|_{\mathbf{E}(\Omega)} = \left\| \mathbf{u} - \mathbf{u}_{\mathbf{m}} \right\|_{\mathbb{L}^{2}(\Omega)}$$

so that  $u_m$  converges to u in  $E(\Omega)$ , and  $\gamma_{\mathcal{V}} u = \lim_{m \to \infty} \gamma_{\mathcal{V}} u_m = 0$  ( $\gamma_{\mathcal{V}} u_m = u_m \cdot \mathcal{V} |_{\Gamma} = 0$ ,  $\forall u_m \in \mathcal{V}$ ). Hence  $u \in H^{\bullet}$ .

Let us suppose that H is not the whole space H<sup>•</sup> and let H<sup>••</sup> be the orthogonal complement of H in H<sup>•</sup>. By (1.28), every  $u \in H^{••}$  is the gradient of some  $p \in H^{1}(\Omega)$ ; other properties of p are:

(1.37) 
$$\Delta p - \operatorname{div} u = 0, \quad u.v|_{\Gamma} = \frac{\partial p}{\partial v}|_{\Gamma} = 0$$

and this implies that p is a constant and u = 0; therefore  $H^{**} = \{0\}$  and  $H = H^{*}$ 

#### Remark 1.4.

If  $\Omega$  is any open set in  $\mathbb{R}^n$ , the proof of (1.28) with very slight modifications shows that

(1.38) 
$$H^{\perp} = \{ u \in L^{2}(\Omega), u = \text{grad } p, p \in L^{2}_{loc}(\Omega) \}.$$

If  $\Omega$  is unbounded but satisfies the condition (1.4) then

(1.39) 
$$H^{\perp} = \{ u \in \mathbb{L}^{2}(\Omega), u = \text{grad } p, p \in L^{2}_{\text{loc}}(\overline{\Omega}) \}.$$

Theorem 1.5.

Let  $\Omega$  be an open bounded set of class  $C^2$ . Then

(1.40) 
$$L^2(\Omega) = H \oplus H, \oplus H_2,$$

where H, H1, H2 are mutually orthogonal spaces

(1.41) H<sub>1</sub> = {
$$u \in \mathbb{L}^2(\Omega)$$
,  $u = \text{grad } p, p \in H^1(\Omega)$ ,  $\Delta p = 0$ },

(1.42) 
$$H_2 = \{ u \in \mathbb{L}^2(\Omega), u = \text{grad } p, p \in H_0^1(\Omega) \}.$$

Proof.

It is clear that  $H_1$  and  $H_2$  are included in  $H^{\perp}$ , and that the intersection of any two of the spaces H,  $H_1$ ,  $H_2$ , reduces to  $\{0\}$ .

The spaces  $H_1$  and  $H_2$  are orthogonal; if  $u = \text{grad } p \in H_1$ ,  $v = \text{grad } q \in H_2$ , then  $u \in E(\Omega)$  and by using the generalized Stokes formula (1.19),

$$(u,v) = (u, \text{ grad } q) = \langle \gamma_{,u}, \gamma_{,u} \rangle - (\text{div } u, q)$$

and this vanishes since  $\gamma_0 q = 0$  and div  $u = \Delta p = 0$ .

It remains to prove that any element u of  $L^2(\Omega)$  can be written as the sum of elements  $u_0$ ,  $u_1$ ,  $u_2$  in H, H<sub>1</sub>, H<sub>2</sub>. For such a u let p be the unique solution of the Dirichlet problem

$$\Delta p = \operatorname{div} u \in \operatorname{H}^{-1}(\Omega), \quad p \in \operatorname{H}^{1}_{0}(\Omega).$$

We take

(1.43)  $u_2 = grad p.$ 

Let then q be the solution of the Neumann problem

(1.44) 
$$\Delta q = 0, \quad \frac{\partial q}{\partial v} \Big|_{\Gamma} = \gamma_{v}(u - \text{grad } p).$$

We notice that div(u - grad p) = 0, so that u - grad  $p \in E(\Omega)$ , and  $\gamma_{v}(u - \text{grad } p)$  is defined as an element of  $H^{-\frac{1}{2}}(\Gamma)$ , and by the Stokes fromula (1.19)

$$\langle \gamma_{v}(u - \text{grad } p), 1 \rangle = \int_{\Omega} \operatorname{div}(u - \text{grad } p) dx = 0.$$

According to a result of Lions-Magenes [1], the Neuman problem (1.44) possesses a

solution which is unique up to some additive constant.

We take

$$(1.45) u_1 = grad q$$

(1.46) 
$$u_0 = u - u_1 - u_2$$
.

It remains to see that  $u_0 \in H$ . But div  $u_0 = \operatorname{div}(u - u_1 - u_2) = \operatorname{div} u - \Delta p = 0$ , and  $\gamma_{\mathcal{V}}u_0 = \gamma_{\mathcal{V}}(u - u_1 - u_2) = \gamma_{\mathcal{V}}(u - \operatorname{grad} p) - \frac{\partial q}{\partial \mathcal{V}} = 0$ .

1.5 Characterization of the Space V.

For the sake of completeness we recall first the following result:

Lemma 1.4.

If  $l \in H^{-1}(\Omega)$  (i.e., the dual space of  $H_0^1(\Omega)$ ), then there exists (n+1) functions  $f_0, \dots, f_n$ , belonging to  $L^2(\Omega)$  such that

(1.47) 
$$\ell = f_0 - \sum_{i=1}^{n} D_i f_i$$

that is to say

(1.48) 
$$< \ell, u > = (f_0, u) + \sum_{i=1}^{n} (f_i, D_i u), \forall u \in H_0^1(\Omega).$$

The functions fo, ... f satisfying (1.47) are not unique.

Proof.

The linear operator  $u \rightarrow \overline{\omega} u = (u, D_1 u, \cdots, D_n u)$  is an isomorphism of  $H_0^1(\Omega)$  into  $L^2(\Omega)^{n+1}$  The linear form  $\ell_0 \overline{\omega}^{-1}$  is continuous on the image space of  $H_0^1(\Omega)$ ; by the Hahn-Banach theorem this form can be extended as a linear continuous form on all the space  $L^2(\Omega)^{n+1}$ ; of course the extended form can be written as

$$g \mapsto (f,g) = \sum_{i=0}^{n+1} (f_i,g_i), f_i \in L^2(\Omega),$$

and (1.48) follows.

Before giving the characterization of V we will state two propositions containing results interesting in themselves.

#### Proposition 1.2.

Let  $\Omega$  be a bounded locally lipschitzian open set in  $\mathbb{R}^n$ .

(i) If a distribution p has all its first order derivatives  $D_{i}p$ ,  $1 \le i \le n$ , in  $L^{2}(\Omega)$ , then  $p \in L^{2}(\Omega)$  and

(1.49) 
$$\|p\|_{L^{2}(\Omega)/\mathbb{R}} \leq c(\Omega) \|\text{grad } p\|_{L^{2}(\Omega)}.$$

(ii) If a distribution p has all its first order derivatives  $D_{i}p$ ,  $1 \le i \le n$ , in  $H^{-1}(\Omega)$ , then  $p \in L^{2}(\Omega)$  and

(1.50) 
$$\|p\|_{L^{2}(\Omega)/R} \leq c(\Omega) \|g\dot{r}ad p\|_{H^{-1}(\Omega)}.$$

In both cases, if  $\Omega$  is any open set in  $\mathbb{R}^n$ ,  $p \in L^2_{100}(\Omega)$ .

#### Proof.

Point (i) and (1.49) are proved in Deny-Lions [1] for a bounded star-shaped open set  $\Omega$ . In our case, because of this result, p is  $L^2$  on every ball contained in  $\Omega$  with its closure, and on all the sets  $\mathcal{O}_j^{i}$  defined after (1.4). Since a finite number of these balls and sets  $\mathcal{O}_j^{i}$  covers  $\Omega$  the result follows.

When  $\Omega$  is a set of class  $C^2$ , a proof of point (ii) due to Lions is partly reported in Magenes-Stampacchia [1]. For  $\Omega$  satisfying only (1.4), the proof of (ii) is given in Temam [1] but is too long and technical to be reproduced here.

For a set  $\Omega$  without any regularity property we apply the preceding results on each ball contained in  $\Omega$  with its closure, and we get  $p \in L^2_{loc}(\Omega)$ .

Proposition 1.3.

Let  $\Omega$  be any open set in  $\mathbb{R}^n$ , let  $u = (u_1, \dots, u_n)$  be a vector distribution with components  $u_i$  belonging to  $H^{-1}(\Omega)$  such that

$$(1.51) \qquad \langle u,v\rangle = 0, \quad \forall v \in \mathcal{V}.$$

Then u is the gradient of some function p belonging to  $L^2_{loc}(\Omega)$ . If  $\Omega$  is locally star-shaped and bounded,  $p \in L^2(\Omega)$ .

#### Proof.

According to Lemma 1.4, for each i = 1, ..., n,

(1.52) 
$$u_i = f_i - \operatorname{div} g_i, \quad f_i \in L^2(\Omega), \quad g_i \in L^2(\Omega).$$

Let  $\rho_{\varepsilon}$  be a sequence of regularizing functions (as in Lemma 1.3 for instance); then for every  $w \in \mathcal{V}$ ,  $\rho_{\varepsilon} * w$  is a vector with components in  $\mathcal{D}(\mathbb{R}^n)$ ; if  $\varepsilon$  is strictly less than the distance between  $\Gamma$  and the support of w,  $\rho_{\varepsilon} * w$  belongs to  $\mathcal{D}(\Omega)$  and moreover to  $\mathcal{V}$  since

$$\operatorname{div}(\rho_*w) = \rho_* \operatorname{div} w = 0.$$

Let us suppose that support  $w \subset \Omega' \subset \overline{\Omega'} \subset \Omega$ ,  $\Omega'$  bounded and  $\varepsilon < d(\Omega', \Gamma)$ , so that  $w_{\varepsilon} = \rho_{\varepsilon} * w \in \mathcal{V}$ .

By (1.51) and (1.52)

$$\langle \mathbf{u}, \mathbf{w}_{\varepsilon} \rangle = 0.$$

$$\sum_{i=1}^{n} \langle \mathbf{f}_{i} - \operatorname{div} \mathbf{g}_{i}, \rho_{\varepsilon} \ast \mathbf{w}_{i} \rangle = 0$$

$$\sum_{i=1}^{n} (\mathbf{f}_{i}, \rho_{\varepsilon} \ast \mathbf{w}_{i}) + \sum_{i=1}^{n} (\mathbf{g}_{i}, \operatorname{grad}(\rho_{\varepsilon} \ast \mathbf{w}_{i})) = 0.$$

$$\sum_{i=1}^{n} (\mathbf{\tilde{f}}_{i}, \rho_{\varepsilon} \ast \mathbf{w}_{i}) + \sum_{i=1}^{n} (\mathbf{\tilde{g}}_{i}, \rho_{\varepsilon} \ast \operatorname{grad} \mathbf{w}_{i}) = 0,$$

where  $\tilde{f}_i(x) = f_i(x), x \in \Omega$ , and  $= 0 \quad x \notin \Omega$  (same definition for  $\tilde{g}_i$ ). It is easy to see that

(f, 
$$\rho_{\epsilon} * g$$
) = ( $\overset{\vee}{\rho_{\epsilon}} * f$ , g),  $\forall f, g \in L^{2}(\mathbb{R}^{n})$ 

where

We then have

$$\sum_{i=1}^{n} (\check{\rho}_{\varepsilon} * \tilde{f}_{i}, w_{i}) + \sum_{i=1}^{n} (\check{\rho}_{\varepsilon} * \tilde{g}_{i}, \text{ grad } w_{i}) = 0,$$

(1.53) 
$$\sum_{i=1}^{n} (\overset{\vee}{\rho}_{\varepsilon} * (\tilde{f}_{i} - \operatorname{div} \tilde{g}_{i}), w_{i}) = 0.$$

Let  $\sigma_{\epsilon i}$  be the restriction to  $\Omega'$  of  $\check{\rho}_{\epsilon}^*(\tilde{f}_i - \operatorname{div} \tilde{g}_i)$ , and  $\sigma_{\epsilon} = (\sigma_{\epsilon 1}, \cdots \sigma_{\epsilon n})$ . By (1.53),  $(\sigma_{\epsilon}, w) = 0$ ,  $\forall w \in \mathcal{V}$  with support in  $\Omega'$ . Hence  $\sigma_{\epsilon}$  belongs to the space  $H^{\perp}$  corresponding to  $\Omega'$  and by (1.28) and Remark 1.5,

(1.54) 
$$\sigma_{g} = \operatorname{grad} p_{g}$$
 in  $\Omega', p_{g} \in L^{2}(\Omega')$ .

As  $\varepsilon \longrightarrow 0$ ,  $\sigma_{\varepsilon}$  converges to u in  $H^{-1}(\Omega')$  and using again the de Rham theorem [1] and (1.54), we see that u is the gradient of some distribution p:

$$(1.55) u = grad p in \Omega'.$$

Since  $\Omega'$  is an arbitrary bounded subset of  $\Omega$ , with  $\Omega' \subset \overline{\Omega'} \subset \Omega$ , it is clear

a shan 1

that u is the gradient of some distribution p in all  $\Omega$ .

Since the derivatives of p are in  $H^{-1}(\Omega)$ , it follows from Proposition 1.2 that  $p \in L^2_{loc}(\Omega)$  in the general case and  $p \in L^2(\Omega)$  if  $\Omega$  is bounded and locally lipschitzian.

#### Remark 1.5.

It is clear that the proof of Proposition 1.3 can be extended to more general vector distributions u: it suffices for this proof that the  $u_i$  are of type (1.52) with  $f_i \in L^{\alpha_0}(\Omega)$ ,  $g_i \in L^{\alpha_1}(\Omega) \times \cdots \times L^{\alpha_n}(\Omega)$ , where  $1 < \alpha_j$ ,  $j = 0, \cdots, n$ . Moreover, the  $\alpha_j$  need not be the same for different components  $u_i$  of u, and each  $u_i$  can be the sum of terms  $f_i - \operatorname{div} g_i$  of preceding type with different  $\alpha_i$  for each term.

Of course we can only show in this case that u is the gradient of some distribution p, and with regularity results similar to those of Proposition 1.2, that p is a function in  $L^1_{loc}(\Omega)$ .

The space V can now be characterized as follows:

#### Theorem 1.6.

Let  $\Omega$  be an open bounded set which is locally lipschitzian. Then

(1.56)  $V = \{ u \in \mathbb{H}_{0}^{1}(\Omega), \text{ div } u = 0 \}.$ 

#### Proof.

Let V<sup>\*</sup> be the space in the right-hand member of (1.56). It is clear that  $V \subset V^*$ , for if  $u \in V$ ,  $u = \lim_{m \to \infty} u_m$ ,  $u_m \in V$ . This convergence in  $\mathbb{H}^1_0(\Omega)$  implies that div  $u_m$  converges to div u as  $m \to \infty$ , and since div  $u_m = 0$ , div u = 0.

To prove that  $V = V^{\circ}$ , we will show that any continuous linear form L on V<sup>•</sup> which vanishes on V is identically equal to 0. With almost exactly the same proof as in Lemma 1.4, noticing that V<sup>•</sup> is a closed subspace of  $\mathbb{H}_0^1(\Omega)$ , one can prove that L admits a (non unique) representation of type

(1.57) 
$$L(v) = \sum_{i=1}^{n} \langle l_i, v_i \rangle, \quad l_i \in H^{-1}(\Omega).$$

The vector distribution  $l = (l_1, \dots, l_n)$  belongs to  $\mathbb{H}^{-1}(\Omega)$ , and  $\langle l, v \rangle = 0$ ,  $\forall v \in \mathcal{P}$ . Proposition 1.3 is applicable and shows that

$$l = \text{grad } p, p \in L^2(\Omega),$$

thus

$$\langle l_i, v_i \rangle = \langle D_i p, v_i \rangle = -(p, D_i v_i), \forall v_i \in H^1_0(\Omega).$$

For  $v \in V^*$ ,

$$L(v) = \sum_{i=1}^{n} < l_{i}, v_{i} > = -(p, div v) = 0,$$

and L vanishes on all of V°

## Remark 1.6.

We can give a more direct and much simpler proof of (1.56) for a set  $\Omega$  which is globally star-shaped with respect to one of its points.

Let us suppose that  $\Omega$  is star-shaped with respect to 0 and let  $\sigma_{\lambda}$  denote as before the linear transformation  $x \mapsto \lambda x$ .

Let  $u \in V^{\bullet}$  Then the function  $\sigma_{\lambda} \circ u$  belongs to  $\mathbb{H}_{0}^{1}(\sigma_{\lambda}\Omega)$  and div  $\sigma_{\lambda}u = 0$ . The function  $u_{\lambda}$  equal to  $\sigma_{\lambda} \circ u$  in  $\sigma_{\lambda}\Omega$  and to 0 in  $\Omega - \sigma_{\lambda}\Omega$  ( $0 < \lambda < 1$ ) is in  $\mathbb{H}_{0}^{1}(\Omega)$  and div  $u_{\lambda}$  equals  $\lambda \sigma_{\lambda}(\operatorname{div} u)$  in  $\sigma_{\lambda}\Omega$  and 0 in  $\Omega - \sigma_{\lambda}\Omega$ ; hence div  $u_{\lambda} = 0$ ,  $u_{\lambda} \in V^{\bullet}$  and has a compact support in  $\Omega$ . In this case it is easy to check by regularization that  $u_{\lambda} \in V$  and since  $u_{\lambda}$  converges to u in  $\mathbb{H}_{0}^{1}(\Omega)$  as  $\lambda \longrightarrow 1$ ,  $u \in V$ , and  $V = V^{\bullet}$ .

#### Remark 1.7.

The results which will be used in the sequel are all the results given in this Section 1.5: Lemma 1.4, Propositions 1.2 and 1.3, Theorem 1.6.

#### \$2. EXISTENCE AND UNIQUENESS FOR THE STOKES EQUATIONS

The Stokes equations are the linearized stationary form of the full Navier-Stokes equations. We give here the variational formulation of Stokes problem, an existence and uniqueness result using the projection theorem, and some other remarks concerning the case of an unbounded domain and the regularity of solutions.

#### 2.1 Variational Formulation of the Problem.

Let  $\Omega$  be an open bounded set in  $\mathbb{R}^n$  with boundary  $\Gamma$ , and let  $f \in L^2(\Omega)$  be a given vector function in  $\Omega$ . We seek a vector function  $u = (u_1, \dots, u_n)$  representing the velocity of the fluid, and a scalar function p representing the pressure, which are defined in  $\Omega$  and satisfy the following equations and boundary conditions ( $\nu$  is the coefficient of kinematic viscosity, a constant):

(2.1) 
$$-v \Delta u + \text{grad } p = f \text{ in } \Omega, \quad (v > 0)$$

$$(2.2) div u = 0 in \Omega,$$

(2.3) 
$$u = 0$$
 on  $\Gamma$ .

If f, u, p are smooth functions satisfying (2.1) - (2.3), then taking the scalar product of (2.1) with a function  $v \in \mathcal{V}$ , we obtain,

$$(-v\Delta u + \text{grad } p, v) = (f, v)$$

and, integrating by parts, the term  $(-\Delta u, v)$  gives

(2.4) 
$$\sum_{i=1}^{n} (\text{grad } u_i, \text{ grad } v_i) = ((u,v)) (1)$$

and the term (grad p,v) gives

$$-(p, div v) = 0$$

and there results

(2.5) 
$$\forall ((u,v)) = (f,v), \forall v \in \mathcal{V}.$$

Since each side of (2.5) depends linearly and continuously on v for the  $\mathbb{H}^1_0(\Omega)$  topology, the equality (2.5) is still valid by continuity for each  $v \in V$ ,

(1) See the notation at the end of Section 1.1.

the closure of  $\mathcal{V}$  in  $\mathrm{H}^{1}_{0}(\Omega)$ . If the set  $\Omega$  is of class  $\mathcal{C}^{2}$ , then due to (2.3) the (smooth) function u belongs to  $\mathrm{H}^{1}_{0}(\Omega)$ , and because of (2.2) and Theorem 1.6,  $u \in V$ . We arrive then at the following conclusion:

(2.6) 
$$\begin{cases} u \text{ belongs to } V \text{ and satisfies} \\ v((u, v)) = (f, v), \forall v \in V. \end{cases}$$

Conversely, let us suppose that u satisfies (2.6), and let us then show that u satisfies (2.1) - (2.3) in some sense. Since u belongs only to  $\mathbb{H}_0^1(\Omega)$ , we have less regularity than before and we can only expect u to satisfy (2.1) - (2.3) in a sense weaker than the classical sense. Actually,  $u \in \mathbb{H}_0^1(\Omega)$  implies that the traces  $\gamma_0 u_i$  of its components are zero in  $\mathbb{H}^2(\Gamma)$ ;  $u \in \mathbb{V}$  implies (using Theorem 1.6) that div u = 0 in the distribution sense; and using (2.6) we have

$$<-v\Delta u - f, v >= 0, \forall v \in \mathcal{V}$$

where, according to Lemma 1.4,  $-\nu\Delta u-f$  belongs to  $\mathbb{H}^{-1}(\Omega)$ . Then because of Proposition 1.3, there exists some distribution  $p \in L^2(\Omega)$ , such that

 $-\nu\Delta u - f = - \text{grad } p$ 

in the distribution sense in  $\Omega$ .

We have thus proved

Lemma 2.1. Let  $\Omega$  be an open bounded set of class  $\mathcal{C}^2$ . The following conditions are equivalent i)  $u \in V$  and satisfies (2.6) ii) u belongs to  $\mathbb{H}^1(\Omega)$  and satisfies (2.1) - (2.3) in the following weak sense: (2.7)  $\begin{cases} \frac{\text{there exists } p \in L^2(\Omega) \quad \text{such that}}{-\nu \Delta u + \text{grad } p = f} \\ \frac{1}{10} \text{ the distribution sense in } \Omega \end{cases}$ 

(2.8)  $\operatorname{div} u = 0$  in distribution sense in  $\Omega$ 

(2.9)  $\gamma_0 u = 0.$ 

Definition 2.1.

The problem: find  $u \in V$  satisfying (2.6) is called the variational formulation of problem (2.1) - (2.3).

#### Remark 2.1.

Before studying existence and uniqueness problems for (2.6), let us make some remarks.

(i) The variational formulation of problem (2.1) - (2.3) was introduced by J. Leray [1][2][3]. It reduces the classical problem (2.1) - (2.3) to the problem of finding only u; then the existence of p is a consequence of Lemma 1.5.

(ii) When the set  $\Omega$  is not smooth, we have two spaces which we called V and V<sup>•</sup> in the proof of Theorem 1.6 and which may be differnt:

 $V = \text{the closure of } \mathcal{V} \text{ in } \mathbb{H}_0^1(\Omega)$  $V' = \{ u \in \mathbb{H}_0^1(\Omega), \text{ div } u = 0 \}$  $V \subset V'.$ 

We can then pose two (perhaps different) variational formulations: either (2.6) exactly, or the same problem with V replaced by V.

Sets  $\Omega$  such that  $V \neq V^{\bullet}$  are not known and, <u>a fortiori</u>, the relation between these two problems is not known. For technical reasons, particularly important in the non linear case, we will always work with the space V and consider only the variational problem (2.6).

Let us remark as a complement of Lemma 2.1, that for any set  $\Omega$ , if u satisfies (2.6) (or (2.6) with V replaced by V'), then it satisfies (2.7) with the restriction that  $p \in L^2_{loc}(\Omega)$  only; it satisfies (2.8) without any modification; and it satisfies (2.9) in the sense that  $u \in H^1_0(\Omega)$ , a more precise meaning depends on the trace theorems available for  $\Omega$ .

2.2. The Projection Theorem.

Let  $\Omega$  be any open set of  $\mathbb{R}^n$  such that

(2.10)  $\Omega$  is bounded in some direction.

According to (1.10),  $\mathbb{H}^1_0(\Omega)$  is a Hilbert space for the scalar product (2.4);  $\mathcal{V}$  is defined in (1.12) and V is the closure of  $\mathcal{V}$  in  $\mathbb{H}^1_0(\Omega)$ .

Theorem 2.1.

For any open set  $\Omega \subset \mathbb{R}^n$  which is bounded in some direction, and for every  $f \in \mathbb{L}^2(\Omega)$ , the problem (2.6) has a unique solution u.

Moreover, there exists a function  $p \in L^2_{loc}(\Omega)$  such that (2.7) - (2.8) are satisfied.

This theorem is a very easy consequence of the preceding lemma and the following classical projection theorem.

Theorem 2.2.

Let W be a separable real Hilbert space (norm  $\|\cdot\|$ ) and let a(u,v) be a bilinear continuous form on W×W, which is coercive, i.e., there exists  $\alpha > 0$  such that

$$(2.11) a(u,u) > \alpha \|u\|_{U}^2, \quad \forall u \in W.$$

Then for each & in W', the dual space of W, there exists one and only one  $u \in W$  such that

(2.12) 
$$a(u,v) = \langle \ell, v \rangle, \forall v \in W.$$

Of course to apply this theorem to (2.6), we take W = the space V equipped with the norm associated with (2.4), a(u,v) = v((u,v)), and for  $v \mapsto \langle l,v \rangle$  the form  $v \mapsto (f,v)$  which is obviously linear and continuous on V. The space V is separable as a closed subspace of the separable space  $\mathbb{H}_0^1(\Omega)$ .

#### Proof of theorem 2.2.

<u>Uniqueness</u>. Let  $u_1$  and  $u_2$  be two solutions of (2.12) and let  $u = u_1 - u_2$ . We have

$$a(u_1,v) = a(u_2,v) = \langle l,v \rangle, \forall v \in W,$$

and

 $a(u_1-u_2,v) = 0 \quad \forall v \in W.$ 

Taking v = u in this equality, we see with (2.11) that

$$\alpha \|u\|^2 \leq a(u,u) = 0,$$

and hence u = 0.

Existence. Since W is separable, there exists a sequence of elements  $w_1, \cdots w_m, \cdots$ , of W which is free and total in W. Let  $W_m$  be the space spanned by  $w_1, \cdots, w_m$ . For each fixed integer m we define an approximate solution of (2.12) in  $W_m$ ; that is, a vector  $u_m \in W_m$ 

(2.13) 
$$u_{m} = \sum_{i=1}^{m} \xi_{i,m} v_{i}, \xi_{i,m} \in \mathbb{R},$$

satisfying

(2.14) 
$$a(u_{m},v) = \langle \ell,v \rangle, \quad \forall v \in W_{m}.$$

Let us show that there exists one and only one  $u_m$  such that (2.14) holds. Equation (2.14) is equivalent to the set of m equations

(2.15) 
$$a(u_{m},w_{j}) = \langle l,w_{j} \rangle, \quad j = 1, \cdots, m,$$

and (2.15) is a linear system of m equations for the m components  $\xi_{i,m}$  of  $u_m$ 

(2.16) 
$$\sum_{i=1}^{m} \xi_{i,m} a(w_i, w_j) = \langle \ell, w_j \rangle, \quad j = 1, \cdots, m.$$

The existence and uniqueness of  $u_m$  will be proved if we show that the linear system (2.16) is regular. In order to show that it is sufficient to prove that the homogeneous linear system associated with (2.16), i.e.,

(2.17) 
$$\sum_{i=1}^{m} \xi_{i} a(w_{i}, w_{j}) = 0, \quad j = 1, \cdots, m,$$

has only one solution  $\xi_1 = \cdots = \xi_m = 0$ . But if  $\xi_1, \cdots, \xi_m$ , satisfy (2.17), then by multiplying each equation (2.17) by the corresponding  $\xi_j$  and adding these equations, we obtain

$$\sum_{i,j=1}^{m} \xi_i \xi_j a(w_i, w_j) = 0$$

or, because of the bilinearity of a,

$$a(\sum_{i=1}^{m} \xi_{i}w_{i}, \sum_{j=1}^{m} \xi_{j}w_{j}) = 0;$$

using (2.11) we find

$$\sum_{i=1}^{m} \xi_{i} w_{i} = 0$$

and finally  $\xi_1 = \cdots = \xi_m = 0$  since  $w_1, \cdots, w_m$  are linearly independent.

## Passage to the Limit.

When we put  $v = u_m$  in (2.14), we obtain

(2.18) 
$$a(u_m, u_m) = < l, u_m^>$$

from which, by (2.11), it follows that

(2.19)  
$$\alpha \|u_{m}\|_{W}^{2} \leq a(u_{m}, u_{m}) = \langle \ell, u_{m} \rangle \leq \|\ell\|_{W}, \|u_{m}\|_{W}$$
$$\|u_{m}\|_{W} \leq \frac{1}{\alpha} \|\ell\|_{W},$$

which proves that the sequence  $u_m$  is bounded independently of m in W. Since the closed balls of a Hilbert space are weakly compact, there exists an element u of W and a sequence  $u_m$ ,  $m' \longrightarrow \infty$ , extracted from  $u_m$ , such that

(2.20) 
$$u_{m} \rightarrow u$$
 in the weak topology of W, as  $m' \rightarrow \infty$ .

Let v be a fixed element of  $W_j$  for some j. As soon as  $m' \ge j$ ,  $v \in W_m'$ , and according to (2.14) we have

$$a(u_{m'})^{v} = \langle \ell, v \rangle$$

By using the following lemma, we can take the limit in this equality as  $m' \rightarrow \infty$ , and we obtain:

(2.21) 
$$a(u,v) = \langle l,v \rangle$$
.  
Equality (2.21) holds for each  $v \in \bigcup_{j=1}^{\infty} W_j$ , and since this set is dense in  $W$ ,  $j=1$ 

equality (2.21) still holds by continuity for v in W. This proves that u is a solution of (2.12).

Lemma 2.2.

Let a(u,v) be a bilinear continuous form on a Hilbert space W.

Let  $\phi_m$  (resp.  $\psi_m$ ) be a sequence of elements of W which converges to  $\phi$  (resp.  $\psi$ ) in the weak (resp. strong) topology of W. Then

(2.22) 
$$\lim_{m \to \infty} a(\psi_m, \phi_m) = a(\psi, \phi)$$

(2.23) 
$$\lim_{m \to \infty} a(\phi_m, \psi_m) = a(\phi, \psi).$$

Proof.

We write

$$a(\psi_m,\phi_m) - a(\psi,\phi) = a(\psi_m-\psi,\phi_m) + a(\psi,\phi_m-\phi).$$

Since the form a is continuous and the sequence  $\phi_m$  is bounded,

$$|a(\psi_{m}-\psi,\phi_{m})| \leq c \|\psi_{m}-\psi\|_{W} \|\phi_{m}\|_{W} \leq c' \|\psi_{m}-\psi\|_{W},$$

and this term converges to 0 as  $m \rightarrow \infty$ .

We notice next that the linear operator  $v \mapsto a(\psi, v)$  is continuous on W and hence there exists some element of W', depending on  $\psi$  and denoted by  $A(\psi)$ , such that

(2.24) 
$$a(\psi, v) = \langle A(\psi), v \rangle, \quad \forall v \in W.$$

We can now write

$$a(\psi,\phi_m-\phi) = \langle A(\psi),\phi_m-\phi \rangle$$

and this converges to 0 as  $m \rightarrow \infty$ , as a consequence of the weak convergence of  $\phi_m$ . This proves (2.22). For (2.23) we only have to apply (2.22) to the bilinear form

 $a^{*}(u,v) = a(v,u).$ 

Remark 2.2.

(i) Theorem 2.1 is also true if f is given  $H^{-1}(\Omega)$ .

(ii) It can be proved that the sequence  $\{u_m\}$  constructed in the proof of Theorem 2.2, as a whole, converges to the solution u of (2.12) in the strong topology of W. We do not prove this result here; it will appear as a consequence of Theorem 3.1.

(iii) Using the form  $A(\psi)$  introduced in the proof of Lemma 2.2 (cf. (2.24)), one can write equation (2.12) in the form

$$< A(u), v > = < \ell, v >$$

which is equivalent to

 $(2.25) A(u) = \ell in W'.$ 

An alternate classical proof of the projection theorem is to show that the operator  $u \mapsto A(u)$  is an isomorphism from W onto W'; see for instance Temam [2].
A Variational Property.

Proposition 2.1.

The solution u of (2.6) is also the unique element of V such that

(2.26)  $E(u) \leq E(v), \forall v \in V$ 

where

(2.27) 
$$E(v) = v ||v||^2 - 2(f,v).$$

Proof.

Let u be the solution of (2.6). Then as

$$\|\mathbf{u} - \mathbf{v}\|^2 > 0, \forall \mathbf{v} \in \mathbf{V},$$

we have

(2.28) 
$$\nabla \| u \|^2 + \nabla \| v \|^2 - 2 \nabla ((u,v)) > 0.$$

Because of (2.6), we have

$$-v \| u \|^{2} = v \| u \|^{2} - 2(f, u) = E(u),$$
$$-2v((u, v)) = -2(f, v)$$

and thus (2.28) gives exactly (2.26).

Conversely, if  $u \in V$  satisfies (2.26), then for any  $v \in V$  and  $\lambda \in \mathbb{R}$ , one has

 $E(u) \leq E(u+\lambda v)$ .

After working this out and simplifying, one finds

(2.29)  $\forall \lambda^2 \|v\|^2 + 2\lambda \nu((u,v)) - 2\lambda(f,v) \ge 0, \forall \lambda \in \mathbb{R}.$ 

This inequality can hold for each  $\lambda \in \mathbb{R}$  only if

$$v((u,v)) = (f,v),$$

and thus u is indeed a solution of (2.6).

Remark 2.3.

If the spaces V and V' of Remark 2.1 are different, Theorem 2.2 gives as well the existence and uniqueness of a  $\tilde{u} \in V^*$  such that

(2.30) 
$$\forall ((\tilde{u},v)) = (f,v), \forall v \in V.$$

Proposition 2.1 is also valid with u and V replaced by  $\widetilde{u}$  and V'.

2.3 The Unbounded Case.

We consider here the case where  $\Omega$  is unbounded, without satisfying (2.10). If  $\Omega$  is of class  $C^2$ , problem (2.1) - (2.3) is not equivalent to (2.6) as in Lemma 2.1. The reason is that if u is a classical solution of (2.1) - (2.3), it is not clear without further information about the behavior of u at infinity, that  $u \in \mathbb{H}^1(\Omega)$ ; hence perhaps  $u \notin \mathbb{V}$  and equation (2.5) cannot be extended by continuity to the closure V of  $\mathbb{V}$ . Besides that, if one would try to solve directly problem (2.6), using Theorem 2.2, the difficulty is that hypothesis (2.11) is not satisfied: V is a Hilbert space for the norm

$$[[u]] = \sqrt{|u|^2 + ||u||^2},$$

which is not equivalent to the norm  $\|u\|$  since we lose the Poincaré inequality.

In order to pose and solve a variational problem in the general case, let us introduce the space

(2.31) 
$$W =$$
 the completion of  $\mathcal{V}$  under the norm  $\|\cdot\|$ .  
It is clear, since  $\|u\| \leq [\![u]\!]$ , that  $W$  is a larger space than  $V$   
(2.32)  $V \subset W$ .  
Lemma 2.3.

(2.33)  $W \subset \{ u \in \mathbb{L}^{\alpha}(\Omega) ; D_{i} u \in \mathbb{L}^{2}(\Omega), 1 \leq i \leq n \}$ 

with a continuous injection, where

(2.34) 
$$\begin{cases} \alpha & \text{is any number}, \quad 1 \leq \alpha < +\infty, \quad \text{if } n = 2 \\ \alpha &= \frac{2n}{n-2}, \quad \text{if } n \geq 3. \end{cases}$$

### Proof.

This is a consequence of Sobloev inequality (see Sobolev [1], Lions [1], or also Chapter II):

(2.35) 
$$\|\phi\|_{L^{\alpha}(\Omega)} \leq c(\alpha,n) \|\text{grad }\phi\|_{L^{2}(\Omega)}, \quad \forall \phi \in \mathcal{D}(\Omega),$$

where  $\alpha$  is given by (2.34).

If  $u \in W$ , there exists a sequence of elements  $u_m \in \mathcal{V}$  converging to u; by (2.35)

$$\left\| \mathbf{u}_{\mathbf{m}} - \mathbf{u}_{\mathbf{p}} \right\|_{\mathbb{L}^{\alpha}(\Omega)} \leq \mathbf{c}'(\alpha, \mathbf{n}) \left\| \mathbf{u}_{\mathbf{m}} - \mathbf{u}_{\mathbf{p}} \right\|$$

and

(2.36) 
$$\|u_{m} - u_{p}\|_{Y} \leq c'' \|u_{m} - u_{p}\|_{Y}$$

where Y stands for the space on the right-hand side of inclusion (2.33) and  $\|\cdot\|_v$  is the natural norm of Y

$$\|\mathbf{u}\| = \|\mathbf{u}\| + \|\mathbf{u}\|.$$

As m and p tend to infinity, the right side of (2.36) converges to 0. Thus  $u_m$  is a Cauchy sequence in Y; its limit u belongs to Y. It is clear also that

$$\| u \|_{Y} \leq c'' \| u \|_{Y}$$

with the same c'' as in (2.36)  $(c'' = c''(\alpha, n))$ .

Theorem 2.3.

Let  $\Omega$  be any open set in  $\mathbb{R}^n$ , and let f be given in W', the dual of the space W in (2.31).

Then, there exists a unique  $u \in W$  such that

$$(2.37) \qquad \qquad \forall v \in W.$$

There exists  $p \in L^2_{100}(\Omega)$ , such that (2.7) is satisfied, and (2.8) is true.

Proof.

We apply Theorem 2.2 with the space W, a(u,v) = v((u,v)), and k replaced by f; we get a unique u satisfying (2.37).

After that, Proposition 1.3 shows the existence of some  $p \in L^2_{loc}(\Omega)$  such that (2.7) is satisfied; (2.8) is of course easily verified. Finally, (2.9) is satisfied in some sense depending on the trace theorem available for W (or for the space Y on the right-hand side of inclusion (2.33)).

If  $\Omega$  is locally lipschitzian, Proposition 1.2 shows that  $p \in L^2_{100}(\overline{\Omega})$ 

Remark 2.4. By Lemma 2.3, for  $f \in L^{\alpha'}(\Omega) \left(\frac{1}{\alpha} + \frac{1}{\alpha'} = 1\right)$ , (2.38)  $v \mapsto \int_{\Omega} f \cdot v \, dx$  is a linear continuous form on W. Thus one can take in Theorem 2.3, any  $f\in L^{\alpha'}(\Omega)$  .

#### 2.4 Regularity Results

A classical result is that the solution  $u \in H_0^1(\Omega)$  of the Dirichlet problem  $-\Delta u+u = f$  belongs to  $H^{m+2}(\Omega)$  whenever  $f \in H^m(\Omega)$  (and  $\Omega$  is sufficiently smooth). One naturally wonders whether similar regularity results exist for the Stokes problem,

#### Theorem 2.4.

If  $\Omega$  is an open bounded set of class  $\mathcal{C}^{m+2}$ , m integer, m  $\geq 0$  and if f belongs to  $H^{m}(\Omega)$ , then the solution u of (2.6) satisfies (2.7) - (2.9) and

(2.39) 
$$u \in H^{m+2}(\Omega), p \in H^{m+1}(\Omega)$$

(2.40) 
$$\| u \|_{H^{m+2}(\Omega)} + \| p \|_{H^{m+1}(\Omega)/\mathbb{R}} \leq c(\nu, m, \Omega) \| f \|_{H^{m}(\Omega)}$$

The theorem follows from the next proposition.

Proposition 2.2.

Let  $\Omega$  be an open bounded set of class  $\mathcal{C}^r$ ,  $r = \max(m+2,2)$ , m integer > 0. Let us suppose that

(2.41) 
$$u \in W^{1,\alpha}(\Omega), p \in L^{\alpha}(\Omega), 1 < \alpha < +\infty,$$

are solutions of the generalized Stokes problem.

 $(2.42) \quad -\nu \Delta u + \text{grad } p = f \quad \text{in } \Omega,$ 

 $(2.43) div u = g in \Omega,$ 

(2.44) 
$$\gamma_0 u = \phi$$
; i.e.,  $u = \phi$  on  $\Gamma$ .

If 
$$f \in W^{m,\alpha}(\Omega)$$
,  $g \in W^{m+1,\alpha}(\Omega)$  and  $\phi \in W^{m+2-\frac{1}{\alpha},\alpha}(\Gamma)$ , <sup>(1)</sup> then

(2.45) 
$$u \in W^{m+2,\alpha}(\Omega), \quad p \in W^{m+1,\alpha}(\Omega)$$

<sup>(1)</sup>  $W^{m+2-\frac{1}{\alpha},\alpha}(\Gamma) = \gamma_0 W^{m+2,\alpha}(\Omega)$  and is equipped with the image norm  $\|\psi\|_{W^{m+2-\frac{1}{\alpha},\alpha}(\Gamma)} = \inf_{\gamma_0 u=\psi} \|u\|_{W^{m+2,\alpha}(\Omega)}$  and there exists a constant  $c_0(\alpha, \nu, m, \Omega)$  such that

$$(2.46) \qquad \|u\|_{W^{m+2},\alpha_{(\Omega)}} + \|p\|_{W^{m+1},\alpha_{(\Omega)}/\mathbb{R}} \leq c_{0} \{\|f\|_{W^{m},\alpha_{(\Omega)}} + \|g\|_{W^{m+1},\alpha_{(\Omega)}} + \|\phi\|_{W^{m+2-\alpha},\alpha_{(\Gamma)}} + d_{\alpha}\|u\|_{\mathbb{L}^{\alpha}(\Omega)} \}$$

 $d_{\alpha} = 0$  for  $\alpha \ge 2$ ,  $d_{\alpha} = 1$  for  $1 < \alpha < 2$ .

# <u>Proof</u>.

This proposition results from the paper of Agmon-Douglis-Nirenberg [2], hereinafter referred to as A.D.N., giving <u>a priori</u> estimates of solutions of general elliptic systems.

Let  $u_{n+1} = \frac{1}{\nu}p$ ,  $u = (u_1, \dots, u_{n+1})$ ,  $f = (+\frac{f_1}{\nu}, \dots, \frac{f_n}{\nu}, g)$ . Then equations (2.42) and (2.43) become

(2.47) 
$$\sum_{j=1}^{n+1} \ell_{j}(D)u_{j} = f_{j}, \quad 1 \le i \le n+1,$$

where  $\ell_{ij}(\xi)$ ,  $\xi = (\xi_1, \dots, \xi_n) \in \mathbb{R}^n$ , is the matrix

(2.48) 
$$\begin{cases} l_{ij}(\xi) = |\xi|^2 \delta_{ij}, \quad 1 \le i, j \le n, \\ l_{n+1,j}(\xi) = -l_{j,n+1}(\xi) = \xi_j, \quad 1 \le j \le n \\ l_{n+1,n+1}(\xi) = 0, \\ |\xi|^2 = \xi_1^2 + \cdots + \xi_n^2. \end{cases}$$

We take (see p. 38),  $s_i = 0$ ,  $t_i = 2$ ,  $1 \le i \le n$ ,  $s_{n+1} = -1$ ,  $t_{n+1} = 1$ . As requested, degree  $\ell_{ij}(\xi) \le s_i + t_j$ ,  $1 \le i, j \le n+1$ , and we have  $\ell'_{ij}(\xi) = \ell_{ij}(\xi)$ . We easily compute  $L(\xi) = \det \ell'_{ij}(\xi) = |\xi|^{2n}$ , so that  $L(\xi) \ne 0$  for real  $\xi \ne 0$ , and this ensures the ellipticity of the system (Condition (1.5), p. 39). It is clear that (1.7) on p. 39 holds with m = n. The Supplementary Condition on L is satisfied:  $L(\xi + \tau\xi') = 0$  has exactly n roots with positive imaginary part and these roots are all equal to

$$\tau^{+}(\xi,\xi') = -\xi \cdot \xi' + i \sqrt{|\xi|^2 |\xi'|^2 - |\xi \cdot \xi'|^2}.$$

Concerning the boundary conditions (see p. 42), there are m boundary conditions and

 $B_{hj} = \delta_{hj}$  (the Kronecker symbol) for  $1 \le h \le n$ ,  $1 \le j \le n+1$ .

We take  $r_h = -2$  for  $h = 1, \dots, n$ . Then, as requested, degree  $B_{hj} \leq r_h + t_j$ and we have  $B'_{hj} = B_{hj}$ .

It remains to check the Complementing Boundary Condition. It is easy to verify that

$$M^{+}(\xi) = (\tau - \tau^{+}(\xi))^{r}$$

where  $\tau^{+}(\xi) = \tau^{+}(\xi, v)$ . The matrix with elements  $\sum_{i=1}^{N} B'_{hj}(\xi) L^{jk}(\xi)$  is simply

the matrix with elements  $\ell_{hk}(\xi)$ ,  $1 \le h, k \le n$ ,  $-\ell_{h,n+1}(\xi)$ ,  $1 \le h \le n$ . A combination  $\sum_{h=1}^{n} C_{h} \sum_{j=1}^{N} B'_{hj} L^{jk}$  is then equal to

$$(C_{i}(\xi+\tau\nu)^{2}, \cdots, C_{n}(\xi+\tau\nu)^{2}, \sum_{i=1}^{n} C_{i}(\xi_{i}+\tau\nu_{i}))$$

and this is zero modulo  $M^+$  only if  $C_1 = \cdots = C_n = 0$ , and the Complementing Condition holds.

We then apply Theorem 2.5, page 78 of A.D.N. in order to get (2.45) and (2.46) with  $d_{\alpha} = 1$  for all  $\alpha$ . According to the remark after Theorem 2.5, one can take  $d_{\alpha} = 0$  for  $\alpha \ge 2$  since the solutions u and p of (2.41) -(2.44) are necessarily unique (p is unique up to an additive constant): if  $(u_*, p_*)$ ,  $(u_{**}, p_{**})$  are two solutions, then  $u = u_* - u_{**}$ ,  $p = p_* - p_{**}$  are solutions of (2.7) - (2.9) with f = 0 and hence u = 0 and p = constant.

Remark 2.5.

For  $\alpha = 2$  and  $m \in \mathbb{R}$ ,  $m \ge -1$ , one has results similar to those in Theorems 2.4 and 2.5 by using the interpolation techniques of Lions-Magenes [1].

Proposition 2.2 does not assert the existence of u,p satisfying (2.42) - (2.46) (for given f,g, $\phi$ ) but gives only a result on the regularity of an eventual solution. The existence is ensured by the variational method if  $\alpha = 2$ , g = 0, and  $\phi = 0$ . The following proposition gives a general existence result for n = 2 or 3.

Proposition 2.3.

Let  $\Omega$  be an open set of class  $\mathcal{C}^{\mathbf{r}}$ ,  $\mathbf{r} = \max(m+2,2)$ , m integer  $\geq -1$ , and let  $f \in \mathbb{W}^{m,\alpha}(\Omega)$ ,  $g \in W^{m+1,\alpha}(\Omega)$ ,  $\phi \in W^{m+2-\alpha,\alpha}(\Gamma)$ , be given satisfying the compatibility condition

(2.49)

 $\int_{\Omega} g dx = \int_{\Gamma} \phi \cdot v d\Gamma.$ 

Then there exist unique functions u and p (p is unique up to a constant) which are solutions of (2.42) - (2.44) and satisfy (2.45) and (2.46) with  $d_{\alpha} = 0$ for any  $\alpha$ .

<u>Proof</u>.

This is precisely the result proved in Cattabriga [1] when n=3 (and even for m=-1).

For n = 2 one can reduce the problem to a classical biharmonic problem. There exists  $v \in W^{m+2,\alpha}(\Omega)$ , such that

(2.50) 
$$div v = g$$

$$(2.51) \qquad \qquad \gamma_0 v = \phi .$$

Such v can be defined by

(2.52) 
$$\mathbf{v} = \operatorname{grad} \theta + \left\{\frac{\partial \sigma}{\partial \mathbf{x}_2}, -\frac{\partial \sigma}{\partial \mathbf{x}_1}\right\}$$

where  $\theta \in \mathbb{W}^{m+3,\alpha}(\Omega)$  is solution of the Neumann problem

$$(2.53) \qquad \qquad \Delta \theta = g \quad \text{in} \quad \Omega$$

(2.54) 
$$\frac{\partial \theta}{\partial v} = \phi \cdot v \quad \text{on} \quad \Gamma$$

and  $\sigma \in W^{m+3,\alpha}(\Omega)$  will be chosen later.

The Neumann problem (2.53) - (2.54) has a solution  $\theta$  because of (2.49), and  $\theta \in W^{m+3,\alpha}(\Omega)$  by the usual regularity results for the Neumann problem. The conditions on  $\sigma$  are only boundary conditions on  $\Gamma$  and these are

 $\frac{\partial \sigma}{\partial \tau}$  = the tangential derivative of  $\sigma = 0$ ,  $\frac{\partial \sigma}{\partial v}$  = the normal derivative of  $\sigma = \phi \cdot \tau - \frac{\partial \theta}{\partial \tau}$ .

Since  $\phi.\tau - \frac{\partial \theta}{\partial \tau} \in W^{m+2-\frac{1}{\alpha},\alpha}(\Gamma)$ , there exists a  $\sigma \in W^{m+3,\alpha}(\Gamma)$  with  $\gamma_0 \sigma = 0$ ,

 $\gamma_1 \sigma = \phi. \tau - \frac{\partial \theta}{\partial \tau}$ . With these definitions of  $\sigma$  and  $\theta$ , the vector v in (2.52) belongs to  $W^{m+2,\alpha}(\Omega)$  and satsifies (2.50) - (2.51). Moreover, the mapping  $\{g,\phi\} \mapsto v$  is linear and continuous.

Setting w = u-v, the problem (2.42) - (2.44) reduces to the problem of finding  $w \in W^{m+2,\alpha}(\Omega)$ ,  $p \in W^{m+1,\alpha}(\Omega)$  such that

 $-\nu\Delta w + \text{grad } p = f', f' = f + \nu\Delta v,$ (2.55)div w = 0, (2.56)(2.57) $\gamma_0 w = 0.$ 

If  $\Omega$  is simply connected then, because of (2.56), there exists a function  $\rho$ such that

(2.58) 
$$w = (D_2 \rho, -D_1 \rho),$$

Condition (2.57) amounts to saying that  $\rho = \frac{\partial \rho}{\partial v} = 0$  on  $\Gamma$  and (2.55) gives after differentiation

$$-\nu\Delta(D_2w_1-D_1w_2) = D_2f'_1-D_1f'_2.$$

Thus we obtain

[1], Lions-Magenes [1]).

(2.59)  $\nu\Delta^{2}\rho = D_{1}f'_{2}-D_{2}f'_{1} \in W^{m-1,\alpha}(\Omega)$ (2.60)  $\rho = 0, \quad \frac{\partial\rho}{\partial \nu} = 0 \quad \text{on} \quad \Gamma.$ 

The biharmonic problem (2.59)(2.60) has a unique solution  $\rho \in W^{m+3,\alpha}(\Omega)$ , and the function w defined by (2.58) is a solution of (2.55) - (2.57) and belongs to  $W^{m+2,\alpha}(\Omega)$ .

If  $\Omega$  is not simply connected we first obtain an existence and uniqueness result for (2.55) - (2.57) by the variational method (Theorem 2.1). We notice that  $f' \in W^{m,\alpha}(\Omega)$ , i.e.,  $f' \in L^{\alpha}(\Omega)$  at least, and  $L^{\alpha}(\Omega) \subset H^{-1}(\Omega)$  since by the Sobolev inclusion theorem  $H_0^1(\Omega) \subset L^{\beta}(\Omega), \forall \beta, 1 < \beta < +\infty$ , in the two dimensional case. The solution w belongs to  $H_0^1(\Omega)$  and  $p \in L^2(\Omega)$ . If  $1 < \alpha \leq 2$ , the proof is achieved using Proposition 2.2. If  $\alpha > 2$ , we obtain more regularity on w by introducing locally the function  $\rho$  and recalling that the regularity property for an elliptic equation is a local property (if  $\rho$  satisfies (2.60) and  $\Delta^2 \rho \in W^{m-1,\alpha}(\partial)$ ,  $\partial \subset \Omega$ , then  $\rho \in W^{m+3,\alpha}_{0}(\partial')$  for  $\partial' = O \cup (\partial \cap \Gamma)$ ; for this see, for instance, the synthesis presentation of Agmon-Douglis-Nirenberg

#### DISCRETIZATION OF THE STOKES EQUATIONS §3.

Section 3.1 deals with the general concept of approximation of a normed space and Section 3.2 contains a general convergence theorem for the approximation of a general variational problem. In the last section and throughout all of Section 4 we then describe some particular approximations of the basic space V of the Navier-Stokes equations. We give the corresponding numerical scheme for the Stokes equations and then apply the general convergence theorem to this case.

In Section 3.3 we consider the finite difference method. The finite element methods will be treated in the next section (Section 4; 4.1 to 4.5).

The approximations of the space V introduced here will be used throughout subsequent chapters, and they will be referred to by (APX1), (APX2), ....

### 3.1 Approximation of a Normed Space.

When computational methods are involved, a normed space W must be a approximated by a family  $(W_h)_{h \in \mathcal{H}}$  of normed spaces  $W_h$ . The set  $\mathcal{H}$  of indexes depends on the type of approximation considered: we will consider in the following the main situations for  $\mathfrak{H}$ , i.e.,  $\mathfrak{H} = \mathbb{N}$  (= positive integers) for the Galerkin method,  $\mathcal{H} = \prod_{i=1}^{n} (0, h_i^0]$  for finite differences, and  $\mathcal{H} = a$  set of triangulations of the domain  $\Omega$  for finite element methods. The precise form of  $\mathcal H$  need not be known; we need only to know the existence of a filter on H, and we are concerned with passing to the limit through this filter. For the sake of simplicity we will always speak about passage to the limit as "h $\rightarrow$  0", which is strictly speaking the correct terminology for finite differences; definitions and results can be easily adapted to the other cases.

Definition 3.1.

An internal approximation of a normed vector space W is a set consisting of a family of triples  $\{W_h, p_h, r_h\}$ ,  $h \in \mathcal{H}$ , where

i) W<sub>h</sub> is a normed vector space;

ii)  $p_h \underline{is a linear continuous operator from } W_h \underline{into} W;$ iii)  $r_h \underline{is a (perhaps nonlinear) operator from } W \underline{into} W_h.$ 

The natural way to compare an element  $u \in W$  and an element  $u_h \in W_h$  is either to compare  $p_h u_h$  and u in W or to compare  $u_h$  and  $r_h u$  in  $W_h$ . The first point of view is certainly more interesting as we make comparisons in a fixed space. Nevertheless comparisons in W<sub>b</sub> can also be useful.

Another way to compare an element  $u \in W$  and an element  $u_h \in W_h$  is to compare a certain image  $\overline{\omega}u$  of u in some other space F, with a certain image  $p_h u_h$  of  $u_h$  in F. This leads to the concept of external approximation of a space W, which contains the concept of internal approximation as a particular case.



An external approximation of a normed space W is a set consisting of i) a normed space F and an isomorphism  $\overline{\omega}$  of W into F. ii) a family of triples  $\{W_h, p_h, r_h\}_{h \in \mathcal{H}}$ , in which, for each h,

- W<sub>h</sub> <u>is a normed space</u>

-  $p_h$  <u>a linear continuous mapping of</u>  $W_h$  <u>into</u> F

- r<sub>h</sub> <u>a (perhaps nonlinear) mapping of</u> W <u>into</u> W<sub>h</sub>

When F = W and  $\overline{w}$  = identity, we get of course an internal approximation of W. It is easy to specialize what follows to internal approximations.

In most cases,  $W_h$  are finite dimensional spaces; rather often the operators  $p_h$  are injective. In some cases the operators  $r_h$  are linear, or only linear on some subspace of W, but there is no need to impose this condition in the general case; also, no continuity property of the  $r_h$  is required.

The operators  $p_h$  and  $r_h$  are called <u>prolongation</u> and <u>restriction</u> <u>operators</u>, respectively. When the spaces W and F are Hilbert spaces, and when the spaces  $W_h$  are likewise Hilbert spaces, the approximation is said to be a Hilbert approximation.

We define now stable and convergent approximations.

<u>Definition 3.4</u>. <u>Stable Approximation</u>. <u>The prolongation operators  $p_h$  are said to be stable if their norms</u>

 $\|\mathbf{p}_{h}\| = \sup_{\mathbf{u}_{h} \in W_{h}} \|\mathbf{p}_{h}\mathbf{u}_{h}\|_{\mathbf{F}}$  $\|\mathbf{u}_{h}\|_{W_{h}} = 1$ 

can be majorized independently of h.

The approximation of the space W is said to be stable if the prolongation operators are stable.

Let us now consider what happens when  $"h \rightarrow 0"$ .

Definition 3.5.

 $\begin{array}{c} \underline{\text{We will say that a family u}_h \quad \text{converges strongly (or weakly) to u if} \\ \textbf{p}_h\textbf{u}_h \quad \underline{\text{converges to } \overline{\textbf{wu}} \quad \underline{\text{when } h \rightarrow 0} \quad \underline{\text{in the strong (or weak) topology of } F. \\ \underline{\text{We will say that the family } u_h \quad \underline{\text{converges discretely to } u \quad \underline{\text{if}} \end{array}$ 

$$\lim_{h \to 0} \left\| u_h - r_h u \right\|_h = 0.$$

Definition 3.6.

We will say that an external approximation of a normed space W is convergent if the two following conditions hold:

(C1) for all  $u \in W$ ,

$$\lim_{h \to 0} p_h r_h u = \overline{\omega} u$$

in the strong topology of F.

(C2) for each sequence  $u_h$ , of elements of  $W_h$ ,  $(h' \rightarrow 0)$ , such that  $p_h, u_h$ , converges to some element  $\phi$  in the weak topology of F, we have  $\phi \in \overline{u}W$ ; i.e.,  $\phi = \overline{u}u$  for some  $u \in W$ .

Remark 3.1.

Condition (C2) disappears in the case of internal approximations.

The following proposition shows that condition. (C1) can be in some sense weakened for internal and external approximations.

Proposition 3.1.

Let there be given a stable external approximation of a space W which is convergent in the following restrictive sense: the operators  $r_h$  are only defined on a dense subset W of W and condition (C1) in Definition 3.6 holds only for the u belonging to W (condition (C2) remains unchanged).

Then it is possible to extend the definition of the restriction operators  $r_h$  to the whole space W so that condition (C1) is valid for each  $u \in W$  and hence the approximation of W is stable and convergent without any restriction.

# Proof.

Let  $u \in W$ ,  $u \notin W$ ; we must define in some way  $r_h u_h \in W_h$ ,  $\forall h$ , so that  $p_h r_h u \rightarrow \overline{w} u$  as  $h \rightarrow 0$ . This element u can be approximated by elements in W, and these elements in turn can be approximated by elements in the  $p_h W_h$ ; we only have to combine suitably these two approximations.

For each integer  $n \ge 1$ , there exists  $u_n \in W$  such that  $||u_n - u||_W \le \frac{1}{n}$ and hence

$$(3.1)  $\|\overline{\omega}u_n - \overline{\omega}u\|_F \leq \frac{c_0}{n};$$$

where  $c_0$  is the norm of the isomorphism  $\overline{\omega}$ .

For each fixed integer n,  $p_h r_h u_n$  converges to  $\overline{\omega} u_n$  in F as  $h \rightarrow 0$ . Thus there exists some  $\eta_n > 0$ , such that  $|h| \le \eta_n$  implies

$$\|\mathbf{p}_{\mathbf{h}}\mathbf{r}_{\mathbf{h}}\mathbf{u}_{\mathbf{n}}-\overline{\boldsymbol{\omega}}\mathbf{u}_{\mathbf{n}}\|_{\mathbf{F}} \leq \frac{1}{n}.$$

We can suppose that  $\eta_n$  is less than both  $\eta_{n-1}$  and  $\frac{1}{n}$  so that the  $\eta_n$  form a strictly decreasing sequence converging to 0:

$$0 < \eta_{n+1} < \cdots < \eta_1; \quad \eta_n \longrightarrow 0.$$

Let us define r<sub>b</sub>u by

$$r_h u = r_h u_n$$
 for  $\eta_{n+1} < |h| \le \eta_n$ .

It is clear that for  $\eta_{n+1} < |h| \le \eta_n$ ,

$$\|\overline{\omega}u-p_{h}r_{h}u\|_{F} \leq \|\overline{\omega}u-\overline{\omega}u_{n}\|_{F} + \|\overline{\omega}u_{n}-p_{h}r_{h}u_{n}\|_{F} + \|p_{h}r_{h}u_{n}-p_{h}r_{h}u\|_{F} \leq \frac{(1+c_{0})}{n}$$

and consequently

$$\left\|\overline{\omega}u-p_{h}r_{h}u\right\|_{F} \leq \frac{1+c_{0}}{n}$$

for  $|h| \leq \eta_n$ . This implies the convergence of  $p_h r_h u$  to  $\overline{\omega} u$  as  $h \to 0$  and completes the proof.

### Remark 3.2.

If the mappings  $r_h$  are defined on the whole space W and condition (C1) holds for all  $u \in W$ , Proposition 3.1 shows us that we can modify the value of  $r_h u$  on the complement of W so that condition (C1) is satisfied for all  $u \in W$ .

### Galerkin Approximation of a Normed Space.

As a very easy example we can define a Galerkin approximation of any separable normed space W.

Let  $W_h$ ,  $h \in \mathbb{N} = \mathcal{H}$ , be an increasing sequence of finite dimensional subspaces of W whose union is dense in W. For each h, let  $p_h$  be the canonical injection of  $W_h$  into W, and for any  $u \in W_{h_0}$ , let  $r_h u = 0$  if  $h \leq h_0$ ,  $r_h u = u$  if  $h \geq h_0$ . It is clear that  $p_h r_h u \rightarrow u$  as  $h \rightarrow \infty$ , for any  $u \in \bigcup_{h \in \mathbb{N}} W_h$ . The  $h \in \mathbb{N}$  operator  $r_h$  is only defined on  $\mathcal{W} = \bigcup_{h \in \mathbb{N}} W_h$  which is dense in W. Since the  $h \in \mathbb{N}$  h prolongation operators have norm one they are stable and according to Proposition 3.1 the definition of the operators  $r_h$  can be extended in some way (which does not matter) to the whole space W so that we get a stable convergent internal approximation of W; this is a Galerkin approximation of W.

#### 3.2 A General Convergence Theorem.

Let us discuss now the approximation of the general variational problem (2.12): W is a Hilbert space, a(u,v) is a coercive bilinear continuous form on  $W \times W$ ,

(3.2) 
$$a(u,u) \geq \alpha \|u\|_{W}^{2}, \quad \forall u \in W, \quad (\alpha > 0),$$

and l is a linear continuous form on W.

Let u denote the unique solution in W of

$$(3.3) a(u,v) = \langle l,v \rangle, \quad \forall v \in W.$$

With respect to the approximation of this element u, let there be given an external stable and convergent Hilbert approximation of the space W, say  $\{W_h, p_h, r_h\}_{h \in \mathcal{H}}$ . Likewise, for each  $h \in \mathcal{H}$ , let there be given

i) a continuous bilinear form  $a_h(u_h,v_h)$  on  $W_h \times W_h$  which is coercive and which, more precisely, satisfies

(3.4) 
$$\begin{cases} \exists \alpha_0 > 0, \text{ independent of } h, \text{ such that} \\ a_h(u_h, u_h) \ge \alpha_0 \|u_h\|_h, \quad \forall u_h \in W_h \end{cases}$$

where  $\|\cdot\|_{h}$  stands for the norm in  $W_{h}$ ,

ii) a continuous linear form on  $W_h, l_h \in W'_h$ , such that:

$$(3.5)  $\| \ell_{\mathbf{h}} \|_{\mathbf{h}} \leq \beta,$$$

in which  $\|\cdot\|_{*h}$  stands for the norm in  $W'_h$ , and in which  $\beta$  is independent of h. We associate now with equation (3.3) the following family of approximate equations:

For fixed 
$$h \in \mathcal{H}$$
, find  $u_h \in W_h$  such that

(3.6) 
$$a_h(u_h, v_h) = \langle l_h, v_h \rangle, \quad \forall v_h \in W_h.$$

By the preceding hypotheses, Theorem 2.2 (in which  $W,W',a,\ell$ , are replaced by  $W_h,W'_h,a_h,\ell_h$ ) asserts now that equation (3.6) has a unique solution; we will say that  $u_h$  is the approximate solution of equation (3.3).

A general theorem on the convergence of the approximate solutions  $u_h$  to the exact solution will be given after making precise the manner in which the forms  $a_h$  and  $\ell_h$  are consistent with the forms a and  $\ell$ . We make the following consistency hypotheses:

(3.7) If the family  $v_h$  converges weakly to v as  $h \rightarrow 0$ , and if the family  $w_h$  converges strongly to w as  $h \rightarrow 0$ ,  $\lim_{h \to 0} a_h (v_h, w_h) = a(v, w)$  $\lim_{h \to 0} a_h (w_h, v_h) = a(w, v)$ .

(3.8) If the family  $v_h$  converges weakly to v as  $h \rightarrow 0$ , then  $\lim_{h \rightarrow 0} \langle \ell_h, v_h \rangle = \langle \ell, v \rangle$ .

The general convergence theorem is then

Theorem 3.1.

Under the hypotheses (3.2), (3.4), (3.5), (3.7) and (3.8), the solution  $u_{h}$  of (3.6) converges strongly to the solution u of (3.3), as  $h \rightarrow 0$ .

# Proof.

Putting  $v_h = u_h$  in (3.6) and using (3.4) and (3.5), we find

(3.9) 
$$\begin{aligned} \mathbf{a}_{h}(\mathbf{u}_{h},\mathbf{u}_{h}) &= \langle \boldsymbol{\ell}_{h},\mathbf{u}_{h} \rangle, \\ \alpha_{0} \| \mathbf{u}_{h} \|_{h}^{2} &\leq \| \boldsymbol{\ell}_{h} \|_{\star h} \| \mathbf{u}_{h} \|_{h} \leq \beta \| \mathbf{u}_{h} \|_{h}; \end{aligned}$$

hence

$$\|\mathbf{u}_{\mathbf{h}}\|_{\mathbf{h}} \leq \beta/\alpha_{0}.$$

As the operators 
$$p_h$$
 are stable, there exists a constant  $c_0$  which majorizes the norm of these operators

(3.11) 
$$\| \mathbf{p}_{h} \| = \| \mathbf{p}_{h} \|_{\mathcal{L}(W_{h},F)} \leq c_{0};$$

and hence

$$\|\mathbf{p}_{h}\mathbf{u}_{h}\|_{\mathbf{F}} \leq \frac{c_{0}\beta}{\alpha_{0}}$$

Under these conditions, there exists some  $\,\phi\,{\in}\,F\,$  and a sequence  $\,h^{\,\prime}\,$  converging to 0, such that

 $\lim_{h' \to 0} p_{h'} u_{h'} = \phi$ 

in the weak topology of F; according to hypothesis (C2) in Definition 3.6,  $\phi \in \overline{\omega}W$ , whence  $\phi = \overline{\omega}u_*$  for some  $u_* \in W$ :

(3.13) 
$$\lim_{h' \to 0} p_{h'} u_{h'} = \overline{\omega} u_{*}, \quad (\text{weak topology of } F).$$

Let us show that  $u_* = u$ . For a fixed  $v \in W$ , we write (3.6) with  $v_h = r_h v_h$ and then take the limit with the sequence h' which gives, by using (3.7), (3.8), and (3.13):

$$a_{h}(u_{h}, r_{h}, v) = \langle \ell_{h}, r_{h}, v \rangle$$

$$\lim_{h' \to 0} a_{h'}(u_{h'}, r_{h'}, v) = a(u_{*}, v)$$

$$\lim_{h' \to 0} \langle \ell_{h'}, r_{h'}, v \rangle = \langle \ell, v \rangle$$

Finally

 $a(u_*,v) = \langle \ell, v \rangle,$ 

and because  $v \in W$  is arbitrary,  $u_*$  is a solution of (3.3) and thus  $u_* = u$ .

One may show in exactly the same way, that from every subsequence of  $p_h u_h^h$  one can extract a subsequence which converges in the weak topology of F to  $\overline{\omega}u$ . This proves that the family  $p_h u_h^h$  as a whole converges to  $\overline{\omega}u$ , in the weak topology, as  $h \rightarrow 0$ .

#### Proof of the Strong Convergence.

Let us consider the expression

$$X_{h} = a_{h}(u_{h}-r_{h}u, u_{h}-r_{h}u),$$

or

$$X_{h} = a_{h}(u_{h}, u_{h}) - a_{h}(u_{h}, r_{h}u) - a_{h}(r_{h}u, u_{h}) + a_{h}(r_{h}u, r_{h}u).$$

By (3.7), (3.8), and (3.9),

$$\lim_{h \to 0} a_h(u_h, u_h) = \langle l, u \rangle$$

 $\lim_{h \to 0} a_h(u_h, r_h u) = \lim_{h \to 0} a_h(r_h u, u_h) = \lim_{h \to 0} a_h(r_h u, r_h u) = a(u, u).$ 

Finally

(3.14) 
$$\lim_{h \to 0} x_h = -a(u,u) + \langle l, u \rangle = 0,$$

according to (3.3) (when v = u).

With (3.4) and (3.11) we get now

$$0 \leq \alpha_0 \| u_h - r_h u \|_h^2 \leq X_h, \quad \text{whence}$$
$$0 \leq \| p_h u_h - p_h r_h u \|_F^2 \leq \frac{c_0^2}{\alpha_0} X_h \longrightarrow 0.$$

Using now condition (C1) of Definition 3.6 and

$$\left\| \mathbf{p}_{\mathbf{h}} \mathbf{u}_{\mathbf{h}} - \overline{\omega} \mathbf{u} \right\|_{\mathbf{F}} \leq \left\| \mathbf{p}_{\mathbf{h}} \mathbf{u}_{\mathbf{h}} - \mathbf{p}_{\mathbf{h}} \mathbf{r}_{\mathbf{h}} \mathbf{u} \right\|_{\mathbf{F}} + \left\| \mathbf{p}_{\mathbf{h}} \mathbf{r}_{\mathbf{h}} \mathbf{u} - \overline{\omega} \mathbf{u} \right\|_{\mathbf{F}},$$

we see that this converges to 0 as  $h \rightarrow 0$ . The theorem is proved.

#### Remark 3.3.

As announced in Remark 2.2, point (ii), Theorem 3.1 is applicable to the Galerkin approximation of (3.3) used in the proof of Theorem 2.2. One takes  $W_h = W_m, \forall h = m \in \mathcal{H}, \mathcal{H} = \mathbb{N}$ , and one gets as in the example at the end of Section 3.2 a Galerkin approximation of W. With

$$a_{h}(v,w) = a(v,w), \langle \ell_{h},v \rangle = \langle \ell,v \rangle, \forall v,w \in W,$$

Theorem 3.1 is applicable and shows that  $u_m$  converges to u in the strong topology of W as  $m \rightarrow \infty$ .

# Remark 3.4.

If  $W_h$  is a finite dimensional space and  $\{w_h\}_{1 \le i \le N(h)}$  is a basis of  $W_h$ , then the approximate problem (3.6) is equivalent to a regular linear system for the components of  $u_h$  in this basis; i.e., if

$$u_{h} = \sum_{i=1}^{N(h)} \xi_{ih} w_{ih}$$

(3.15)  $\sum_{i=1}^{N(h)} \xi_{ih} a_{h} (w_{ih}, w_{jh}) = \langle \ell_{h}, w_{jh} \rangle, \quad 1 \leq j \leq N(h).$ 

The solution of (3.15) is effected by the usual methods for algebraic linear systems.

When no basis of  $W_h$  can be easily constructed (and this happens sometimes for the Stokes problem), some special method must be found to actually solve (3.6).

# 3.3 <u>Approximation by Finite Differences.</u>

We study the approximation by finite differences of the space  $H_0^1(\Omega)$ , then the same for the space V, and finally the approximation of Stokes problem by the corresponding scheme. The approximation of V considered here will be denoted by (APX1). Notation.

When working with finite differences, h denotes the vector-mesh,  $h = (h_1, \dots, h_n)$  where  $h_i$  is the mesh in the  $x_i$  direction and thus

$$0 \leq h_{i} \leq h_{i}^{0},$$

for some strictly positive numbers  $h_i^0$ ; hence

 $\mathcal{H} = \prod_{i=1}^{n} (0, h_i^0].$ (3.16)

We are interested in passing to the limit  $h \rightarrow 0$ .

For all  $h \in \mathcal{H}$  we define:

i)  $\vec{h}_i$  is the vector  $h_i e_i$ , where  $e_i$  has for j h coordinate  $\delta_{ii}$  = the symbol of Kronecker.

ii)  $\mathcal{R}_h$  is the set of points of  $\mathbb{R}^n$  of the form  $j_1h_1 + \cdots + j_nh_n$ , in which the  $j_i$  are integers of arbitrary sign  $(j_i \in Z)$ .

iii)  $\sigma_{h}(M)$ ,  $M = (\mu_{1}, \dots, \mu_{n})$ , is the set

$$\prod_{i=1}^{n} [\mu_i - \frac{h_i}{2}, \mu_i + \frac{h_i}{2}]$$

and is called a <u>block</u>.

iv)  $\sigma_{h}(M,r)$  is the set  $\bigcup_{1 \le i \le n} \sigma_{h}(M + \frac{\alpha}{2} \vec{h}_{i})$ 

of course  $\sigma_h(M,0) = \sigma_h(M)$ .

v)  $w_{hM}$  is the characteristic function of the block  $\sigma_{h}(M)$ .

vi)  $\delta_{ih}$  (or  $\delta_i$  if no confusion can arise) is the finite difference operator

(3.17) 
$$(\delta_{i}\phi)(x) = \frac{\phi(x + \frac{1}{2}\dot{h}_{i}) - \phi(x - \frac{1}{2}\dot{h}_{i})}{h_{i}}$$

If  $j = (j_1, \dots, j_n) \in \mathbb{N}^n$  is a multi-index, then  $\delta_h^j$  (or simply  $\delta^j$ ) will denote the operator

$$(3.18) \qquad \qquad \delta^{\mathbf{j}} = \delta_1^{\mathbf{j}_1} \cdots \delta_n^{\mathbf{j}_n}.$$

vii) With each open set  $\Omega$  of  $\mathbb{R}^n$  and each non-negative integer r we associate the following point sets



# Examples of Sets $\sigma_h(M,r)$ in the Plane.

viii) Sometimes we will use other finite difference operators such as  $\nabla_{ih}$ and  $\overline{\nabla}_{ih}$  (also denoted  $\nabla_i$  and  $\overline{\nabla}_i$ );

(3.21) 
$$\nabla_{ih}\phi(x) = \frac{\phi(x+\vec{h}_i) - \phi(x)}{h_i}$$

(3.22) 
$$\overline{\nabla}_{ih}\phi(x) = \frac{\phi(x) - \phi(x-h_i)}{h_i}$$

# External Approximation of $H_0^1(\Omega)$ .

Let  $\Omega$  be a lipschitzian open bounded set in  $\mathbb{R}^n$ . Let  $W = H_0^1(\Omega)$ ,  $F = L^2(\Omega)^{n+1}$  equipped with the natural hilbertian scalar product, and let  $\overline{\omega}$  be the mapping

$$(3.23) u \mapsto \overline{\omega} u = (u, D_1 u, \cdots, D_n u)$$

from W into F. It is clear that

$$\|\overline{\omega}\mathbf{u}\| = \|\mathbf{u}\| \\ \mathbf{F} \qquad \mathbb{H}_0^1(\Omega)$$

so that  $\overline{\omega}$  is an isomorphism from W into F.

Space  $W_h$ : With the preceding notation,  $W_h$  will be the space of step functions

(3.24) 
$$u_{h}(x) = \sum_{M \in \widehat{\Omega}_{h}^{1}} u_{h}(M) w_{hM}(x), u_{h}(M) \in \mathbb{R}$$

The functions  $w_{hM}$  for  $M \in \hat{\Omega}_{h}^{1}$  are linearly independent and span the whole space  $W_{h}$ ; they form a basis of  $W_{h}$ . The dimension of  $W_{h}$  is n times the number N(h) of points  $M \in \hat{\Omega}_{h}^{1}$ ;  $W_{h}$  is finite dimensional. This space is provided with the scalar product

(3.25) 
$$\llbracket u_h, v_h \rrbracket_h = \int_{\Omega} u_h(x) v_h(x) dx + \sum_{i=1}^n \int_{\Omega} \delta_i u_h(x) \delta_i v_h(x) dx$$

which makes it a Hilbert space.

The functions  $u_h$  and  $\delta_i u_h$ ,  $1 \leq i \leq n$ , have compact supports in  $\Omega$ , by the definition of  $W_h$  and the set  $\hat{\Omega}_h^1$ . Hence they will be considered as vector functions defined on  $\Omega$  or on  $\mathbb{R}^n$ .

<u>Operators</u>  $p_h$  The prolongation operators  $p_h$  are the discrete analog of  $\overline{\omega}$ :

(3.26) 
$$p_h u_h = (u_h, \delta_1 u_h, \cdots, \delta_h u_h), \quad \forall u_h \in W_h.$$

The norm of p<sub>b</sub> is exactly one,

$$\|\mathbf{p}_{\mathbf{h}}\mathbf{u}_{\mathbf{h}}\|_{\mathbf{F}} = [\mathbf{u}_{\mathbf{h}}]_{\mathbf{h}}$$

and they are stable.

Operators  $r_h$ : As a consequence of Proposition 3.1 we need only define the operator  $r_h$  on  $\mathcal{W} = \mathcal{D}(\Omega)$  which is a dense subspace of  $H_0^1(\Omega)$ ; we put

(3.27) 
$$(\mathbf{r}_{h}\mathbf{u})(\mathbf{M}) = \mathbf{u}(\mathbf{M}), \quad \forall \mathbf{M} \in \hat{\Omega}_{h}^{1}, \quad \forall \mathbf{u} \in \mathfrak{D}(\Omega)$$

which completely defines  $r_h u \in W_h$ .

Proposition 3.2.

The preceding external approximation of  $H_0^1(\Omega)$  is stable and convergent. Proof.

The approximation is stable since the prolongation operators are stable. We must check now conditions (C1) and (C2) of Definition 3.6.

Lemma 3.1.

<u>Condition</u> (C1) is satisfied:  $\forall u \in \mathfrak{D}(\Omega)$ ,

- (3.28)  $r_{\rm h} u \rightarrow u \quad \underline{\rm in} \quad L^2(\Omega),$
- (3.29)  $\delta_i r_h u \longrightarrow D_i u \quad \underline{in} \quad L^2(\Omega),$

as  $h \rightarrow 0$ .

Proof.

Let  $u \in \mathscr{D}(\Omega)$  and let h be sufficiently small so that the support of u is included in the set

(3.30) 
$$\Omega(h) = \bigcup_{M \in \Omega_h^1} \sigma_h(M)$$

For any  $M \in \hat{\Omega}_{h}^{1}$ , for any  $x \in \sigma_{h}(M)$ , the Taylor formula gives (with  $u_{h} = r_{h}u$ )  $|u_{h}(x)-u(x)| = |u(M)-u(x)| \leq c_{1}(u)|M-x| \leq \frac{c_{1}(u)}{2}|h|$ 

where

(3.31) 
$$c_1(u) = \sup_{x \in \Omega} |grad u(x)|$$
  
(3.32)  $|h| = (\sum_{i=1}^n h_i^2)^{\frac{1}{2}}.$ 

Then

(3.33) 
$$\begin{array}{c} \sup_{\mathbf{x} \in \Omega(\mathbf{h})} \left| u_{\mathbf{h}}(\mathbf{x}) - u(\mathbf{x}) \right| \leq \frac{c_1(u)}{2} \left| \mathbf{h} \right|, \end{array}$$

On the set  $\Omega-\Omega(h)$ ,

$$|u(x)| \leq c_1(u)d(x,\Gamma)$$

(3.34)  $|u_h(x)-u(x)| \leq c_1(u)d(\Omega(h),\Gamma)$ .

Hence

(3.35) 
$$\sup_{\mathbf{x}\in\Omega} |\mathbf{u}_{\mathbf{h}}(\mathbf{x})-\mathbf{u}(\mathbf{x})| \leq c_1(\mathbf{u})\{\frac{|\mathbf{h}|}{2}+d(\Omega(\mathbf{h}),\Gamma)\}$$

which converges to 0 as  $h \rightarrow 0$ ;  $u_h$  converges to u in  $\mathbb{L}^{\infty}(\Omega)$  and then in  $L^2(\Omega)$  since  $\Omega$  is bounded.

To prove (3.29), we use again Taylor's formula:  $\forall M \in \hat{\Omega}_h^1$ ,  $\forall x \in \sigma_h(M)$ ,

(3.36) 
$$|\delta_{i}u_{h}(x) - D_{i}u(x)|$$
  
=  $|\frac{1}{h_{i}}[u_{h}(M + \frac{1}{2}\dot{h}_{i}) - u_{h}(M - \frac{1}{2}\dot{h}_{i})] - D_{i}u(x)| \leq c_{2}(u)|h|,$ 

where  $c_2(u)$  depends only on the maximum norm of the second derivatives of u. On the set  $\Omega-\Omega(h)$ ,

$$|D_{\mathbf{f}}\mathbf{u}(\mathbf{x})| \leq c'_{2}(\mathbf{u})d(\Omega(\mathbf{h}),\Gamma)$$

and then on the whole set  $\ \Omega$ 

(3.37) 
$$|\delta_{i}u_{h}(x)-D_{i}u(x)| \leq c_{2}(u)|h| + c'_{2}(u)d(\Gamma).$$

This converges to 0 as  $h \rightarrow 0$ , and shows the convergence of  $\delta_{ih}$  to  $D_{iu}$  in the uniform and  $L^2$  norms.

# Lemma 3.2.

Condition (C2) is satisfied.

#### Proof.

Let there be given a sequence  $u_h \in W_h$ ,  $h' \rightarrow 0$ , such that  $p_h, u_h$ , converges to  $\phi$  in the weak topology of F, as  $h' \rightarrow 0$ . This means

(3.38) 
$$\begin{cases} \lim_{h' \to 0} u_{h'} = \phi_0 \\ \lim_{h' \to 0} \delta_{ih'} u_{h'} = \phi_i, \quad 1 \le i \le n \\ h' \longrightarrow 0 \quad \lim_{h' \to 0} \delta_{ih'} u_{h'} = \phi_i, \quad 1 \le i \le n \end{cases}$$

in the weak topology of  $\mathbb{L}^2(\Omega)$ ;  $\phi = (\phi_0, \dots, \phi_n)$ . As the functions  $u_h$ ,  $\delta_{ih}$ , have compact support in  $\Omega$ , we also have

(3.39) 
$$\begin{cases} \lim_{h' \to 0} u_{h'} = \tilde{\phi}_{0} \\ \lim_{h' \to 0} \delta_{i} u_{h'} = \tilde{\phi}_{i}, \quad 1 \leq i \leq n, \\ h' \to 0 \end{cases}$$

in the weak topology of  $L^2(\mathbb{R}^n)$ ; here  $\tilde{g}$  means the function equal to g in  $\Omega$ and equal to 0 in the complement of  $\Omega$ .

A discrete integration by parts formula gives

(3.40) 
$$\int_{\mathbb{R}^n} \delta_{ih} u_{h'}(x) \sigma(x) dx = - \int_{\mathbb{R}^n} u_{h'}(x) \delta_{ih} \sigma(x) dx,$$

for each  $\sigma \in \mathcal{D}(\mathbb{R}^n)$ 

As  $h' \rightarrow 0$ , the left-hand side of (3.40) converges to

$$\int_{\mathbb{R}^n} \tilde{\phi}_i(\mathbf{x}) \sigma(\mathbf{x}) d\mathbf{x};$$

the right-hand side converges to

$$-\int_{\mathbb{R}^n} \tilde{\phi}_0(\mathbf{x}) D_{\mathbf{i}} \sigma(\mathbf{x}) d\mathbf{x},$$

since  $\delta_{ih}\sigma$  converges to  $D_i\sigma$  in  $L^2(\mathbb{R}^n)$  as shown in Lemma 3.1. Then

$$\int_{\mathbb{R}^n} \tilde{\phi}_i(\mathbf{x}) \sigma(\mathbf{x}) d\mathbf{x} = - \int_{\mathbb{R}^n} \tilde{\phi}_0(\mathbf{x}) D_i \sigma(\mathbf{x}) d\mathbf{x}, \quad \forall \sigma \in \mathcal{D}(\mathbb{R}^n)$$

which amounts saying that

(3.41) 
$$\tilde{\phi}_{i} = D_{i} \tilde{\phi}_{0}, \quad 1 \leq i \leq n,$$

in the distribution sense.

It is clear now that  $\tilde{\phi}_0 \in \mathbb{H}^1(\mathbb{R}^n)$ , and since  $\tilde{\phi}_0$  vanishes in the complement of  $\Omega$ ,  $\phi_0$  belongs to  $\mathbb{H}^1_0(\Omega)$ . Thus  $\phi \in \widetilde{\omega} W$ ;

(3.42) 
$$\phi = \overline{\omega}\phi_0, \quad \phi_0 \in \mathbb{H}^1_0(\Omega).$$

# Discrete Poincaré Inequality.

The following discrete Poincaré inequality (see (1.9)) will allow us to equip the space  $W_h$  in (3.24) with another scalar product  $((\cdot, \cdot))_h$ , the discrete analog of scalar product  $((\cdot, \cdot))$  (see (1.11)).

Proposition 3.3.

Let  $\Omega$  be a set bounded in the  $x_i$  direction, and let  $u_h$  be a scalar step function of type (3.24) (with  $u_h(M) \in \mathbb{R}$ ). Then

$$|u_{h}| \leq 2\mathcal{L}|\delta_{ih}u_{h}|$$

where  $\ell$  is the width of  $\Omega$  in this direction.

### Proof.

For the sake of simplicity we take i = 1. Since  $u_{h}$  has a compact support, for any  $M \! \in \! \mathcal{R}_{h}$  ,

$$\begin{split} u_{h}(M)^{2} &= \sum_{j=0}^{+\infty} \left\{ \left[ u_{h}(M-j\vec{h}_{1}) \right]^{2} - \left[ u_{h}(M-(j+1)\vec{h}_{1}) \right]^{2} \right\} \\ &= h_{1} \sum_{j=0}^{+\infty} \left[ \delta_{1h} u_{h}(M-(j+\frac{1}{2})\vec{h}_{1}) \right] \left[ u_{h}(M-j\vec{h}_{1}) + u_{h}(M-(j+1)\vec{h}_{1}) \right] , \end{split}$$

$$(3.44) \quad u_{h}(M)^{2} \leq I = h_{1} \sum_{j=-\infty}^{+\infty} |\delta_{1h}u_{h}(M-(j+\frac{1}{2})\vec{h}_{1})| [|u_{h}(M-j\vec{h}_{1})|+|u_{h}(M-(j+1)\vec{h}_{1})|].$$

The sums are actually finite. Now for any  $i \in \mathbb{Z}$ ,  $u_h (M+i\dot{h_1})^2$  is majorized by exactly the same expression I. There are less than  $\frac{\lambda}{h_1}$  values of i such that  $u_h(M+i\dot{h_1}) \neq 0$  since the  $x_1$ -width of  $\Omega$  is less than  $\ell$ . Hence

(3.45) 
$$\sum_{i=-\infty}^{+\infty} u_h (M+ih_1)^2 \leq \frac{\ell}{h_1} I.$$

Let  $\mathcal{F}_{h}(M)$  denote the tube  $\bigcup_{i=-\infty}^{+\infty} \sigma_{h}(M+i\vec{h}_{1})$ . We can interpret (3.45) as follows:  $\begin{pmatrix} u_{h}(x)^{2}dx = (h_{1}\cdots h_{n}) & \sum_{i=-\infty}^{+\infty} u_{i}(M+i\vec{h}_{1})^{2} \end{pmatrix}$ 

$$\int_{\mathcal{F}_{h}(M)} u_{h}(x)^{2} dx = (h_{1} \cdots h_{n}) \sum_{i=-\infty} u_{h}(M+ih_{1})^{2}$$

$$\leq \ell(h_{1} \cdots h_{n}) \frac{1}{h_{1}}$$

$$= \ell \int_{\mathcal{F}_{h}(M)} |\delta_{1h}u_{h}(x)| \{ |u_{h}(x+2\dot{h}_{1})| + |u_{h}(x-2\dot{h}_{1})| \} dx.$$

We take summations of the last inequality for all tubes  $\mathcal{F}_{h}(M)$  and obtain

$$\int_{\mathbb{R}^{n}} u_{h}(x)^{2} dx \leq \ell \int_{\mathbb{R}^{n}} |\delta_{1h}u_{h}(x)| \{ |u_{h}(x+\frac{1}{2h_{1}})| + |u_{h}(x-\frac{1}{2h_{1}})| \} dx.$$

Applying Schwarz's inequality, we obtain

$$|u_{h}|^{2} = \int_{\mathbb{R}^{n}} u_{h}(x)^{2} dx \leq \ell |\delta_{1h}u_{h}| \cdot \{2 \int_{\mathbb{R}^{n}} [|u_{h}(x+\frac{1}{2h_{1}})|^{2} + |u_{h}(x-\frac{1}{2h_{1}})|^{2}] dx\}^{\frac{1}{2}}$$
  
$$\leq 2\ell |\delta_{1h}u_{h}| \cdot |u_{h}|$$

and (3.43) follows.

Proposition 3.4.

Let  $\Omega$  be a bounded lipschitzian set. If we equip the space  $W_h$  with the scalar product

(3.46) 
$$((u_h, v_h))_h = \sum_{i=1}^n \int_{\Omega} \delta_{ih} u_h \delta_{ih} v_h dx,$$

we get again a stable convergent approximation of  $H_0^1(\Omega)$ .

Proof.

The prolongation operators are stable as a consequence of Proposition 3.3. Remark 3.5.

Using the difference operators  $\nabla_{ih}$ , or  $\overline{\nabla}_{ih}$ , or even any "reasonable" approximation of the differentiation operator  $\frac{\partial}{\partial x_i}$ , one can define many other similar approximations of the space  $H_0^1(\Omega)$ . The modifications arise then in (3.25) where  $\delta_{ih}$  is replaced by  $\nabla_{ih}$ , ...; in the set of points  $\hat{\Omega}_h^1$  which must be suitably defined, and in some points of the proof of Lemmas 3.1 and 3.2.

The same Poincaré inequality is valid for the operators  $\nabla_{ih}$  and  $\overline{\nabla}_{ih}$  but not for more general operators.

Remark 3.6.

When  $\Omega$  is unbounded, one can define an external approximation of  $H_0^1(\Omega)$  with a space  $W_h$  consisting of either:

- step functions  $\sum_{M \in \Omega_h^1} \lambda_M^w_{hM}$ , which have compact support (we restrict the sum

sum to a finite number of points  $M \in \hat{\Omega}_{h}^{1}$ ),

- or step functions  $\sum_{M} \lambda_{M} w_{hM}$  for the M in the intersection of  $\hat{\Omega}_{h}^{1}$  with some "large" ball:  $|\mathbf{x}| \leq \rho(h)$ , where  $\rho(h) \longrightarrow +\infty$  as  $h \longrightarrow 0$ .

In the second case  $W_h$  is finite dimensional but not in the first case.

Without any modification for the other elements of the approximation, one can see that we get a stable convergent approximation of  $H_0^1(\Omega)$  for an unbounded locally lipschitzian set  $\Omega$ .

The discrete Poincaré inequality is available if  $\Omega$  is bounded in one of the directions  $x_1, \dots, x_n$ .

#### Approximation of the Space V (APX1).

Let  $\Omega$  be a lipschitzian bounded set in  $\mathbb{R}^n$  and let  $\mathcal{V}$  be the usual space (1.12) and V its closure in  $\mathbb{H}^1_0(\Omega)$ .

We define now an approximation of V using finite differences (which will be denoted by (APX1)).

Let  $F = L^2(\Omega)^{n+1}$  equipped with the natural hilbertian scalar product and let us define the mapping  $\overline{\omega} \in \chi(V,F)$ :

$$\mathbf{u} \rightarrow \widetilde{\omega}\mathbf{u} = (\mathbf{u}, \mathbf{D}_1\mathbf{u}, \cdots, \mathbf{D}_n\mathbf{u}),$$

It is clear that

$$\|\overline{\omega}u\|_{\mathbf{F}} = [[u]]$$

so that  $\overline{\omega}$  is an isomorphism from V into F.

Space  $V_h$ . We take the following space  $V_h$  as the space  $W_h$ :  $V_h$  is the space of step functions

(3.47) 
$$u_{h}(x) = \sum_{M \in \widehat{\Omega}_{h}^{1}} u_{h}(M) w_{hM}(x), u_{h}(M) \in \mathbb{R}^{n},$$

which are discretely divergence free in the following sense:

(3.48) 
$$\sum_{i=1}^{n} \nabla_{ih} u_{h}(M) = 0, \quad \forall M \in \hat{\Omega}_{h}^{1}$$

which amounts to saying

(3.49) 
$$\sum_{i=1}^{n} \nabla_{ih} u_{h}(x) = 0, \quad \forall x \in \Omega(h).$$

No basis of  $V_h$  is available; it is clear that  $V_h$  is a finite dimensional space with dimension less than or equal to nN(h)-N(h) = (n-1)N(h) since all the functions in (3.47) form a space of dimension nN(h) and there are at most N(h) independent linear constraints in (3.48); it is not clear whether the constraints (3.48) are always linearly independent so that  $V_h$  is not necessarily of dimension (n-1)N(h).

The space  $V_h$  is equipped with one of the scalar products

(3.50) 
$$((\mathbf{u}_{\mathbf{h}},\mathbf{v}_{\mathbf{h}}))_{\mathbf{h}} = \sum_{i=1}^{n} \int_{\Omega} \delta_{i} \mathbf{u}_{\mathbf{h}}(\mathbf{x}) \cdot \delta_{i} \mathbf{u}_{\mathbf{h}}(\mathbf{x}) d\mathbf{x}$$

(3.51) 
$$[ u_h, v_h]_h = \int_{\Omega} u_h(x) v_h(x) dx + ((u_h, v_h))_h.$$

Because of Proposition 3.3 (discrete Poincaré inequality),  $\rm V_h$  equipped with either one of these scalar products is a Hilbert space.

Operators  $p_h$ . These are the discrete analog of  $\overline{\omega}$ :

$$(3.52) p_h u_h = \{u_h, D_i u_h, \cdots, D_n u_h\}$$

These operators are stable since by (3.43)

(3.53) 
$$\| \mathbf{p}_{h} \mathbf{u}_{h} \|_{\mathbf{F}} = \| \mathbf{u}_{h} \|_{h} \leq c \| \mathbf{u}_{h} \|_{h}, \quad \forall \mathbf{u}_{h} \in \mathbf{V}_{h}.$$

Operator  $r_h$  We only define  $r_h u$  for  $u \in \mathcal{V}$ , a dense subspace of V;  $\forall M = (m_1 h_1, \cdots, m_h h_n) \in \hat{\Omega}_h^1$ ,  $(r_h u) (M) = u_h(M)$  is defined by

(3.54) 
$$u_{ih}(M) = i^{th} \text{ component of } u_h$$
  
= the average value of  $u_i$  on the face  
 $x_i = (m_i - \frac{1}{2})h_i$  of  $\sigma_h(M)$ .

This complicated definition of  $r_{\rm h}\,u$  is necessary if we want  $u_{\rm h}$  to belong to  $V_{\rm h}\,$  ; actually

# <u>Lemma 3.3</u>. <u>For each</u> $u \in V$ , $r_h u \in V_h$

# Proof.

One has

$$\nabla_{ih} u_{ih}(M) = \frac{1}{h_1 \cdots h_n} \left\{ \int_{\Sigma_i} u_i(x) dx - \int_{\Sigma'_i} u_i(x) dx \right\}$$

where  $\Sigma_{i}$  and  $\Sigma'_{i}$  respectively are the faces  $x_{i} = (m_{i} + \frac{1}{2})h_{i}$  and  $x_{i} = (m_{i} - \frac{1}{2})h_{i}$  of  $\sigma_{h}(M)$ .

This gives also

$$\nabla_{ih} u_{ih}(M) = \frac{1}{h_1 \cdots h_n} \int_{\Sigma_i \cup \Sigma'_i} u \cdot \nu d\Gamma_i$$

where  $\nu$  stands for the unit vector normal to the boundary of  $\sigma_h(M)$  and pointing in the outward direction. Then, for each  $M \in \hat{\Omega}_h^1$ ,

$$\sum_{i=1}^{n} \nabla_{ih} u_{ih}(M) = \frac{1}{h_1 \cdots h_n} \int_{\partial \sigma_h(M)} u \cdot v d\Gamma$$
  
= (by Stokes formula)  
$$= \frac{1}{h_1 \cdots h_n} \int_{\sigma_h(M)} div \ u \ dx = 0, \text{ since } div \ u = 0$$

Conditions (3.48) are met.

Proposition 3.5. The preceding external approximation of V is stable and convergent. Proof.

Stability has been shown already.

The proof of condition (C1) is very similar to the proof of Lemma 3.1 and we do not repeat all the details; for example, for  $x \in \sigma_h(M)$ ,  $M \in \hat{\Omega}_h^1$ ,

$$|u_{ih}(x)-u_{i}(x)| = |u_{ih}(\xi)-u_{i}(x)|,$$

where  $\xi$  is some point of the face  $x_i = (m_i - \frac{1}{2})h_i$  of  $\sigma_h(M)$ , and hence

The proof for condition (C2) is similar to the proof of Lemma 3.2; more precisely, the same proof as for Lemma 3.2 shows that if

$$p_h, u_h \xrightarrow{} \phi$$
 in the weak topology of F,

as  $h' \longrightarrow 0$ , then  $\phi = \overline{\omega} u = (u, D_1 u, \dots, D_n u)$ , where  $u \in H_0^1(\Omega)$ . Because of Theorem 1.6, proving that  $u \in V$  amounts now to proving that div u = 0. This follows from (3.49) as we now show. Let  $\sigma$  be any test function in  $\mathcal{D}(\Omega)$ , and let us suppose that h is small enough that the support of  $\sigma$  is included in  $\Omega(h)$ ; then (3.49) shows that

$$\int_{\Omega} (\sum_{i=1}^{n} (\nabla_{ih} u_{ih})(x)) \sigma(x) dx = 0$$

or

(3.56) 
$$\int_{\mathbb{R}^n} \left( \sum_{i=1}^n \left( \nabla_{ih} u_{ih} \right)(x) \right) \sigma(x) dx = 0.$$

It is easy to check the discrete integration by parts formula

(3.57) 
$$\int_{\mathbb{R}^n} (\nabla_{ih} \theta)(x) . \sigma(x) dx = - \int_{\mathbb{R}^n} \theta(x) (\overline{\nabla}_{ih} \sigma)(x) dx;$$

then (3.56) becomes

(3.58) 
$$\int_{\mathbb{R}^n} \sum_{i=1}^n [u_{ih}(x), (\overline{\nabla}_{ih}\sigma)(x)] dx = 0.$$

With a proof similar to that of Lemma 3.1 (based on Taylor's formula) we see that

$$(3.59) \qquad \overline{\nabla}_{ih} \sigma \rightarrow -D_i \sigma, \text{ as } h \rightarrow 0$$

in the (uniform and)  $L^2$  norm. Since  $u_{ih}$  converges to  $u_i$  for the weak topology of  $L^2(\mathbb{R}^n)$ , letting  $h \rightarrow 0$  in (3.58) gives the result

(3.60) 
$$\int_{\mathbb{R}^n} \left[ \sum_{i=1}^n u_i(x) \cdot D_i \sigma(x) \right] dx = 0.$$

The equality (3.60), true for any  $\sigma \in \mathcal{D}(\Omega)$ , implies that div u = 0 and then  $\phi = \overline{\omega}u$ , with  $u \in V$ .

Remark 3.7.

The Remark 3.5 can be extended to the present case with, however, one restriction: condition (3.48) - (3.49) in the definition of the spaces  $V_h$  cannot be replaced by similar relations involving other finite difference operators; for example, it seems impossible to replace (3.49) by

$$\sum_{i=1}^{n} \delta_{ih} u_{ih}(x) = 0,$$

since (3.61) requires many more algebraic relations than (3.49) and probably too many relations.

#### Remark 3.8.

When  $\Omega$  is unbounded one can define, by using the methods mentioned in Remark 3.5, a stable and convergent external approximation of the space W introduced in (2.31).

# Approximation of Stokes Problem.

Using the preceding approximation of V and the results of Section 3.2, we can propose a finite difference scheme for the approximation of Stokes problem. Let us take, for (3.6),

(3.62) 
$$a_{h}(u_{h},v_{h}) = v((u_{h},v_{h}))_{h}$$

(3.63) 
$$< \ell_h, v_h^> = (f, v_h),$$

where  $V_h$  and  $((\cdot, \cdot))_h$  are the space and scalar product just defined, and  $\nu$  and f are given as in Section 2.1.

The approximate problem corresponding to (2.6) is then:

Proof.

Stability has been shown already.

The proof of condition (C1) is very similar to the proof of Lemma 3.1 and we do not repeat all the details; for example, for  $x \in \sigma_{h}(M)$ ,  $M \in \hat{\Omega}_{h}^{1}$ ,

$$|u_{ih}(x)-u_{i}(x)| = |u_{ih}(\xi)-u_{i}(x)|,$$

where  $\xi$  is some point of the face  $x_i = (m_i - \frac{1}{2})h_i$  of  $\sigma_h(M)$ , and hence

The proof for condition (C2) is similar to the proof of Lemma 3.2; more precisely, the same proof as for Lemma 3.2 shows that if

$$p_{h'}u_{h'} \rightarrow \phi$$
 in the weak topology of F,

as  $h' \rightarrow 0$ , then  $\phi = \overline{\omega}u = (u, D_1 u, \dots, D_n u)$ , where  $u \in H_0^1(\Omega)$ . Because of Theorem 1.6, proving that  $u \in V$  amounts now to proving that div u = 0. This follows from (3.49) as we now show. Let  $\sigma$  be any test function in  $\mathcal{P}(\Omega)$ , and let us suppose that h is small enough that the support of  $\sigma$  is included in  $\Omega(h)$ ; then (3.49) shows that

$$\int_{\Omega} \left( \sum_{i=1}^{n} (\nabla_{ih} u_{ih})(x) \right) \sigma(x) dx = 0$$

or

(3.56) 
$$\int_{\mathbb{R}^n} \left( \sum_{i=1}^n \left( \nabla_{ih} u_{ih} \right)(x) \right) \sigma(x) dx = 0.$$

It is easy to check the discrete integration by parts formula

(3.57) 
$$\int_{\mathbb{R}^{n}} (\nabla_{ih} \theta)(x) \cdot \sigma(x) dx = - \int_{\mathbb{R}^{n}} \theta(x) (\overline{\nabla}_{ih} \sigma)(x) dx;$$

then (3.56) becomes

(3.58) 
$$\int_{\mathbb{R}^n} \sum_{i=1}^n [u_{ih}(x).(\overline{\nabla}_{ih}\sigma)(x)] dx = 0.$$

With a proof similar to that of Lemma 3.1 (based on Taylor's formula) we see that

$$(3.59) \qquad \qquad \overline{\nabla}_{ih} \sigma \longrightarrow -D_i \sigma, \text{ as } h \longrightarrow 0,$$

in the (uniform and)  $L^2$  norm. Since  $u_{ih}$  converges to  $u_i$  for the weak topology of  $L^2(\mathbb{R}^n)$ , letting  $h \rightarrow 0$  in (3.58) gives the result

(3.60) 
$$\int_{\mathbb{R}^n} \left[ \sum_{i=1}^n u_i(x) \cdot D_i \sigma(x) \right] dx = 0.$$

The equality (3.60), true for any  $\sigma \in \mathcal{D}(\Omega)$ , implies that div u = 0 and then  $\phi = \overline{\omega}u$ , with  $u \in V$ .

Remark 3.7.

The Remark 3.5 can be extended to the present case with, however, one restriction: condition (3.48) - (3.49) in the definition of the spaces  $V_h$  cannot be replaced by similar relations involving other finite difference operators; for example, it seems impossible to replace (3.49) by

(3.61) 
$$\sum_{i=1}^{n} \delta_{ih} u_{ih}(x) = 0,$$

since (3.61) requires many more algebraic relations than (3.49) and probably too many relations.

### Remark 3.8.

When  $\Omega$  is unbounded one can define, by using the methods mentioned in Remark 3.5, a stable and convergent external approximation of the space W introduced in (2.31).

#### Approximation of Stokes Problem.

Using the preceding approximation of y and the results of Section 3.2, we can propose a finite difference scheme for the approximation of Stokes problem. Let us take, for (3.6),

(3.62) 
$$a_{h}(u_{h},v_{h}) = v((u_{h},v_{h}))_{h}$$

(3.63) 
$$< l_h, v_h^> = (f, v_h),$$

where  $V_h$  and  $((\cdot, \cdot))_h$  are the space and scalar product just defined, and  $\nu$  and f are given as in Section 2.1.

The approximate problem corresponding to (2.6) is then:

(3.64) 
$$\begin{cases} \frac{\text{To find}}{(u_h, v_h)} & \frac{u_h \in V_h}{(u_h, v_h)} & \frac{\text{such that}}{(f, v_h)}, \quad \forall v_h \in V_h. \end{cases}$$

# Proposition 3.6.

For all  $h \in \mathcal{H}$ , the solution  $u_h$  of (3.64) exists and is unique; moreover as  $h \rightarrow 0$  the solution  $u_h$  of (3.64) converges to the solution u of (2.6) in the following sense:

$$(3.65) u_h \xrightarrow{} u \text{ in } \mathbb{L}^2(\Omega),$$

(3.66) 
$$\delta_{ih} u_{h} \longrightarrow D_{i} u \quad in \quad \mathbb{L}^{2}(\Omega).$$

Proof.

We only have to check that Theorem 3.1 is applicable. Condition (3.4) is obvious ( $\alpha_0 = 1$ ); for (3.5) we notice that

$$\begin{split} |\langle \boldsymbol{\mathfrak{l}}_{h}, \boldsymbol{v}_{h}^{}\rangle| &= |(\boldsymbol{\mathtt{f}}, \boldsymbol{v}_{h}^{})| \leq |\boldsymbol{\mathtt{f}}| . |\boldsymbol{v}_{h}^{}| \\ &\leq (by \text{ the discrete Poincaré inequality}) \\ &\leq c(\Omega) |\boldsymbol{\mathtt{f}}| ||\boldsymbol{v}_{h}^{}||_{h}, \quad \forall \boldsymbol{v}_{h}^{} \in \boldsymbol{\mathtt{V}}_{h}. \end{split}$$

Hence

$$\|\boldsymbol{\ell}_{h}\|_{sh} \leq c(\Omega) \|\boldsymbol{f}\|,$$

and (3.5) is satisfied.

For (3.7) - (3.8) we notice that

$$p_h v_h \longrightarrow \overline{\omega} v$$
 weakly (resp.  $p_h w_h \longrightarrow \overline{\omega} w$  strongly)

means

$$v_h \longrightarrow v$$
, and  $\delta_{ih} v_h \longrightarrow D_i v$  in  $\mathbb{L}^2(\Omega)$  weakly,

(resp.

$$w_h \rightarrow w$$
, and  $\delta_{ih} w_h \rightarrow D_i w$  in  $\mathbb{L}^2(\Omega)$  strongly),

and it is clear that this implies,

$$(\delta_{ih}v_{h}, \delta_{ih}w_{h}) \longrightarrow (D_{i}v, D_{i}w)$$
$$((v_{h}, w_{h}))_{h} \longrightarrow ((v, w)),$$
$$(f, v_{h}) \longrightarrow (f, v).$$

Approximation of the Pressure.

We want to present the "approximate" pressure which is implicitly contained in (3.64) as the exact pressure p is implicitly contained in (2.6).

The space  $V_h$  in (3.47) is a subspace of the space  $W_h$  in (3.24); namely, the space of  $v_h \in W_h$  satisfying the linear constraint (3.68).

The form  $v_h \mapsto v((u_h, v_h))_h - (f, v_h)$  appears as a linear form defined on  $W_h$  which vanishes on  $V_h$ . Hence introducing the Lagrange multipliers corresponding to the linear constraints (3.48) we find, with the aid of a classical theorem of linear algebra, that there exist numbers  $\lambda_M \in \mathbb{R}, M \in \hat{\Omega}_h^1$ , such that the equation

(3.68) 
$$\nu((u_h, v_h))_h - (f, v_h) = \sum_{M \in \hat{\Omega}_h^1} \lambda_M \sum_{i=1}^n (\nabla_{ih} v_{ih}(M)),$$

holds for each  $v_h \in W_h$ .

Let us now introduce the operator  $D_h \in \mathcal{K}(W_h, L^2(\Omega))$ :

(3.69) 
$$D_h v_h(x) = \sum_{i=1}^n \nabla_{ih} v_{ih}(x), \quad \forall v_h \in W_h,$$

its adjoint  $D_h^* \in \mathcal{K}(L^2(\Omega), W_h)$  is defined by

(3.70) 
$$(D_h^{*\theta}, v_h) = (\theta, D_h v_h) \quad \forall v_h \in W_h, \forall \theta \in L^2(\Omega).$$

Let  $\pi_h$  be the step function which vanishes outside  $\Omega(h) = \bigcup_{M \in \Omega_h^1} \sigma_h(M)$ , and which satisfies

(3.71) 
$$\pi_{h}(x) = \pi_{h}(M) = \frac{\lambda_{M}}{h_{1}\cdots h_{n}}, \quad \forall x \in \sigma_{h}(M), M \in \hat{\Omega}_{h}^{1}.$$

Then (3.68) can be written as

$$v((u_h, v_h))_h - (f, v_h) = (\pi_h, D_h v_h),$$

or equivalently,

(3.72) 
$$\nu((u_h, v_h))_h - (D_h^* \pi_h, v_h) = (f, v_h), \quad \forall v_h \in W_h.$$

Taking successively  $v_h = w_{hM}e_j$  for  $M \in \hat{\Omega}_h^1$ ,  $j = 1, \dots, n$ , we can interpret (3.72) as

(3.73) 
$$-\nu \sum_{i=1}^{n} \delta_{ih}^{2} u_{h}(M) + (\overline{\nabla}_{h} \pi_{h})(M) = f_{h}(M), \quad M \in \hat{\Omega}_{h}^{1}$$

where  $\overline{\mathbb{V}}_{h}(\pi_{h}(M))$  is the vector  $(\overline{\mathbb{V}}_{1h}\pi_{h}(M), \cdots, \overline{\mathbb{V}}_{nh}\pi_{h}(M))$  and

(3.74) 
$$f_{h}(M) = \frac{1}{h_{1}\cdots h_{n}} \int_{\sigma_{h}(M)} f(x) dx.$$

The equations (3.73), and

(3.75) 
$$\sum_{i=1}^{n} (\nabla_{ih} u_{ih}) (M) = 0, \quad M \in \mathring{\Omega}_{h}^{1},$$

are the discrete form of the equations (2.7) - (2.8);  $-D_h^*$  is the "approximation" of p,  $-D_h^*$  is a discrete gradient operator.

Remark 3.9.

As indicated in Remark 3.4, the solution of (3.64) is not easy since we do not know any simple basis of  $V_h$ . One possiblility for solving (3.64) would be to solve the system (3.73), (3.75), which is a linear system with unknowns

$$u_{1h}(M), \cdots, u_{nh}(M); \pi_{h}(M), M \in \mathring{\Omega}_{h}^{1}.$$

This system has a unique solution up to an additive constant for the  $\pi_h(M)$ ; this non-uniqueness makes the resolution of this linear system difficult; moreover, the matrix of the system is ill-conditionned.

More efficient ways for actually computing the approximate solution will be given in Section 5.

#### The Error.

Let us suppose that the exact solution satisfies  $u \in \mathbb{C}^3(\overline{\Omega})$  and  $p \in \mathcal{C}^2(\overline{\Omega})$ . Then by using Taylor's formula, we have

(3.76) 
$$-\nu \sum_{i=1}^{n} (\delta_{ih}^{2} r_{h} u) (M) - (\overline{\nabla}_{h} p) (M) = f(M) + \varepsilon_{h} (M)$$

where  $r_h^u$  is the function of  $V_h^u$  defined by (3.54) and where  $\varepsilon_h^u(M)$  is a "small" vector:

$$|\varepsilon_{h}(M)| \leq c(u,p)|h|,$$

c(u,p) depending only on the maximum norms of third derivatives of u and second derivatives of p. Let us denote by  $\pi'_h$  the function  $\sum_{\substack{M \in \Omega_h^1 \\ h}} p(M) w_{hM}$ . Then,

the equality (3.76) is equivalent to

(3.78) 
$$\nu((\mathbf{r}_{h}\mathbf{u},\mathbf{v}_{h}))_{h} + (\pi^{*}_{h},\mathbf{D}_{h}\mathbf{v}_{h}) = (\mathbf{f}+\varepsilon_{h},\mathbf{v}_{h})$$

for each  $v_h \in W_h$  (space (3.24)) and implies

for each  $v_h \in V_h$ .

Subtracting this equality from (3.64) we get

and then taking  $v_h = u_h - r_h u$ , we see that

$$\begin{array}{l} & \left\| u_{h} - r_{h} u \right\|_{h}^{2} = (\varepsilon_{h}, u_{h} - r_{h} u) \\ & \leq c(\Omega, u, p) \left\| h \right\| \left\| u_{h} - r_{h} u \right\|_{h}. \end{array}$$

Hence we find the following estimates for the discrete error:

(3.81) 
$$\|u_{h} - r_{h}u\|_{h} \leq \frac{1}{\nu} c(\Omega, u, p) |h|$$
  
(3.82)  $|u_{h} - r_{h}u| \leq \frac{1}{\nu} c'(\Omega, u, p) |h|.$ 

### \$4. DISCRETIZATION OF STOKES EQUATIONS (II)

We study here the discretization of Stokes equations by means of finite element methods. The results are less general here than in the previous section and vary according to the dimension. We successively consider conforming finite elements which are first piecewise polynomials of degree two in the two-dimensional case (Section 4.2), next piecewise polynomials of degree three in the three-dimensional case (Section 4.3), and then piecewise polynomials of degree four in the twodimensional case (Section 4.4). Finally we consider an external approximation by non-conforming finite elements (any dimension) in Section 4.5.

#### 4.1 Preliminary Results.

We will have to work with peicewise polynomial functions defined on n-simplices. For that purpose, we recall here some definitions and introduce some notations adapted to the situation.

#### Barycentric Coordinates.

Let there be given in  $\mathbb{R}^n$ , (n+1) points  $A_1, \dots, A_{n+1}$ , (1) with coordinates  $a_{1,i}, \dots, a_{n,i}, 1 \leq i \leq n+1$ , which do not lie in the same hyperplane; this amounts saying that the n vectors  $A_1A_2, \dots, A_1A_{n+1}$  are independent, or that the matrix

(4.1) 
$$\mathcal{A} =$$

$$\begin{vmatrix} a_{1,1} & a_{1,2} & \cdots & a_{1,n+1} \\ a_{2,1} & a_{2,2} & \cdots & a_{2,n+1} \\ \vdots & & & \vdots \\ a_{n,1} & a_{n,2} & a_{n,n+1} \\ 1 & 1 & 1 \end{vmatrix}$$

is non-singular. Given any point  $P \in \mathbb{R}^n$ , with coordinates  $x_1, \dots, x_n$ , there exists (n+1) real numbers

$$\lambda_i = \lambda_i(P), \quad 1 \leq i \leq n+1$$

such that

(4.2) 
$$OP = \sum_{i=1}^{n+1} \lambda_i OA_i,$$
  
(4.3)  $\sum_{i=1}^{n+1} \lambda_i = 1,$ 

<sup>(1)</sup> In this section dealing with finite elements, the capital letters A,B,M,P,..., (sometimes with subscripts) will denote points of the affine space R<sup>n</sup>. Couples of such letters, like AB,..., denote the vector of R<sup>n</sup> with origin A and terminal point B.
where 0 is the origin of  $\mathbb{R}^n$ .

To see this it suffices to remark that (4.2)(4.3) are equivalent to the linear system

(4.4) 
$$\mathcal{A} \begin{pmatrix} \lambda_{1} \\ \vdots \\ \lambda_{n} \\ \lambda_{n+1} \end{pmatrix} = \begin{pmatrix} x \\ \vdots \\ \vdots \\ x_{n} \\ 1 \end{pmatrix}$$

which has a unique solution since the matrix  $\mathcal{A}$  is non-singular, by hypothesis. The quantities  $\lambda_i$  are called the <u>barycentric coordinates of P</u>, with respect to the (n+1) points  $A_1, \dots, A_{n+1}$ . As a consequence of (4.4), the numbers  $\lambda_i$  appear as linear, generally nonhomogeneous functions of the coordinates  $x_1, \dots, x_n$  of P:

(4.5) 
$$\lambda_{i} = \sum_{j=1}^{n} b_{i,j} x_{j} + b_{i,n+1}, \quad 1 \le i \le n+1,$$

where the matrix  $\mathcal{B} = (b_{i,j})$  is the inverse of the matrix  $\mathcal{A}$ . It is easy to see that the point 0 in (4.2) can be replaced by another point of  $\mathbb{R}^n$  without changing the value of the barycentric coordinates; hence

(4.6) 
$$\sum_{i=1}^{n+1} \lambda_i PA_i = 0.$$

The barycentric coordinates are also clearly independent of the choice of basis in  $\mathbb{R}^n$ .

The convex hull of the (n+1) points  $A_i$  is exactly the set of points of  $R^n$  with barycentric coordinates satisfying the conditions:

$$(4.7) 0 \leq \lambda_i \leq 1, \quad 1 \leq i \leq n+1.$$

This convex hull  $\mathcal{N}$  is the <u>n-simplex generated by the points</u>  $A_i$ , which are <u>called the vertices of the n-simplex</u>. The <u>barycenter</u> G of  $\mathcal{N}$  is the point of  $\mathcal{N}$  whose barycentric coordinates are all equal and hence equal  $\frac{1}{n+1}$ . An m-<u>dimensional face</u> of  $\mathcal{N}$  is any m-simplex  $(1 \leq m \leq n-1)$  generated by m+1 of the vertices of  $\mathcal{N}$  (of course these vertices do not lie in a (m-1)-dimensional subspace of  $\mathbb{R}^n$ ). A <u>1-dimensional face is an edge</u>.

In the two-dimensional case (n=2) the 2-simplices are triangles; the vertices and edges of the simplex are simply the vertices and edges of the triangle. In the three dimensional case, the 3-simplices are tetrahedrons, the two-faces

are the four triangles which form its boundary.

An Interpolation Result.

Proposition 4.1.

Let  $A_1, \dots, A_{n+1}$  be (n+1) points of  $\mathbb{R}^n$  which are not included in a hyperplane. Given (n+1) real numbers  $\alpha_1, \dots, \alpha_{n+1}$ , there exists one and only linear function u such that  $u(A_i) = \alpha_i$ ,  $1 \le i \le n+1$ , and

(4.8) 
$$u(P) = \sum_{i=1}^{n+1} \alpha_i \lambda_i(P), \quad \forall P \in \mathbb{R}^n,$$

where the  $\lambda_i(P)$  are the barycentric coordinates of P with respect to  $A_1, \dots, A_{n+1}$ .

Proof. Let

$$u(x) = \sum_{j=1}^{n} \beta_{j} x_{j} + \beta_{n+1},$$

be this function. The unknowns are  $\beta_1, \dots, \beta_{n+1}$  which satisfy the following equations asserting that  $u(A_i) = \alpha_i$ :

$$\sum_{j=1}^{n} \beta_{j}a_{j,i} + \beta_{n+1} = \alpha_{i}, \quad 1 \le i \le n+1.$$

The matrix of the system is the transposed matrix  ${}^{t}\mathcal{A}$  of  $\mathcal{A}$  and thus the function u exists and is unique.

It remains to see that (4.8) is the required function; actually

$$u(A_j) = \alpha_j, \quad 1 \leq j \leq n+1,$$

since  $\lambda_i(A_i) = \delta_{ii}$  = the Kronecker symbol, for each i and j.

#### Remark 4.1.

Higher order interpolation formulas using the barycentric coordinates will be given later (see Sections 4.2, 4.3, 4.4).

#### Differential Properties.

We give some differential properties of the  $\lambda_i$  considered as functions of the cartesian coordinates  $x_1, \dots, x_n$ , of P; here we denote by D the gradient operator  $D = (D_1, \dots, D_n)$ .

$$\frac{\text{Lemma}}{(4.9)} \frac{4.1}{\sum_{i=1}^{n+1} D\lambda_i = 0}$$

(4.10) 
$$D\lambda_{i}(P) \cdot PA_{j} = \delta_{ij} - \lambda_{i}(P), 1 \leq i, j \leq n+1.$$

Proof.

The identity (4.3) immediately implies (4.9). According to (4.5)

$$\frac{\partial \lambda_{i}}{\partial x_{k}} = b_{i,k}, \ 1 \le i \le n+1, \ 1 \le k \le n,$$

and then

$$D\lambda_{i}(P) \cdot PA_{j} = D\lambda_{i}(P) \cdot OA_{j} - D\lambda_{i}(P) \cdot OP$$
$$= \sum_{k=1}^{n} b_{i,k}a_{k,j} - \sum_{k=1}^{n} b_{i,k}x_{k}$$
$$= \sum_{k=1}^{n} b_{i,k}a_{k,j} + b_{i,n+1} - \lambda_{i}$$
$$= \delta_{ij} - \lambda_{i};$$

for the last equality we note that  $\mathcal{B} = \mathcal{Q}^{-1}$ .

<u>Lemma 4.2</u>.

Let  $\mathcal{D}$  be an n-simplex with vertices  $A_1, \dots, A_{n+1}$  and let  $\rho'$  be the least upper bound of the diameters of all balls included in  $\mathcal{D}$ . Then

(4.11) 
$$|D\lambda_{i}| \leq \frac{1}{\rho'}, 1 \leq i \leq n+1,$$

where  $|D\lambda_i|$  is the euclidian norm of the constant vector  $D\lambda_i$ . <u>Proof</u>. We have

$$|D\lambda_{i}| = \sup_{\substack{\mathbf{x} \in \mathbb{R}^{n} \\ |\mathbf{x}|=1}} D\lambda_{i} \cdot \mathbf{x}.$$

Any vector x, with norm 1, can be written as

$$x = \frac{1}{\rho'} PQ$$
,

where P and Q belong to  $\mathcal{L}$ ; denoting by  $\mu_1, \dots, \mu_{n+1}$ , the barycentric

coordinates of Q with respect to  $A_1, \dots, A_{n+1}$ , we have because of (4.2)-(4.3),

$$PQ = \sum_{j=1}^{n+1} \mu_j PA_j,$$
  
$$\prod_{j=1}^{n+1} \mu_j = 1.$$
  
$$j=1$$

Then

$$D\lambda_{i} \cdot \mathbf{x} = \frac{1}{\rho'} (D\lambda_{i}) \cdot (\sum_{j=1}^{n+1} \mu_{j} PA_{j})$$
$$= \frac{1}{\rho'} \sum_{j=1}^{n+1} \mu_{j} D\lambda_{i} \cdot PA_{j}$$

= (according to (4.10))

$$= \frac{1}{\rho'} \sum_{j=1}^{n+1} \mu_j (\delta_{ij} - \lambda_i) = \frac{1}{\rho'} (\mu_i - \lambda_i).$$

Since P and Q belong to  $\Delta$ ,  $0 \le \lambda_i \le 1$ ,  $0 \le \mu_i \le 1$  for each i,  $1 \le i \le n+1$ , and then  $-1 \le \mu_i - \lambda_i \le 1$ , so that

$$|D\lambda_i \cdot x| \leq \frac{1}{\rho'}$$

and (4.11) follows.

# Norms of Some Linear Transformations.

Let & and  $\overline{\&}$  be two n-simplices with vertices  $A_1, \dots, A_{n+1}$ , and  $\overline{A_1}, \dots, \overline{A_{n+1}}$ . We denote by  $\rho$  (resp.  $\rho$ ') the diameter of the smallest ball containing & (resp. the diameter of largest ball contained in &); with a similar meaning for  $\overline{\rho}$ ,  $\overline{\rho'}$ .

We can suppose that, up to a translation,  $A_1 = \overline{A}_1 = 0$ , the origin in  $\mathbb{R}^n$ , and we denote then by  $\Lambda$  the linear mapping in  $\mathbb{R}^n$  such that

(4.13)  $A_i = \Lambda \overline{A}_i, \quad 2 \leq i \leq n+1.$ 

The norms of  $\Lambda$  and  $\Lambda^{-1}$  can be majorized as follows in terms of  $\rho, \rho', \overline{\rho, \rho'}$ :

Lemma 4.3.

(4.14)  $\|\Lambda\| \leq \frac{\rho}{\rho}, \|\Lambda^{-1}\| \leq \frac{\rho}{\rho^{\dagger}}$ 

Proof.

As in the proof of Lemma 4.2, let x be any vector in  $\mathbb{R}^n$  with norm 1. Then

$$x = \frac{1}{\rho'} PQ,$$

where P and Q belong to  $\mathcal{A}$ . It is clear that

$$\Lambda x = \frac{1}{\rho'} \overline{PQ},$$

where  $\overline{P} = \Lambda P$ ,  $\overline{Q} = \Lambda Q$ . But  $\overline{P}$  and  $\overline{Q}$  belong to  $\overline{\Delta}$  too, since

$$OP = \sum_{i=1}^{n+1} \lambda_i OA_i, \quad 0 \le \lambda_i \le 1,$$

implies

$$AOP = \sum_{i=1}^{n+1} \lambda_i \quad O\overline{A}_i,$$

so that the barycentric coordinates of  $\overline{P}$  with respect to  $\overline{A}_1, \dots, \overline{A}_{n+1}$ , are the same as the barycentric coordinates of P with respect to  $A_1, \dots, A_{n+1}$ . Hence  $|\overline{PQ}| < \overline{\rho}$ , and

$$|\Lambda \mathbf{x}| \leq \frac{\overline{\rho}}{\rho'}$$
.

The first inequality (4.14) is proved. The second inequality is obvious when interchanging the role of  $\Delta$  and  $\overline{\Delta}$ .

When handling divergence free vector functions, the following lemma will be useful:

Lemma 4.4.

Let  $x \mapsto u(x)$  be a divergence free vector function defined on  $\mathcal{A}$  (or on  $\mathbb{R}^n_x$ ) and let  $\overline{x \mapsto u(x)}$  be defined on  $\overline{\mathcal{A}}$  by

(4.15)  $\overline{u(x)} = \Lambda u(\Lambda^{-1}\overline{x}), \quad \forall \overline{x} \in \overline{\lambda}(\text{or } \mathbb{R}^{n}_{\overline{x}}).$ 

Then  $\overline{u}$  is a divergence free vector function too. Proof.

Let  $(\alpha_{ij})$  and  $(\beta_{kl})$  denote the elements of  $\Lambda$  and  $\Lambda^{-1}$ . Then

$$\frac{\partial u_{i}}{\partial \overline{x}_{j}} (\overline{x}) = \frac{\partial}{\partial \overline{x}_{j}} \sum_{\ell} \alpha_{i\ell} u_{\ell} (\Lambda^{-1} \overline{x})$$
$$= \sum_{\ell,k} \alpha_{i\ell} \frac{\partial u_{\ell}}{\partial x_{k}} \cdot \frac{\partial x_{k}}{\partial \overline{x}_{j}}$$
$$= \sum_{\ell,k} \alpha_{i\ell} \beta_{kj} \frac{\partial u_{\ell}}{\partial x_{k}} (\Lambda^{-1} \overline{x})$$

and

$$(\operatorname{div} \overline{u})(\overline{x}) = \sum_{i} \frac{\partial \overline{u}_{i}}{\partial \overline{x}_{i}} (\overline{x}) = \sum_{i,k,\ell} \alpha_{i\ell} \beta_{kj} \frac{\partial u_{\ell}}{\partial x_{k}} (\Lambda^{-1} \overline{x})$$
$$= \sum_{k} \frac{\partial u_{k}}{\partial x_{k}} (\Lambda^{-1} \overline{x}) = 0.$$

Regular Triangulations of an Open Set Ω.

Let  $\Omega$  be an open bounded set in  $\mathbb{R}^n$ .

Let  $\mathcal{C}_h$  be a family of n-simplices; such a family will be called an admissible triangulation of  $\Omega$  if the following conditions are satisfied:

(4.16) 
$$\Omega(h) = \bigcup_{h \in \mathcal{N} \subset \Omega} h$$

(4.17) If  $\mathcal{A}$  and  $\mathcal{A} \in \mathcal{T}_h$ , then  $\mathcal{A} \cap \mathcal{A}' = \phi$ , (where  $\mathcal{A}$  is the interior of  $\mathcal{A}^{(1)}$ ) and, either  $\mathcal{A} \cap \mathcal{A}'$  is empty or  $\mathcal{A} \cap \mathcal{A}'$  is exactly a whole m-face for both  $\mathcal{A}$  and  $\mathcal{A}'$  (any m,  $0 \le m \le n-1$ ).

We will denote by  $\{\Im_h\}_{h\in\mathcal{H}}$  the family of all admissible triangulations of  $\Omega$ ; with each admissible triangulation  $\Im_h$  we associate the following three numbers

(4.18) 
$$\rho(h) = \sup_{\substack{h \in \mathcal{T}_h \\ b \in \mathcal{T}_h}} \rho_{b}$$
(4.19) 
$$\rho'(h) = \inf_{\substack{h \in \mathcal{T}_h \\ b \in \mathcal{T}_h}} \rho_{b}$$

(4.20) 
$$\sigma(h) = \sup_{\substack{h \in \mathcal{T} \\ h}} \frac{\rho_{\lambda}}{\rho_{\lambda}}$$

(1) i.e., the points of  $\mathcal{N}$  with barycentric coordinates, with respect to the vertices of  $\mathcal{N}$ , satisfying  $0 < \lambda_i < 1$ ,  $1 \leq i \leq n+1$ . where, as before,  $\rho = \rho_{\infty}$  is the diameter of the smallest ball containing A, and  $\rho' = \rho'_{\infty}$  is the diameter of the greatest ball contained in A.

For finite element methods, we are concerned with passage to the limit  $\rho(h) \rightarrow 0$ . It will appear later that some restrictions on  $\sigma(h)$  are necessary to get convergent approximations.

A subfamily of the family of admissible triangulations  $\{\widehat{J}_h\}_{h\in\mathcal{N}}$  will be called a regular triangulation of  $\Omega$  if  $\sigma(h)$  remains bounded as  $\rho(h) \longrightarrow 0$ ,

(4.21)  $\sigma(h) \leq \alpha < +\infty, \rho(h) \longrightarrow 0$ 

and  $\Omega(h)$  converges to  $\Omega$  in the following sense:

(4.22) For each compact set 
$$K \subseteq \Omega$$
, there exists  $\delta = \delta(K) > 0$  such that  
 $\rho(h) < \delta(K) \Longrightarrow \Omega(h) \supset K$ .

 $\Re_{\alpha}$  will denote the set of admissible triangulations of  $\Omega$  satsifying (4.21) and (4.22).

Remark 4.2.

In the two dimensional case the 2-simplex is a triangle and it is known that

$$\frac{1}{2 \tan \frac{\theta}{2}} \le \frac{\rho_{\infty}}{\rho'_{\infty}} \le \frac{2}{\sin \theta}$$

where  $\theta$  is the smallest angle of  $\lambda$ .

The condition (4.21) amounts then to saying that the smallest angle of all the triangles  $\Delta \in \mathcal{T}_h$  remains bounded from below

 $(4.23) \qquad \qquad \theta \ge \theta_0 > 0.$ 

Our purpose now will be to associate to a regular family of triangulations  $\{\mathcal{T}_h\}_{h\in\mathcal{H}_\alpha}$  of  $\Omega$ , various types of approximations of the function spaces we are concerned with.

## 4.2 Finite Elements of Degree 2(n=2).

Let  $\Omega$  be a lipschitzian open bounded set in  $\mathbb{R}^n$ . We describe an internal approximation of  $\mathbb{H}^1_0(\Omega)$  (any n) and then an external approximation of V (n=2 only). The approximate functions are piecewise polynomials of degree 2.

Let  $\mathfrak{T}_{\mathbf{h}}$  be any admissible triangulation of  $\,\Omega\,.\,$ 

# Space Wh

This is the space of continuous vector functions, which vanish outside  $\Omega(h)$ 

(4.24) 
$$\Omega(h) = \bigcup_{\substack{b \in \mathcal{T} \\ h}} \mathcal{N}$$

and whose components are polynomials of degree two (1) on each simplex  $\Delta \in \mathbb{T}_{h}$ .

This space  $W_h$  is a finite dimensional subspace of  $\mathbb{H}^1_0(\Omega)$ . We equip it with the scalar product induced by  $\mathbb{H}^1_{\Omega}(\Omega)$ :

(4.25) 
$$((u_h, v_h))_h = ((u_h, v_h)), \quad \forall u_h, v_h \in W_h$$

A Basis of  $W_h$ . If  $\lambda$  is an n-simplex we denote as before by  $A_1, \dots, A_{n+1}$ , the vertices of  $\mathcal{L}$ ; we denote also by  $A_{ij}$  the mid-point of  $A_i A_j$ .

We first have

Lemma 4.5.

A polynomial of degree less than or equal to two is uniquely defined by its values at the points  $A_{i,A_{ij}}$ ,  $1 \le i,j \le n+1$  (the vertices and the mid-points of the edges of an n-simplex .

Moreover, this polynomial is given in terms of the barycentric coordinates with respect to  $A_1, \dots, A_{n+1}$ , by the formula:

(4.26) 
$$\phi(\mathbf{x}) = \sum_{i=1}^{n+1} (2(\lambda_i(\mathbf{x}))^{2} - \lambda_i(\mathbf{x}))\phi(\mathbf{A}_i) + 2\sum_{\substack{i,j=1\\i\neq j}}^{n+1} \lambda_i(\mathbf{x})\lambda_j(\mathbf{x})\phi(\mathbf{A}_{ij}).$$

#### Proof.

Let us show first that (4.26) satisfies the requirements. The function on the right-hand side of (4.26) is a polynomial of degree two since the  $\lambda_i(x)$ are linear nonhomogeneous functions of  $x_1, \dots, x_n$  (see (4.5)). Besides, if  $\psi(x)$ denotes this function,

<sup>(1)</sup> Roughly speaking, a polynomial of degree two means a polynomial of degree less than or equal to two.

$$\psi(A_{k}) = \phi(A_{k}) \text{ since } \lambda_{i}(A_{k}) = \delta_{ik}$$
$$\psi(A_{kl}) = \phi(A_{kl}) \text{ since } \lambda_{i}(A_{kl}) = \frac{\delta_{ik} + \delta_{il}}{2}$$

Thus  $\psi$  is as required.

Now, a polynomial of degree two has the form

(4.27) 
$$\phi(\mathbf{x}) = \alpha_0 + \sum_{i=1}^{n} (\alpha_i \mathbf{x}_i + \beta_i \mathbf{x}_i^2) + \sum_{\substack{i,j=1 \\ i < j}}^{n} \alpha_{ij} \mathbf{x}_i \mathbf{x}_j.$$

and  $\phi$  is defined by  $\frac{(n+1)(n+2)}{2}$  unknown coefficients  $\alpha_0, \alpha_i, \beta_i, \alpha_{ij}$ . There are (n+1) points  $A_i, \frac{n(n+1)}{2}$  points  $A_{ij}$ , and hence the conditions on  $\phi$ : (4.28)  $\phi(A_i) = given, \phi(A_{ij}) = given,$ 

are  $\frac{(n+1)(n+2)}{2}$  linear equations for the unknown coefficients. According to (4.26) this system has a solution for any set of data in (4.28); then the linear system is a regular system <sup>(1)</sup>, and the solution found in (4.26) is unique.

Now let us denote by  $\mathcal{U}_h$  the set of vertices and mid-edges of the n-simplices  $\mathcal{A} \in \mathcal{T}_h$ . We denote also by  $\mathcal{U}_h$  those points of  $\mathcal{U}_h$  which belong to the interior of  $\Omega(h)$ . According to the preceding lemma there is at most one function  $u_h$  in  $\mathcal{W}_h$  which takes given values at the points  $A \in \mathcal{U}_h$ . Actually we have more.

Lemma 4.6.

There exists one and only one function  $u_h \stackrel{\text{in } W}{=} h \stackrel{\text{which takes given}}{\overset{\text{values at the points}}{\longrightarrow}} M \in \hat{\mathcal{U}}_h$ .

#### Proof.

We saw that such a function is necessarily unique. Now, by Lemma 4.5, there exists a function  $u_h$  whose components are piecewise polynomials of degree two, which takes given values at the points  $M \in \mathcal{U}_h$ , and vanish at the points  $M \in \mathcal{U}_h \cap \mathcal{U}_h$  and outside  $\Omega(h)$ . We just have to check that this function is continuous. On each (n-1)-face  $\Delta'$  of a simplex  $\Delta \in \mathcal{T}_h$ , each component  $u_{ih}$ 

<sup>(1)</sup> We use the well-known property that, in finite dimensional spaces, the linear operators which are onto, are one-to-one and onto.

of  $u_h$  is a polynomial of degree two which has two (perhaps different) values  $u_{ih}^+$  and  $u_{ih}^-$ . But  $u_{ih}^+$  and  $u_{ih}^-$  are polynomials of degree less than or equal to two in (n-1) variables, which are equal at the vertices and the mid-points of the edges of  $\mathcal{N}$ ; then Lemma 4.5 applied to a (n-1) dimensional simplex shows that  $u_{ih}^+ = u_{ih}^-$  on  $\mathcal{N}$ . Therefore  $u_h$  is continuous, and  $u_h$  belongs to  $W_h$ .

Repeating the argument of the preceding proof, we see that there exists a unique scalar continuous function, which is a polynomial of degree two on each simplex  $\mathcal{D} \in \mathcal{T}_h$ , and which takes on given values at the points  $M \in \mathfrak{V}_h$ , and vanishes outside  $\Omega(h)$ . Let us denote by  $w_{hM}$  the function of this type defined by

Finally, we have

Lemma 4.7.

<u>The functions</u>  $W_{hM}e_i$ ,  $M \in \mathcal{N}_h$ ,  $i = 1, \dots, n$ , form a basis of  $W_h$ , and the dimension of  $W_h$  is nN(h) where N(h) is the number of points in  $\mathcal{N}_h$ .

## Proof.

These functions are linearly independent, and, clearly, each function  $\mathbf{u}_{h} \in \mathbf{W}_{h}$  can be written

$$u_{h}(x) = \sum_{M \in \mathcal{U}_{h}} \sum_{i=1}^{n} u_{ih}(M) e_{i} w_{hM}(x)$$

or

(4.30) 
$$u_{h} = \sum_{M \in \mathcal{O}_{h}} u_{h}(M) w_{hM}.$$

Operator  $p_h$ . The prolongation operator  $p_h$  is the identity,

$$(4.31) p_h u_h = u_h, \quad \forall u_h \in W_h.$$

The ph's have norm one and thus are stable.

Operator  $r_h$ . We define  $r_h u$  for  $u \in D(\Omega)$ ; we set:

(4.32)  $(\mathbf{r}_{h}\mathbf{u})(\mathbf{M}) = \mathbf{u}(\mathbf{M}), \quad \forall \mathbf{M} \in \mathfrak{U}_{h}.$ 

Proposition 4.2.

The preceding internal approximation of  $\mathbb{H}_0^1(\Omega)$  is stable and convergent, provided h belongs to a regular triangulation  $\mathcal{H}_{\alpha}$  of  $\Omega$ .

Proof.

We only have to prove that for each  $u \in \mathcal{D}(\Omega)$ ,

$$p_h r_h u \rightarrow u$$
 in  $\mathbb{H}^1_0(\Omega)$ ,

as  $\rho(h) \longrightarrow 0$ ,  $h \in \mathcal{H}_{\alpha}$ .

If h is sufficiently small,  $\Omega(h)$  contains the support of u, and then the next lemma will show that

(4.33) 
$$\| \mathbf{p}_{h} \mathbf{r}_{h} \mathbf{u} - \mathbf{u} \| \leq c(\mathbf{u}) \rho^{2}(\mathbf{h}) \cdot \sigma(\mathbf{h}) \leq c(\mathbf{u}) \alpha \rho^{2}(\mathbf{h}),$$

and the result follows.

Lemma 4.8.

Let  $\mathcal{D}$  be an n-simplex,  $\phi$  a scalar function in  $\mathcal{C}^3(\mathcal{D})$ , and let  $\phi$  be the interpolating polynomial of degree two such that,

$$\tilde{\phi}(A_{i}) = \phi(A_{i}), \quad \tilde{\phi}(A_{ij}) = \phi(A_{ij})$$

for  $1 \le i, j \le n+1$ .

Then, we have

(4.34) 
$$\sup_{\mathbf{x}\in\mathcal{D}} |\phi(\mathbf{x}) - \tilde{\phi}(\mathbf{x})| \leq c(\phi)\rho_{\mathbf{x}}^{3}$$

(4.35) 
$$\sup_{\mathbf{x} \in \mathcal{D}} \left| \frac{\partial \phi}{\partial \mathbf{x}_{\mathbf{i}}} (\mathbf{x}) - \frac{\partial \widetilde{\phi}}{\partial \mathbf{x}_{\mathbf{i}}} (\mathbf{x}) \right| \leq c(\phi) \frac{\rho^{3}}{\rho'_{\mathcal{D}}}$$

where  $c(\phi)$  depends on the maximum norm of the third derivatives of  $\phi$ .

This lemma is a particular case of general theorems concerning polynomial interpolation on a simplex in connection with finite elements.

# Polynomial Interpolation on a Simplex.

Let  $\gg$  be an n-simplex and let  $\mathcal{E}$  be a finite set of points having the following property: for any family of given numbers  $\gamma_M \in \mathbb{R}$ ,  $M \in \mathcal{E}$ , there exists a unique polynomial p of degree less than or equal to k such that

$$(4.36) p(M) = \gamma_M, \quad \forall M \in \mathcal{E}.$$

75

Such a set  $\xi$  is called k-unisolvent by Ciarlet-Raviart [1]; for example, according to Proposition 4.1 and Lemma 4.5, the vertices  $A_1, \dots, A_{n+1}$  of  $\mathcal{D}$  are 1-unisolvent, the points  $A_i, A_{ij}, 1 \leq i, j \leq n+1$ , are 2-unisolvent.

Let us denote by p, the polynomial of degree k such that

(4.37) 
$$p_{i}(M_{i}) = 1, p_{i}(M_{j}) = 0, M_{j} \neq M_{i}, M_{j} \in \mathcal{E}.$$

Then the polynomial p in (4.36) can be written as

$$(4.38) p = \sum_{M_i \in \mathcal{E}} \gamma_{M_i} p_i$$

Now let us suppose that a function  $\phi$  is given,  $\phi \in e^{k+1}(\Sigma)$ , and let  $\tilde{\phi}$  be the interpolating polynomial of degree k defined by

$$(4.39) \qquad \qquad \qquad \widetilde{\phi} (M) = \phi(M), \ \forall M \in \mathcal{E}$$

i.e.,

(4.40) 
$$\widetilde{\phi} = \sum_{M_i \in \mathcal{E}} \phi(M_i) p_i.$$

Using Taylor's formula it is proved that for any multi-index  $j = (j_1, \dots, j_n)$ with  $[j] = j_1 + \dots + j_n \leq k$ , one has

(4.41) 
$$D^{j}\tilde{\phi}(P) = D^{j}\phi(P)$$
  
+  $\frac{1}{(k+1)!}\sum_{M_{i}\in \xi}\sum_{\lfloor \ell \rfloor = k+1} \{D^{\ell}\phi(P_{i})\cdot M_{i}P^{\ell}\}D^{j}p_{i}(P),$ 

where  $P_i$  is some point of the open interval  $(M_i, P)$ ,

$$\mathbf{D}^{\ell} = \mathbf{D}_{1}^{\ell_{1}} \cdots \mathbf{D}_{n}^{\ell_{n}}; \mathbf{M}_{i}\mathbf{P}^{\ell} = \varepsilon_{1i}^{\ell_{1}} \cdots \varepsilon_{ni}^{\ell_{n}}$$

for  $M_i P = (\varepsilon_{1i}, \cdots, \varepsilon_{ni}), \quad \ell = (\ell_1, \cdots, \ell_n).$ 

The error between  $\phi$  and  $\delta$  is majorized on  $\mathcal{A}$  by

(4.42) 
$$\sup_{\mathbf{x} \in \mathcal{D}} |D^{\mathbf{j}}\phi(\mathbf{P}) - D^{\mathbf{j}}\phi(\mathbf{P})| \leq cn_{k+1}(\phi) \frac{\rho_{\mathcal{D}}^{k+1}}{\rho_{\mathcal{D}}^{m}}$$

for  $[j] = j_1 + \cdots + j_n = m \le k$ , where  $\rho_{\infty}$  and  $\rho'_{\infty}$  are defined in Section 4.1<sup>(1)</sup>.

(1)  

$$n_{k}(\phi) = \sup \sup \{ |D^{j}\phi(x)| \}.$$

$$x [j]=k$$

The supremum in x is taken on  $\mathcal{D}$ ; elsewhere when using this notation, the supremum is understood on the whole support of  $\phi$ .

This is a consequence of (4.41) and the following estimation of  $p_{i}$ 

(4.43) 
$$\sup_{\mathbf{x}\in\mathcal{Y}} |D^{j}p_{\mathbf{i}}(\mathbf{x})| \leq \frac{c}{\rho_{\mathcal{W}}^{im}} \text{ for } [\mathbf{j}] = \mathbf{m} \leq \mathbf{k}.$$

For the proofs of (4.41) and (4.43), the reader is referred to Ciarlet-Raviart [1]; for the particular case of Lemma 4.8, see also Ciarlet-Wagshall [1].

## Approximation of V (APK 2)

Here  $\Omega$  is an open bounded set in  $\mathbb{R}^2$ ; we shall define an external approximation of the space V.

Space F, Operator W.

The space F is  $\mathbb{H}^1_0(\Omega)$  and  $\overline{\omega}$  is the identity

$$(4.44) \qquad \overline{\omega}u = u, \quad \forall u \in V;$$

 $\overline{\omega}$  is an isomorphism from V into F.

Let  $\boldsymbol{\mathfrak{T}}_h$  be any admissible triangulation of  $\,\Omega.$ 

## Space V<sub>h</sub>.

 $V_h$  is a subspace of the space  $W_h$  previously defined. It is the space of continuous vector functions which vanish outside

and whose components are polynomials of degree two on each simplex  ${\overset{\scriptstyle{}}{\sim}} \in {\overset{\scriptstyle{}}{\mathbb{T}}}_h$  such that

(4.46) 
$$\int_{\mathcal{W}} \operatorname{div} u_{h} dx = 0, \quad \forall \mathcal{W} \in \mathcal{T}_{h}.$$

The condition (4.46) is a discrete form of the condition div u = 0. The functions  $u_h \in V_h$  belong to  $\mathbb{H}_0^1(\Omega)$ , but not to V,  $V_h \notin V$ . We do not have a simple basis of  $V_h$ ; according to Lemma 4.7, any function  $u_h \in V_h$  can be written as

$$u_h = \sum_{M \in \mathcal{N}_h} u_h^{(M)} w_{hM}$$

but the functions  $w_{hM}$  do not belong to  $V_h$ . Lemma 4.7 and (4.46) shows also that

dim 
$$V_h \leq 2N(h) - N'(h)$$
,

where N(h) is the number of points in  $\mathcal{N}_h$  and N'(h) is the number of triangles  $\mathcal{A} \in \mathcal{T}_h$ .

We provide the space  $V_h$  with the scalar product of  $\mathbb{H}_0^1(\Omega)$  (as  $W_h$ ):

(4.47) 
$$((u_h, v_h))_h = ((u_h, v_h)).$$

Operator ph

The operator  $p_h$  is the identity (recall that  $V_h \subset \mathbb{H}^1_0(\Omega)$ ). The prolongation operators have norm one and are thus stable.

#### Operator r.

The restriction operators are more difficult to define because of condition (4.46) which must be satisfied by  $r_{\rm h}^{\rm u}$ .

Let u be an element of  $\mathcal{V}$ ; we set

(4.48) 
$$r_h u = u_h = u_h^1 + u_h^2$$
,

where  $u_h^1$  and  $u_h^2$  belong separately to  $W_h$ ;  $u_h^1$  is defined like in (4.32) by (4.49)  $u_h^1(M) = u(M), \forall M \in \mathcal{O}_{\tilde{h}}.$ 

There is no reason for  $u_h^1$  to belong to  $V_h$ , and actually  $u_h^2$  will be a "small corrector" so that  $u_h^{1+}u_h^2 \in V_h$ . We define  $u_h^2$  by its values at the points  $M \in \mathcal{V}_h^1$ ; if  $M = A_i$  is the vertex of a triangle then  $u_h^2(A_i) = 0$ ; if  $M = A_{ij}$  is the mid-point of an edge, then, letting  $v_{ij}$  denote one of the two unit vectors orthogonal to  $A_i A_i$ , we set:

$$(4.50) \begin{cases} u_{h}^{2}(A_{ij}) \cdot A_{i}A_{j} = 0 \\ u_{h}^{2}(A_{ij}) \cdot v_{ij} \\ = -\{u_{h}^{1}(A_{ij}) + \frac{1}{4} u_{h}^{1}(A_{i}) + \frac{1}{4} u_{h}^{1}(A_{j})\} v_{ij} + \frac{3}{2} \int_{0}^{1} u(tA_{i} + (1-t)A_{j}) v_{ij} dt \end{cases}$$

Lemma 4.9.

$$u_h \frac{\text{defined by } (4.48)(4.49)(4.50)}{\text{belongs to } V_h}$$
.

Proof.

The main idea in (4.50) was to choose  $u_h^2$  so that

(4.51) 
$$\int_{A_{i}}^{A_{j}} u_{h} \cdot v_{ij} d\ell = \int_{A_{i}}^{A_{j}} u \cdot v_{ij} d\ell.$$

If we show that (4.51) is satisfied, we will then have for any triangle  $\mathcal{A}$ :

$$\int_{\mathcal{N}} \operatorname{div} u_{h} \, \mathrm{dx} = \int_{\partial \mathcal{N}} u_{h} \cdot \mathrm{vdl} = \int_{\partial \mathcal{N}} u \cdot \mathrm{vdl} = \int_{\mathcal{N}} \operatorname{div} u \, \mathrm{dx} = 0,$$

since  $u \in \mathcal{V}$  (v = unit vector normal to  $\partial \mathcal{D}$  pointing outward with respect to  $\mathcal{D}$ ). Let us prove (4.51). The function  $u_h^2$  is equal to

(4.52) 
$$u_{h}^{2} = \sum_{\substack{M \in \mathcal{W} \\ M = A_{k} g_{h}}} u_{h}^{2} (M) w_{hM}$$

On the segment  $\overline{A_iA_j}$ , the function  $w_{hA_{ij}}$  is the only function  $w_{hM}$  in the preceding sum which is not identically equal to 0. By the definition of  $w_{hA_{ij}}$  one easily checks that

(4.53) 
$$w_{hA_{ij}}(tA_i+(1-t)A_j) = 4t(1-t), 0 < t < 1.$$

Likewise

$$u_{h}^{1} = \sum_{M \in \mathcal{O}_{h}} u_{h}^{1} \quad (M)w_{hM}$$

where the only functions  $w_{hM}$  which do not vanish on  $A_{ij}A_{j}$  are  $w_{hA_{j}}, w_{hA_{j}}, w_{hA_{ij}}$ . It is easily shown that

(4.54) 
$$\begin{cases} w_{hA_{i}}(tA_{i}+(1-t)A_{j}) = (t-1)(2t-1) \\ w_{hA_{j}}(tA_{i}+(1-t)A_{j}) = t(2t-1). \end{cases}$$

Then

$$\frac{1}{|A_{i}A_{j}|} \int_{A_{i}}^{A_{j}} u_{h}(x) \cdot v_{ij} d\ell = \int_{0}^{1} u_{h}(tA_{i}+(1-t)A_{j}) \cdot v_{ij} dt$$

$$= \frac{2}{3} u_{h}^{2}(A_{ij}) \cdot v_{ij} + \frac{2}{3} u_{h}^{1}(A_{ij}) \cdot v_{ij} + \frac{1}{6} \{u_{h}^{1}(A_{i})+u_{h}^{1}(A_{j})\} \cdot v_{ij}$$

$$= (by (4.50))$$

$$= \int_{0}^{1} u(tA_{i}+(1-t)A_{j}) \cdot v_{ij} dt = \frac{1}{|A_{i}A_{j}|} \int_{A_{i}}^{A_{j}} u(x) \cdot v_{ij} d\ell.$$

The lemma is proved.

Proposition 4.3.

The preceding external approximation of V is stable and convergent, provided h belongs to a regular triangulation  $\mathcal{H}_{\alpha}$  of  $\Omega$ .

Proof.

Let us check first the condition (c2) of Definition 3.6.

We have to show that if a sequence  $p_h, u_h, u_h \in V_h$ , converges weakly to  $\phi$  in F, then  $\phi = u \in V$ . According to Theorem 1.6, we need only show that

(4.55) div u = 0.

Let  $\theta$  be any function of  $\mathcal{D}(\Omega)$ ; by (4.46), we have

(4.56) 
$$\int_{\Omega} (\operatorname{div} u_{h}) \theta_{h} dx = 0,$$

where  $\theta_h$  is the step function which is equal on each  $\Delta \in \mathcal{T}_h$  to the average value of  $\theta$  on  $\Delta$ , and which vanishes outside  $\Omega(h)$ . It is easy to see that when support  $\theta \in \Omega(h)$ ,

 $\sup_{\mathbf{x}\in\Omega} |\theta_{h}(\mathbf{x}) - \theta(\mathbf{x})| \leq c(\theta)\rho_{h},$ 

so that  $\theta_h$  converges to  $\theta$  in the  $L^{\infty}$  and  $L^2$  norms; thus we can pass to the limit in (4.56) and obtain

$$\int_{\Omega} \operatorname{div} \mathbf{u} \cdot \boldsymbol{\theta} \, \mathrm{dx} = \mathbf{0}, \quad \forall \boldsymbol{\theta} \in \mathcal{D}(\Omega)$$

This proves (4.55).

The condition (c1) of Definition 3.6 is

(4.57) 
$$\lim_{h \to 0} p_h r_h u = \overline{\omega} u, \quad \forall u \in \mathcal{V};$$

This is equivalent to

(4.58) 
$$\lim_{h \to 0} \|\mathbf{u} - \mathbf{r}_h \mathbf{u}\| = 0, \quad \forall \mathbf{u} \in \mathcal{V}.$$

Let us suppose that  $\rho_h$  is sufficiently small so that  $\Omega(h)$  contains the support of u. Because of Lemma 4.8 and (4.42), on each triangle  $\mathcal{LET}_h$ :

(4.59) 
$$\begin{cases} \sup_{\mathbf{x} \in \mathcal{N}} |u(\mathbf{x}) - u_{h}^{1}(\mathbf{x})| \leq \operatorname{cn}_{3}(u) \rho_{\mathcal{N}}^{3} \\ \sup_{\mathbf{x} \in \mathcal{N}} |D_{1}u(\mathbf{x}) - D_{1}u_{h}^{1}(\mathbf{x})| \leq \operatorname{cn}_{3}(u) \frac{\rho_{\mathcal{N}}^{3}}{\rho_{\mathcal{N}}^{3}} \end{cases}$$

By the proof of Lemma 4.9,

(4.60) 
$$\frac{2}{3} u_{h}^{2} (A_{ij}) \cdot v_{ij} = \frac{1}{|A_{i}A_{j}|} \int_{A_{i}}^{A_{j}} u_{h}^{2} (x) \cdot v_{ij} dt$$
$$= \frac{1}{|A_{i}A_{j}|} \int_{A_{i}}^{A_{j}} [u(x) - u_{h}^{1} (x)] \cdot v_{ij} dt.$$
$$= \int_{0}^{1} (u - u_{h}^{1}) (tA_{i} + (1 - t)A_{j}) \cdot v_{ij} dt$$

Hence, with (4.60) estimated by (4.59),

(4.61) 
$$|u_h^2(A_{ij})| = |u_h^2(A_{ij}) \cdot v_{ij}| \leq cn_3(u)\rho_{\diamond}^3$$

Now by (4.43), we obtain

(4.62)  

$$\begin{cases}
\sup_{x \in \mathcal{A}} |w_{hM}(x)| \leq c \\
\sup_{x \in \mathcal{A}} |D_{i}w_{hM}(x)| \leq \frac{c}{\rho' \mathcal{A}}, \quad i = 1, \cdots, n \\
\text{Next, combining (4.61)} - (4.62) \text{ with (4.52), we get}
\end{cases}$$

$$\begin{split} \sup_{\mathbf{x} \in \mathcal{D}} & \left| u_{h}^{2}(\mathbf{x}) \right| \leq \operatorname{cn}_{3}(\mathbf{u}) \rho_{\mathcal{D}}^{3} \\ & \mathbf{x} \in \mathcal{D} \\ \\ & \operatorname{Sup} & \left| D_{1} u_{h}^{2}(\mathbf{x}) \right| \leq \operatorname{cn}_{3}(\mathbf{u}) \frac{\rho_{\mathcal{D}}^{3}}{\rho_{\mathcal{D}}^{*}} \end{split}$$

Finally, combining (4.59) and last inequalities, it follows that

(4.63) 
$$\begin{cases} \sup_{\mathbf{x}\in\Omega} |\mathbf{u}(\mathbf{x})-\mathbf{u}_{h}(\mathbf{x})| \leq \operatorname{cn}_{3}(\mathbf{u})\rho(h)^{3} \\ \sup_{\mathbf{x}\in\Omega} |\mathbf{D}_{\mathbf{i}}\mathbf{u}(\mathbf{x})-\mathbf{D}_{\mathbf{i}}\mathbf{u}_{h}(\mathbf{x})| \leq \operatorname{cn}_{3}(\mathbf{u})\rho(h)^{2}\sigma(h) \leq \operatorname{cn}_{3}(\mathbf{u})\alpha\rho(h)^{2}. \end{cases}$$

The proof is achieved.

Remark 4.3.

If  $\Omega$  is a polygon, it is possible to choose the triangulation  $\mathcal{T}_h$  such that  $\Omega(h) = \Omega$ , and this is usually done in practical computations. In this case we can extend the preceding computation to any  $u \in \mathbb{H}^1_0(\Omega) \cap \mathbb{C}^3(\overline{\Omega})$  and we find

(4.64) 
$$||u-r_{h}u|| < cn_{3}(u)\sigma(h)\rho(h)^{2}$$
.

### Approximation of Stokes Problem.

Using the preceding approximation of V and Section 3.2 we can propose a finite element scheme for the approximation of a two-dimensional Stokes problem. Let us take in (3.6)

(4.65) 
$$a_h(u_h, v_h) = v((u_h, v_h)),$$

$$(4.66) \qquad < \ell_h, v_h > = (f, v_h)$$

where  $\nu$  and f are given as in Section 2.1 (see Theorem 2.1).

The approximate problem (3.6) is then

(4.67) 
$$\begin{cases} \frac{\text{To find}}{v((u_h, v_h))} &= (f, v_h), \forall v_h \in V_h. \end{cases}$$

The solution  $u_{\rm b}$  of (4.67) exists and is unique; moreover, we have

#### Proposition 4.4.

If  $\rho(h) \longrightarrow 0$ , with  $\sigma(h) \le \alpha$  (i.e.,  $h \in \mathcal{H}_{\alpha}$ ), the solution  $u_h$  of (4.67) converges to the solution u of (2.6) in the  $\mathbb{H}_0^1(\Omega)$  norm.

#### Proof.

It is easy to see that Theorem 3.1 is applicable, and the conclusion gives exactly the convergence result announced.

#### Approximation of the Pressure.

We introduce the approximation of the pressure, as in Section 3.3. The form

$$v_h \mapsto v((u_h, v_h)) - (f, v_h)$$

is a linear form on  $W_h$ , which vanishes on  $V_h$ . Since  $V_h$  is characterized by the set of linear constraints (4.46), we know that there exists a family of

numbers  $\lambda_{\lambda}$ ,  $\Delta \in \mathcal{T}_{h}$ , which are the Lagrange multipliers associated with the constraints (4.46), such that

$$\forall ((u_h, v_h)) - (f, v_h) = \sum_{w \in \mathcal{Q}_h} \lambda_w (\int_{\mathcal{W}} div v_h dx), \forall v_h \in W_h.$$

Let  $\chi_{h,b}$  denote the characteristic function of b and let  $\pi_h$  denote the step function

(4.69) 
$$\begin{cases} \pi_{h} = \sum_{\substack{\lambda \in \mathcal{X} \\ h}} \pi_{h}(\mathcal{W}) \chi_{n,\mathcal{W}} \\ \pi_{h}(\mathcal{W}) = \frac{\lambda_{\mathcal{W}}}{(\text{meas }\mathcal{W})} \end{cases}$$

1

We then have

(4.69) 
$$\nu((u_h, v_h)) - (\pi_h, \text{ div } v_h) = (f, v_h), \forall v_h \in W_h,$$

which is the discrete analog of equation

(4.70) 
$$v((u,v)) - (p, \operatorname{div} v) = (f,v), \quad \forall v \in \mathbb{H}_0^1(\Omega).$$

## Remark 4.4.

Since no basis of  $V_h$  is available, the solution of (4.67) is not easy. The computation can be effected by the algorithms studied in Section 5.

# The Error between u and uh.

Let us suppose that  $\Omega$  has a polygonal boundary  $(\Omega \subset \mathbb{R}^2)$  and that  $u \in \mathfrak{C}^3(\overline{\Omega})$  and  $p \in \mathfrak{C}^1(\overline{\Omega})$ . Then according to Remark 4.3,

(4.71) 
$$\|\mathbf{u}-\mathbf{r}_{\mathbf{h}}\mathbf{u}\| \leq c(\mathbf{u},\alpha)\rho(\mathbf{h})^{2}.$$

We can take  $v = v_h = u_h - r_h u$  in (4.69) and (4.70); subtracting then (4.70) from (4.69) there remains

(4.72) 
$$\nu((u_h - u, u_h - r_h u)) = (\pi_h - p, div(u_h - r_h u)).$$

Let  $\pi'_{h}$  denote the step function

$$\pi'_{h} = \sum_{\lambda \in \mathcal{I}_{h}} \frac{1}{(\text{meas } \lambda)} \left( \int_{\lambda} p(\mathbf{x}) d\mathbf{x} \right) \chi_{h \lambda}.$$

Then the right-hand side of (4.72) is equal to

$$(\pi'_{h}-p, div(u_{h}-r_{h}u))$$

and is majorized by

$$|\pi'_{h}-p|\cdot |div(u_{h}-r_{h}u)| \le \sqrt{2}|\pi'_{h}-p|||u_{h}-r_{h}u||.$$

1-1

111

Hence

It is easy to see that  $|\pi'_h - p|$  is majorized by  $cn_1(p)\rho(h)$  and then we have at least

$$(4.74) \|u_h - r_h u\| \leq cn_1(u,p)\rho(h).$$

This estimation is perhaps not optimal.

When the boundary of  $\Omega$  is not a polygon, an additional error of order  $\rho(h)$  appears in the right-hand side of (4.74).

### 4.3 Finite Elements of Degree 3(n = 3).

Let  $\Omega$  be a lipschitzian open bounded set in  $\mathbb{R}^3$ . We describe first an internal approximation of  $\mathbb{H}^1_0(\Omega)$  and then an external approximation of V. The approximate functions are piecewise polynomials of degree 3.

# <u>Approximation of</u> $H_0^1(\Omega)$ .

Let  $\mathcal{T}_{\mathbf{b}}$  be an admissible triangulation of  $\Omega$  and let

(4.75) 
$$\Omega(h) = \bigcup_{h \in \mathcal{J}} \mathcal{N}$$

If  $\mathcal{N}$  is a 3-simplex (i.e., a tetrahedron) we denote by  $A_1, \dots, A_4$ , the vertices of  $\mathcal{N}$  and by  $B_1, \dots, B_4$ , the barycenter of the 2-faces  $\mathcal{N}_1, \dots, \mathcal{N}_4$ . We denote by  $\mathcal{E}_h^1$  the set of vertices of the simplices  $\mathcal{N} \in \mathcal{T}_h$  and by  $\mathcal{E}_h^2$  the set of barycenters of the 2-faces of the simplices  $\mathcal{N}$  belonging to  $\mathcal{T}_h$ ;  $\mathcal{E}_h = \mathcal{E}_h^1 \cup \mathcal{E}_h^2$ .

We first prove the following result.

Lemma 4.10.

<u>A polynomial of degree three in</u>  $\mathbb{R}^3$  is uniquely defined by its values at the points  $A_i, B_i, 1 \le i \le 4$ , and the values of its first derivatives at the points  $A_i$ . Moreover, the polynomial is given in terms of the barycentric coordinates with respect to  $A_1, \dots, A_4$ , by the formula

$$(4.76) \quad \phi(\mathbf{x}) = \sum_{i=1}^{4} \left[ -2(\lambda_{i}(\mathbf{x}))^{3} + 3(\lambda_{i}(\mathbf{x}))^{2} \right] \phi(\mathbf{A}_{i}) \\ + \frac{1}{6} \sum_{i=1}^{4} \frac{\lambda_{i}(\mathbf{x}) \cdots \lambda_{i}(\mathbf{x})}{\lambda_{i}(\mathbf{x})} \left[ 27\phi(\mathbf{B}_{i}) - 7 \sum_{\substack{\alpha=1 \\ \alpha\neq i}}^{4} \phi(\mathbf{A}_{\alpha}) \right] \\ + \sum_{\substack{i,j=1 \\ i\neq j}}^{4} (\lambda_{i}(\mathbf{x}))^{2}\lambda_{j}(\mathbf{x}) \left[ D\phi(\mathbf{A}_{i}) \cdot \mathbf{A}_{i}\mathbf{A}_{j} \right] \\ - \sum_{i=1}^{4} \frac{\lambda_{1}(\mathbf{x}) \cdots \lambda_{i}(\mathbf{x})}{\lambda_{i}(\mathbf{x})} \left[ D\phi(\mathbf{A}_{i}) \cdot \mathbf{A}_{i}\mathbf{A}_{j} \right].$$

Proof.

The proof is exactly the same as the proof of Lemma 4.5. The coefficients of  $\phi$  are the solutions of a linear system with as many equations as unknowns; we just have to check that the polynomial on the right-hand side of (4.76) fulfills all the required conditions for any set of given data  $\phi(A_i)$ ,  $\phi(B_i)$ ,  $D\phi(A_i)$ .

It follows from this Lemma that a scalar function  $\phi_h$  which is defined on  $\Omega(h)$  and is a polynomial of degree three on each simplex  $\mathcal{M} \in \mathcal{T}_h^{-1}$  is completely known if the values of  $\phi_h$  are given at the points  $A_i \in \mathcal{C}_h^1$  and  $B_i \in \mathcal{C}_h^2$  and also the values of  $D\phi_h$  are given at the points  $A_i \in \mathcal{C}_h^1$ . Such a function  $\phi_h$  is differentiable on each  $\mathcal{M}, \mathcal{M} \in \mathcal{T}_h^{-1}$ , but there is no reason for such a function to be differentiable or even continuous in all of  $\Omega(h)$ . Actually, this function  $\phi_h$  is at least continuous: on a two face  $\mathcal{M}$  of a tetrahedron  $\mathcal{M} \in \mathcal{T}_h^{-1}, \phi_h$  has two, perhaps different, values  $\phi_h^+$  and  $\phi_h^-$ ; but  $\phi_h^+$  and  $\phi_h^$ are polynomials of degree three which take the same values at the vertices and at the barycenter of  $\mathcal{M}$ ; the first derivatives of  $\phi_h^+$  and  $\phi_h^-$  at the vertices of  $\mathcal{M}$  are also equal (these are the derivatives of  $\phi_h$  which are tangential with respect to  $\mathcal{M}$ ). Then it can be proved exactly as in Lemmas 4.5 and 4.6 that  $\phi_h^+ = \phi_h^-$ .

We denote then by  $w_{hM}^{}, M \in \mathcal{E}_h^{}$ , the scalar function which is a piecewise polynomial of degree three on  $\Omega(h)$  with

(4.77) 
$$\begin{cases} w_{hM}(M) = 1, & w_{hM}(P) = 0 \quad \forall P \in \mathcal{E}_{h} \quad P \neq M \\ & Dw_{hM}(P) = 0, \quad \forall P \in \mathcal{E}_{h}^{1}. \end{cases}$$

For  $M \in \xi_h$ , i = 1, 2, 3,  $w_{hM}^{(i)}$  is the scalar function which is a piecewise polynomial of degree three on  $\Omega(h)$  such that

(4.78)  
$$\begin{cases} w_{hM}^{(i)}(P) = 0, \quad \forall P \in \mathcal{E}_{h} \\ D_{w}_{hM}^{(i)}(P) = 0 \quad \forall P \in \mathcal{E}_{h}^{1}, \quad P \neq M \\ D_{w}_{hM}^{(i)}(M) = e_{i}, \quad i = 1, 2, 3. \end{cases}$$

All of the functions  $w_{hM}^{(i)}$ ,  $w_{hM}^{(i)}$  are continuous on  $\Omega(h)$ .

Space Wh

The space  $W_h$  is the space of continuous vector functions  $u_h$  from  $\Omega$ into  $\mathbb{R}^3$ , of type

(4.79) 
$$u_{h} = \sum_{M \in \mathcal{E}_{h}} u_{h}(M) w_{hM} + \sum_{M \in \mathcal{E}_{h}^{1}} \sum_{i=1}^{3} D_{i} u_{h}(M) w_{hM}^{(i)},$$

which vanish outside  $\Omega(h)$ .

It is clear that  $u_h(M) = 0$  for any  $M \in \mathcal{E}_h \cup \partial \Omega(h)$ ; but since the tangential derivatives of  $u_h$  vanish on the faces of the tetrahedrons  $\mathcal{D}$  which are included in  $\Omega(h)$ , the derivatives  $D_{i}u_{h}(M)$ ,  $M \in \mathcal{E}_{h}^{1} \cup \partial \Omega(h)$  are not independent.

The space  $W_h$  is a finite dimensional subspace of  $H^1_0(\Omega)$ ; we provide it with the scalar product induced by  $\mathbb{H}^1_0(\Omega)$ 

(4.80) 
$$((u_h, v_h))_h = ((u_h, v_h)), \forall u_h, v_h \in W_h.$$

Operator ph.

The prolongation operator  $p_h$  is the identity; the  $p_h$  are stable.

<u>Operator</u>  $r_h$ . For  $u \in \mathscr{D}(\Omega)$ , we define  $u_h = r_h u$  by  $u_h = 0$  if the support of u is not included in  $\Omega(h)$ , and if the support is included in  $\Omega(h)$ ,

(4.81) 
$$\begin{cases} u_{h}(M) = u(M), \quad \forall M \in \mathcal{E}_{h}, \\ Du_{h}(M) = Du(M), \quad \forall M \in \mathcal{E}_{h}^{1}. \end{cases}$$

Proposition 4.5.

The preceding internal approximation of  $\mathbb{H}_0^1(\Omega)$  is stable and convergent, provided h belongs to a regular triangulation  $\mathcal{H}_{\alpha}$  of  $\Omega$ .

Proof.

We only have to prove that for each  $u \in \mathfrak{D}(\Omega)$ ,

$$u_h = r_h u \longrightarrow u$$
 in  $\mathbb{H}^1_0(\Omega)$ 

as  $\rho(h) \rightarrow 0$ ,  $h \in \mathcal{H}_{\alpha}$ .

This is proved like Proposition 4.2. The analog of (4.42) for Hermite type interpolation polynomials (see Ciarlet-Raviart [1]) shows that

(4.82) 
$$\begin{cases} \sup_{\mathbf{x} \in \mathcal{D}} |u(\mathbf{x}) - u_{h}(\mathbf{x})| \leq cn_{4}(u)\rho_{\mathcal{D}}^{4} \\ x \in \mathcal{D} \end{cases}$$
$$Sup_{\mathbf{x} \in \mathcal{D}} |Du(\mathbf{x}) - Du_{h}(\mathbf{x})| \leq cn_{4}(u)\frac{\rho_{\mathcal{D}}^{4}}{\rho_{\mathcal{D}}^{2}} \end{cases}$$

Hence

(4.83) 
$$\begin{cases} \sup_{\mathbf{x} \in \Omega} |\mathbf{u}(\mathbf{x}) - \mathbf{u}_{h}(\mathbf{x})| \leq c(\mathbf{u})\rho^{4}(\mathbf{h}) \\ \sup_{\mathbf{x} \in \Omega} |\mathbf{D}\mathbf{u}(\mathbf{x}) - \mathbf{D}\mathbf{u}_{h}(\mathbf{x})| \leq c(\mathbf{u})\rho^{3}(\mathbf{h})\sigma(\mathbf{h}) \\ \mathbf{x} \in \Omega \end{cases}$$

and in particular

$$(4.84) ||u-u_h|| \leq \alpha c(u) \rho^3(h)$$

provided supp  $u \subset \Omega(h)$ .

# Approximation of V (APX 3).

We recall that  $\Omega$  is a lipschitzian bounded set in  $\mathbb{R}^3$ .

Space F, Operator w.

The space F is  $\mathbb{H}^1_0(\Omega)$  and  $\overline{\omega}$  is the identity. Let  $\mathcal{T}_h$  be an admissible triangulation of  $\Omega$ .

Space V<sub>h</sub>.

 $V_h$  is a subspace of the previous space  $W_h$ ; it is the space of  $u_h \in W_h$ 

(4.85) 
$$\int_{\mathcal{D}} \operatorname{div} u_{h} dx = 0, \quad \forall \mathcal{D} \in \mathcal{J}_{h}.$$

We provide the space  $\,V_{\rm h}^{}\,$  with the scalar product (4.80) induced by  $\,\mathbb{H}^1_0(\Omega)\,$  and  $\,W_{\rm h}^{}\,.$ 

# Operator ph.

This operator is the identity (recall that  $V_h \subset \mathbb{H}^1_0(\Omega)$ ).

# Operator r.

The construction of  $r_h$  is based on the same principle as in Section 4.2. Let u be an element of  $\mathcal{V}$ ; we set

(4.86) 
$$r_h u = u_h^1 + u_h^2$$

where  $u_h^1$  and  $u_h^2$  separately belong to  $W_h$ ;  $u_h^1$  is defined exactly as in (4.81)

(4.87) 
$$\begin{cases} u_h^1(M) = u(M), \forall M \in \mathcal{E}_h \\ Du_h^1(M) = Du(M), \forall M \in \mathcal{E}_h^1 \end{cases}$$

The corrector  $u_h^2$  is defined by

(4.88) 
$$u_h^2(M) = 0, \quad Du_h^2(M) = 0, \quad \forall M \in \mathcal{E}_h^1$$

and at the points  $M \in \xi_h^2$ , the component of  $u_h^2(M)$  which is tangent to the face  $\mathcal{N}$  whose M is the barycenter, is zero; the normal components  $u_h^2(M) \cdot v$  is characterized by the condition that

(4.89) 
$$\int_{\mathcal{M}} u_{h}(x) \cdot v d\Gamma = \int_{\mathcal{M}} u(x) \cdot v d\Gamma$$

One can prove <sup>(1)</sup> that there exists some constant d such that

$$\int_{\mathcal{M}} u_h^2(\mathbf{x}) \cdot v d\Gamma = d \cdot (\operatorname{area} \mathcal{M}) u_h^2(\mathbf{M}) \cdot v$$

and (4.89) means that

(4.90)  $u_h^2(M) \cdot v = \frac{1}{d \cdot (\text{area } N')} \int_{N'} (u - u_h^1)(x) \cdot v d\Gamma.$ 

(1) The principle of the proof is similar to that of Lemma 4.9.

It is clear then that  $u_h$  belongs to  $V_h$  since, for each  $\mathcal{A} \in \mathcal{J}_h$ ,

$$\int_{\mathcal{D}} \operatorname{div} \mathbf{u}_{h} d\mathbf{x} = \int_{\partial \mathcal{D}} \mathbf{u}_{h} \cdot v d\Gamma = \int_{\partial \mathcal{D}} \mathbf{u} \cdot v d\Gamma = \int_{\mathcal{D}} \operatorname{div} \mathbf{u} d\mathbf{x} = 0$$

Proposition 4.6.

The preceding external approximation of V is stable and convergent, provided h belongs to a set of regular triangulations  $\mathcal{H}_{\alpha}$  of  $\Omega$ .

The proof of this proposition follows the same lines as the proof of Proposition 4.3.

The approximation of Stokes problem can then be studied exactly as in Section 4.2.

## 4.4 An Internal Approximation of V.

We suppose here that  $\Omega$  is a bounded and simply-connected open subset of  $\mathbb{R}^2$  with a lipschitzian boundary.

In the two dimensional case, the condition div u = 0 is

(4.91) 
$$\frac{\partial u_1}{\partial x_1} + \frac{\partial u_2}{\partial x_2} = 0,$$

and implies that there exists a function  $\psi$  (the stream function), such that

(4.92) 
$$u_1 = \frac{\partial \psi}{\partial x_2}$$
,  $u_2 = -\frac{\partial \psi}{\partial x_1}$ ;

the function  $\psi$  exists locally for any set  $\Omega$ , and globally for a simply connected set  $\Omega$ .

In the present case we can associate to each function u in V the corresponding stream function  $\psi$ . The condition u = 0 on  $\partial\Omega$  amounts to saying that the tangential and normal derivatives of  $\psi$  on  $\partial\Omega$  vanish. Then  $\psi$  is constant on  $\partial\Omega$  and since  $\psi$  is only defined up to an additive constant, we can suppose that  $\psi = 0$  on  $\Gamma$  and hence  $\psi \in H_0^2(\Omega)$ .

Therefore the mapping

(4.93) 
$$\psi \longrightarrow u = \left\{ \frac{\partial \psi}{\partial x_2}, - \frac{\partial \psi}{\partial x_1} \right\}$$

is an isomorphism from  $H^2_{\Omega}(\Omega)$  onto V.

Our purpose is now to construct an approximation of  $H_0^2(\Omega)$  by piecewise polynomial functions of degree 5 and then to obtain with the isomorphism (4.93) an <u>internal</u> approximation of V.

Internal Approximation of  $H_0^2(\Omega)$ .

Let  $\mathfrak{I}_{\mathbf{h}}$  be an admissible triangulation of  $\Omega$ , and let

(4.94) 
$$\Omega(h) = \bigcup_{h \in \mathcal{J}_{h}} \mathcal{L}$$

A 2-simplex is a triangle. Let & be some triangle with vertices  $A_1, A_2, A_3$ ; we denote by  $B_1, B_2, B_3$ , or by  $A_{23}, A_{13}, A_{12}$ , the mid-points of the edges  $A_2A_3, A_1A_3$ , and  $A_1A_2$ ;  $v_{ij}$  denotes one of the unit vectors normal to the edge  $A_iA_j$ ,  $1 \le i, j \le 3$ .

We first notice the following result:

Lemma 4.11.

<u>A polynomial</u>  $\phi$  of degree 5 in  $\mathbb{R}^2$  is uniquely defined by the following values of  $\phi$  and its derivatives

(4.95)  $D^{\alpha}\phi(A_{i}), 1 \leq i \leq 3, [\alpha] \leq 2,$ 

(4.96) 
$$\frac{\partial \phi}{\partial v_{ij}}$$
 (A<sub>ij</sub>),  $1 \leq i, j \leq 3, i \neq j,$ 

where the  $A_i$  are the vertices of a triangle  $\lambda$  and the  $A_{ij}$  are the midpoints of the edges.

Principle of the Proof.

We see that there are as many unknowns (21 coefficients for  $\phi$ ) as linear equations for these unknowns (the 21 conditions corresponding to (4.95)-(4.96)).

As in Lemmas 4.5 and 4.10 it is then sufficient to show that a solution does in fact exist for any set of data in (4.95)-(4.96) and this can be proved by an explicit construction of  $\phi$  leading to a formula similar to (4.26) or (4.76). We omit the very technical proof of this point which can be found in A.Ženišek [1] or M. Zlámal [1]<sup>(1)</sup>

It follows from this lemma that a scalar function  $\psi_h$  which is defined on  $\Omega(h)$  and is a polynomial of degree five on each triangle  $\mathcal{A} \in \mathcal{T}_h$ , is completely known if the values of  $\psi_h$  are given at the points  $A_i \in \mathcal{E}_h^1$ ,  $B_i \in \mathcal{E}_h^2$ ,

(1) The principle of the construction is the following: let  $\lambda_1, \lambda_2, \lambda_3$  denote the barycentric coordinates with respect to  $A_1, A_2, A_3$ . The affine mapping  $x \mapsto y = (\lambda_1(x), \lambda_2(x))$ , maps the triangle  $\Delta$ , on the triangle  $\Delta$ :

 $y_1 = \lambda_1 \ge 0$ ,  $y_2 = \lambda_2 \ge 0$ ,  $0 \le y_1 + y_2 = \lambda_1 + \lambda_2 \le 1$ . The construction of  $\phi(x(y))$  on  $\hat{\mathcal{S}}$  is elementary; then using the inverse mapping  $y \mapsto x$ , we obtain the function  $\phi(x)$ . and if also the values of the first and second derivatives are given at the points  $A_i\!\in\!\xi_h^l,$  where

(4.97) 
$$\begin{cases} \xi_h^1 = \text{set of vertices of the triangles } \mathcal{A} \in \mathcal{T}_h^*.\\ \xi_h^2 = \text{set of mid-points of the edges of the triangles } \mathcal{A} \in \mathcal{T}_h^*.\\ \xi_h = \xi_h^1 \cup \xi_h^2. \end{cases}$$

Such a function  $\phi_h$  is infinitely differentiable on each  $\overline{J}$ ,  $\overline{J} \in \mathcal{T}_h$ , but there is no reason for such a function to be so smooth in all of  $\Omega(h)$ . Actually, the function  $\phi_h$  is continuously differentiable in  $\Omega(h)$ . Let  $\phi_h^+$  and  $\phi_h^-$  denote the values of  $\phi_h$  on two sides of the edge  $A_1A_2$  of a triangle  $\overline{J} \in \mathcal{T}_h$ ;  $\phi_h^+$ ,  $\phi_h^-$  are polynomials of degree less than or equal to 5 on  $A_1A_2$  and they are equal together with their first and second derivatives at the points  $A_1$  and  $A_2$ (six independent conditions) and hence  $\phi_h^+ = \phi_h^-$ . The tangential derivatives  $\frac{\partial \phi_h^+}{\partial \tau}$  and  $\frac{\partial \phi_h^-}{\partial \tau}$ ,  $\tau = \frac{A_1A_2}{|A_1A_2|}$  are also necessarily equal. Let us show then that the normal derivatives  $\frac{\partial \phi_h^+}{\partial \upsilon_{12}}$  and  $\frac{\partial \phi_h^-}{\partial \upsilon_{12}}$  are equal on  $A_1A_2$ . These derivatives are polynomials of degree less than or equal to 4 on  $A_1A_2$ ; they are equal at  $A_1$  and  $A_2$  together with their first derivatives, and they are equal at  $A_{12}$ . Therefore they are equal on  $A_1A_2$ . This shows that  $\phi_h$  is continuously differentiable on  $\Omega(h)$ .

To each point  $M \in \mathbb{C}_h^2$  we associate the function  $\psi_{hM}^0$  which is a piecewise polynomial of degree 5 on  $\Omega(h)$  and such that,

(4.98)  $\begin{cases} \frac{\partial}{\partial \nu} \psi_{hM}^{0}(M) = 1 \text{ and all the other nodal values of } \psi_{hM}^{0} \\ \text{are zero, i.e.,} \\ \frac{\partial \psi_{hM}^{0}}{\partial \nu} (P) = 0, \quad \forall P \in \mathcal{E}_{h}^{2}, P \neq M, \\ D^{\alpha} \psi_{hM}^{0}(P) = 0, \quad \forall P \in \mathcal{E}_{h}^{1}, [\alpha] \leq 2. \end{cases}$ 

To each point  $M \in \mathcal{E}_h^1$ , we associate the six functions  $\psi_{hM}^1, \cdots, \psi_{hM}^6$  defined as follows: they are piecewise polynomials of degree 5 on  $\Omega(h)$  and

(4.101) 
$$\begin{cases} D_1^2 \psi_{hM}^4(M) = 1, \quad D_1 D_2 \psi_{hM}^5(M) = 1, \quad D_2^2 \psi_{hM}^6(M) = 1\\ \text{and all the other nodal values of } \psi_{hM}^4\\ (\text{resp. } \psi_{hM}^5, \text{ resp. } \psi_{hM}^6) \text{ are zero.} \end{cases}$$

All these functions are continuously differentiable on  $\Omega(h)$ .

# Space X<sub>h</sub>.

The space  $X_h$  is the space of continuously differentiable scalar functions on  $\Omega$  (or  $\mathbb{R}^2)$  of type:

(4.102) 
$$\psi_{h} = \sum_{M \in \mathcal{E}_{h}^{2}} \xi_{M}^{0} \psi_{hM}^{0} + \sum_{i=1}^{6} \sum_{M \in \mathcal{E}_{h}^{1}} \xi_{M}^{i} \psi_{hM}^{i}, \ \xi_{M}^{j} \in \mathbb{R}.$$

These functions vanish outside  $\Omega(h)$ , and since they are continuously differentiable in  $\Omega$ ,

(4.103) 
$$\begin{cases} D^{\alpha}\psi_{h}(M) = 0, \quad \forall M \in \mathcal{E}_{h}^{1} \cap \partial\Omega(h), \quad [\alpha] \leq 1, \\ \frac{\partial\psi_{h}}{\partial\nu}(M) = 0, \quad \forall M \in \mathcal{E}_{h}^{2} \cap \partial\Omega(h). \end{cases}$$

The space  $X_h$  is a finite dimensional subspace of  $H^2_0(\Omega)$ ; we provide it with the scalar product induced by  $H^2_0(\Omega)$ 

(4.104) 
$$((\psi_h, \phi_h))_h = ((\psi_h, \phi_h))_{H^2_0(\Omega)}, \quad \forall \psi_h, \phi_h \in X_h.$$

Operator p<sub>h</sub>.

 $p_h = \text{the identity, as } X_h \subset H_0^2(\Omega).$ 

Operator r.

For  $\psi \in \mathcal{D}(\Omega)$  (a dense subspace of  $H^2_0(\Omega)$ ), we define  $r_h \psi = \psi_h$  by its nodal values

(4.105) 
$$\begin{cases} D^{\alpha}\psi_{h}(M) = D^{\alpha}\psi(M), \quad \forall M \in \mathcal{E}_{h}^{1}, \quad [\alpha] \leq 2\\ \frac{\partial\psi_{h}}{\partial\nu_{ij}}(A_{ij}) = \frac{\partial\psi}{\partial\nu_{ij}}(A_{ij}), \quad \forall A_{ij} \in \mathcal{E}_{h}^{2}. \end{cases}$$

Proposition 4.7.

 $(X_h, p_h, r_h)$  defines a stable and convergent internal approximation of  $H_0^2(\Omega)$ , provided h belongs to a regular triangualtion  $\mathcal{H}_{\alpha}$  of  $\Omega$ .

Proof.

We only have to prove that, for each  $\psi \in \mathcal{D}(\Omega)$ ,

(4.106) 
$$r_h \psi \longrightarrow \psi \text{ in } H_0^2(\Omega), \text{ as } \rho(h) \longrightarrow 0.$$

This follows from an analog of (4.42) and (4.82) for Hermite type polynomial interpolation (see Ciarlet-Raviart [1], G. Strang and G. Fix [1], A. Ženíšek [1], M. Zlámal [1]):

(4.107) 
$$\sup_{\mathbf{x} \in \mathcal{X}} |\psi_{\mathbf{h}}(\mathbf{x}) - \psi(\mathbf{x})| \leq \operatorname{cn}_{6}(\psi) \rho_{\mathcal{X}}^{5}$$

(4.108) 
$$\sup_{\mathbf{x}\in\mathcal{D}} |\mathbf{D}_{\mathbf{i}}\psi_{\mathbf{h}}(\mathbf{x})-\mathbf{D}_{\mathbf{i}}\psi(\mathbf{x})| \leq cn_{6}(\psi) \frac{\rho^{5}}{\rho^{7}}, \quad \mathbf{i} = 1,2$$

(4.109) 
$$\sup_{\mathbf{x}\in\mathcal{Y}} |D^{\alpha}\psi_{h}(\mathbf{x})-D^{\alpha}\psi(\mathbf{x})| \leq cn_{6}(\psi) \frac{\rho^{5}}{\rho^{1/2}}, \quad [\alpha] = 2.$$

Therefore

(4.110) 
$$\|\psi_{h} - \psi\|_{H^{2}_{0}(\Omega(h))} \leq c(\psi) \alpha^{2} \rho(h)^{3}$$

and it is clear by (4.22) that

(4.111) 
$$\|\psi\| \xrightarrow{} 0 \text{ as } \rho(h) \longrightarrow 0.$$
  
 $H_0^2(\Omega - \Omega(h))$ 

# Internal Approximation of V (APX 4)

We recall that  $\Omega$  is a bounded simply connected set of  $\mathbb{R}^2$ . We define an <u>internal</u> approximation of V, using the preceding approximation of  $H_0^2(\Omega)$  and the isomorphism (4.93).

Let there be given an admissible triangulation  $\Im_h$  of  $\Omega$ . We associate with  $\Im_h$ , the space  $V_h$ , and the operators  $p_h, r_h$ , as follows.

# Space V<sub>h</sub>.

It is the space of continuous vector functions  $\boldsymbol{u}_h^{}$  defined on  $\Omega$  (or  $\mathbb{R}^2),$  of type

(4.112) 
$$u_{h} = \left\{ \frac{\partial \psi_{h}}{\partial x_{2}}, -\frac{\partial \psi_{h}}{\partial x_{1}} \right\},$$

 $\psi_h$  belonging to the previous space  $X_h^{},$ 

It is clear that  $u_h$  vanishes outside  $\Omega(h)$  and is continuous since  $\psi_h$  is continuously differentiable, and that div  $u_h = 0$ . Therefore  $u_h \in V$ , and  $V_h$  is a <u>finite dimensional subspace</u> of V. We provide it with the scalar product induced by V

(4.113) 
$$((u_h, v_h))_h = ((u_h, v_h)).$$

In particular,

(4.114) 
$$\|u_{h}\|_{h} = \left(\sum_{[\alpha]=2} |D^{\alpha}\psi_{h}|^{2}_{L^{2}(\Omega)}\right)^{\frac{1}{2}} \leq \|\psi_{h}\|_{H^{2}_{0}(\Omega)}$$

Operator p<sub>h</sub>: the identity.

# <u>Operator</u> r<sub>h</sub>:

Let u belong to  $\mathcal{V}$ , and let  $\psi$  denote the corresponding stream function (see (4.93)); clearly,  $\psi \in \mathcal{D}(\Omega)$  and we can define  $\psi_h \in X_h$  by (4.105). Then we set

(4.115) 
$$u_h = r_h u = \left\{ \frac{\partial \psi_h}{\partial x_2}, - \frac{\partial \psi_h}{\partial x_1} \right\} \in V_h.$$

Proposition 4.8.

The preceding internal approximation of V is stable and convergent if  $\rho(h) \rightarrow 0$ , with h belonging to a regular triangulation  $\mathcal{H}_{\alpha}$  of  $\Omega$ .

Proof.

We just have to show that

(4.116) 
$$u_h = r_h u \rightarrow u \text{ in } V, \forall u \in \mathfrak{P}.$$

According to (4.93), (4.114), (4.115), we have

$$\|\mathbf{u}_{\mathbf{h}}^{-}\mathbf{u}\| \leq \|\psi_{\mathbf{h}}^{-}\psi\|_{\mathbf{H}^{2}_{0}(\Omega)}$$

The convergence (4.116) follows then from (4.110) and (4.111).

# Approximation of Stokes Problem.

We take for (3.6),

(4.117) 
$$a_h(u_h, v_h) = v((u_h, v_h))_h = v((u_h, v_h))$$
  
(4.118)  $< \ell_h, v_h > = < f, v_h >,$ 

where  $\nu$  and f are given as in Section 2.1 (see Theorem 2.1). The approximate problem associated with (2.6) is

(4.119) 
$$\begin{cases} \underline{\text{To find}} & u_h \in V_h & \underline{\text{such that}} \\ & \nu((u_h, v_h)) = \langle f, v_h \rangle, & \forall v_h \in V_h \end{cases}$$

The solution  $u_h$  of (4.119) exists and is unique and, according to Theorem 3.1 and Proposition 4.8,

(4.120) 
$$u_{h} \longrightarrow u$$
 in V strongly if  $\rho(h) \longrightarrow 0$ ,  $h \in \mathcal{H}_{\alpha}$ .

# The Error between u and uh.

Let us suppose that  $\Omega$  has a polygonal boundary, so that we can choose triangualtions  $\mathcal{J}_h$  such that  $\Omega(h) = \Omega$ . Let us suppose that the solution u of Stokes problem is so smooth that  $u \in \mathbb{C}^5(\Omega)$ ; then, by (4.110) and (4.114), <sup>(1)</sup>

$$(4.121) ||u-r_hu|| \leq c(u,\alpha)\rho(h)^3.$$

The equations

give

$$v((u-u_h,r_hu-u_h)) = 0.$$

 $\vee((u,v)) = \langle f,v \rangle, \forall v \in V,$ 

 $v((u_h, v_h)) = \langle f, v_h \rangle, \forall v_h \in V,$ 

Therefore,

(4.122)

$$||u-u_h||^2 = ((u-u_h, u-r_hu)) \le ||u-u_h|| ||u-r_hu||$$
  
 $||u-u_h|| \le ||u-r_hu||$ 

and by (4.121), we obtain

(4.123) 
$$||u-u_h|| \leq c(u,\alpha)\rho(h)^3$$
.

Remark 4.5.

(i) An internal approximation of V with piecewise polynomials of degree6 is constructed in F. Thomasset [1].

(ii) Internal approximations of V are not available if n = 3 or if  $\Omega$  is not simply connected.

(1) For the sake of simplicity (4.110) was proved for  $\psi \in \mathcal{D}(\Omega)$ ; the proof is valid for any  $\psi \in \mathcal{C}^6(\overline{\Omega}) \cap \mathrm{H}^2_0(\Omega)$ .

4.5 Non-Conforming Finite Elements.

Because of the condition div u = 0, it is not possible to approximate V by the most simple finite elements, the piecewise linear continuous functions. This was shown by M. Fortin [2]. Our purpose here is to describe an approximation of V by linear, non-conforming finite elements, which means in this case, piecewise linear but discontinuous functions. This leads to the approximation of V denoted by (APX 5). Then we associate with this approximation of V a new approximation scheme for Stokes problem.

# <u>Approximation of</u> $\mathbb{H}^{1}_{0}(\Omega)$ .

We suppose that  $\Omega$  is a bounded lipschitzian open set in  $\mathbb{R}^n$  and in this section we will approximate  $\mathrm{H}^1_0(\Omega)$  by non-conforming piecewise linear finite elements.

Let  $\mathcal{T}_h$  denote an admissible triangulation of  $\Omega$ . If  $\mathcal{A} \in \mathcal{T}_h$ , we denote by  $A_1, \dots, A_{n+1}$  its vertices, by  $\mathcal{A}_i$  the (n-1)-face which does not contain  $A_i$ , and by  $B_i$  the barycenter of the face  $\mathcal{A}_i$ . If G denotes the barycenter of  $\mathcal{A}_i$ , then since the barycentric coordinates of  $B_i$  with respect to the  $A_j$ ,  $j \neq i$ , are equal to 1/n, we have

(4.124) 
$$GB_{i} = \sum_{j \neq i} \frac{GA_{j}}{n} = \sum_{j=1}^{n+1} \frac{GA_{i}}{n} - \frac{GA_{i}}{n}$$

or

(4.125) 
$$GB_i = -\frac{1}{n} GA_i$$
,

since  $\sum_{j=1}^{n+1} GA_j = 0$  (the barycentric coordinates of G with respect to  $A_1, \dots, A_{n+1}, \dots, A_{n+1}$ 

are equal to  $\frac{1}{n+1}$  ). We deduce from this, that

(4.126)  $nB_{i}B_{j} = n(GB_{j}-GB_{i}) = GA_{i}-GA_{j} = -A_{i}A_{j},$ 

and therefore the vectors  $B_1B_j$ ,  $j = 2, \cdots, n+1$ , are linearly independent like the vectors  $A_1A_j$ ,  $j = 2, \cdots, n+1$ . Because of this, the barycentric coordinates of a point P, with respect to  $B_1, \cdots, B_{n+1}$ , can be defined, and we denote by  $\mu_1, \cdots, \mu_{n+1}$ , these coordinates. We remark also that for each given set of (n+1) numbers  $\beta_1, \cdots, \beta_{n+1}$ , there exists one and only one linear function taking on at the points  $B_1, \cdots, B_{n+1}$ , the values  $\beta_1, \cdots, \beta_{n+1}$ , and this function u is

(4.127) 
$$u(P) = \sum_{i=1}^{n+1} \beta_i \mu_i(P);$$

(see Proposition 4.1).

96

Space Wh.

 $W_h$  is the space of vector-functions  $u_h$  which are linear on each  $\mathcal{D} \in \mathcal{T}_h$ , vanish outside  $\Omega(h)^{(1)}$  and such that the value of  $u_h$  at the barycenter  $B_i$  of some (n-1) dimensional face  $\mathcal{D}_i$  of a simplex  $\mathcal{D} \in \mathcal{T}_h$  is zero, if this face belongs to the boundary of  $\Omega(h)$ ; if this face intersects the interior of  $\Omega(h)$  then the values of  $u_h$  at  $B_i$  are the same when  $B_i$  is considered as a point of the two different adjacent simplices.

Let  $\mathfrak{V}_{h}$  denote the set of points  $B_{i}$  which are barycenters of an (n-1) dimensional face of a simplex  $\mathfrak{M} \in \mathfrak{T}_{h}$  and which belong to the interior of  $\Omega(h)$ . A function  $u_{h} \in W_{h}$  is completely characterized by its values at the points  $B_{i} \in \mathfrak{V}_{h}$ .

We denote by w<sub>hB</sub>, B a point of  $\mathcal{V}_h$ , the <u>scalar</u> function which is linear on each simplex  $\mathcal{D} \in \mathcal{T}_h$ , satisfies the same boundary and matching condition that the functions of  $W_h$  satisfy and, moreover,

(4.128)  $w_{hB}(B) = 1, \quad w_{hB}(M) = 0 \quad \forall M \in \mathbb{Q}_{h}, \quad M \neq B.$ 

Such a function  $w_{hB}$  has a support equal to the two simplices which are adjacent to B.  $A_{a} = A_{a}^{2}$ 



# <u>Lemma 4.12</u>.

It is clear that these functions are linearly independent and that they span the whole space  $W_h$ : by Proposition 4.1, any  $u_h \in W_h$  can be written as

(4.129) 
$$u_{h} = \sum_{B \in \mathcal{U}_{h}} u_{h}(B)w_{hB}.$$

<sup>(1)</sup>As before,  $\Omega(h) = \bigcup_{\substack{\lambda \in \mathcal{J}_h}} \mathcal{N}$ .

The space  $W_h$  is not included in  $H^1_0(\Omega)$ ; actually the derivative  $D_i u_h$  of some function  $u_h \in W_h$  is the sum of Dirac distributions located on the faces of the simplices and of a step function  $D_{ih}u_h$  defined almost everywhere by

(4.130) 
$$D_{ih}u_h(x) = D_iu_h(x), \forall x \in \mathcal{D}, \forall \mathcal{D} \in \mathcal{T}_h.$$

Since  $u_h$  is linear on  $\mathcal{D}$ ,  $D_{ih}u_h$  is constant on each simplex.

We equip  $W_h$  with the following scalar product:

(4.131) 
$$[[u_h, v_h]]_h = (u_h, v_h) + \sum_{i=1}^n (D_{ih}u_h, D_{ih}v_h)$$

which is the discrete analog of the scalar product of  $H_0^1(\Omega)$ :

(4.132) 
$$[[u,v]] = (u,v) + \sum_{i=1}^{n} (D_i u, D_i v).$$

<u>Space F, Operators  $\overline{\omega}$ , p<sub>h</sub>.</u>

We take, as in Section 3.3,  $F = L^2(\Omega)^{n+1}$ , and for  $\overline{\omega}$  the isomorphism

(4.133) 
$$u \in \mathbb{H}_0^1(\Omega) \longrightarrow \overline{\omega} u = (u, D_1 u, \cdots, D_n u) \in \mathbb{F}.$$

Similarly, the operator  $p_{\rm b}$  is defined by

(4.134) 
$$u_h \in W_h \longrightarrow p_h u_h = (u_h, \tilde{D}_{1h} u_h, \cdots, \tilde{D}_{nh} u_h) \in F.$$

The operators  $p_h$  each have norm equal to 1 and are stable.

We define  $r_h u = u_h$ , for  $u \in \mathfrak{D}(\Omega)$ , by

(4.135) 
$$u_{h}(B) = u(B), \forall B \in \mathcal{U}_{h}.$$

Proposition 4.9.

If h belongs to a regular triangulation  $\mathcal{H}_{\alpha}$  of  $\Omega$ , the preceding approximation of  $\mathbb{H}^1_0(\Omega)$  is stable and convergent.

#### Proof.

We have to check the conditions (cl) and (c2) of the Definition 3.6. For condition (c2) we have to prove that, for each  $u \in \mathfrak{D}(\Omega)$ ,

(4.136) 
$$u_{h} \longrightarrow u \text{ in } L^{2}(\Omega) \text{ as } \rho(h) \longrightarrow 0,$$

(4.137) 
$$D_{ih}u_{h} \rightarrow D_{i}u \text{ in } L^{2}(\Omega) \text{ as } \rho(h) \rightarrow 0.$$

On each simplex  $\mathcal{A}$  we can apply the result (4.42) to each component of u; this gives:

(4.138) 
$$\begin{cases} \sup_{\mathbf{x}\in\mathcal{Y}} |u(\mathbf{x})-u_{h}(\mathbf{x})| \leq cn_{2}(u)\rho_{\mathcal{Y}}^{2}, \\ \sup_{\mathbf{x}\in\mathcal{Y}} |D_{\mathbf{i}}u(\mathbf{x}) - D_{\mathbf{i}}u_{h}(\mathbf{x})| \leq cn_{2}(u) \frac{\rho_{\mathcal{Y}}^{2}}{\rho_{\mathcal{Y}}^{2}} \end{cases}$$

Therefore

(4.139) 
$$\| p_h u_h - \overline{\omega} u \|_F \leq c(u) \alpha \rho(h) + \| u \|_{H^1_0(\Omega - \Omega(h))}$$

and this goes to 0 as  $\rho(h) \rightarrow 0$ .

To prove the condition (cl) let us suppose that  $p_h, u_h$ , converges weakly in F to  $\phi = (\phi_0, \dots, \phi_n)$ ; this means that

$$(4.140) u_h, \longrightarrow \phi_0 in L^2(\Omega) weakly,$$

(4.141) 
$$D_{ih} u_{h} \longrightarrow \phi_{i}$$
 in  $\mathbb{L}^{2}(\Omega)$  weakly,  $1 \le i \le n$ .

Since the functions have compact supports included in  $\Omega$ , (4.140) and (4.141) amount to saying that:

(4.142) 
$$\tilde{u}_{h}, \longrightarrow \tilde{\phi}_{0}$$
 in  $\mathbb{L}^{2}(\mathbb{R}^{n})$  weakly,  
(4.143)  $D_{ih}\tilde{u}_{h}, \longrightarrow \tilde{\phi}_{i}$  in  $\mathbb{L}^{2}(\mathbb{R}^{n})$  weakly,  $1 \le i \le n$ ,

(§ is the function equal to g in  $\Omega$  and to 0 in  $\Omega$ ). If we show that

(4.144)  $\tilde{\phi}_{i} = D_{i}\tilde{\phi}_{0}, \quad 1 \le i \le n,$ 

it will follow that  $\tilde{\phi}_0 \in \mathbb{H}^1(\mathbb{R}^n)$  and hence  $\phi_0 \in \mathbb{H}^1_0(\Omega)$  with  $\phi_1 = D_1 \phi_0$ , which amounts to saying that  $\phi = \overline{\omega}u$ ,  $u = \phi_0$ .

Let  $\theta$  be any test function in  $\mathscr{P}(\mathbb{R}^n)$ ; then:

$$\begin{split} & \int_{\mathbb{R}^{n}} \tilde{\mathbf{u}}_{h}, (\mathbf{D}_{i}\theta) \, \mathrm{dx} \longrightarrow \int_{\mathbb{R}^{n}} \tilde{\boldsymbol{\phi}}_{0}(\mathbf{D}_{i}\theta) \, \mathrm{dx}, \\ & \int_{\mathbb{R}^{n}} (\mathbf{D}_{ih}\tilde{\mathbf{u}}_{h}, ) \, \theta \, \mathrm{dx} \longrightarrow \int_{\mathbb{R}^{n}} \tilde{\boldsymbol{\phi}}_{i} \, \theta \, \mathrm{dx}. \end{split}$$

The equality (4.144) is proved if we show that

$$\int_{\mathbb{R}^n} \tilde{\phi}_{\mathbf{i}} \theta d\mathbf{x} = - \int_{\mathbb{R}^n} \tilde{\phi}_{\mathbf{0}}(D_{\mathbf{i}} \theta) d\mathbf{x}, \quad \forall \theta \in \mathcal{D}(\mathbb{R}^n),$$

or that

for each  $\theta \in \mathscr{D}(\mathbb{R}^n)$ .

The proof of (4.145) is the goal of the next three lemmas.

(4.146)  $\begin{cases} \frac{\text{Lemma}}{h} = \sum_{\lambda \in \mathcal{T}_{h}} \sum_{\lambda' \in \partial^{+} \lambda'} \int_{\mathcal{M}} u_{h} \theta \dot{v}_{i,\lambda'} d\Gamma, \\ u_{h} \theta \dot{v}_{i,\lambda'} d\Gamma, \end{cases}$ 

where  $\partial^+ \mathcal{W}$  is the set of the (n-1)-dimensional faces of  $\mathcal{W}^{(1)}$ , and  $v_{i,\mathcal{W}}$  is the i<sup>th</sup> component of the unit vector  $v_{\mathcal{W}}$ , which is normal to  $\mathcal{W}$  and is pointing outward with respect to  $\mathcal{W}$ .

Proof.

Since the functions vanish outside  $\Omega(h)$ , we have

$$\begin{aligned} \begin{aligned} & \bigvee_{h} = \int_{\Omega(h)} \left[ u_{h}(D_{i}\theta) + (D_{ih}u_{h})\theta \right] dx = \sum_{\mathcal{N} \in \mathcal{J}_{h}} \int_{\mathcal{N}} \left[ u_{h}(D_{i}\theta) \frac{d}{dx} + (D_{i}u_{h})\theta \right] dx \\ & = \sum_{\mathcal{N} \in \mathcal{J}_{h}} \int_{\mathcal{N}} D_{i}(u_{h}\theta) dx. \end{aligned}$$

The Green-Stokes formula gives

$$\int_{\mathcal{D}} D_{\mathbf{i}}(\mathbf{u}_{\mathbf{h}}^{\theta}) d\mathbf{x} = \sum_{\mathcal{N}' \in \partial^{+} \mathcal{N}} \int_{\mathcal{N}'} u_{\mathbf{h}}^{\theta} v_{\mathbf{i}, \mathcal{N}'} d\Gamma,$$

(1) There are (n+1) such faces.
and (4.146) follows.

Lemma 4.14.

(4.147) 
$$\mathcal{J}_{h} = \sum_{\mathcal{D} \in \mathfrak{I}_{h}} \sum_{\mathcal{D} \in \mathfrak{I}_{h}} \int_{\mathcal{D}} (u_{h}(x) - u_{h}(B))(\theta(x) - \theta(B))v_{i,\mathcal{D}} d\Gamma,$$

where  $B = B_{\mathcal{M}}$  is the barycenter of  $\mathcal{M}$ .

Proof.

We show first that

(4.148) 
$$\int h = \sum_{\mathcal{D} \in \mathcal{J}_{h}} \sum_{\mathcal{U} \in \partial^{+} \mathcal{D}} \int_{\mathcal{U}} (u_{h}(x) - u_{h}(B)) \theta(x) v_{i,\mathcal{U}} d\Gamma.$$

To prove this equality we just have to show that

(4.149) 
$$\sum_{\mathcal{D} \in \mathcal{I}_{h}} \sum_{\mathcal{D} \in \partial^{+} \mathcal{D}} \int_{\mathcal{M}} u_{h}(B) \theta(\mathbf{x}) v_{\mathbf{i}, \mathcal{M}} d\Gamma = 0.$$

But for a face  $\mathcal{X}$  belonging to the boundary of  $\Omega(h)$ ,  $u_h(B) = 0$  and the contribution of this face in the sum is zero. If  $\mathcal{X}$  belongs to the boundary of two adjacent simplices, this face contributes to the sum two opposite terms: the  $u_h(B)$  and  $\theta(x)$  are the same and the  $v_{1,\mathcal{X}}$  are equal but with opposite signs when  $\mathcal{X}$  is considered as part of the boundary of the two simplices. Hence the sum (4.149) is zero.

The equality (4.147) is then easily deduced from (4.148) if we prove that

(4.150) 
$$\sum_{\mathcal{A} \in \mathcal{T}_{h}} \sum_{\mathcal{B} \in \partial^{+} \mathcal{D}} \int_{\mathcal{B}} (u_{h}(x) - u_{h}(B)) \theta(B) v_{i,\mathcal{D}} d\Gamma = 0.$$

But to prove (4.150) we simply note that

$$[u_h(x)-u_h(B)]\theta(B)v_{i,\lambda}d\Gamma = 0,$$

since  $\theta(B)$  and  $v_{i, \mathcal{N}}$  are constant on  $\mathcal{N}$  and since

(4.151) 
$$\int_{\mathcal{M}} u_h(\mathbf{x}) d\Gamma = u_h(B) \int_{\mathcal{M}} d\Gamma,$$

because  $u_h$  is linear on  $\mathcal{N}$ , and B is the barycenter of  $\mathcal{N}$ .

Lemma 4.15.

(4.152)  $\oint_{h'} \longrightarrow 0$  as  $\rho(h') \longrightarrow 0$ , with  $\sigma(h') \le \alpha$ .

Proof.

Since  $|v_{i,\mathcal{Y}}| \leq 1$  and

 $|\theta(\mathbf{x})-\theta(\mathbf{B})| \leq c(\theta)\rho_{\lambda}, \forall \mathbf{x} \in \mathcal{M},$ 

we get the estimate,

(4.153) 
$$\left| \int_{\mathcal{Y}} (u_{h}(x) - u_{h}(B)) (\theta(x) - \theta(B)) v_{i,\mathcal{Y}} d\Gamma \right| \\ \leq c(\theta) \rho_{\mathcal{Y}} \int_{\mathcal{Y}} |u_{h}(x) - u_{h}(B)| d\Gamma$$

Since  $u_h(x)-u_h(B)$  is a linear function on  $\mathcal{N}$  which vanishes at x = B, we can write it on  $\mathcal{N}$  as,

$$u_{h}(x)-u_{h}(B) = \sum_{i=1}^{n} \frac{\partial u_{h}}{\partial x_{i}} \cdot (x_{i}-\beta_{i}),$$

where  $\beta_1, \dots, \beta_n$ , are the coordinates of B. Therefore

and

$$\begin{aligned} \left| u_{h}(x) - u_{h}(B) \right| &\leq \rho_{\mathcal{N}} \sum_{i=1}^{n} \left| \frac{\partial u_{h}}{\partial x_{i}} \right|, \quad \forall x \in \mathcal{N}, \\ c(\theta) \rho_{\mathcal{N}} \int_{\mathcal{N}} \left| u_{h}(x) - u_{h}(B) \right| d\Gamma &\leq c(\theta) \rho_{\mathcal{N}}^{2} \int_{\mathcal{N}} \left( \sum_{i=1}^{n} \left| \frac{\partial u_{h}}{\partial x_{i}} \right| \right) d\Gamma \\ &= c(\theta) \rho_{\mathcal{N}}^{2} (\text{meas}_{n-1} \mathcal{M}) \left\{ \sum_{i=1}^{n} \left| \frac{\partial u_{h}}{\partial x_{i}} \right| \right\} \end{aligned}$$

Let us denote by  $A_1, \dots, A_{n+1}$ , the vertices of  $\Sigma_i$  and let us suppose that  $\Sigma$  is the face containing  $A_2, \dots, A_{n+1}$ ; let  $\xi$  = the distance between  $A_1$  and  $\Sigma$ . The following elementary formula is well known,

(4.154) 
$$\operatorname{meas}_{n}(\mathfrak{A}) = \frac{1}{n} \xi \operatorname{meas}_{n-1}(\mathfrak{A})$$

Hence

(4.155) 
$$\operatorname{meas}_{n-1}(\mathcal{W}) = \frac{n}{\xi} \operatorname{meas}_{n}(\mathcal{W}) \leq \frac{n}{2\rho_{\mathcal{W}}'} \operatorname{meas}_{n}(\mathcal{W}),$$

since

(4.156)  $2\rho'_{N} \leq \xi$ 

The reason (4.156) holds is that the largest ball included in  $\mathcal{N}$  has a diameter

equal to  $2p'_{N}$ , and this ball is included in the set bounded by the hyperplane containing  $A_2, \dots, A_{n+1}$ , and the parallel hyperplane issued from  $A_1$ .

We then have

$$c(\theta) \rho_{\mathcal{J}} \int_{\mathcal{U}} |u_{h}(x) - u_{h}(B)| d\Gamma \leq \frac{c(\theta)n}{2} \frac{\rho_{\mathcal{J}}^{2}}{\rho_{\mathcal{J}}^{1}} (\text{meas}_{n} \mathcal{D}) \left\{ \begin{array}{c} n \\ \sum \\ i=1 \end{array} \left| \frac{\partial u_{h}}{\partial x_{i}} \right| \right\}$$
$$\leq \frac{c(\theta)n\alpha}{2} \rho(h) \int_{\mathcal{D}} \sum \limits_{i=1}^{n} \frac{\partial u_{h}}{\partial x_{i}} dx.$$

Combining this with (4.153) and (4.147) we obtain the following estimates for  $h_{h}$ :

$$\left| \frac{1}{h} \right| \leq \frac{c(\theta)n(n+1)}{2} \alpha \rho(h) \sum_{\lambda \in \mathcal{J}_{h}} \sum_{i=1}^{n} \int \frac{\partial u_{h}}{\partial x_{i}} dx$$
$$\left| \frac{1}{h} \right| \leq c(\theta; n, \alpha) \rho(h) \int_{\Omega} |\text{grad } u_{h}| dx$$

,

(4.157) 
$$\left| \int_{\mathbf{h}} \right| \leq c(\theta, \mathbf{n}, \alpha, \Omega) \rho(\mathbf{h}) \left[ u_{\mathbf{h}} \right]_{\mathbf{h}}.$$

According to (4.136)(4.137), the sequence  $\llbracket u_h, \rrbracket_h$ , is bounded and therefore  $h' \longrightarrow 0$  as  $\rho(h') \longrightarrow 0$ .

## Discrete Poincaré Inequality.

The following discrete Poincaré inequality will allow us to endow the space  $W_h$  described above, with another scalar product  $((\cdot, \cdot))_h$ , the discrete analog of the scalar product  $((\cdot, \cdot))$  of  $\mathbb{H}^1_0(\Omega)$  (see (1.11) and Proposition 3.3).

# Proposition 4.10.

Let us suppose that  $\Omega$  is a bounded set in  $\mathbb{R}^n$ . Then there exists a constant  $c(\Omega, \alpha)$  depending only on  $\Omega$  and the constant  $\alpha$  in (4.21) such that the inequality

(4.158) 
$$|u_h|_{L^2(\Omega)} \leq c(\Omega) \sum_{i=1}^n |D_{ih}u_h|_{L^2(\Omega)}^2$$

holds for any scalar function of type (4.129)

(4.159) 
$$u_{h} = \sum_{B \in \mathcal{V}_{h}} u_{h}(B) w_{hB}$$

103

A similar inequality holds for the vector functions of type (4.129):

(4.160) 
$$|\mathbf{u}_{h}|_{\mathbf{L}^{2}(\Omega)} \leq c'(\Omega) ||\mathbf{u}_{h}||_{h}, \quad \forall \mathbf{u}_{h} \in \mathbf{W}_{h}$$

where

(4.161) 
$$\|u_h\|_h = \{\sum_{i=1}^n |D_{ih}u_h|_{L^2(\Omega)}^2\}^{\frac{1}{2}}$$

Proof.

The inequality (4.160) follows immediately from (4.158). In order to prove (4.158), we will show that

(4.162) 
$$\left|\int_{\Omega} u_{h} \theta dx\right| \leq c(\Omega) \left|\theta\right|_{L^{2}(\Omega)} \left\|u_{h}\right\|_{h},$$

for each  $u_h$  of type (4.159) and for each  $\theta$  in  $\mathcal{P}(\Omega)$ ; (4.162) implies (4.158) since  $\mathcal{P}(\Omega)$  is dense in  $L^2(\Omega)$ .

Let us denote by  $\,\chi\,$  the solution of the Dirichlet problem

$$\Delta \chi = \theta \quad \text{in} \quad \Omega, \quad \chi \in \mathrm{H}^{1}_{0}(\Omega).$$

The function  $\chi$  is  $\boldsymbol{\mathcal{C}}^{\infty}$  on  $\boldsymbol{\Omega}$  and

(4.163) 
$$\|\chi\|_{H^{2}(\Omega)} \leq c_{0}(\Omega) |\theta| \qquad (1)$$

We then have

$$\int_{\Omega} \mathbf{u}_{h} \boldsymbol{\theta} \, d\mathbf{x} = \sum_{\boldsymbol{\lambda} \in \mathbf{J}_{h}} \int \mathbf{u}_{h} \, \boldsymbol{\Delta} \boldsymbol{\chi} \, d\mathbf{x}$$

and the Green formula implies

$$\int_{\mathcal{A}} u_h \Delta \chi \, dx = \int_{\partial \mathcal{A}} u_h \frac{\partial \chi}{\partial \nu} \, d\Gamma - \int_{\mathcal{A}} \operatorname{grad} u_h \operatorname{grad} \chi \, dx.$$

Hence

$$\left|\int_{\Omega} u_{h} \theta d\mathbf{x}\right| \leq \left|\left(\left(u_{h},\chi\right)\right)_{h}\right| + \left|\mathcal{O}_{h}\right|$$

Ň

<sup>&</sup>lt;sup>(1)</sup>Strictly speaking this inequality is true only if  $\Omega$  is smooth enough; in the general case (4.163) is valid if we define  $\chi$  by  $\Delta \chi = \theta$ ,  $\chi \in H_0^1(\Omega')$  where  $\Omega'$  is smooth and  $\Omega' \supset \overline{\Omega}$ . This makes no change in the following.

104

$$(4.164) \qquad \left| \int_{\Omega} u_{h} \theta dx \right| \leq \left\| u_{h} \right\|_{h} \left\| \chi \right\|_{H^{1}(\Omega)} + \left\| \mathcal{H}_{h} \right\| \\ \leq c(\Omega) \left\| \theta \right\|_{L^{2}(\Omega)} \left\| u_{h} \right\|_{h} + \left\| \mathcal{H}_{h} \right\|$$

where

$$\int_{\mathbf{h}} = \sum_{\lambda \in \mathcal{J}_{\mathbf{h}}} \int_{\partial \mathcal{N}} u_{\mathbf{h}} \frac{\partial \chi}{\partial \nu} d\Gamma.$$

The next lemmas will give an estimate for  $\mathcal{F}_h$  which together with (4.164) will give (4.162).

## Lemma 4.16.

Using the same notations as in Lemma 4.14, we have

 $\begin{cases} h = \sum_{i=1}^{n} f_{h}^{i}, \end{cases}$ 

<u>with</u>

$$(4.165) \quad \begin{cases} \mathbf{i} \\ \mathbf{h} \end{cases} = \sum_{\lambda \in \mathcal{J}_{h}} \sum_{\lambda' \in \partial^{+} \lambda} \int_{\lambda'} (\mathbf{u}_{h}(\mathbf{x}) - \mathbf{u}_{h}(\mathbf{B})) (\chi_{\mathbf{i}}(\mathbf{x}) - \chi_{\mathbf{i}}(\mathbf{B})) v_{\mathbf{i}, \lambda'} d\Gamma, \end{cases}$$

where

$$\chi_i = \frac{\partial \chi}{\partial x_i}$$

Proof.

We write

$$\frac{\partial \chi}{\partial \nu} = \sum_{i=1}^{n} \chi_{i} v_{i, \lambda}$$

on the face  $\mathcal{N}$ , and we then proceed exactly as in Lemma 4.14.

(4.166) 
$$\left\| \mathcal{Y}_{h}^{i} \right\| \leq c(n) \sqrt{\alpha \rho(h)} \left\| u_{h} \right\|_{h} \left( \sum_{\lambda \in \mathcal{J}_{h}} \sum_{\lambda' \in \partial^{+} \lambda} \int_{\mathcal{Y}} (\chi_{i}(x) - \chi_{i}(B))^{2} d\Gamma \right)^{\frac{1}{2}}$$

Proof.

The proof is similar to that of Lemma 4.15. We write on the face  $\mathcal{N}$ ,

(4.167) 
$$u_{h}(x)-u_{h}(B) = \sum_{i=1}^{n} \frac{\partial u_{h}}{\partial x_{i}} (x_{i}-\beta_{i}),$$

where  $\beta_1, \dots, \beta_n$ , are the coordinates of B. Hence

(4.168) 
$$|u_h(\mathbf{x})-u_h(\mathbf{B})| \leq \rho_{\lambda}|\text{grad } u_h|, \quad \forall \mathbf{x} \in \lambda,$$

and

$$\begin{split} \left| \int_{\mathcal{M}} \left( u_{h}(\mathbf{x}) - u_{h}(\mathbf{B}) \right) \left( \chi_{i}(\mathbf{x}) - \chi_{i}(\mathbf{B}) \right) v_{i, \mathcal{M}} d\Gamma \right| \\ & \leq \rho_{\mathcal{M}} \left( \int_{\mathcal{M}} |\operatorname{grad} u_{h}|^{2} d\Gamma \right)^{\frac{1}{2}} \left( \int_{\mathcal{M}} \left( \chi_{i}(\mathbf{x}) - \chi_{i}(\mathbf{B}) \right)^{2} d\Gamma \right)^{\frac{1}{2}}. \end{split}$$

But, since grad  $u_h$  is contant on  $\lambda$ ,

$$\int_{\mathcal{M}} (\operatorname{grad} u_{h})^{2} d\Gamma = \operatorname{meas}_{n-1}(\mathcal{M}) \cdot |\operatorname{grad} u_{h}|^{2}$$

$$\leq (\operatorname{because of} (4.155))$$

$$\leq \frac{n}{2\rho' \mathcal{M}} \operatorname{meas}_{n}(\mathcal{M}) \cdot |\operatorname{grad} u_{h}|^{2}$$

$$\leq \frac{n}{2\rho' \mathcal{M}} \int_{\mathcal{M}} |\operatorname{grad} u_{h}|^{2} dx$$

and

$$(4.169) | \mathfrak{f}_{h}^{i} | \leq c(n) \sum_{\lambda \in \mathfrak{I}_{h}} \sum_{\lambda' \in \mathfrak{d}^{+} \mathfrak{L}} \frac{\rho_{\lambda}}{\sqrt{\rho_{\lambda}}} \cdot \left( \int_{\lambda'} |\operatorname{grad} u_{h}|^{2} dx \right)^{\frac{1}{2}} \left( \int_{\lambda'} |\chi_{i}(x) - \chi_{i}(B)|^{2} d\Gamma \right)^{\frac{1}{2}}.$$

We then obtain (4.166) by using (4.20), (4.21) and applying the Schwarz inequality to (4.169).

$$\underbrace{\text{Lemma } 4.18}_{(4.170)} \int_{\mathcal{N}} (\chi_{i}(x) - \chi_{i}(B))^{2} d\Gamma \leq c(n) \alpha^{2} \int_{\mathcal{N}} (\operatorname{grad} \chi_{i})^{2} dx ,$$

$$\underbrace{(4.171)}_{\mathcal{N} \in \mathfrak{J}_{h}} \sum_{\lambda' \in \mathfrak{d}^{+} \mathfrak{Z}} \int_{\mathcal{N}} (\chi_{i}(x) - \chi_{i}(B))^{2} d\Gamma$$

$$\leq c(n) \alpha^{2} \int_{\Omega} (\operatorname{grad} \chi_{i})^{2} dx \leq c(n, \Omega) \alpha^{2} |\theta|^{2}_{L^{2}(\Omega)}.$$

106

Proof.

The inequalities (4.171) follow directly from (4.170) and (4.164).

The inequality (4.170) is an obvious consequence of the trace theorems in  $H^1(\Delta)$  if we replace the constant  $c(n)\alpha^2$  in the right-hand side of (4.170) by some constant  $c(\Delta)$  depending on the particular simplex  $\Delta$ ; the interest of (4.170) is that this inequality is uniformly valid with respect to the simplices  $\Delta$  in  $\mathcal{J}_{\rm b}$ .

To prove (4.170) we make some transformation in the coordinates which maps  $\Delta$  on a fixed simplex  $\overline{\Delta}$  and then we apply the trace theorem inequality in  $\overline{\Delta}$  and come back to  $\Delta$ .

For simplicity we suppose that  $A_1 = 0$ , that  $\mathcal{N}'$  is contained in the hyperplane x = 0, and that the vertices of  $\mathcal{N}'$  are  $A_1, \dots, A_n$ ; the referential simplex is the simplex  $\overline{\mathcal{N}}$  with vertices  $\overline{A_1}, \dots, \overline{A_{n+1}}, \overline{A_1} = 0$ , and  $\overline{A_1A_{i+1}} = e_i$ ,  $i = 1, \dots, n$ . The face corresponding to  $\mathcal{N}'$  is the face  $\overline{\mathcal{N}}'$  with vertices  $\overline{A_1}, \dots, \overline{A_n}$ . Let  $\Lambda$  denote the linear operator in  $\mathbb{R}^n$  which is defined by

$$A_i = \Lambda \overline{A}_i, \quad i = 2, \cdots, n+1$$

and let  $\Lambda'$  be the linear mapping in  $\mathbb{R}^{n-1}$ , which is defined by

$$A_i = \Lambda' \overline{A}_i, \quad i = 2, \cdots, n.$$

A change of coordinates for the integral in the left hand side of (4.170) gives

$$\int_{\mathcal{D}} \sigma(\mathbf{x})^2 d\Gamma_{\mathcal{D}'} = \frac{1}{|\det \Lambda'|} \int_{\overline{\mathcal{D}}} \overline{\sigma}^2(\overline{\mathbf{x}}) d\Gamma_{\overline{\mathcal{D}}}$$

where

(4.172) 
$$\sigma(x) = \chi_{i}(x) - \chi_{i}(B)$$

and

(4.173)  $\overline{\sigma(x)} = \sigma(x), \quad \overline{x} = \Lambda^{-1}x.$ 

For the simplex  $\Delta_{s}$ , the trace theorem inequality and the Poincaré inequality give (recall that  $\sigma(B) = 0$ )

$$\int_{\overline{\lambda}} \overline{\sigma^2}(\overline{x}) d\Gamma \leq c(\overline{\lambda}) \int_{j=1}^n \int (\frac{\partial \overline{\sigma}}{\partial \overline{x_j}}(\overline{x}))^2 d\overline{x}.$$

We come back to  $\mathcal{A}$  and the coordinates  $x_i$ . We write

$$\frac{\partial \overline{\sigma}}{\partial \overline{x}_{j}} (\overline{x}) = \sum_{k=1}^{n} \Lambda_{kj}^{-1} (\frac{\partial \sigma}{\partial x_{k}}) (\Lambda \overline{x}),$$
$$\sum_{j=1}^{n} |\frac{\partial \overline{\sigma}}{\partial \overline{x}_{j}} (\overline{x})|^{2} \leq ||\Lambda^{-1}||^{2} \sum_{k=1}^{n} |\frac{\partial \sigma}{\partial x_{k}} (\Lambda \overline{x})|^{2},$$

and hence

$$\begin{split} \int_{\overline{\lambda}} \overline{\sigma^2}(\overline{x}) d\Gamma &\leq c(\overline{\lambda}) \|\Lambda^{-1}\|^2 \int_{\overline{\lambda}} (\sum_{k=1}^n |\frac{\partial \sigma}{\partial x_k} (\Lambda |\overline{x})|^2) d\overline{x} \\ &\leq c(\overline{\lambda}) |\det \Lambda| \|\Lambda^{-1}\|^2 \int_{\overline{\lambda}} (\operatorname{grad} \sigma)^2 dx. \end{split}$$

We arrive at

$$\int_{\mathcal{M}} (\chi_{i}(\mathbf{x}) - \chi_{i}(B))^{2} d\Gamma \leq c \frac{|\det \Lambda|}{|\det \Lambda^{*}|} \|\Lambda^{-1}\|^{2} \int_{\mathcal{M}} (\operatorname{grad} \chi_{i})^{2} d\mathbf{x}.$$

In order to prove (4.170), it remains to show that

(4.174) 
$$\frac{|\det \Lambda|}{|\det \Lambda|} \|\Lambda^{-1}\|^2 \leq c\alpha^2.$$

Since

$$\frac{\det \Lambda}{\det \Lambda'} = \frac{\det (\Lambda')^{-1}}{\det \Lambda^{-1}} = \Lambda_{nn}$$
(1)

and

 $|\Lambda_{nn}| \leq ||\Lambda||,$ 

the left-hand side of (4.174) is majorized by

$$\|\Lambda\|^2 \|\Lambda^{-1}\|^2$$

Because of Lemma 4.3, this term is majorized by

<sup>&</sup>lt;sup>(1)</sup> $\Lambda_{nn}$  is the (n,n) element of  $\Lambda$ ; note that  $\Lambda_{in} = 0$ ,  $1 \le i \le n-1$ , due to our choice of the coordinate axes.

$$\frac{\left(\rho_{\underline{\lambda}},\rho_{\underline{\lambda}}\right)^{2}}{\left(\rho_{\underline{\lambda}}^{\dagger},\rho_{\underline{\lambda}}^{\dagger}\right)^{2}} \leq c(\overline{\lambda})\alpha^{2}$$

and (4.174) follows.

The proof of Lemma 4.18 is complete.

$$\frac{\text{Lemma } 4.19}{(4.175)} \qquad \left| \right\rangle_{h} \right| \leq c(n) \sqrt{\rho(h)} \alpha^{\frac{3}{2}} \left\| u_{h} \right\|_{h} \left| \theta \right|_{L^{2}(\Omega)}$$

The proof of this lemma is now obvious. As mentioned above, this finishes the proof of (4.158) and, more precisely, shows that

(4.176) 
$$|u|_{L^{2}(\Omega)} \leq (c(\Omega) + c(n)\alpha^{\frac{3}{2}}\rho(h)^{\frac{1}{2}} ||u_{h}||_{h};$$

 $\rho(h)$  is bounded by the diameter of  $\Omega$  and actually goes to 0 as  $h \rightarrow 0$ .

## Proposition 4.11.

Let  $\Omega$  be a bounded lipschitzian set. Let us suppose that we equip the space  $W_h$  with the scalar product

(4.177) 
$$((u_h, v_h))_h = \sum_{i=1}^n (D_{ih}u_h, D_{ih}v_h),$$

and leave the other unchanged in the statement of Proposition 4.9. Then this approximation of  $\mathbb{H}^1_0(\Omega)$  is again stable and convergent.

### Proof.

The only difference between this and Proposition 4.9 comes from the stability of the operators  $p_h$  and this difficulty is completely overcome by Proposition 4.10 and (4.160), which give

(4.178)  $\|\mathbf{u}_{h}\|_{h} \leq \|\mathbf{u}_{h}\|_{h} \leq c(\Omega) \|\mathbf{u}_{h}\|_{h}, \quad \forall \mathbf{u}_{h} \in \mathbf{W}_{h}.$ 

## Approximation of V (APX 5)

Let  $\Omega$  be a lipschitzian bounded set in  $\mathbb{R}^n$  and let  $\mathcal{V}$  be the usual space (1.12) and V its closure in  $\mathbb{H}^1_0(\Omega)$ .

We now define an approximation of V similar to the preceding approximation of  $H^1_0(\Omega)$ .

As previously we take  $F = L^2(\Omega)^{n+1}$ , and  $\overline{\omega} \in \mathfrak{L}(V,F)$  as the linear operator

(4.179) 
$$u \in V \longmapsto \overline{\omega}u = \{u, D_1u, \cdots, D_nu\}.$$

Space V<sub>h</sub>.

We take as space  $V_h$  a subspace of the preceding space  $W_h$ 

(4.180) 
$$V_h = \{u_h \in W_h \mid \sum_{i=1}^n D_{ih}u_{ih} = 0\}$$

The condition in (4.180) concerning the divergence of  $u_{h}$  is equivalent to

(4.181) 
$$\operatorname{div} u_{h} = 0 \quad \text{in } \mathcal{S}, \quad \forall \mathcal{S} \in \mathcal{J}_{h}.$$

We equip the space  $V_h$  with the scalar product  $((u_h, v_h))_h$  induced by  $W_h$ .

<u>Operator</u> p<sub>h</sub>.

As before,

(4.181)  $p_h u_h = \{u_h, D_{1h} u_h, \cdots, D_{nh} u_h\}.$ 

The operators  $p_h$  are stable because of the inequality (4.160) (or (4.178)). Operator  $r_h$ .

We have to define  $r_h u = u_h \in V_h$ , for  $u \in \mathcal{V}$ . Since  $u_h$  must satisfy the condition (4.181), the operator  $r_h$  used for the approximation of  $\mathbb{H}_0^1(\Omega)$  does not satisfy all the requirements. We choose instead the following operator  $r_h$ :  $u_h = r_h u$  is characterized by the values of  $u_h(B)$ ,  $B \in \mathcal{U}_h$ ; if  $B \in \mathcal{U}_h$ , B is the barycenter of some (n-1)-face  $\mathcal{N}$  of some n-simplex  $\mathcal{N} \in \mathcal{J}_h$ ; we set

(4.182) 
$$u_{h}(B) = \frac{1}{\max_{n-1}(\Delta')} \int_{\Delta'} u \, d\Gamma.$$

Let us show that  $u_h \in V_h$ ; since div  $u_h$  is constant on each simplex  $\mathcal{A}_h$ , the condition (4.181) is equivalent to

$$\int_{\mathcal{D}} \operatorname{div} u_h \, dx = 0, \quad \forall \mathcal{D} \in \mathcal{J}_h.$$

Applying the Green formula, we get

$$\int_{\mathcal{A}} \operatorname{div} u_{h} \, \mathrm{dx} = \sum_{\mathcal{N}' \in \partial^{+} \mathcal{A}} \int_{\mathcal{N}'} u_{h} \cdot v_{\mathcal{A}'} \, \mathrm{d\Gamma}$$
$$= (by (4.182))$$
$$= \sum_{\mathcal{N}' \in \partial^{+} \mathcal{A}} \int_{\mathcal{N}'} u \cdot v_{\mathcal{N}'} \, \mathrm{d\Gamma}$$
$$= \int_{\partial \mathcal{A}} u \cdot v \, \mathrm{d\Gamma} = \int_{\mathcal{A}} \operatorname{div} u \, \mathrm{dx},$$

and this last integral is zero since div u = 0.

Proposition 4.12.

The previous external approximation of V is stable and convergent, provided h belongs to a regular triangulation  $\mathscr{H}_{\alpha}$  of  $\Omega$ .

## Proof.

We noticed already that the  $p_h$  are stable. Let us check the condition (C2) of Definition 3.6. For that, let us suppose that

(4.183)  $p_{h}, - \phi$  in F, weakly.

Exactly as in Proposition 4.9 we see that

(4.184)  $\phi = \overline{\omega}u, \quad u \in \mathbb{H}_0^1(\Omega)$ 

Here, we must prove moreover that  $u \in V$ , i.e., div u = 0. But (4.183) means in particular n

$$D_{ih}, u_{ih}, \longrightarrow div u in L^2(\Omega)$$
, weakly,

and since  $\sum_{i=1}^{n} D_{ih}'u_{ih}'$  is identically zero, div u is zero.

Let us check the condition (C1); if  $u\in \mathcal{V}$  we denote by  $u_h$  the function  $r_h^{}u$  and by  $v_h^{}$  the function of  $W_h^{}$  defined by

$$v_h(B) = u(B), \quad \forall B \in \mathcal{U}_h.$$

It was proved in Proposition 4.9 that

(4.185) 
$$\| \mathbf{p}_{h} \mathbf{v}_{h} - \overline{\omega} \mathbf{u} \|_{F} \leq c(\mathbf{u}) \alpha \rho(\mathbf{h}) + [\mathbf{u}]_{H^{1}(\Omega - \Omega(\mathbf{h}))}$$

It suffices now to show that

(4.186) 
$$\| \mathbf{p}_h \mathbf{u}_h - \mathbf{p}_h \mathbf{v}_h \|_F = \| \mathbf{u}_h - \mathbf{v}_h \|_h \longrightarrow 0$$
, as  $\rho(h) \longrightarrow 0$ 

Because of the inequality (4.178), it suffices to prove that

$$\|\mathbf{u}_{\mathbf{h}} - \mathbf{v}_{\mathbf{h}}\|_{\mathbf{h}} \longrightarrow 0$$
, as  $\rho(\mathbf{h}) \longrightarrow 0$ .

Each B of  $\mathcal{U}_h$  is the barycenter of some face  $\mathscr{A}'$  of some simplex  $\mathscr{A};$  we can write

(4.187) 
$$u(x) = u(B) + \sum_{i=1}^{n} \frac{\partial u}{\partial x_{i}} (B) \cdot (x_{i} - \beta_{i}) + \sigma(x),$$

where  $(\beta_1, \dots, \beta_n)$  are the coordinates of B and

(4.188) 
$$|\sigma(\mathbf{x})| \leq c(\mathbf{u})\rho_{\mathcal{X}}^2, \quad \forall \mathbf{x} \in \mathcal{L},$$

Integrating (4.187) on  $\mathcal{N}$ , we find

$$u_{h}(B) = v_{h}(B) + (\int_{\mathcal{M}} \sigma(x) dx) (\int_{\mathcal{M}} dx)^{-1}$$

since  $\int_{\mathcal{N}} (x_i - \beta_i) dx = 0$ . Because of (4.188),

(4.189) 
$$u_h(B) - v_h(B) = \varepsilon_h(B),$$

with

$$|\epsilon_{h}(B)| \leq c(u)\rho_{\lambda}^{2}.$$

Inside the simplex  $\lambda$  with faces  $\lambda_1, \dots, \lambda_{n+1}$ ,

$$u_h(x) - v_h(x) = \sum_{i=1}^{n+1} \varepsilon_h(B_i) \mu_i(x)$$

where  $\mu_1, \dots, \mu_n$ , are the barycentric coordinates of x with respect to  $B_1, \dots, B_{n+1}$ . Therefore, in  $A_n$ 

$$|\operatorname{grad}(u_{h}^{-v_{h}})| \leq c(u)\rho_{\mathcal{X}}^{2}\sum_{i=1}^{n+1} |\operatorname{grad} \mu_{i}|$$

and by Lemma 4.2 and (4.21),

$$|\operatorname{grad}(u_{h}-v_{h})| \leq c(u) \frac{\rho^{2}}{\rho'}.$$

Therefore in all  $\Omega$ ,

(4.191) 
$$|D_{ih}(u_h - v_h)(x)| \leq c(u)\alpha\rho(h)$$

and this implies

(4.192) 
$$\|\mathbf{u}_{h}-\mathbf{v}_{h}\|_{h} \leq c(\mathbf{u},\alpha,\Omega)\rho(\mathbf{h})$$

so that

(4.193) 
$$\| \mathbf{p}_{\mathbf{h}} \mathbf{u}_{\mathbf{h}} - \overline{\boldsymbol{\omega}} \mathbf{u} \|_{\mathbf{F}} \leq c(\mathbf{u}) \alpha \rho(\mathbf{h}) + \| \mathbf{u} \|_{\mathbf{H}^{1}(\Omega - \Omega(\mathbf{h}))} .$$

#### Approximation of the Stokes Problem

Using the preceding approximation of V and the general results of Section 3.2, we can propose another approximation scheme of the Stokes problem.

Let f belong to  $\mathbb{L}^{2}(\Omega)$ , and  $\nu > 0$ . We set, with the preceding notations,

(4.194) 
$$a_h(u_h, v_h) = v((u_h, v_h))_h, \quad \forall u_h, v_h \in V_h$$

(4.195) 
$$\langle l_h, v_n \rangle = (f, v_h), \forall v_h \in V_h.$$

The approximation problem is

(4.196) 
$$\begin{cases} \text{To find } u_h \in V_h, \text{ such that} \\ \nu((u_h, v_h))_h = (f, v_h), \quad \forall v_h \in V_h. \end{cases}$$

The solution  $u_h$  of (4.196) exists and is unique. If  $\rho(h) \rightarrow 0$ , with h belonging to a regular triangulation  $\mathcal{H}_{\alpha}$ , then the following convergence results hold

(4.197) 
$$\begin{cases} u_{h} \longrightarrow u \quad \text{in } \mathbb{L}^{2}(\Omega) \quad \text{strongly,} \\ D_{ih} u_{h} \longrightarrow D_{i} u \quad \text{in } \mathbb{L}^{2}(\Omega) \quad \text{strongly,} \quad 1 \leq i \leq n. \end{cases}$$

This follows of course from Theorem 3.1.

We can, as in Section 3.3 and as for the other approximations, introduce the discrete pressure. It is a step function  $\pi_h$  of the type

(4.198) 
$$\pi_{h} = \sum_{\lambda \in \mathcal{I}_{h}} \pi_{h}(\lambda) \chi_{h,\lambda},$$

where  $\pi_h(\lambda)$  is the value of  $\pi_h$  on  $\lambda$ ,  $\pi_h(\lambda) \in \mathbb{R}$ , and  $\chi_{h,\lambda}$  is the characteristic function of  $\lambda$ . This function  $\pi_h$  is such that

(4.199) 
$$\nu((u_h, v_h))_h - (\pi_h, \operatorname{div}_h v_h) = (f, v_h), \quad \forall v_h \in W_h,$$

where

(4.200) 
$$\operatorname{div}_{h}v_{h} = \sum_{i=1}^{n} D_{ih}v_{ih}$$

The error between u and  $u_h$ , the solutions of (2.6) and (4.199) respectively, can be estimated as in Section 3.3. Let us suppose that  $\Omega$  has a polygonal boundary, that  $\Omega(h) = \Omega$ , and that  $u \in \mathbb{C}^3(\overline{\Omega})$ ,  $p \in \mathbb{C}^1(\overline{\Omega})$ . We can define an approximation  $r_h$  by a formula similar to (4.182), and it is not difficult to see that the estimation (4.193) still holds:

(4.201) 
$$\| \mathbf{p}_{\mathbf{h}} \mathbf{r}_{\mathbf{h}} \mathbf{u} - \overline{\omega} \mathbf{u} \|_{\mathbf{F}} \leq c(\mathbf{u}, \alpha) \rho(\mathbf{h}).$$

We will prove later the following lemma.

Lemma 4.20.

Let u,p denote the exact solution of (2.6)-(2.8) and let us suppose that  $u \in \mathcal{C}^3(\overline{\Omega})$ ,  $p \in \mathbb{C}^1(\overline{\Omega})$ . Then,

(4.202) 
$$a_h(u,v_h) = (f,v_h) + \ell_h(v_h), \quad \forall v_h \in V_h,$$

where

(4.203) 
$$|\ell_h(v_h)| \leq c(u,p)\rho(h) ||v_h||_h.$$

If we admit this lemma temporarily, we see that

$$a_h(u_h-u,v_h) = -\ell_h(v_h), \quad \forall v_h \in V_h,$$

$$a_h(u_h-r_hu,v_h) = a_h(u-r_hu,v_h) - \ell_h(v_h).$$

Taking  $v_h = u_h - r_h u$  and using (4.195) we obtain

$$\| u_{h} - r_{h} u \|_{h}^{2} \leq \| u - r_{h} u \|_{h} \| u_{h} - r_{h} u \|_{h} + \| \ell_{h} (u_{h} - r_{h} u) |.$$

The estimates (4.201) and (4.203) then give

More precisely, the constant c in (4.204) depends only on the norms of u in  $\mathbb{C}^3(\overline{\Omega})$  and of p in  $\mathcal{C}^1(\overline{\Omega})$ .

Proof of Lemma 4.20.

We take the scalar product, in  $L^2(\Omega)$ , of the equation

$$(4.205) \qquad -\nu\Delta u + \text{grad } p = f,$$

with  $v_h$ ; since  $\Omega = \Omega(h)$ , we find

$$\sum_{\substack{b \in \mathcal{I}_h}} \{-\nu(\Delta u, v_h)_{\underline{b}} + (\text{grad } p, v_h)_{\underline{b}} - (f, v_h)_{\underline{b}}\} = 0.$$

The Green formula applied in each simplex  $\lambda$  gives

$$\sum_{k=\mathfrak{I}_{h}} \{-\nu(\Delta u, v_{h})_{\lambda} + (\operatorname{grad} p, v_{h}) - (f, v_{h})_{\lambda}\}$$
$$= a_{h}(u, v_{h}) - (f, v_{h}) - \lambda_{h}(v_{h}) = 0,$$

where

$$\ell_{h}(v_{h}) = \sum_{\mathcal{N} \in \mathcal{J}_{h}} \int_{\partial \mathcal{N}} (v \frac{\partial u}{\partial v} v_{h} - pv_{h} \vec{v}) d\Gamma$$
(1)

The estimate (4.203) of  $l_h$  is then proved exactly as in Lemmas 4.13, 4.14 and 4.15.

## Remark 4.6.

A simple basis for  $v_h$  is available in the two-dimensional case. See the work of Crouzeix [1].

Non-conforming finite elements which are piecewise polynomials of degree

(1) The unit vector normal to  $\partial \Delta$  is denoted  $\vec{v}$  in this formula, to avoid any confusion with the constant v > 0.

 $k \geq 1$  have also been studied for the approximation of, either  $H^1_0(\Omega)$  , or the space V.

M. Fortin [2] has pointed out that the space V cannot be approximated by conforming finite elements of degree one (i.e., piecewise linear functions). For this reason the approximation studied in this section is certainly very useful.

#### **§5.** NUMERICAL ALGORITHMS

We saw that discretization of the Stokes equations does not solve completely the problem of numerical approximation of these equations; for the actual computation of the solution, we must have a basis of the space  $V_h$  such that the analog of (3.6) leads to an algebraic linear system for the components of  $u_h$ , with a sufficiently sparse matrix. This occurs only with the schemes corresponding to (APX 4) and (APX 5); for the discrete problem associated with (APX 1) --(APX 3) we do not even have a basis of  $V_h$ .

In Sections 5.1 to 5.3 we will study two algorithms which are very useful for the practical solution of the discretized equations. In Sections 5.1 and 5.2 we consider the continuous case and in Section 5.3 we show rapidly how they can be adapted to the discrete problems.

The results proved in Section 5.4 are related to this problem but they also show how incompressible fluids can be considered as the limit of "slightly" compressible fluids.

## 5.1 Uzawa Algorithm

We interpreted in Proposition 2.1 the Stokes problem as a variational problem, an optimization problem with linear constraints. The algorithms described in this section and the next are classical algorithms of optimization. We will present these algorithms and study their convergence without any direct reference to optimization theory, although the idea of the algorithm and the proof of the convergence result from optimization theory.

Let us consider the functions u and p defined by Theorem 2.1; we will obtain u,p as limits of sequences  $u^{m}, p^{m}$  which are much easier to compute than u and p.

We start the algorithm with an arbitrary element  $p^0$ ,

When  $p^{m}$  is known, we define  $u^{m+1}$  and  $p^{m+1}$   $(m \ge 0)$ , by the conditions (5.2)  $\begin{cases} u^{m+1} \in H_{0}^{1}(\Omega) \text{ and} \\ v((u^{m+1},v)) - (p^{m}, \operatorname{div} v) = (f,v), \quad \forall v \in H_{0}^{1}(\Omega), \end{cases}$ 

(5.3) 
$$\begin{cases} p^{m+1} \in L^2(\Omega) \text{ and} \\ (p^{m+1}-p^m,q) + \rho(\operatorname{div} u^{m+1},q) = 0, \quad \forall q \in L^2(\Omega). \end{cases}$$

We suppose that  $\rho > 0$  is a fixed number; a condition that  $\rho$  must satisfy will be given later.

The existence and uniqueness of the solution  $u^{m+1}$  of (5.2) is very easy, because of the projection theorem (Theorem 2.2). Actually  $u^{m+1}$  is simply a solution of the Dirichlet problem

(5.4) 
$$\begin{cases} u^{m+1} \in \mathbb{H}_0^1(\Omega) \\ -\nu \Delta u^{m+1} = \text{grad } p^m + f \in \mathbb{H}^{-1}(\Omega). \end{cases}$$

When  $u^{m+1}$  is known,  $p^{m+1}$  is explicitly given by (5.3) which is equivalent to

(5.5) 
$$p^{m+1} = p^m - \rho \operatorname{div} u^{m+1} \in L^2(\Omega).$$

Convergence of the Algorithm

Theorem 5.1. If the number  $\rho$  satisfies (5.6)  $0 < \rho < \frac{2\nu}{p}$ ,

then as  $m \longrightarrow \infty$ ,  $u^{m+1}$  converges to u in  $\mathbb{H}_0^1(\Omega)$  and  $p^{m+1}$  converges to p weakly in  $L^2(\Omega)/\mathbb{R}$ .

# <u>Proof</u>.

The equation (2.7) which is satisfied by u and p is equivalent to

(5.7) 
$$v((u,v)) - (p, \operatorname{div} v) = (f,v), \quad \forall v \in \mathbb{H}_0^1(\Omega)$$

Let us take  $v = u^{m+1}-u$  in equations (5.2) and (5.7) and then let us subtract the resulting equations; this gives:

$$v \| u^{m+1} - u \|^2 = (p^m - p, div u^{m+1})$$

9

or

(5.8) 
$$v ||v^{m+1}||^2 = (q^m, \text{ div } v^{m+1})$$

where we have set

(5.9) 
$$v^{m+1} = u^{m+1} - u$$
.

(5.10) 
$$q^{m} = p^{m} - p.$$

Taking  $q = p^{m+1} - p$  in (5.3), we get:  $(q^{m+1}-q^m, q^{m+1}) + \rho(\text{div v}^{m+1}, q^{m+1}) = 0,$ 

or equivalently

(5.11) 
$$|q^{m+1}|^2 - |q^m|^2 + |q^{m+1}-q^m|^2 = -2\rho(\text{div v}^{m+1}, q^{m+1}).$$

We next multiply equation (5.8) by  $2\rho,\,$  and then add equation (5.11), obtaining

(5.12) 
$$|q^{m+1}|^2 - |q^m|^2 + |q^{m+1}-q^m|^2 + 2\rho v ||v^{m+1}||^2$$
  
=  $-2\rho(\text{div } v^{m+1}, q^{m+1}-q^m).$ 

We majorize the right-hand side of (5.12) by

$$2\rho | \operatorname{div} v^{m+1} | | q^{m+1} - q^{m} |$$

which is less than or equal to

$$2\rho\sqrt{n}\|\mathbf{v}^{m+1}\| |q^{m+1}-q^{m}|$$

since

(5.13) 
$$|\operatorname{div} v| \leq \sqrt{n} ||v||, \quad \forall v \in \mathbb{H}_0^1(\Omega).$$

We can then majorize the last expression by

$$\delta |q^{m+1}-q^{m}|^{2} + \frac{\rho^{2}n}{\delta} ||v^{m+1}||^{2},$$

where  $0 < \delta < 1$  is arbitrary at the present time. Hence

$$(5.14) |q^{m+1}|^2 - |q^m|^2 + (1-\delta)|q^{m+1}-q^m|^2 + \rho(2\nu - \frac{\rho_n}{\delta})||v^{m+1}||^2 \le 0.$$

If we add the inequalities (5.14) for  $m = 0, \dots, N$ , we find

(5.15) 
$$|q^{N+1}|^2 + (1-\delta) \sum_{m=0}^{N} |q^{m+1}-q^m|^2 + (2\nu - \frac{\rho_n}{\delta})\rho \sum_{m=0}^{N} ||v^{m+1}||^2 \le |q^0|^2.$$

Because of condition (5.6), there exists some 
$$\delta$$
 such that

$$0 < \frac{\rho n}{2\nu} < \delta < 1,$$

and hence

$$(2\nu - \frac{\rho n}{\delta}) > 0.$$

With such a  $\,\delta\,$  fixed, the inequality (5.15) shows that

(5.16) 
$$\begin{cases} |q^{m+1}-q^{m}|^{2} = |p^{m+1}-p^{m}|^{2} \longrightarrow 0 \text{ as } m \longrightarrow \infty, \\ ||v^{m+1}||^{2} = ||u^{m+1}-u||^{2} \longrightarrow 0 \text{ as } m \longrightarrow \infty. \end{cases}$$

The convergence of  $u^{m+1}$  to u is thereby proved. Now by (5.14) we see also that the sequence  $p^m$  is bounded in  $L^2(\Omega)$ . We can then extract from  $p^m$  a subsequence  $p^{m^*}$  converging weakly in  $L^2(\Omega)$  to some element  $p_*$ . The equation (5.2) gives in the limit

$$\forall ((u,v)) - (p_*, div v) = (f,v), \quad \forall v \in H^1_0(\Omega),$$

and by comparison with (5.7), we get

$$(p-p_*, \text{ div } v) = 0, \quad \forall v \in \mathbb{H}^1_0(\Omega),$$

whence

$$grad(p-p_*) = 0$$
,  $p_* = p + const.$ 

From any subsequence of  $p^m$ , we can extract a subsequence converging weakly in  $L^2(\Omega)$  to p + c; hence the sequence  $p^m$  converges as a whole to p for the weak topology of  $L^2(\Omega)/\mathbb{R}$ .

Remark 5.1.

Let us choose p by demanding the condition

$$\int_{\Omega}^{\infty} p(\mathbf{x}) d\mathbf{x} = 0.$$

Let us suppose that  $p^0$  in  $L^2(\Omega)$  is chosen so that

$$\int_{\Omega} p^{0}(x) dx = 0.$$

Then clearly we have

$$\int_{\Omega} p^{m}(x) dx = 0, \quad \forall m \ge 1$$

and the whole sequence  $p^m$  converges to p, weakly, in the space  $L^2(\Omega)$ .

## 5.2 Arrow-Hurwicz Algorithm.

In this case too, the functions u and p are the limits of two sequences  $u^m, p^m$  which are recursively defined.

We start the algorithm with arbitrary elements  $u^0, p^0$ ,

(5.17)  $u^0 \in \mathbb{H}^1_0(\Omega), \quad p^0 \in L^2(\Omega).$ 

When  $p^{m}$  and  $u^{m}$  are known, we define  $p^{m+1}$  and  $u^{m+1}$  as the solutions of

(5.18) 
$$\begin{cases} u^{m+1} \in \mathbb{H}_0^1(\Omega) \text{ and} \\ ((u^{m+1}-u^m,v)) + \rho \vee ((u^m,v)) - \rho(p^m, \operatorname{div} v) \\ = (f,v), \quad \forall v \in \mathbb{H}_0^1(\Omega), \end{cases}$$

(5.19) 
$$\begin{cases} p^{m+1} \in L^2(\Omega) \text{ and} \\ \alpha(p^{m+1}-p^m,q) + \rho(\operatorname{div} u^{m+1},q) = 0, \quad \forall q \in L^2(\Omega). \end{cases}$$

We suppose that  $\rho$  and  $\alpha$  are two striclty positive numbers; conditions on  $\rho$  and  $\alpha$  will appear later.

The existence and uniqueness of  $u^{m+1} \in H_0^1(\Omega)$  satisfying (5.18) is easily established with the projection theorem;  $u^{m+1}$  is the solution of the Dirichlet problem

(5.20) 
$$\begin{cases} -\Delta u^{m+1} = -\Delta u^{m} + \rho v \Delta u^{m} - \rho \text{ grad } p^{m} + f \\ u^{m+1} \in \mathbb{H}_{0}^{1}(\Omega). \end{cases}$$

Then  $p^{m+1}$  is explicitly given by (5.19) which is equivalent to

(5.21) 
$$p^{m+1} = p^m - \frac{\rho}{\alpha} \operatorname{div} u^{m+1} \in L^2(\Omega).$$

Convergence of the Algorithm.

 $\frac{\text{Theorem } 5.2}{\text{If the numbers } \alpha \text{ and } \rho \text{ satisfy}}$ (5.22)  $0 < \rho < \frac{2\alpha\nu}{\alpha\nu^2 + n} ,$ 

then, as  $m \longrightarrow \infty$ ,  $u^m$  converges to u in  $H^1_0(\Omega)$  and  $p^m$  converges to p weakly in  $L^2(\Omega)/R$ .

Proof.

Let

(5.23) 
$$v^{m} = u^{m} - u$$
,

(5.24) 
$$q^{m} = p^{m} - p.$$

Equations (5.18) and (5.7) give

$$((v^{m+1}-v^{m},v)) + \rho v((v^{m},v)) = \rho(q^{m}, \operatorname{div} v), \quad \forall v \in H^{1}_{0}(\Omega),$$

and taking  $v = v^{m+1}$  we obtain

(5.25) 
$$\|v^{m+1}\|^2 - \|v^m\|^2 + \|v^{m+1} - v^m\|^2 + 2\rho v\|v^{m+1}\|^2$$
  
=  $2\rho v((v^{m+1}, v^{m+1} - v^m)) + 2\rho(q^m, div v^{m+1})$   
 $\leq \delta \|v^{m+1} - v^m\|^2 + \frac{\rho^2 v^2}{\delta} \|v^{m+1}\|^2 + 2\rho(q^m, div v^{m+1}),$ 

where  $\delta > 0$  is arbitrary at the present time.

Equation (5.19), with  $q = q^{m+1}$  can be written as

$$\begin{aligned} \alpha |q^{m+1}|^2 &- \alpha |q^{m}|^2 + \alpha |q^{m+1} - q^{m}|^2 &= -2\rho(q^{m+1}, \text{ div } u^{m+1}) \\ &= -2\rho(q^{m}, \text{ div } v^{m+1}) - 2\rho(q^{m+1} - q^{m}, \text{ div } v^{m+1}) \\ &\leq -2\rho(q^{m}, \text{ div } v^{m+1}) + 2\rho |q^{m+1} - q^{m}| ||\text{ div } v^{m+1}| \\ &\leq (\text{by } (5.13)) \\ &\leq -2\rho(q^{m}, \text{ div } v^{m+1}) + 2\rho \sqrt{n} |q^{m+1} - q^{m}| ||v^{m+1}||. \end{aligned}$$

Finally, with the same  $\delta$  as before,

(5.26) 
$$\alpha |q^{m+1}|^2 - \alpha |q^m|^2 + \alpha |q^{m+1} - q^m|^2 \leq -2\rho(q^m, \text{ div } v^{m+1})$$
  
  $+ \alpha \delta |q^{m+1} - q^m|^2 + \frac{\rho^2 n}{\alpha \delta} ||v^{m+1}||^2.$ 

Adding inequalities (5.25) and (5.26), we get

(5.27) 
$$\alpha |q^{m+1}|^2 + ||v^{m+1}||^2 - \alpha |q^m|^2 - ||v^m||^2 + \alpha(1-\delta) |q^{m+1}-q^m|^2 + (1-\delta) ||v^{m+1}-v^m||^2 + \rho(2\nu - \frac{\rho\nu^2}{\delta} - \frac{\rho n}{\alpha\delta}) ||v^{m+1}||^2 \le 0.$$

If condition (5.22) holds, then

$$2\nu > \rho\nu^2 + \frac{\rho n}{\alpha}$$
,

and for some  $0 < \delta < 1$  sufficiently close to 1, we have again

$$2\nu > \frac{1}{\delta}(\rho\nu^2 + \frac{\rho n}{\alpha})$$

so that

(5.28) 
$$\rho(2\nu - \frac{\rho\nu^2}{\delta} - \frac{\rho n}{\alpha\delta}) > 0.$$

By adding inequalities (5.27) for  $m = 0, \dots, N$ , we obtain an inequality of the same type as (5.15), and the proof is completed as for Theorem 5.1.

#### Remark 5.2.

It is easy to extend the Remark 5.1 to this algorithm.

## 5.3 Discrete Form of these Algorithms

We describe the discrete form of these algorithms in the case of finite differences (approximation APX 1).

In order to actually compute the step functions  $u_h$  and  $\pi_h$  which are solutions of (3.64), (3.71), (3.73), we define two sequences of step functions  $u_h^m$ ,  $\pi_h^m$ , of the type

(5.29) 
$$u_h^m = \sum_{M \in \Omega_h^1} \xi_M w_{hM}, \quad \xi_M \in \mathbb{R}^n \quad (i.e., u_h^m \in W_h)$$

(5.30) 
$$\pi_{h}^{m} = \sum_{M \in \hat{\Omega}_{h}^{1}} \eta_{M}^{w}{}_{hM}, \quad \eta_{M} \in \mathbb{R},$$

which are recursively defined by the analog of one of the preceding algorithms.

#### Uzawa Algorithm

We start with an arbitrary  $\pi_h^0$  of type (5.30). When  $\pi_h^m$  is known, we define  $u_h^{m+1}$  and  $\pi_h^{m+1}$  by

(5.31) 
$$\begin{cases} u_h^{m+1} \in W_h \text{ and} \\ \nu((u_h^{m+1}, v_h))_h - (\pi_h^m, D_h v_h) = (f, v_h), \forall v_h \in W_h \end{cases}$$

(5.32) 
$$\pi_h^{m+1}(M) = \pi_h^m(M) - \rho(D_h u_h^{m+1})(M)$$

where  $D_h$  is the discrete divergence operator defined by (3.69).

If  $\rho$  satisfies the same condition (5.6), a repetition of the proof of Theorem 5.1 shows that, as  $m \longrightarrow \infty$ 

(5.33) 
$$u_h^m \longrightarrow u_h \text{ in } W_h,$$

(5.34) 
$$\pi_h^m \longrightarrow \pi_h$$
 up to a constant;

the convergence holds for any norm on the finite dimensional spaces considered.

#### Arrow-Hurwicz Algorithm

We start with arbitrary  $u_h^0, \pi_h^0$  of type (5.29) and (5.30) respectively. When  $u_h^m, \pi_h^m$  are known, we define  $u_h^{m+1}, \pi_h^{m+1}$  by

(5.35) 
$$u_{h}^{m+1} \in W_{h} \text{ and}$$

$$((u_{h}^{m+1} - u_{h}^{m}, v_{h}))_{h} + \rho \vee ((u_{h}^{m}, v_{h}))_{h}$$

$$-\rho(\pi_{h}^{m}, D_{h}v_{h}) = (f, v_{h}), \quad \forall v_{h} \in W_{h}.$$
(5.36) 
$$\pi_{h}^{m+1}(M) = \pi_{h}^{m}(M) - \frac{\rho}{\alpha} D_{h}u_{h}^{m+1}(M), \quad \forall M \in \hat{\Omega}_{h}^{1}.$$

If  $\rho$  satisfies the condition (5.22), an extension of the proof of Theorem 5.2 gives the convergence (5.33) - (5.34).

### Discrete Arrow-Hurwicz Algorithm

The problems (5.31) and (5.35) are discrete Dirichlet problems and their solution is easy and quite standard. Nevertheless, it is interesting to notice that in the finite dimensional case, we can use another form of Arrow-Hurwicz algorithm, for which we do not have any boundary value problem to solve during the iteration process.

When 
$$u_h^m$$
,  $\pi_h^m$  are known, we define  $u_h^{m+1}$  by  
(5.37)  $u_h^{m+1} \in W_h$  and,  
 $(u_h^{m+1}-u_h^m, v_h) + \rho v ((u_h^m, v_h))_h - \rho (\pi_h^m, D_h v_h) = (f, v_h), \quad \forall v_h \in W_h,$ 

and then  $\pi_h^{m+1}$  is again defined by (5.36). The variational equation (5.37) is equivalent to the following equations

(5.38) 
$$u_{h}^{m+1}(M) = u_{h}^{m}(M) + \rho v \sum_{i=1}^{n} (\delta_{ih}^{2} u_{h}^{m})(M)$$
$$-\rho(\overline{V}_{h} \pi_{h}^{m})(M) + f_{h}(M), \quad \forall M \in \mathcal{A}_{h}^{1},$$

where  $\overline{V}_h$  and  $f_h$  were defined in (3.74)

The proof of Theorem 5.2 can be extended to this situation as follows. Since  $W_h$  is a finite dimensional space, all the norms defined on  $W_h$  are equivalent, and hence there exists some constant S(h) depending on h, such that

(5.39) 
$$\|u_h\|_h \leq S(h) |u_h|$$
,  $\forall u_h \in W_h$ .  
We will compute  $S(h)$ , and use this remark extensively in Chapter III ( $S(h) = 2(\sum_{i=1}^{n} \frac{1}{h_i^2})^{\frac{1}{2}}$ .  
 $i=1$   $h_i^2$  Now we have: if  $\rho$  satisfies  
(5.40)  $0 < \rho < \frac{2\alpha\nu}{\alpha\nu^2S^2(h)+n}$ ,

then the convergences (5.33) - (5.34) are also true for the algorithm (5.36) - (5.37).

The proof is the same as for Theorem 5.2. The inequality (5.25) is just replaced by

$$(5.41) \qquad (v_{h}^{m} = u_{h}^{m} - u_{h}, \quad \kappa_{h}^{m} = \pi_{h}^{m} - \pi_{h}): \\ |v_{h}^{m+1}|^{2} - |v_{h}^{m}|^{2} + |v_{h}^{m+1} - v_{h}^{m}| + 2\rho \langle |v_{h}^{m+1}||_{h}^{2} \\ = 2\rho \vee (\langle v_{h}^{m+1}, v_{h}^{m+1} - v_{h}^{m}\rangle)_{h} + 2\rho \langle \kappa_{h}^{m}, D_{h} v_{h}^{m+1}\rangle \\ \leq 2\rho \vee |v_{h}^{m+1}||_{h} ||v_{h}^{m+1} - v_{h}^{m}||_{h} + 2\rho \langle \kappa_{h}^{m}, D_{h} v_{h}^{m+1}\rangle \\ \leq 2\rho \vee S(h) ||v_{h}^{m+1}||_{h} ||v_{h}^{m+1} - v_{h}^{m}|| + 2\rho \langle \kappa_{h}^{m}, D_{h} v_{h}^{m+1}\rangle \\ \leq \delta |v_{h}^{m+1} - v_{h}^{m}|^{2} + \frac{\rho^{2} \vee^{2} S^{2}(h)}{\delta} ||v_{h}^{m+1}||_{h}^{2} \\ + 2\rho \langle \kappa_{h}^{m}, D_{h} v_{h}^{m+1}\rangle.$$

The inequality (5.27) is accordingly changed to

(5.42)  

$$\alpha |\kappa_{h}^{m+1}|^{2} + |v_{h}^{m+1}|^{2} - \alpha |\kappa_{h}^{m}|^{2} - |v_{h}^{m}|^{2}$$

$$+ \alpha (1-\delta) |\kappa_{h}^{m+1} - \kappa_{h}^{m}|^{2} + (1-\delta) |v_{h}^{m+1} - v_{h}^{m}|^{2}$$

$$+ \rho (2\nu - \frac{\rho \nu^{2} S^{2}(h)}{\delta} - \frac{\rho n}{\alpha \delta}) ||v_{h}^{m+1}||_{h}^{2} \leq 0;$$

and because of (5.40), inequality (5.42) leads to the same conclusion as (5.27).

## \$6. SLIGHTLY COMPRESSIBLE FLUIDS

The stationary linearized equations of slightly compressible fluids are

(6.1) 
$$-\nu\Delta u_{\varepsilon} - \frac{1}{\varepsilon} \operatorname{grad} \operatorname{div} u_{\varepsilon} = f \quad \text{in } \Omega,$$

$$u_{c} = 0 \quad \text{on} \quad \partial\Omega,$$

where  $\varepsilon > 0$  is "small". Equations (6.1) - (6.2) are also the stationary Lamé equations of elasticity.

We will show that equations (6.1) - (6.2) have a unique solution  $u_{\varepsilon}$  for  $\varepsilon > 0$  fixed, and that  $u_{\varepsilon}$  converges to the solution u of the Stokes equations as  $\varepsilon \longrightarrow 0$ .

At first, equations (6.1) - (6.2) were used as "approximate" equations for Stokes equations -- one way to overcome the difficulty "div u = 0", was to solve equations (6.1) - (6.2) with  $\varepsilon$  sufficiently small in place of solving the Stokes equations. Nowadays, since many efficient algorithms are known for solving Stokes equations, and since the discretization of (6.1)-(6.2) leads to a very ill-conditioned matrix for very small  $\varepsilon$ , one can try to do the converse: compute  $u_{\varepsilon}$  for small  $\varepsilon$  by using Stokes equations.

In Section 6.1 we show the relation between  $u_{\varepsilon}$  and u and in Section 6.2 we give an asymptotic development of  $u_{\varepsilon}$  as  $\varepsilon \longrightarrow 0$ . Then in Section 6.3 we show how one can proceed to compute  $u_{\varepsilon}$ , for small  $\varepsilon$ , using this asymptotic development.

6.1 <u>Convergence of</u>  $u_{\varepsilon}$  to u. <u>Theorem 6.1</u>. <u>Let  $\Omega$  be a bounded lipschitzian domain in</u>  $\mathbb{R}^{n}$ . <u>For  $\varepsilon > 0$  fixed, there exists a unique</u>  $u_{\varepsilon} \in \mathbb{H}_{0}^{1}(\Omega)$  which satisfies (6.1). <u>When</u>  $\varepsilon \rightarrow 0$ , (6.3)  $u_{\varepsilon} \rightarrow u$  in the norm of  $\mathbb{H}_{0}^{1}(\Omega)$ , (6.4)  $-\frac{\operatorname{div} u_{\varepsilon}}{\varepsilon} \rightarrow p$  in the norm of  $L^{2}(\Omega)$ , <u>where</u> u and p are defined by (2.6) - (2.9) and moreover

(6.5) 
$$\int_{\Omega} p(x) dx = 0.$$

Proof.

It is easy to show that the problem (6.1) - (6.2) is equivalent to the following variational problem:

<u>To find</u>  $u_{\rho} \in \mathbb{H}_{0}^{1}(\Omega)$  such that

(6.6) 
$$\nu((u_{\varepsilon},v)) + \frac{1}{\varepsilon} (\operatorname{div} u_{\varepsilon}, \operatorname{div} v) = (f,v), \quad \forall v \in \mathbb{H}_0^1(\Omega).$$

Actually, if  $u_{\varepsilon}$  satisfies (6.1) - (6.2) then  $u_{\varepsilon} \in \mathbb{H}_{0}^{1}(\Omega)$  and satisfies (6.6) for each  $v \in \mathfrak{D}(\Omega)$ . By a continuity argument, it satisfies (6.6) too, for each  $v \in \mathbb{H}_{0}^{1}(\Omega)$ . Conversely, if  $u_{\varepsilon} \in \mathbb{H}_{0}^{1}(\Omega)$  is a solution of (6.6) then  $u_{\varepsilon}$  satisfies (6.1) in the distribution sense and (6.2) in the sense of trace theorems.

The existence and uniqueness of  $u_{\epsilon}$  satisfying (6.6) results from the projection theorem: we apply Theorem 2.2 with

$$W = H_0^1(\Omega), a(u,v) = v((u,v)) + \frac{1}{\varepsilon} (div u, div v),$$
$$\langle \ell, v \rangle = (f,v).$$

The coercivity of a and the continuity of a and & are obvious.

To prove (6.3) let us substract (2.7) from (6.1); this gives

(6.7) 
$$-\nu\Delta(u_{\varepsilon}-u) - \frac{1}{\varepsilon} \text{ grad div } u_{\varepsilon} = + \text{ grad } p$$

and thus

(6.8) 
$$\nu((u_{\varepsilon}-u, v)) + \frac{1}{\varepsilon} (\operatorname{div} u_{\varepsilon}, \operatorname{div} v) = -(p, \operatorname{div} v), \quad \forall v \in \mathbb{H}_{0}^{1}(\Omega).$$

Equation (6.8) follows easily from (6.7) for  $v \in \mathfrak{D}(\Omega)$ ; by a continuity argument, (6.8) is satisfied for each  $v \in H_0^1(\Omega)$ .

Let us put  $v = u_c - u$  in (6.8); we obtain

$$V \| u_{\varepsilon}^{-} u \|^{2} + \frac{1}{\varepsilon} | \operatorname{div} u_{\varepsilon} |^{2}$$
  
= -(p, div u\_{\varepsilon})  $\leq |p| | \operatorname{div} u_{\varepsilon} |$   
 $\leq \frac{1}{2\varepsilon} | \operatorname{div} u_{\varepsilon} |^{2} + \frac{\varepsilon}{2} |p|^{2}$ 

so that

This proves (6.3). Consequently, (6.7) shows that

(6.10) 
$$\frac{\partial}{\partial \mathbf{x}_{i}} \xrightarrow{\operatorname{div} \mathbf{u}_{\varepsilon}} \longrightarrow - \frac{\partial \mathbf{p}}{\partial \mathbf{x}_{i}},$$

in the norm of  $H^{-1}(\Omega)$ , for  $i = 1, \dots, n$  ( $\Delta(u_{\varepsilon}-u)$  converges to 0 in  $H^{-1}(\Omega)$ , because of (6.3)).

According to the following lemma,

(6.11) 
$$|p + \frac{\operatorname{div} u_{\varepsilon}}{\varepsilon}| \leq \sum_{i=1}^{n} || \frac{\partial}{\partial x_{i}} (p + \frac{\operatorname{div} u_{\varepsilon}}{\varepsilon})||_{H^{-1}(\Omega)},$$

since, because u vanishes on  $\partial\Omega$  and (6.5) holds,

$$\int_{\Omega} (p + \frac{\operatorname{div} u_{\varepsilon}}{\varepsilon}) dx = 0$$

The convergence (6.4) is proved.

Lemma 6.1.

Let  $\Omega$  be a bounded lipschitzian domain in  $\mathbb{R}^n$ . Then there exists a constant  $c = c(\Omega)$  depending only on  $\Omega$ , such that

(6.12) 
$$|\sigma|_{L^{2}(\Omega)} \leq c(\Omega) \left\{ \left| \int_{\Omega} \sigma dx \right| + \sum_{i=1}^{n} \left\| \frac{\partial \sigma}{\partial x_{i}} \right\|_{H^{-1}(\Omega)} \right\}$$

for every  $\sigma \ in \ L^2(\Omega)$ .

## Proof.

Let us denote by  $[\sigma]$  the expression between the brackets on the righthand side of (6.12);  $[\sigma]$  is a norm on  $L^2(\Omega)$ : it is obviously a semi-norm and, if  $[\sigma] = 0$ , then  $\sigma$  is a constant since  $\frac{\partial \sigma}{\partial \mathbf{x}_i} = 0$ ,  $i = 1, \dots, n$ , and

this constant is zero since  $\int_{\Omega} \sigma \, d\mathbf{x} = 0$ .

It is clear that there exists a constant  $c' = c'(\Omega)$  such that

(6.13) 
$$[\sigma] \leq c'(\Omega) |\sigma|_{L^{2}(\Omega)}, \quad \forall \sigma \in L^{2}(\Omega).$$

If we show that  $L^2(\Omega)$  is complete for the norm  $[\sigma]$  then, by the closed graph theorem,  $[\sigma]$  and  $|\sigma|$  will be two equivalent norms on  $L^2(\Omega)$  and (6.12) will be proved.

In order to show that  $L^2(\Omega)$  is complete for the norm  $[\sigma]$ , let us consider a sequence  $\sigma_m$ , which is a Cauchy sequence for this norm. Then the

 $\frac{\partial \mathbf{x}_{m}}{\partial \sigma}$ integrals  $\int_{\Omega} \sigma_{m} dx$  form a Cauchy sequence in R and the derivatives are Cauchy sequences in  $H^{-1}(\Omega)$ :

(6.14) 
$$\int_{\Omega} \sigma_{m} dx \longrightarrow \lambda \quad \text{as } m \longrightarrow \infty,$$
  
(6.15) 
$$\frac{\partial \sigma_{m}}{\partial x_{i}} \longrightarrow \chi_{i} \quad \text{as } m \longrightarrow \infty, \quad \text{in } H^{-1}(\Omega), \quad 1 \le i \le n.$$

Because of (6.15) and the de Rham [1] theorem, there exists some distribution  $\sigma$  such that

$$\chi_i = \frac{\partial \sigma}{\partial x_i}$$
,  $1 \le i \le n$ .

Proposition 1.2 shows that  $\sigma \in L^2(\Omega)$ . We can then choose  $\sigma$  so that

$$\int_{\Omega} \sigma \, d\mathbf{x} = \lambda$$

and it is easy to see that the sequence  $\sigma_m$  converges to this element  $\sigma$  of  $L^{2}(\Omega)$  in the norm  $[\sigma]$ .

Remark 6.1. If  $\Omega$  is not connected, (6.12) is true if we replace  $\int_{\Omega} \sigma d\mathbf{x} d\mathbf{x}$  by

 $\sum_{j} \left| \int_{\Omega_{j}} \sigma \, d\mathbf{x} \right|$ 

where the  $\Omega_i$  are the connected components of  $\Omega$ . For extending Theorem 6.1 to this case we just have to define p by

(6.16) 
$$\int_{\Omega_j} p dx = 0, \quad \forall \Omega_j.$$

6.2 <u>Asymptotic Development of</u>  $u_{\epsilon}$ . From now on we denote by  $u^{0}$  and  $p^{0}$  the solution of Stokes problem (2.6) -(2.9) which satisfies (6.5) (in place of u and p).

We will show that u has an asymptotic development

(6.17) 
$$u_{\varepsilon} = u^{0} + \varepsilon u + \varepsilon u^{2} + \cdots + \varepsilon^{N} u^{N} + \cdots$$

where all the  $u^{i}$  belong to the space  $H_{0}^{1}(\Omega)$ .

(6.18)  $u^0$ ,  $p^0$ , are already known; when  $u^{m-1}$ ,  $p^{m-1}$  are known (m  $\geq$  1), we define  $u^m$  and  $p^m$  as the solutions of the nonhomogeneous Stokes problem

- (6.19)  $u^{m} \in \mathbb{H}_{0}^{1}(\Omega)$ ,  $p^{m} \in L^{2}(\Omega)$ ,
- $(6.20) \qquad -\nu \Delta u^{m} + \text{grad } p^{m} = 0$
- (6.21)  $\operatorname{div} u^{m} = -p^{m-1}$

(6.22) 
$$\int_{\Omega} p^{\mathfrak{m}}(\mathbf{x}) \, d\mathbf{x} = 0.$$

The existence and uniqueness of  $u^m$  and  $p^m$  follow Proposition 2.3 in the two and three-dimensional cases (see also Remark 2.6). The condition (6.22) is useful in two ways: it ensures the complete uniqueness of  $p^m$  which is otherwise only unique up to an additive constant; it also ensures the compatibility condition necessary for the level m + 1:

(6.23) 
$$\int_{\Omega} \operatorname{div} u^{m+1} dx = \int_{\Gamma} u^{m+1} \cdot v d\Gamma = 0 - \int_{\Omega} p^{m} dx.$$
  
We denote by  $u_{\varepsilon}^{N}, p_{\varepsilon}^{N}, N \ge 1$ , the quantities  
(6.24)  $u_{\varepsilon}^{N} = \sum_{m=0}^{N} \varepsilon^{m} u^{m},$   
(6.25)  $p_{\varepsilon}^{N} = \sum_{m=0}^{N} \varepsilon^{m} p^{m}.$ 

Theorem 6.2.

Let  $\Omega$  be a bounded domain of class  $\mathcal{C}^2$  in  $\mathbb{R}^2$  or  $\mathbb{R}^3$ .

Then for each  $m \ge 1$ , there exist functions  $u^m$ ,  $p^m$ , uniquely defined by (6.19) - (6.22).

For each  $N \ge 0$ , as  $\varepsilon \longrightarrow 0$ ,

(6.26) 
$$\frac{\frac{u_{\varepsilon}-u_{\varepsilon}^{N}}{\varepsilon^{N}} \longrightarrow 0 \quad \underline{\text{in the}} \quad \mathbb{H}_{0}^{1}(\Omega) \quad \underline{\text{norm}},}{\varepsilon^{N}}$$
(6.27) 
$$\frac{1}{\varepsilon^{N}} \left(-\frac{\operatorname{div} u_{\varepsilon}}{\varepsilon} - p_{\varepsilon}^{N}\right) \longrightarrow 0 \quad \underline{\text{in the}} \quad L^{2}(\Omega) \quad \underline{\text{norm}}.$$

Proof.

The existence and uniqueness results have been established as previously remarked.

We multiply (6.20) by  $\varepsilon^m$  and add these equations for  $m = 1, \dots, N$ . We then add the resulting equation to the equation satisfied by  $u^0$  and  $p^0$  (formerly denoted by u and p)

$$-\nu\Delta u^0$$
 + grad  $p^0$  = f.

After expanding we obtain

$$-\nu \Delta u_{\epsilon}^{N} - \frac{1}{\epsilon} \text{ grad div } u_{\epsilon}^{N} = f - \epsilon^{N} \text{ grad } p^{N}.$$

By comparison with (6.3) we find

(6.28) 
$$u_{\varepsilon} - u_{\varepsilon}^{N} \in \mathbb{H}_{0}^{1}(\Omega), \quad p^{N} \in L^{2}(\Omega),$$
  
(6.29)  $-\nu \Delta (u_{\varepsilon} - u_{\varepsilon}^{N}) - \frac{1}{\varepsilon} \text{ grad } \operatorname{div} (u_{\varepsilon} - u_{\varepsilon}^{N}) = + \varepsilon^{N} \text{ grad } p^{N}$ 

As done for (6.8), we show that (6.29) is equivalent to

(6.30) 
$$\vee((u_{\varepsilon}-u_{\varepsilon}^{N}, v)) + \frac{1}{\varepsilon} (\operatorname{div}(u_{\varepsilon}-u_{\varepsilon}^{N}), \operatorname{div} v) = -\varepsilon^{N}(p^{N}, \operatorname{div} v), \quad \forall v \in \mathbb{H}_{0}^{1}(\Omega).$$

Putting  $v = u_{\varepsilon} - u_{\varepsilon}^{N}$  in (6.30) we get

$$\begin{split} \| \mathbf{u}_{\varepsilon} - \mathbf{u}_{\varepsilon}^{N} \|^{2} &+ \frac{1}{\varepsilon} \left\| \operatorname{div}(\mathbf{u}_{\varepsilon} - \mathbf{u}_{\varepsilon}^{N}) \right\|^{2} \\ &= -\varepsilon^{N}(\mathbf{p}^{N}, \operatorname{div}(\mathbf{u}_{\varepsilon} - \mathbf{u}_{\varepsilon}^{N})) \\ &\leq \varepsilon^{N} |\mathbf{p}^{N}| \left\| \operatorname{div}(\mathbf{u}_{\varepsilon} - \mathbf{u}_{\varepsilon}^{N}) \right\| \\ &\leq \frac{1}{2\varepsilon} \left\| \operatorname{div}(\mathbf{u}_{\varepsilon} - \mathbf{u}_{\varepsilon}^{N}) \right\|^{2} + \frac{\varepsilon^{2N+1}}{2} \| \mathbf{p}^{N} \|^{2} \end{split}$$

so that

(6.31) 
$$\nabla \| u_{\varepsilon} - u_{\varepsilon}^{N} \|^{2} + \frac{1}{2\varepsilon} |\operatorname{div}(u_{\varepsilon} - u_{\varepsilon}^{N})|^{2} \leq \frac{\varepsilon^{2N+1}}{2} |p^{N}|^{2}.$$

The inequality (6.31) clearly implies (6.26). This, in turn, implies that  $\frac{1}{e^N} \Delta(u_{\varepsilon} - u_{\varepsilon}^N) \longrightarrow 0$  in  $\mathbb{H}^{-1}(\Omega)$  and hence (6.29) shows that

(6.32) 
$$\frac{1}{\varepsilon^{N+1}} \frac{\partial}{\partial x_i} \operatorname{div}(u_\varepsilon - u_\varepsilon^N) \longrightarrow \frac{\partial p^N}{\partial x_i}$$
 in  $H^{-1}(\Omega)$ ,  $1 \le i \le n$ .

But

$$-\frac{1}{\varepsilon} \operatorname{grad} \operatorname{div} u_{\varepsilon}^{N} = -\frac{1}{\varepsilon} \sum_{m=1}^{N} \varepsilon^{m} \operatorname{grad} \operatorname{div} u^{m}$$
$$= \frac{1}{\varepsilon} \sum_{m=1}^{N} \varepsilon^{m} \operatorname{grad} p^{m-1} = \operatorname{grad}(p_{\varepsilon}^{N} - \varepsilon^{N} p^{N}),$$

which along with (6.32) implies that

$$\frac{1}{\varepsilon^{N}} \frac{\partial}{\partial x_{i}} \left( - \frac{\operatorname{div} u_{\varepsilon}}{\varepsilon} - p_{\varepsilon}^{N} \right) \longrightarrow 0 \quad \text{in } \mathbb{H}^{-1}(\Omega)$$

as  $\epsilon \rightarrow 0$ . Finally (6.27) results from (6.22), (6.25), and Lemma 6.1.

Remark 6.2.

Remark 6.1 can be easily adapted to Theorem 6.2: this theorem holds for non-connected sets  $\Omega$ , provided we replace condition (6.22) by the conditions

(6.33) 
$$\int_{\Omega_{j}} p^{m} dx = 0$$

on each connected component  $\Omega_{i}$  of  $\Omega$ .

#### Remark 6.3.

Theorem 6.2 can be extended to higher dimensions, provided one proves the existence of  $u^m$ ,  $p^m$ , satisfying (6.19) - (6.22).

It suffices to show that the divergence operator maps  $\mathbb{H}^1_0(\Omega)$  onto  $L^2(\Omega)/\mathbb{R}$  and this is most probably true although unproved (except for n = 2,3). See also Remark 2.6.

#### 6.3 <u>Numerical Algorithms</u>.

Let us show rapidly how one can extend the algorithms described in Section 5 to algorithms for solving the nonhomogeneous Stokes problems (6.19) - (6.22) which is, at this point, the only difficulty for practical computation of the asymptotic development (6.17) of  $u_c$ .

We only describe the adaptation of the Uzawa algorithm.

Changing our notation, we write problem (6.19) - (6.22) as the problem: to find v,p such that

(6.34)  $\mathbf{v} \in \mathbb{H}_0^1(\Omega), \quad \mathbf{p} \in L^2(\Omega)$ 

 $(6.35) \qquad -\nu\Delta v + \text{grad } p = 0,$ 

(6.36) 
$$div v = \phi$$
,

(6.37) 
$$\int_{\Omega} p(\mathbf{x}) d\mathbf{x} = 0,$$

where  $\phi$  is given with

(6.38) 
$$\int_{\Omega} \phi(\mathbf{x}) \, d\mathbf{x} = 0.$$

The existence and uniqueness of v and p are known. We start the algorithm with any

(6.39) 
$$p^{0} \in L^{2}(\Omega)$$
, such that  $\int_{\Omega} p^{0}(x) dx = 0$ .  
When  $p^{m}$  is known, we define  $v^{m+1}$  and  $p^{m+1}$   $(m \geq 0)$  by  
(6.40)  $\begin{cases} v^{m+1} \in \mathbb{H}_{0}^{1}(\Omega) \text{ and} \\ v((v^{m+1}, w)) - (p^{m}, \operatorname{div} w) = 0, \forall w \in \mathbb{H}_{0}^{1}(\Omega) \end{cases}$ 

(6.41) 
$$\begin{cases} p^{m+1} \in L^{2}(\Omega) \text{ and} \\ (p^{m+1}-p^{m}, \theta) + \rho(\operatorname{div} v^{m+1}-\phi, \theta) = 0, \quad \forall \theta \in L^{2}(\Omega). \end{cases}$$

The equation (6.40) is a Dirichlet problem for  $v^{m+1}$ :

(6.42) 
$$\begin{cases} v^{m+1} \in \mathbb{H}_0^1(\Omega) \\ -v\Delta v^{m+1} = - \text{ grad } p^m \in \mathbb{H}^{-1}(\Omega) \end{cases}$$

and (6.41) gives  $p^{m+1}$  directly as

(6.43) 
$$p^{m+1} = p^m - \rho(\text{div } v^{m+1} - \phi) \in L^2(\Omega).$$

We notice that

(6.44) 
$$\int_{\Omega} p^{m} dx = 0, \quad \forall m \geq 0.$$

Exactly as for Theorem 5.1 (see also Remark 5.1), one can prove the following result.

Theorem 6.3.

If the number  $\rho$  satisfies

$$(6.45) 0 < \rho < \frac{2\nu}{n}$$

then, as  $m \to \infty$ ,  $v^{m+1}$  converges to v in the norm of  $H_0^1(\Omega)$  and  $p^{m+1}$  converges to p in  $L^2(\Omega)$  weakly.

#### CHAPTER II

### Stationary Navier-Stokes Equations

#### INTRODUCTION

In this chapter we will be concerned with the stationary Navier-Stokes equations from the same point of view as in the previous chapter, i.e., existence, uniqueness, and numerical approximation of the solution. However, there are three important differences from the linear case; these are:

- The introduction of the <u>compactness methods</u>. For passing to the limit in the nonlinear term we need strong convergence results; these are obtained by compactness arguments.

- Some technical difficulties related to the nonlinear term and connected with the Sobolev inequalities. Their consequence is a treatment of the equation which varies slightly according to the dimension of the space.

- <u>The non-uniqueness of solutions</u>, in general. Uniqueness occurs only when "the data are small enough, or the viscosity is large enough."

In Section 1 we describe some existence, uniqueness, and regularity results in various situations ( $\Omega$  bounded or not, homogeneous or inhomogeneous equations; ...). In Section 2 we prove a discrete Sobolev inequality and a discrete compactness theorem for step function spaces considered in the approximation (APX 1) of V (approximation of V by finite differences). The similar results for the approximations (APX 2), ..., (APX 4), are already available as consequences of the theorem in the continuous case. (1) Section 3 deals with the approximation of the stationary problem: discretization and resolution of the discretized problems.

(1) For the approximation (APX 5), such results are not proved yet.
#### \$1. EXISTENCE AND UNIQUENESS THEOREMS.

In this section we study some existence and uniqueness results for the stationary (nonlinear) Navier-Stokes equations. The existence results are obtained by constructing approximate solutions to the equation by the Galerkin Method, and then passing to the limit, as in the linear case. As we already said, for passing to the limit, we need, in the nonlinear case, some strong convergence properties of the sequence and these are obtained by compactness methods.

In Section 1.1 we recall the Sobolev inequalities and a compactness theorem for the Sobolev spaces; this theorem is of course the basic tool for the compactness method. In Section 1.2 we give a variational formulation of the homogeneous Navier-Stokes equations (i.e., the Navier-Stokes equations with homogeneous boundary conditions); we study some properties of a nonlinear (trilinear) form which occurs in the variational formulation. We then give a general existence theorem and a rather restricted uniqueness result. In Section 1.3 we consider the case where the set  $\Omega$  is unbounded and we give regularity results for solutions. Section 1.4 deals with inhomogeneous Navier-Stokes equations.

## 1.1 <u>Sobolev Inequalities and Compactness Theorems</u>. Imbedding Theorems.

We recall the Sobolev imbedding theorems which will be used frequently from now on. Let m be an integer and p any finite number greater than or equal to one,  $p \ge 1$ ; then, if  $\frac{1}{p} - \frac{m}{n} = \frac{1}{q} > 0$  the space  $W^{m,p}(\mathbb{R}^n)$  is included in  $L^q(\mathbb{R}^n)$  and the injection is continuous. If  $u \in W^{m,p}(\mathbb{R}^n)$  and  $\frac{1}{p} - \frac{m}{n} = 0$  then u belongs to  $L^q(\partial)$  for any bounded set  $\partial^{-}$  and any q,  $1 \le q < \infty$ . If  $\frac{1}{p} - \frac{m}{n} < 0$  then a function in  $W^{m,p}(\mathbb{R}^n)$  is almost everywhere equal to a continuous function; such a function has also some Hölder or Lipschitz continuity properties but such properties will not be used here; if a function belongs to  $W^{m,p}(\mathbb{R}^n)$  with  $\frac{1}{p} - \frac{m}{n} < 0$  then the derivatives of order  $\alpha$  belong to  $W^{m-\alpha,p}(\mathbb{R}^n)$  and some imbedding results of preceding type hold for these derivatives if  $\frac{1}{p} - \frac{m-\alpha}{n} > 0$ . For  $u \in W^{m,p}(\mathbb{R}^n)$ ,  $m \ge 1$ ,  $1 \le p < \infty$ 

136

$$(1.1) \begin{cases} \text{if } \frac{1}{p} - \frac{m}{n} = \frac{1}{q} > 0, \quad |u|_{L^{q}(\mathbb{R}^{n})} \leq c(m,p,n) ||u||_{W^{m,p}(\mathbb{R}^{n})}, \\ \text{if } \frac{1}{p} - \frac{m}{n} = 0, \quad |u|_{L^{q}(\mathcal{O})} \leq c(m,p,n,q,\mathcal{O}) ||u||_{W^{m,p}(\mathbb{R}^{n})} \\ \forall \text{ bounded set } \mathcal{O} \subset \mathbb{R}^{n}, \forall q, \quad 1 \leq q < \infty, \\ \text{if } \frac{1}{p} - \frac{m}{n} < 0, \quad |u|_{\mathcal{O}^{0}}(\mathcal{O}) \leq c(m,n,p,\mathcal{O}) ||u||_{W^{m,p}(\mathbb{R}^{n})}, \\ \forall \text{ bounded set } \mathcal{O}, \quad \mathcal{O} \subset \mathbb{R}^{n}. \end{cases}$$

If  $\Omega$  is any open set of  $\mathbb{R}^n$ , results similar to (1.1) can usually be obtained if  $\Omega$  is sufficiently smooth so that

(1.2) There exists a continuous linear prolongation operator  

$$\Pi \in \mathbb{Z}(W^{m,p}(\Omega), W^{m,p}(\mathbb{R}^{n})).$$

Property (1.2) is satisfied by a locally lipschitzian set  $\Omega$ . When (1.2) is satisfied, the properties (1.1) applied to  $\Pi u$ ,  $u \in W^{m,p}(\Omega)$  give in particular, assuming that  $u \in W^{m,p}(\Omega)$ ,  $m \geq 1$ , 1 , and (1.2) holds:

$$(1.3)\left\{\begin{array}{l} \text{if } \frac{1}{p} - \frac{m}{n} = \frac{1}{q} > 0, \quad |u|_{L^{q}(\Omega)} \leq c(m,p,n,\Omega) \, ||u||_{W^{m},P(\Omega)},\\ \text{if } \frac{1}{p} - \frac{m}{n} = 0, \quad |u|_{L^{q}(\partial)} \leq c(m,p,n,q,\partial,\Omega) \, ||u||_{W^{m},P(\Omega)},\\ \text{any } q, \quad 1 \leq q < \infty, \text{ any bounded set } \theta \subset \overline{\Omega},\\ \text{if } \frac{1}{p} - \frac{m}{n} < 0, \quad |u|_{\mathcal{C}^{0}(\partial)} \leq c(m,p,n,q,\Omega,\partial) \, ||u||_{W^{m},P(\Omega)},\\ \text{any bounded set } \theta, \quad \theta \subset \overline{\Omega}.\end{array}\right\}$$

When  $u \in \tilde{W}^{m,p}(\Omega)$ , the function  $\tilde{u}$  which is equal to u in  $\Omega$  and to 0 in  $(\Omega)$ , belongs to  $W^{m,p}(\mathbb{R}^n)$  and hence the properties (1.3) are valid without any hypothesis on  $\Omega$ .

The case of particular interest for us is the case p = 2, m = 1, i.e., the case  $H_0^1(\Omega)$ . Without any regularity property required for  $\Omega$  we have for  $\mathbf{u} \in \mathbf{H}_0^1(\Omega)$ 

$$(1.4) \qquad \begin{pmatrix} n = 2, |u|_{L^{q}(\Omega)} \leq c(q, \mathcal{O}, \Omega) ||u||_{H^{1}_{0}(\Omega)} \\ \forall \text{ bounded set } \mathcal{O} \subseteq \Omega, \forall q, 1 \leq q < \infty \\ n = 3, |u|_{L^{6}(\Omega)} \leq c(\Omega) ||u||_{H^{1}_{0}(\Omega)} \\ n = 4, |u|_{L^{4}(\Omega)} \leq c(\Omega) ||u||_{H^{1}_{0}(\Omega)} \\ n \geq 3, |u|_{L^{\frac{2n}{n-2}(\Omega)}} \leq c(\Omega) ||u||_{H^{1}_{0}(\Omega)}.$$

Compactness Theorems.

 $\frac{\text{Theorem: } 1.1}{\text{Let } \Omega \text{ be any bounded open set of } \mathbb{R}^{n} \text{ satisfying (1.2). Then the imbedding}}$   $(1.5) \qquad \qquad \mathbb{W}^{1,p}(\Omega) \subset L^{q_{1}}(\Omega)$ 

#### is compact for any bounded open set $\Omega$ .

As a particular case of Theorem 1.1 we notice that for any unbounded set  $\Omega$ , if  $u \in W^{1,p}(\Omega)$ , then the restriction of u to  $\mathcal{O}$ ,  $\mathcal{O} \subset \overline{\mathcal{O}} \subset \Omega$ ,  $\mathcal{O}$  bounded, belongs to  $L^{q_1}(\mathcal{O})$  and this restriction mapping is compact

(1.7) 
$$W^{1,p}(\Omega) \longrightarrow L^{q_1}(\Omega)$$

(same values of p and  $q_1$ ).

For all the preceding properties of Sobolev spaces, the reader is referred to the references mentioned in the first section of Chapter I (see also at the end the comments on Chapter I).

#### 1.2 The Homogeneous Navier-Stokes Equations.

Let  $\Omega$  be a lipschitzian, bounded open set in  $\mathbb{R}^n$  with boundary  $\Gamma$ , let  $f \in L^2(\Omega)$  be a given vector function. We are looking for a vector function

 $u = (u_1, \dots, u_n)$  and a scalar function p, representing the velocity and the pressure of the fluid, which are defined in  $\Omega$  and satisfy the following equations and boundary conditions:

(1.8) 
$$-\nu\Delta u + \sum_{i=1}^{n} u_{i}D_{i}u + \text{grad } p = f \text{ in } \Omega$$

$$(1.9) \qquad \text{div } u = 0 \quad \text{in } \Omega$$

(1.10) 
$$u = 0$$
 on  $\Gamma$ .

Exactly as in Section 2.1 of Chapter I, if f,u,p are smooth functions satisfying (1.8) - (1.10) then  $u \in V$  and, for each  $v \in \mathcal{D}$ ,

(1.11) 
$$V((u,v)) + b(u,u,v) = (f,v)$$

where

(1.12) 
$$b(u,v,w) = \sum_{i,j=1}^{n} \int_{\Omega} u_{i}(D_{i}v_{j})w_{j} dx.$$

A continuity argument shows moreover that equation (1.11) is satisfied by any  $v \in V$ . Conversely, if u is a smooth function in V such that (1.11) holds for each  $v \in V$ ; then because of Proposition 1.3, Chapter I, there exists a distribution p such that (1.8) is satisfied, and (1.9)-(1.10) are satisfied since  $u \in V$ .

For u and v in V, the expression b(u,u,v) does not necessarily make sense and then the variational formulation of (1.8) - (1.10) is not exactly: "to find  $u \in V$  such that (1.11) holds for each  $v \in V$ ." The variational formulation will be slightly different, and this will be stated after studying some properties of the form b(u,v,w).

Let us introduce first the following space:

(1.13)  $\tilde{V}$  = the closure of  $\mathcal{V}$  in  $\mathbb{H}^1_0(\Omega) \cap \mathbb{L}^n(\Omega)$ ; of course  $\mathbb{H}^1_0(\Omega) \cap \mathbb{L}^n(\Omega)$  and  $\tilde{V}$  are equipped with the norm

(1.14) 
$$\|\mathbf{u}\| + \|\mathbf{u}\|_{\mathbf{H}_0^1(\Omega)} = \mathbb{L}^n(\Omega)$$

In general  $\tilde{V}$  is a subspace of V, different from V but, because of (1.4),  $\tilde{V} = V$  for n = 2,3, or 4 (and  $\Omega$  bounded). (1.15)  $V_s =$  the closure of  $\mathcal{V}$  in  $\mathbb{H}^1_0(\Omega) \cap \mathbb{H}^S(\Omega)$ ,  $(s \ge 1)$ ;

it is understood again that  $\mathbb{H}^1_0(\Omega)\cap\mathbb{H}^S(\Omega)$  and  $V_S$  are equipped with the hilbertian norm

(1.16) 
$$\{ \| u \|^{2} + \| u \|^{2} \}^{\frac{1}{2}};$$
 
$$\mathbb{H}^{1}_{0}(\Omega) = \mathbb{H}^{S}(\Omega) \}^{\frac{1}{2}};$$

 $V_{s}$  is included in V.

#### The Trilinear Form b.

The form b is trilinear and continuous on various spaces among the spaces V,  $\tilde{V}$ ,  $V_s$ . The most convenient result concerning b is the following result which is independent of any property of  $\Omega$ .

# Lemma 1.1. The form b is defined and trilinear continuous on $H_0^1(\Omega) \times H_0^1(\Omega) \times (H_0^1(\Omega) \cap \mathbb{L}^n(\Omega)), \Omega$ bounded or unbounded, any dimension of space $\mathbb{R}^n$ .

Proof.  
If 
$$u, v \in V$$
 and  $w \in \tilde{V}$ , then because of (1.4)  $(n \ge 3)$ :  
 $u_i \in L^{\frac{2n}{n-2}}(\Omega)$ ,  $D_i v_j \in L^2(\Omega)$ ,  $w_j \in L^n(\Omega)$ ,  $1 \le i, j \le n$ .

By the Hölder inequality,  $u_i(D_iv_j)w_j$  belongs to  $L^1(\Omega)$  and

$$(1.17) \qquad \left| \int_{\Omega} u_{\mathbf{i}} D_{\mathbf{i}} v_{\mathbf{j}} w_{\mathbf{j}} d\mathbf{x} \right| \leq \left| u_{\mathbf{i}} \right|_{\mathbf{L}^{n-2}(\Omega)} \left| D_{\mathbf{i}} v_{\mathbf{j}} \right|_{\mathbf{L}^{2}(\Omega)} \left| w_{\mathbf{j}} \right|_{\mathbf{L}^{n}(\Omega)}$$

Then b(u,v,w) is well defined and

(1.18) 
$$|b(\mathbf{u},\mathbf{v},\mathbf{w})| \leq c(\mathbf{n}) \|\mathbf{u}\| \|\mathbf{v}\| \|\mathbf{w}\| \|\mathbf{w}\|$$
  
 $\mathbf{H}_0^1(\Omega) = \mathbf{H}_0^1(\Omega) = \mathbf{H}_0^1(\Omega) \cap \mathbf{L}^n(\Omega)$ 

The form b is obviously trilinear and (1.18) ensures the continuity of b. When n = 2, we have the same result,  $(\mathbb{H}_0^1(\Omega) \cap \mathbb{L}^2(\Omega) = \mathbb{H}_0^1(\Omega))$ , but

(1.17) must be replaced by

(1.19) 
$$\left|\int_{\Omega} u_{\mathbf{i}} D_{\mathbf{i}} v_{\mathbf{j}} w_{\mathbf{j}} d\mathbf{x}\right| \leq \left|u_{\mathbf{i}}\right|_{L^{4}(\Omega)} \left|D_{\mathbf{i}} v_{\mathbf{j}}\right|_{L^{2}(\Omega)} \left|w_{\mathbf{j}}\right|_{L^{4}(\Omega)}$$

In particular one has

Lemma 1.2.

For any open set  $\Omega$ , b is a trilinear continuous form on  $V \times V \times \tilde{V}$ . If  $\Omega$  is bounded and  $n \leq 4$ , b is trilinear continuous on  $V \times V \times V$ .

We will prove, when needed, other properties of b similar to those given before; the proof will be always the same as in Lemma 1.1 (use of Hölder's inequality and the imbedding theorem (1.4)).

We denote by B(u,v),  $u,v \in \mathbb{H}_0^1(\Omega)$ , the linear continuous form on  $\widetilde{V}$  defined by

(1.20) 
$$\langle B(u,v),w \rangle = b(u,v,w), \quad u,v \in \mathbb{H}_0^1(\Omega), \quad \forall w \in \mathbb{V}.$$

For u = v, we write

(1.21)  $B(u) = B(u,u), u \in \mathbb{H}_0^1(\Omega).$ 

Another fundamental property of b is the following

<u>Lemma 1.3</u>.

For any open set  $\Omega$ ,

(1.22) 
$$b(u,v,v) = 0, \quad \forall u \in V, \quad v \in \mathbb{H}_0^1(\Omega) \cap \mathbb{L}^1(\Omega)$$

(1.23)  $b(u,v,w) = -b(u,w,v), \quad \forall u \in V, v,w, \in \mathbb{H}_0^1(\Omega) \cap \mathbb{L}^n(\Omega).$ 

Proof.

Property (1.23) is a consequence of (1.22) when we replace v by v + w, and we use the multilinear properties of b.

In order to prove (1.22), it suffices to show this equality for  $u \in \mathcal{V}$ and  $v \in \mathfrak{D}(\Omega)$ . But for such u and v

(1.24)  
$$\int_{\Omega} u_{i} D_{i} v_{j} v_{j} dx = \int_{\Omega} u_{i} D_{i} \frac{(v_{j})^{2}}{2} dx$$
$$= -\frac{1}{2} \int_{\Omega} D_{i} u_{i} (v_{j})^{2} dx ,$$
$$b(u, v, v) = -\frac{1}{2} \sum_{j=1}^{n} \int_{\Omega} div u(v_{j})^{2} dx = 0.$$

#### Variational Formulation.

For  $\Omega$  bounded, and n arbitrary, we associate with (1.8) - (1.10) the problem

(1.25) To find 
$$u \in V$$
 such that

$$V((u,v)) + b(u,u,v) = (f,v), \forall v \in V,$$

(f given in  $\mathbb{L}^2(\Omega)$ ). It is clear from (1.11) and (1.13) that if u and p are smooth functions satisfying (1.8) - (1.10), then u satisfies (1.25). Conversely if  $u \in V$  satisfies (1.25), then

(1.26) 
$$\langle -v\Delta u + \sum_{i} u_{i}D_{i}u - f, v \rangle = 0, \quad \forall v \in \mathcal{V};$$

 $\Delta u \in H^{-1}(\Omega)$ ,  $f \in L^{2}(\Omega)$ , and  $u_{i}D_{i}u \in L^{n'}(\Omega)$   $(\frac{1}{n'} = 1 - \frac{1}{n})$ , since  $u_{i} \in \mu^{\frac{24u}{n-2}}(\Omega)$  by (1.4) and  $D_{i}u \in L^{2}(\Omega)$ . According then to Proposition 1.3 and Remark 1.5 of Chapter I, there exists a distribution  $p \in L^{1}_{loc}(\Omega)$ , such that (1.8) is satisfied in the distribution sense; then (1.9) and (1.10) are satisfied respectively in the distribution and the trace theorem senses.

Theorem 1.2.

 $\begin{array}{cccc} \underline{\text{Let}} & \Omega & \underline{\text{be a bounded set in }} \mathbb{R}^n & \underline{\text{and let}} & f & \underline{\text{be given in }} \mathbb{H}^{-1}(\Omega) \, . \\ \hline \underline{\text{Then Problem (1.25) has at least one solution }} & u \in \mathbb{V} & \underline{\text{and there exists a}} \\ \underline{\text{distribution }} & p \in L^1_{loc}(\Omega) & \underline{\text{such that (1.8)-(1.9) are satisfied.}} \end{array}$ 

## Proof.

We have only to prove the existence of u; the existence of p and the interpretation of (1.8)-(1.9) have already been shown.

The existence of u is proved by the Galerkin method: we construct an approximate solution of (1.25) and then pass to the limit.

The space  $\tilde{V}$  is separable as a subspace of  $\mathbb{H}^1_0(\Omega)$ . Because of (1.13) there exists a sequence  $w_1, \dots, w_m, \dots$ , of elements of  $\mathcal{V}$  which is free and total in  $\tilde{V}$ . This sequence is also free and total in V.

For each fixed integer  $\,m\geq 1\,,\,$  we would like to define an approximate solution  $u_{m}^{}$  of (1.25) by

(1.27) 
$$u_{m}^{i} = \sum_{i=1}^{m} \xi_{i,m} w_{i}, \quad \xi_{i,m} \in \mathbb{R}$$

(1.28)  $v((u_m, w_k)) + b(u_m, u_m, w_k) = \langle f, w_k \rangle, k = 1, \dots, m.$ 

The equations (1.27)-(1.28) are a system of nonlinear equations for  $\xi_{1,m}, \dots, \xi_{m,m}$ , and the existence of a solution of this system is not obvious, but follows from the next lemma.

Lemma 1.4.

Let X be a finite dimensional Hilbert space with scalar product [•,•] and norm [•] and let P be a continuous map from X into itself such that

(1.29)  $[P(\xi),\xi] > 0 \quad for \quad [\xi] = k > 0.$ 

Then there exists  $\xi \in X$ ,  $[\xi] \leq k$ , such that

(1.30) 
$$P(\xi) = 0.$$

The proof of Lemma 1.4 follows the proof of Theorem 1.2. We apply this lemma for proving the existence of  $u_m$ , as follows:

X = the space spanned by  $w_1, \dots, w_m$ ; the scalar product on X is the scalar product ((.,.)) induced by V, and P = P<sub>m</sub> is defined by

$$[P_{m}(u),v] = ((P_{m}(u),v)) = v((u,v)) + b(u,u,v) - (f,v), \quad \forall u,v \in X.$$

The continuity of the mapping  $P_m$  is obvious; let us show (1.29).

$$[P_{m}(u), u] = v || u ||^{2} + b(u, u, u) - \langle f, u \rangle$$
  
= (by (1.22))  
= v || u ||^{2} - \langle f, u \rangle  
$$\geq v || u ||^{2} - || f ||_{V}, || u ||_{V},$$

(1.31)  $[P_{m}(u), u] \geq ||u|| (v||u|| - ||f||_{V'}).$ 

It follows that  $[P_m u, u] > 0$  for ||u|| = k, and k large enough; more precisely,  $k > \frac{1}{v} ||f||_{V}$ . The hypotheses of Lemma 1.4 are satisfied and there exists a solution  $u_m$  of (1.27)-(1,28).

Passage to the Limit.

We multiply (1.28) by  $\xi_{k,m}$  and add the corresponding equalities for  $k = 1, \dots, m$ ; this gives

$$v \| u_{m} \|^{2} + b(u_{m}, u_{m}, u_{m}) = \langle f, u_{m} \rangle$$

or, because of (1, 24),

$$v \| u_{m} \|^{2} = \langle f, u_{m} \rangle \leq \| f \|_{V}, \| u_{m} \|$$

We obtain then the a priori estimate:

(1.32) 
$$\|u_{m}\| \leq \frac{1}{\nu} \|f\|_{V}$$

Since the sequence  $u_m$  remains bounded in.V, there exists some u in V and a subsequence  $m' \longrightarrow \infty$  such that

(1.33) 
$$u_m, \longrightarrow u$$
 for the weak topology of V.

The compactness theorem 1.1 shows in particular that the injection of V into  $L^2(\Omega)$  is compact, so we have also

(1.34) 
$$u_{m} \rightarrow u$$
 in the norm of  $\mathbb{L}^{2}(\Omega)$ .

Let us admit for a short time the following lemma .

Lemma 1.5.  
If 
$$u_{\mu}$$
 converges to  $u$  in  $V$  weakly and in  $\mathbb{L}^{2}(\Omega)$  strongly, then  
(1.35)  $b(u_{\mu}, u_{\mu}, v) \longrightarrow b(u, u, v), \forall v \in \mathcal{V}.$ 

Then we can pass to the limit in (1.28) with the subsequence  $m' \rightarrow \infty$ . From (1.33), (1.34), (1.35) we find that

(1.36) 
$$v((u,v)) + b(u,u,v) = \langle f,v \rangle$$

for any  $v = w_1, \dots, w_m, \dots$  Equation (1.36) is also true for any v which is a linear combination of  $w_1, \dots, w_m, \dots$ . Since these combinations are dense in  $\tilde{V}$ , a continuity argument shows finally that (1.36) holds for each  $v \in \tilde{V}$  and that u is a solution of (1.25).

#### Proof of Lemma 1.4.

This is an easy consequence of the Brouwer fixed point theorem.

Suppose that P has no zero in the ball D of X centered at 0 and with radius k. Then the following application

$$\xi \mapsto S(\xi) = -k \frac{P(\xi)}{[P(\xi)]}$$

maps D into itself and is continuous. The Brouwer theorem implies then that S has a fixed point in D: there exists  $\xi_0 \in D$ , such that

$$-k \frac{P(\xi_0)}{[P(\xi_0)]} = \xi_0$$

If we take the norm of both sides of this equation we see that  $[\xi_0] = k$ , and if we take the scalar product of each side with  $\xi_0$ , we find

$$[\xi_0]^2 = k^2 = -k \frac{[P(\xi_0), \xi_0]}{[P(\xi_0)]}$$

This equality contradicts (1.29) and thus  $P(\xi)$  must vanish at some point of D.

#### Proof of Lemma 1.5.

It is easy to show, as for (1.22)-(1.23), that

$$b(u_{\mu}, u_{\mu}, v) = -b(u_{\mu}, v, u_{\mu})$$
$$= -\sum_{i,j=1}^{n} \int_{\Omega} u_{\mu i} u_{\mu j} D_{i} v_{j} dx.$$

But  $u_{\mu i}$  converges to  $u_i$  in  $L^2(\Omega)$  strongly; since  $D_i v_j \in L^{\infty}(\Omega)$ , it is easy to check that

$$\int_{\Omega} u_{\mu i} u_{\mu j} D_{i} v_{j} dx \longrightarrow \int_{\Omega} u_{i} u_{j} D_{i} v_{j} dx.$$

Hence  $b(u_{ij},v,u_{ij})$  converges to b(u,v,u) = -b(u,u,v).

#### Uniqueness.

For uniqueness we only have the following result:

Theorem 1.3.If 
$$n \le 4$$
 and if  $\nu$  is sufficiently large or f "sufficiently small" so that(1.37) $\nu^2 > c(n) ||f||_V$ 

then there exists a unique solution u of (1.25).

The constant c(n) in (1.37) is the constant c(n) in (1.18); its estimation is connected with the estimation of the constants in (1.4) and this is given for instance in Lions [1].

Proof of Theorem 1.3. We can take v = u in (1.25) since  $\tilde{V} = V$  for  $n \le 4$ ; we obtain with (1.22) (1.38)  $v \|u\|^2 = \langle f, u \rangle \le \|f\|_{V^1} \|u\|$  so that any solution u of (1.25) satisfies

(1.39) 
$$||u|| \leq \frac{1}{\nu} ||f||_{V}$$

Now let  $u_*$  and  $u_{**}$  be two different solutions of (1.25) and let  $u = u_* - u_{**}$ . We substract the equations (1.25) corresponding to  $u_*$  and  $u_{**}$ and we obtain

(1.40) 
$$v((u,v)) + b(u,u,v) + b(u,u,v) = 0, \forall v \in V.$$

We take v = u in (1.40) and use again (1.22); hence:

$$v | u |^{2} = -b(u, u_{*}, u).$$

With (1.18) and (1.39) (for  $u = u_*$ ) this gives

$$\begin{array}{c|c} v \| \mathbf{u} \|^{2} \leq c(\mathbf{r}) \| \mathbf{u} \|^{2} \| \mathbf{u}_{*} \| \\ \leq \frac{c(\mathbf{n})}{v} \| \mathbf{f} \|_{V}, \| \mathbf{u} \|^{2}, \\ (v - \frac{c(\mathbf{n})}{v} \| \mathbf{f} \|_{V}) \| \mathbf{u} \|^{2} \leq 0 \end{array}$$

Because of (1.37) this inequality implies  $\|u\| = 0$ , which means  $u_* = u_{**}$ .

#### Remark 1.1.

The solution of (1.25) is probably not unique if (1.37) is not satisfied or at least for  $\nu$  small enough (f fixed).

Using the bifurcation theory, it has been proven for problems very similar to (1.25) that for  $\nu$  greater than some critical value  $\nu_{\star}$ , the solution is unique and, for  $\nu$  smaller than this value there exist several solutions. This was proved by P. Rabinowitz for the stationary convection problem and by Velte for stationary inhomogeneous Navier-Stokes equations. See Remark 1.7 and the references cited in the "Comments and Bibliography".

Remark 1.2.

For n > 4, Theorem 1.2 shows the existence of solutions u of (1.25) satisfying (1.39): the majoration (1.32) and (1.33) give indeed:

(1.41) 
$$\|\mathbf{u}\| \leq \lim_{m' \to \infty} \|\mathbf{u}_m\| \leq \frac{1}{\nu} \|\mathbf{f}\|_{V'}$$

Nevertheless, the proof of Theorem 1.3 cannot be extended to this case; (1.40) holds for each  $v \in \tilde{V}$  and it is not possible to take v = u.

1.3 <u>The Homogeneous Navier-Stokes Equations</u> (continued). The Unbounded Case.

We can study the case for  $\ \Omega$  unbounded by introducing the same spaces as in Section 2.3, Chapter I. Let us recall that

(1.42) Y =the completed space of  $\mathcal{V}$  for the norm  $\|\cdot\|$ .

Let us consider also the space Y

(1.43)  $\tilde{Y}$  = the closure of  $\mathcal{V}$  in the space  $Y \cap \mathbb{L}^{n}(\Omega)$  equipped with the norm

(1.44) 
$$\|u\| + \|u\|_{\mathbb{L}^{n}(\Omega)}$$
 (1)

We recall that because of Lemma 2.3, Chapter I, we have the continuous injection

(1.45) 
$$Y \subset \{u \in \mathbb{L}^{\alpha}(\Omega), D_{i} u \in \mathbb{L}^{2}(\Omega), 1 \leq i \leq n\}$$

for  $n \geq 3$ , where

$$(1.46) \qquad \qquad \alpha = \frac{2n}{n-2} .$$

Since  $\Omega$  is unbounded the spaces  $L^{\gamma}(\Omega)$  are not decreasing with increasing  $\gamma$  as in the bounded case and  $\tilde{Y} \neq Y$  even for  $n \leq 4$ . Lemma 1.2 cannot be extended to the unbounded case; however, we have:

Lemma 1.6. For  $n \ge 3$ , the form b is defined and trilinear continuous on  $Y \times Y \times \tilde{Y}$  and (1.47)  $b(u,v,v) = 0, \forall u \in Y, v \in \tilde{Y},$ (1.48)  $b(u,v,w) = -b(u,w,v), \forall u \in Y, v,w \in \tilde{Y}.$ Proof.

The inequality (1.17) (n > 3) is valid; for  $u, v \in Y$ ,  $w \in \widetilde{Y}$  we then have  $\left| \int_{\Omega} u_i D_i v_j w_j \, dx \right| \le c \|u\|_Y \, \|v\|_Y \, \|w\|_{\widetilde{Y}}$ 

so that

$$(1.49) |b(u,v,w)| \leq c(n) ||u||_{Y} ||v||_{\widetilde{Y}} .$$

(1) For a smooth open set  $\Omega$ ,  $\tilde{Y}$  is probably equal to  $Y \cap L^{n}(\Omega)$ , but this result is not proved.

The relations (1.47) and (1.48) are proved exactly as (1.22) and (1.23): we prove them for  $u,v,w \in \mathbb{Q}^{\times}$  and then pass to the limit.

The variational formulation of Problem (1.8) - (1.10) for  $\Omega$  unbounded and  $n \ge 3$  is set as follows:

(1.50)

To find  $u \in Y$  such that '

 $v((u,v)) + b(u,u,v) = \langle f,v \rangle, \quad \forall v \in \tilde{Y}.$ 

Theorem 1.4.

Let  $\Omega$  be any open set in  $\mathbb{R}^n$ ,  $n \ge 3$ , and let f be given in Y', the dual space of Y.

Then there exists at least one u in Y which satisfies (1.50).

Proof.

The proof is very similar to the proof of Theorem 1.2 (the bounded case). There exists a sequence  $w_1, \dots, w_m, \dots$ , of elements of  $\mathcal{Y}$  which is free and total in  $\tilde{Y}$  and hence in Y; this sequence is not perhaps the same sequence as before.

We define an approximate solution  $\mathbf{u}_{m}$  by

(1.51) 
$$u_{m} = \sum_{i=1}^{m} \xi_{i,m} w_{i}, \quad \xi_{i,m} \in \mathbb{R},$$

(1.52) 
$$\nu((u_m, w_k)) + b(u_m, u_m, w_k) = \langle f, w_k \rangle, \quad k = 1, \cdots, m.$$

The existence of  $u_m$  satisfying (1.51)-(1.52) is proved exactly as before, using Lemma 1.4. We have then an <u>a priori</u> estimate analogous to (1.32):

(1.53) 
$$\|u_m\| \leq \frac{1}{\nu} \|f\|_Y$$
,  $(\|\cdot\| = \text{the norm in } Y)$ .

There exists therefore a subsequence  $m' \longrightarrow \infty$  and an element  $u \in Y$  such that

$$(1.54) \qquad \qquad u_{m} \xrightarrow{} u \text{ weakly in } Y.$$

The proof finishes as in the bounded case, except for the passage to the limit in the nonlinear term  $b(u_{m'}, u_{m'}, v)$ ; it is not true that  $u_{m'}$  converges to u in  $\mathbb{L}^2(\Omega)$  strongly since u does not even belong to  $\mathbb{L}^2(\Omega)$ , in general  $(Y \in \mathbb{L}^2(\Omega))$ . Nevertheless, we have <u>Lemma</u> <u>1.7</u>.

$$\underline{\text{If}} \quad u_{\mu} \quad \underline{\text{converges to}} \quad u \quad \underline{\text{in Y weakly, then}},$$

(1.55) 
$$b(u_{\mu}, u_{\mu}, v) \longrightarrow b(u, u, v), \quad \forall v \in \mathcal{V}.$$

Proof.

We can show that

(1.56) 
$$u_{\mu} \longrightarrow u \text{ in } L^{2}_{loc}(\overline{\Omega}) \text{ strongly,}$$

which means that

(1.57) 
$$u_{1} \longrightarrow u \text{ in } L^{2}(\mathcal{O}),$$

for each bounded set  $\partial \subset \Omega$ .

Actually, let  $\psi \in \mathcal{P}(\mathbb{R}^n)$ ,  $\psi = 1$  on  $\mathcal{O}$ , and let  $\Omega'$  be a bounded subset of  $\Omega$  containing the support of  $\psi$ . Then the functions  $\psi_{\mu}$  belong to  $H^1_0(\Omega')$  and since  $u_{\mu}$  converges to u weakly in Y,

$$\psi u_{\mu} \longrightarrow \psi u$$
, weakly in  $H_0^1(\Omega^{\prime})$ .

Hence  $\psi u_{\mu} \longrightarrow \psi u$  strongly in  $L^{2}(\Omega')$ ; in particular

$$\int_{\mathcal{O}} |u_{\mu} - u|^2 dx \leq \int_{\Omega'} \psi^2 |u_{\mu} - u|^2 dx \longrightarrow 0,$$

and (1.57) follows.

Since  $u_{\mu}$  converges to u for the L<sup>2</sup> norm, on the support of v, the convergence (1.55) is now proved as in the bounded case:

$$b(u_{\mu}, u_{\mu}, v) = -b(u_{\mu}, v, u_{\mu}) \longrightarrow -b(u, v, u) = b(u, u, v).$$

Remark 1.3.

For n = 2, an element u of Y does not belong in general to any space  $\mathbb{L}^{\beta}(\Omega)$ . For this reason the proof of Lemma 1.6 fails and b is not defined on  $Y \times Y \times \tilde{Y}$ .

We can replace (1.50) by the problem: to find  $u \in Y$  such that

$$\forall ((u,v)) + b(u,u,v) = \langle f,v \rangle, \quad \forall v \in \mathcal{V}.$$

The same proof as for Theorem 1.4 shows that such a u always exists provided f is given in Y'.

Remark 1.4.

Since  $Y \neq \tilde{Y}$  (for any n), we cannot put v = u in (1.50). Therefore the proof of Theorem 1.2 cannot be extended to the unbounded case even for n < 4.

#### Regularity of the Solution.

If the dimension n of the space is less than or equal to three, we can obtain some information about any solution u of (1.25) or (1.50) by reiterating the following simple procedure: the information we have on u gives us some regularity property of the nonlinear term

$$\sum_{i=1}^{n} u_{i} D_{i} u.$$

We then write (1.8) - (1.10) as

(1.58)	-v∆u +	grad	p	a	f	- 1	Σ [=1	<sup>u</sup> i	D <sub>i</sub> u,	in	Ω,
--------	--------	------	---	---	---	--------	----------	----------------	-------------------	----	----

(1.59)  $\operatorname{div} u = 0$ , in  $\Omega$ ,

(1.60) 
$$u = 0$$
 on  $\Gamma$ ;

using the available regularity properties of f and Proposition 2.2 and Theorem 2.4 of Chapter I we obtain new information on the regularity of u. If the properties of u thus obtained are better than before, we can reiterate the procedure.

Let us show, for example, the following result:

Proposition 1.1.

 $\begin{array}{c} \underline{\operatorname{Let}} \ \Omega \ \underline{\operatorname{be}} \ \underline{\operatorname{an open set}} \ \underline{\operatorname{of}} \ \underline{\operatorname{class}} \ \mathcal{V}^{\infty} \ \underline{\operatorname{in}} \ \mathbb{R}^2 \ \underline{\operatorname{or}} \ \mathbb{R}^3 \ \underline{\operatorname{and}} \ \underline{\operatorname{let}} \ f \ \underline{\operatorname{be}} \ \underline{\operatorname{given}} \\ \underline{\operatorname{in}} \ \mathcal{O}^{\infty}(\overline{\Omega}) \ . \\ \underline{\operatorname{Then any solution}} \ \{u,p\} \ \underline{\operatorname{of}} \ (1.8), (1.25) \ \underline{\operatorname{belongs to}} \ \mathbb{C}^{\infty}(\overline{\Omega}) \times \mathcal{V}^{\infty}(\overline{\Omega}) \ . \\ \underline{\operatorname{Proof.}} \\ \underline{\operatorname{Proof.}} \\ \underline{\operatorname{Let}} \ us \ \mathrm{start} \ first \ with \ the \ case \ where \ \Omega \ is \ \mathrm{bounded.} \\ \mathrm{The nonlinear term} \ \sum_{i=1}^n u_i D_i u \ is \ also \ equal \ to \ \sum_{i=1}^n D_i (u_i u), \ because \ of \ i=1 \\ (1.59). \ \mathrm{If} \ n = 2, \ u_i \ \mathrm{belongs to} \ L^{\alpha}(\Omega) \ for \ \mathrm{any} \ \alpha, \ 1 \leq \alpha < +\infty \ (by \ (1.4)), \\ \mathrm{and \ then} \ u_i u_j \ \mathrm{belongs to} \ L^{\alpha}(\Omega) \ for \ \mathrm{any such} \ \alpha, \ \mathrm{and} \ D_i (u_i u_j) \ \mathrm{belongs to} \\ \mathbb{W}^{-1,\alpha}(\Omega) \ for \ \mathrm{any such} \ \alpha. \ \mathrm{Proposition} \ 2.2, \ \mathrm{Chapter} \ \mathrm{I}, \mathrm{shows us \ that} \ u \ \mathrm{belongs} \\ \mathrm{then to} \ \mathbb{W}^{1,\alpha}(\Omega), \ \mathrm{and} \ p \ \mathrm{belongs to} \ L^{\alpha}(\Omega), \ \mathrm{for \ any } \alpha. \ \mathrm{For} \ \alpha > 2, \\ \mathbb{W}^{1,\alpha}(\Omega) \subset \mathbb{L}^{\infty}(\Omega) \ \mathrm{because \ of} \ (1.3); \ \mathrm{hence} \ u_i D_i u \in \mathbb{L}^{\alpha}(\Omega) \ \mathrm{for \ any } \alpha. \ \mathrm{Then}, \end{array} \right$ 

Proposition I.2.2 shows us that  $u \in W^{2,\alpha}(\Omega)$ ,  $p \in W^{1,\alpha}(\Omega)$  for any  $\alpha$ . It is easy to check that  $u_i D_i u \in W^{1,\alpha}(\Omega)$ , so that  $u \in W^{3,\alpha}(\Omega)$ . Repeating this procedure we find in particular that

(1.61) 
$$u \in \mathbb{H}^{m}(\Omega), p \in \mathbb{H}^{m}(\Omega), \text{ for any } m \geq 1.$$

The same properties hold for any derivative of u or p; (1.3) implies therefore that any derivative of u or p belongs to  $\mathcal{C}(\overline{\Omega})$ , and this is the property announced.

For n = 3, we notice that  $u_i \in L^6(\Omega)$  (by (1.4)) and then

$$u_{j}D_{j}u_{j} \in L^{\frac{3}{2}}(\Omega).$$

Proposition I.2.2 implies that  $u \in \mathbb{W}^{2,\frac{3}{2}}(\Omega)$ ; but (1.3) shows us that  $u \in \mathbb{L}^{\alpha}(\Omega)$  for any  $\alpha$ ,  $1 \leq \alpha < +\infty$  (p =  $\frac{3}{2}$ , m = 2, n = 3). Therefore

$$D_{i}(u_{i}u_{j}) \in W^{-1,\alpha}(\Omega),$$

for any  $\alpha$ , and at this point we only need to repeat the proof given for n = 2.

If  $\Omega$  is unbounded, we obtain the same regularity on any compact subset of  $\overline{\Omega}$  by applying the preceding technique to  $\psi u$  where  $\psi$  is a cut-off function,  $\psi \in \mathscr{D}(\mathbb{R}^n)$ ,  $\psi = 1$  on the compact subset of  $\Omega$ .

#### Remark 1.4.

(i) It is clear that we can assume less regularity for f and obtain less regularity for u and p.

(ii) The same technique gives nothing for  $n \ge 4$ . For instance, for  $\Omega$  bounded and n = 4, if we write the nonlinear term as  $D_i(u_i u)$ , we just have  $u_i u_j \in L^2(\Omega)$ ,  $D_i(u_i u) \in H^{-1}(\Omega)$ , so that  $u \in H_0^1(\Omega)$ ; if we write the nonlinear terms as  $u_i D_i u$ , we have  $u_i D_i u \in L^{\frac{1}{3}}(\Omega)$ , so that  $u \in W^{2,\frac{3}{3}}(\Omega)$ ; but this gives nothing more than  $u_i \in L^4(\Omega)$ ,  $D_i u_i \in L^2(\Omega)$ , which was known before.

## 1.4 The Inhomogeneous Navier-Stokes Equations.

Let  $\Omega$  be an open bounded set in  $\mathbb{R}^n$ . We consider here the following inhomogeneous Navier-Stokes problem: let there be given two vector functions f and  $\phi$  defined respectively on  $\Omega$  and  $\Gamma$  and satisfying some conditions which will be specified later; to find u and p such that

(1.62) 
$$-\nu\Delta u + \sum_{i=1}^{n} u_{i}D_{i}u + \text{grad } p = f, \text{ in } \Omega,$$

(1.63) div u = 0, in  $\Omega$ ,

 $(1.64) u = \phi, on \Gamma.$ 

We will suppose that  $\Omega$  is of class  $\mathcal{C}^2$ , that f is given in  $\mathbb{H}^{-1}(\Omega)$  and that  $\phi$  is given in the following slightly restrictive way:

$$(1.65) \qquad \qquad \phi = \operatorname{rot} \zeta$$

where

(1.66) 
$$\zeta \in \mathbb{H}^{2}(\Omega), \quad D_{j}\zeta \in \mathbb{L}^{n}(\Omega), \quad \zeta \in \mathbb{L}^{\infty}(\Omega),$$

and rot denotes the usual rotational operator for n = 2,3; and for  $n \ge 4$ , rot denotes a linear differential operator with constant coefficients, such that div(rot  $\zeta$ )  $\equiv 0$  <sup>(1)</sup>.

#### Theorem 1.5.

Under the above hypotheses, there exists at least one  $u \in \mathbb{H}^{1}(\Omega)$ , and a distribution p on  $\Omega$ , such that (1.62) - (1.64) hold.

#### Proof.

Let  $\Psi$  be any vector function belonging to  $\mathbb{H}^1(\Omega) \cap \mathbb{L}^n(\Omega)$  such that

(1.67) 
$$\begin{cases} \psi \in \mathbb{H}^{1}(\Omega) \cap \mathbb{L}^{n}(\Omega), & \operatorname{div} \psi = 0 \\ \psi = \left[ \phi \quad \operatorname{on} \ \Gamma \right]. \end{cases}$$

Let us set

 $\hat{\mathbf{u}} = \mathbf{u} - \boldsymbol{\psi}.$ 

Then u belongs to  $\mathbb{H}^1(\Omega)$  and satisfies (1.63); (1.64) amounts to saying that (1.68)  $\hat{u} \in V_*$ 

Equation (1.62) is equivalent to

(1.69) 
$$-\nu\Delta\hat{u} + \sum_{i=1}^{n} \hat{u}_{i}D_{i}\hat{u} + \sum_{i=1}^{n} \hat{u}_{i}D_{i}\psi + \sum_{i=1}^{n}\psi_{i}D_{i}\hat{u} + \text{grad } p = \hat{f}$$

 $\frac{1}{(1)} \operatorname{rot} \zeta = (R_1 \zeta, \cdots, R_n \zeta), \quad R_i \zeta = \sum_{j,k} \alpha_{ijk} D_j \zeta_k; \quad \text{it suffices that} \quad \sum_{i=1}^n \alpha_{ijk} = 0,$  $\forall j,k, \quad 1 \leq j,k \leq n.$  where

$$\hat{f} = f + v\Delta \psi - \sum_{i=1}^{n} \psi_i D_i \psi.$$

We remark that

 $\hat{f} \in \mathbb{H}^{-1}(\Omega)$ .

which we show as follows. It is clear that  $f + \nu \Delta \psi \in H^{-1}(\Omega)$ ; we notice moreover that  $\psi_i D_i \in \mathbb{L}^{\alpha'}(\Omega)$ ,  $1/\alpha' + 1/\alpha = 1$ ,  $\alpha = \frac{2n}{n-2}$  if  $n \ge 3$ , any  $\alpha > 2$  if n = 2; since  $\mathbb{H}^1_0(\Omega) \subset \mathbb{L}^{\alpha}(\Omega)$ ,  $\psi_i D_i \psi$  belongs to  $\mathbb{H}^{-1}(\Omega)$  too.

As in Section 1.2 we can show that Problem (1.68)-(1.69) is solved if we find a  $\hat{u}$  in V such that

$$(1.70) \quad v((\hat{u},v)) + b(\hat{u},\hat{u},v) + b(\hat{u},\psi,v) + b(\psi,\hat{u},v) = \langle \hat{f},v \rangle, \quad \forall v \in \widetilde{V}.$$

The existence of a  $\hat{u} \in V$  satisfying (1.70) can be proved exactly as in Theorem 1.2, provided there exists some  $\beta > 0$ , such that

$$|\mathbf{v}||^{2} + \mathbf{b}(\mathbf{v},\mathbf{v},\mathbf{v}) + \mathbf{b}(\mathbf{v},\psi,\mathbf{v}) + \mathbf{b}(\psi,\mathbf{v},\mathbf{v}) \geq \beta ||\mathbf{v}||^{2} \quad \forall \mathbf{v} \in \widetilde{\mathbf{v}},$$

or because of (1.22)

(1.71) 
$$||v||^2 + b(v,\psi,v) \geq \beta ||v||^2, \quad \forall v \in \widetilde{V}.$$

(1.71) will certainly be satisfied if we can find  $\psi$  which satisfies (1.67) Now and

(1.72) 
$$|b(v,\psi,v)| \leq \frac{v}{2} ||v||^2, \quad \forall v \in \widetilde{V}.$$

In order to show this, we will prove the following lemma:

Before this we prove two other lemmas.

Lemma 1.9.

Let  $\rho(x) = d(x,\Gamma) = \underline{the \ distance \ from \ x \ to \ \Gamma}$ . For any  $\varepsilon > 0$ , there exists a function  $\theta_{\varepsilon} \in \mathcal{C}^2(\overline{\Omega})$  such that

(1.74)  $\theta_{\varepsilon} = 1$  in some neighborhood of  $\Gamma$  (which depends on  $\varepsilon$ ).

(1.75) 
$$\theta_{\varepsilon} = 0 \quad \underline{\text{if}} \quad \rho(\mathbf{x}) \geq 2\delta(\varepsilon), \quad \delta(\varepsilon) = \exp(-\frac{1}{\varepsilon})$$

(1.76) 
$$|D_k \theta_{\varepsilon}(\mathbf{x})| \leq \frac{\varepsilon}{\rho(\mathbf{x})} \quad \text{if } \rho(\mathbf{x}) \leq 2\delta(\varepsilon), \quad k = 1, \cdots, n.$$

Proof.

Let us consider with E. Hopf [2], the function  $\lambda \longrightarrow \xi_{\varepsilon}(\lambda)$  defined for  $\lambda \ge 0$  by

(1.77) 
$$\xi_{\varepsilon}(\lambda) = \begin{cases} 1 & \text{if } \lambda < \delta(\varepsilon)^{2} \\ \varepsilon & \log(\frac{\delta(\varepsilon)}{\lambda}) & \text{if } \delta(\varepsilon)^{2} < \lambda < \delta(\varepsilon) \\ 0 & \text{if } \lambda > \delta(\varepsilon) \end{cases}$$

and let us denote by  $\ \chi_{_{{\color{red} {\mathcal E}}}}$  the function

(1.78) 
$$\chi_{\varepsilon}(\mathbf{x}) = \xi_{\varepsilon}(\rho(\mathbf{x})).$$

Since the function  $\rho$  belongs to  $\mathcal{C}^2(\overline{\Omega})$ , the function  $\chi_{\varepsilon}$  satisfies (1.74) - (1.76) and  $\theta_{\varepsilon}$  is obtained by regularization of  $\chi_{\varepsilon}$ .

$$\frac{\text{Lemma 1.10.}}{\text{There exists a positive constant } c_1 \quad \frac{\text{depending only on } \Omega \quad \text{such that}}{\left|\frac{1}{\rho} v\right|_{L^2(\Omega)} \leq c_1 \|v\|_{H^1_0(\Omega)}, \quad \forall v \in H^1_0(\Omega).$$

Proof.

By using a partition of unity subordinated to a covering of  $\Gamma$ , and local coordinates near the boundary, we reduce the problem to the same problem with  $\Omega$  = a half-space = {x = (x\_n,x'), x\_n > 0, x' = (x\_1, \dots, x\_{n-1}) \in \mathbb{R}^{n-1}}. In this case  $\rho(x) = \chi_n$ , and it is sufficient to check that

(1.80) 
$$\int_{\Omega} \frac{|\mathbf{v}(\mathbf{x})|^2}{\mathbf{x}_n^2} d\mathbf{x} \leq c_1 \int_{\Omega} |\mathbf{D}_n \mathbf{v}(\mathbf{x})|^2 d\mathbf{x}, \quad \forall \mathbf{v} \in \mathcal{D}(\Omega).$$

This inequality is obvious if one proves the following one dimensional inequality:

(1.81) 
$$\int_{0}^{+\infty} \left|\frac{\mathbf{v}(\mathbf{s})}{\mathbf{s}}\right|^{2} d\mathbf{s} \leq 2 \int_{0}^{+\infty} |\mathbf{v}'(\mathbf{s})|^{2} d\mathbf{s}, \quad \forall \mathbf{v} \in \mathcal{D}(0, +\infty).$$

This is a classical Hardy inequality. In order to prove it, we write  $s = e^{\sigma}$ ,  $t = e^{T}$  and

$$\frac{\mathbf{v}(\mathbf{s})}{\mathbf{s}} = \frac{1}{\mathbf{s}} \int_{0}^{\mathbf{s}} \mathbf{w}(t) dt, \quad \mathbf{v}' = \mathbf{w},$$

$$\int_{0}^{+\infty} \frac{|\mathbf{v}(\mathbf{s})|^{2}}{|\mathbf{s}|^{2}} d\mathbf{s} = \int_{-\infty}^{+\infty} e^{-\sigma} \left( \int_{0}^{e^{\sigma}} \mathbf{w}(t) dt \right)^{2} d\sigma$$

$$= \int_{-\infty}^{+\infty} \left( \int_{-\infty}^{+\infty} \mathbf{q}(\sigma - \tau) e^{-\frac{\sigma - \tau}{2}} \mathbf{w}(e^{\tau}) e^{\frac{\tau}{2}} d\tau \right)^{2} d\sigma,$$

where  $\Psi$  stands for the Heaviside function,  $\Psi(\sigma) = 1$ , for  $\sigma > 0$  and  $\Psi(\sigma) = 0$  for  $\sigma < 0$ . By the usual convolution inequality we majorize the last quantity by

$$\left(\int_{-\infty}^{+\infty} \mathcal{Y}(\sigma) e^{-\frac{\sigma}{2}} d\sigma\right)^{2} \cdot \int_{-\infty}^{+\infty} |w(e^{\tau})|^{2} e^{\tau} d\tau = 4 \int_{0}^{+\infty} |w(t)|^{2} dt,$$

and (1.81) follows.

Proof of Lemma 1.8.

Let us now show that

 $\psi = \operatorname{rot}(\theta_{\varepsilon}\zeta)$ 

satisfies (1.67) and (1.73); (1.67) is obvious because of (1.65) and (1.74),

$$\psi_j(\mathbf{x}) = 0$$
 if  $\rho(\mathbf{x}) > 2\delta(\varepsilon)$ 

and

82) 
$$|\psi_{j}(x)| \leq c_{2} \left(\frac{\varepsilon}{\rho(x)} |\zeta(x)| + |D\zeta(x)|\right)$$
 if  $\rho(x) \leq 2\delta(\varepsilon)$ 

where

(1.

$$|D\zeta(x)| = \{\sum_{i,j=1}^{n} |D_{i}\zeta_{j}(x)|^{2}\}^{\frac{1}{2}}.$$

As we supposed that  $\zeta_i \in L^{\infty}(\Omega)$ , we deduce from (1.82) that

$$|\psi_{j}(\mathbf{x})| \leq c_{3}(\frac{\varepsilon}{\rho(\mathbf{x})} + |D\rho(\mathbf{x})|), \quad \forall j, \rho(\mathbf{x}) \leq 2\delta(\varepsilon).$$

We have therefore

(1.83) 
$$|\mathbf{v}_{\mathbf{i}}\psi_{\mathbf{j}}|_{\mathbf{L}^{2}(\Omega)} \leq c_{\mathbf{i}} \{\varepsilon |\frac{\mathbf{v}_{\mathbf{i}}}{\rho}|_{\mathbf{L}^{2}} + \left(\int_{\rho \leq 2\delta(\varepsilon)} |\mathbf{v}_{\mathbf{i}}^{2}| |D\zeta|^{2} |dx|^{\frac{1}{2}}\right).$$

But, using Hölder's inequality, we see that

$$\left(\int_{\rho\leq 2\delta(\varepsilon)} \mathbf{v}_{\mathbf{i}}^{2} |D\zeta|^{2} d\mathbf{x}\right)^{\frac{1}{2}} \leq \mu(\varepsilon) |\mathbf{v}_{\mathbf{i}}|_{\mathbf{L}^{\alpha}(\Omega)}$$

where  $\frac{1}{\alpha} = \frac{1}{2} - \frac{1}{n}$  and

$$\mu(\varepsilon) = \left\{ \int_{\rho(\mathbf{x}) \leq 2\delta(\varepsilon)} |D\zeta(\mathbf{x})|^n d\mathbf{x} \right\}^{\frac{1}{n}};$$

 $\mu(\varepsilon) \longrightarrow 0$  as  $\varepsilon \longrightarrow 0$  since  $D_i \zeta_j \in L^n(\Omega)$ ,  $1 \le i, j \le n$ . With this last majorization, (1.4), and Lemma 1.10, (1.83) gives

(1.84) 
$$|v_{\mathbf{j}}\psi_{\mathbf{j}}|_{L^{2}(\Omega)} \leq c_{5}(\varepsilon ||v|| + \mu(\varepsilon) ||v||) \leq c_{6}(\varepsilon + \mu(\varepsilon)) ||v||, \quad 1 \leq \mathbf{i}, \mathbf{j} \leq \mathbf{n}.$$

Now it is easy to check (1.73); for each  $v \in \mathcal{V}$ ,

$$b(\mathbf{v}, \psi, \mathbf{v}) = -b(\mathbf{v}, \mathbf{v}, \psi)$$

$$(1.85) \qquad |b(\mathbf{v}, \mathbf{v}, \psi)| \leq ||\mathbf{v}|| \{\sum_{i,j=1}^{n} |\mathbf{v}_{i}\psi_{j}|\}$$

$$\leq (by \ (1.84))$$

$$\leq c_{7}(\varepsilon + \mu(\varepsilon)) ||\mathbf{v}||^{2}.$$

If  $\varepsilon$  is sufficiently small so that

$$c_{\gamma}(\varepsilon + \mu(\varepsilon)) \leq \gamma$$
,

we obtain (1.73) for each  $v \in \mathbb{V}$  and by continuity, for each  $v \in \widetilde{\mathbb{V}}$ .

#### Remark 1.5.

(i) For  $n \leq 3$ , the conditions (1.66) reduce to

$$\zeta \in \mathbb{H}^{2}(\Omega),$$

because of the Sobolev imbedding theorems (see (1.3)).

(ii) It is easy to write the boundary condition in the form (1.65) for the classical problems of hydrodynamics, such as the cavitation problem, the Taylor problem ....

It is easy to extend Proposition 1.1 to inhomogeneous problems. With the hypotheses of this proposition and moreover assuming that  $\phi \in \mathfrak{C}^{\infty}(\Gamma)$  the solution  $\{u,p\}$  of (1.62) - (1.64) belongs to  $\mathfrak{C}^{\infty}(\overline{\Omega}) \times \mathfrak{C}^{\infty}(\overline{\Omega})$ .

To prove this we proceed as in Proposition 1.1, directly on the equations (1.62) - (1.64) (i.e., without introducing  $\hat{u}$ ).

A uniqueness result similar to Theorem 1.3 holds: for  $n \le 4$ , v "large", and f "small", there is uniqueness:

Theorem 1.6.

We suppose that  $n \leq 4$ , that the norm of  $\phi$  in  $\mathbb{L}^{n}(\Omega)$  is sufficiently small so that

(1.86) 
$$|b(v,\phi,v)| \leq \frac{\nu}{2} ||v||^2, \quad \forall v \in \mathbb{V},$$
 (1)

and v is sufficiently large so that

(1.87) 
$$v^2 \ge 4c(n) \|\hat{f}\|_{v}$$

where c(n) is the constant in (1.18) and

(1.88) 
$$\hat{\mathbf{f}} = \mathbf{f} + \nu \Delta \phi - \sum_{i=1}^{n} \phi_i \mathbf{D}_i \phi.$$

Then, there exists a unique solution u,p of (1.62) - (1.64) (2).

Proof.

It was proved in Lemmas 1.1, 1.2, 1.3 that

$$b(v,\phi,v) = -b(v,v,\phi),$$

and

$$|\mathbf{b}(\mathbf{v},\mathbf{v},\mathbf{\phi})| \leq c \|\mathbf{v}\|^2 |\mathbf{\phi}|_{\mathbb{L}^n(\Omega)}$$

Therefore condition (1.86) is satisfied if  $|\phi|$  is small enough: this  $\mathbb{L}^{n}(\Omega)$ 

(1) For n = 2 replace  $\mathbb{L}^{n}(\Omega)$  by  $L^{\alpha}(\Omega)$  for some  $\alpha > 2$ . (2) As always, p is unique up to a constant. means that (1.72) is satisfied with  $\psi = \phi$  and we do not need in this case the previous construction of  $\psi$ . Nevertheless, the proof of existence goes along the same lines, with  $\psi = \phi$ .

If  $u_1$  is a solution of (1.62) - (1.64), then  $\hat{u}_1 = u_1 - \phi$  is a solution of (1.70) with  $\psi = \phi$ . Taking  $v = \hat{u}_1$  in (1.70) we get

$$\sqrt{\|\hat{\mathbf{u}}_1\|^2} = -b(\hat{\mathbf{u}}_1,\phi,\hat{\mathbf{u}}_1) + \langle \hat{\mathbf{f}},\hat{\mathbf{u}}_1 \rangle \leq \frac{\sqrt{2}}{2} \|\hat{\mathbf{u}}_1\|^2 + \|\hat{\mathbf{f}}\|_{\mathbf{V}}, \|\hat{\mathbf{u}}_1\|$$

by using (1.86), and therefore

(1.89) 
$$\|\hat{u}_1\| \leq \frac{2}{\nu} \|\hat{f}\|_{V},.$$

Let us suppose that  $u_0, u_1$  are two solutions of (1.62) - (1.64); let  $\hat{u}_0 = u_0 - \phi$ ,  $\hat{u}_1 = u_1 - \phi$ ,  $\hat{u} = \hat{u}_0 - \hat{u}_1$ ;  $\hat{u}_0$  and  $\hat{u}_1$  satisfy (1.70) with  $\psi = \phi$ :

$$v((\hat{u}_{0}, v)) + b(\hat{u}_{0}, \hat{u}_{0}, v) + b(\hat{u}_{0}, \phi, v) + b(\phi, \hat{u}_{0}, v) = \langle \hat{f}, v \rangle, \quad v \quad V,$$

$$v((\hat{u}_{1}, v)) + b(\hat{u}_{1}, \hat{u}_{1}, v) + b(\hat{u}_{1}, \phi, v) + b(\phi, \hat{u}_{1}, v) = \langle \hat{f}, v \rangle, \quad v \quad V.$$

We take  $v = \hat{u}$  in these equations, and take the difference between them; after expanding and using (1.22) we find:

(1.90) 
$$v \| \hat{u} \|^2 = -b(\hat{u}, \hat{u}, \hat{u}) - b(\hat{u}, \phi, \hat{u}).$$

Because of (1.86),

$$-b(\hat{u},\phi,\hat{u}) \leq \frac{\nu}{2} \|\hat{u}\|^2.$$

By (1.18),

 $-b(\hat{u},\hat{u}_{1},\hat{u}) \leq c(n) \|\hat{u}_{1}\| \|\hat{u}\|^{2},$ 

and because of (1.89), this is majorized by

$$\frac{2}{v} c(n) \| \hat{f} \|_{V}, \| \hat{u} \|^2.$$

We finally arrive at the inequality

$$\left(\frac{\nu}{2}-\frac{2}{\nu}c(n)\right\|\hat{f}\|_{V}, \quad \|\hat{u}\|^{2} \leq 0,$$

and because of (1.87), this implies  $\hat{u} = 0$ .

If  $\hat{u}_0 = \hat{u}_1$ , it is clear that grad  $p_0 = \text{grad } p_1$ , so the difference between  $p_0$  and  $p_1$  is constant.

## Remark 1.7.

Nonuniqueness results for Problem (1.62) - (1.64) have been proved, in the two-dimensional case, for certain configurations (cf., Velte [1], [2]).

\$2. DISCRETE INEQUALITIES AND COMPACTNESS THEOREMS.

Before going through the numerical approximation of the stationary Navier-Stokes equations, we must introduce new tools: the discrete analog, for step functions, of the Sobolev inequalities and of the compactness theorem, Theorem 1.1. These and some further Sobolev-type inequalities are the goals set for this aection.

#### 2.1 Discrete Sobolev Inequalities.

The notations are those used for finite differences; see Section 3.3, Chapter I. We recall in particular that  $\mathcal{R}_h$  is the set of points with coordinates  $m_1h_1, \cdots, m_hh_n, m_i \in \mathbb{Z}, h = (h_1, \cdots, h_n) h_i > 0; w_{hM}$  is the characteristic function of the block

(2.1) 
$$\sigma_{h}(M) = \frac{n}{\prod_{i=1}^{n}} [\mu_{i} - \frac{h_{i}}{2}, \mu_{i} + \frac{h_{i}}{2}], M = (\mu_{i}, \cdots, \mu_{n}),$$

and  $\delta_{ih}$  is the difference operator

(2.2) 
$$\delta_{ih}\phi(x) = \frac{1}{h_i} \left[\phi(x + \frac{1}{2}\vec{h}_i) - \phi(x - \frac{1}{2}\vec{h}_i)\right]$$

where  $\vec{h}_i$  is the vector with i<sup>th</sup> component  $h_i$ , and all other components 0.

Theorem 2.1.

Let p denote some number such that  $1 \le p \le n$ , and let q be defined by  $\frac{1}{q} = \frac{1}{p} - \frac{1}{n}$ .

There exists a constant c = c(n,p) depending only on n and p such that

(2.3) 
$$|u_h|_{L^q(\mathbb{R}^n)} \leq c(n,p) \sum_{i=1}^n |\delta_{ih}u_h|_{L^p(\mathbb{R}^n)}$$

for each step function uh

(2.4) 
$$\mathbf{u}_{h} = \sum_{M \in \mathcal{R}_{h}} \mathbf{u}_{h}^{(M)} \mathbf{w}_{hM}^{*},$$

with compact support.

Proof.

i) Let us consider the scalar function

$$s \mapsto g(s) = |s|^{\frac{(n-1)p}{n-p}}$$

Since  $\frac{(n-1)p}{n-p} \ge 1$ , this function is differentiable with derivative

g'(s) = 
$$\frac{(n-1)p}{n-p} |s| \frac{(n-1)p}{n-p} -2$$
  
s.

The Taylor formula can be written

$$g(s_1)-g(s_2) = (s_1-s_2)g'(\lambda s_1+(1-\lambda)s_2), \quad \lambda \in (0,1)$$

and gives

$$\begin{aligned} |g(s_1) - g(s_2)| &\leq |s_1 - s_2| \frac{(n-1)p}{n-p} |\lambda s_1 + (1-\lambda)s_2| \frac{n(p-1)}{n-p} \\ &\leq \frac{(n-1)p}{n-p} |s_1 - s_2| \{|s_1| + |s_2|\}^{n-p}, \end{aligned}$$

(2.5) 
$$|g(s_1)-g(s_2)| \leq c_1(n,p) |s_1-s_2| \{ |s_1| \frac{n(p-1)}{n-p} + |s_2| \frac{n(p-1)}{n-p} \}.$$

ii) Let M belong to  $R_h$ ; we apply (2.5) with

$$s_{1} = u_{h}(M-r\dot{h}_{1}), \quad s_{2} = u_{h}(M-(r+1)\dot{h}_{1});$$

$$|u_{h}(M-r\dot{h}_{1})| \xrightarrow{(n-1)p} - |u_{h}(M-(r+1)\dot{h}_{1})| \xrightarrow{(n-1)p}$$

$$\leq c_{1}(n,p) |u_{h}(M-r\dot{h}_{1})-u_{h}(M-(r+1)\dot{h}_{1})|$$

$$\cdot \{|u_{h}(M-r\dot{h}_{1})| \xrightarrow{n(p-1)} + |u_{h}(M-(r+1)\dot{h}_{1})| \xrightarrow{n(p-1)} \}$$

Summing these inequalities for  $r \ge 0$ , we find (the sum is actually finite):

(2.6) 
$$|u_{h}(M)| \stackrel{(n-1)p}{=} \leq c_{1}(n,p)h_{i} \sum_{r=0}^{+\infty} |\delta_{ih}u_{h}(M-(r+\frac{1}{2})\vec{h}_{i})| \cdot \{|u_{h}(M-r\vec{h}_{i})| \stackrel{n(p-1)}{=} + |u_{h}(M-(r+1)\vec{h}_{i})| \stackrel{n(p-1)}{=}.$$

We strengthen inequality (2.6) by replacing the sum on the right-hand side by the sum for  $r \in Z$ ; we can then interpret the sum as an integral and majorize it by

$$c_{1}(n,p)\int_{-\infty}^{+\infty} |\delta_{ih}u_{h}(\hat{\mu}_{i},\xi_{i})| \cdot \{\sum_{\alpha=-1}^{1} |u_{h}(\hat{\mu}_{i},\xi_{i}+\frac{\alpha}{2}h_{i})|^{\frac{n(p-1)}{n-p}}\}d\xi_{i},$$

where  $(\mu_1, \dots, \mu_n)$  are the coordinates of M and  $\hat{\mu}_i = (\mu_1, \dots, \mu_{i-1}, \mu_{i+1}, \dots, \mu_n)$ . In a similar way we denote by  $\hat{x}_i$  the vector  $(x_1, \dots, x_{i-1}, x_{i+1}, \dots, x_n)$  and then write  $x = (\hat{x}_i, x_i)$ .

For any  $x \in \sigma_h(M)$ , inequality (2.6) gives now

(2.7) 
$$|u_{h}(x)|^{\frac{(n-1)p}{n-p}} = |u_{h}(M)|^{\frac{(n-1)p}{n-p}} \leq c_{1}(n,p) \int_{-\infty}^{+\infty} |\delta_{1h}u_{h}(\hat{x}_{1},\xi_{1})| \cdot \{\sum_{\alpha = -1}^{1} |u_{h}(\hat{x}_{1},\xi_{1} + \frac{\alpha h_{1}}{2})|^{\frac{n(p-1)}{n-p}} \} d\xi_{1}.$$

Let us now set

(2.8) 
$$w_{i}(x) = w_{i}(\hat{x}_{i}) = \sup_{x_{i} \in \mathbb{R}} |u_{h}(x)|^{\frac{p}{n-p}}$$

Then,  $|w_i(\hat{x}_i)|^{n-1}$  is majorized by the right-hand side of (2.7), hence

$$\sum_{\mathbb{R}^{n-1}}^{w_i(\hat{x}_i)^{n-1}} d\hat{x}_i$$

$$\leq c_1(n,p) \int_{\mathbb{R}^n} |\delta_{ih} u_h(\hat{x}_i,\xi_i)|$$

$$\cdot \{\sum_{\alpha=-1}^{1} |u_h(\hat{x}_i,\xi_i + \frac{\alpha h_i}{2})|^{\frac{n(p-1)}{n-p}} \} d\hat{x}_i d\xi_i.$$

$$\leq (by H\"{o}lder's inequality)$$

+1 -1

$$\leq c_{1}(n,p) \left| \delta_{ih} u_{h} \right|_{L^{p}(\mathbb{R}^{n})} \cdot \left( \sum_{\alpha=-1}^{+1} \int_{\mathbb{R}^{n}} \left| u_{h}(\hat{x}_{i},\xi_{i}+\alpha h_{i}) \right|^{q} d\hat{x}_{i} d\xi_{i} \right)^{p};$$

therefore

(2.9) 
$$\int_{\mathbb{R}^{n-1}} w_i(\hat{x}_i)^{n-1} d\hat{x}_i \leq c_2(n,p) \left| \delta_{ih} u_h \right|_{L^p(\mathbb{R}^n)} \left| u_h \right|_{L^q(\mathbb{R}^n)} \frac{(p-1)q}{p}$$

Now we have

$$\int_{\mathbb{R}^{n}} |\mathbf{u}_{h}|^{q} d\mathbf{x} \leq \int_{\mathbb{R}^{n}} \frac{\prod_{i=1}^{n} \sup_{\mathbf{x}_{i}} |\mathbf{u}_{h}(\hat{\mathbf{x}}_{i},\mathbf{x}_{i})|^{\frac{p}{n-p}} d\mathbf{x}$$
$$\leq \int_{\mathbb{R}^{n}} \frac{\prod_{i=1}^{n} w_{i}(\hat{\mathbf{x}}_{i}) d\mathbf{x}.$$

According to the inequality given in the next lemma, this is majorized by

$$\frac{1}{\prod_{i=1}^{n}} \left\{ \int_{\mathbb{R}^{n-1}} |w_i(\hat{x}_i)|^{n-1} d\hat{x}_i \right\}^{\frac{1}{n-1}}$$

and, because of (2.9),

$$\begin{aligned} \left| u_{h} \right|_{L^{q}(\mathbb{R}^{n})}^{q} &\leq c_{3}(n,p) \cdot \left\{ \prod_{i=1}^{n} \left| \delta_{ih} u_{h} \right|_{L^{p}(\mathbb{R}^{n})}^{\frac{1}{n-1}} \right\} \left| u_{h} \right|_{L^{q}(\mathbb{R}^{n})}^{\frac{n}{n-1}} \\ \left| u_{h} \right|_{L^{q}(\mathbb{R}^{n})}^{\frac{n}{n-1}} &\leq c_{3}(n,p) \left\{ \prod_{i=1}^{n} \left| \delta_{ih} u_{h} \right|_{L^{p}(\mathbb{R}^{n})}^{\frac{1}{n-1}} \right\} \\ \left| u_{h} \right|_{L^{q}(\mathbb{R}^{n})} &\leq c_{4}(n,p) \left\{ \prod_{i=1}^{n} \left| \delta_{ih} u_{h} \right|_{L^{p}(\mathbb{R}^{n})}^{\frac{1}{n}} \right\} \\ \left| u_{h} \right|_{L^{q}(\mathbb{R}^{n})} &\leq c_{5}(n,p) \left\{ \prod_{i=1}^{n} \left| \delta_{ih} u_{h} \right|_{L^{p}(\mathbb{R}^{n})}^{\frac{1}{n}} \right\}. \end{aligned}$$

<u>Lemma 2.1</u>.

Let  $w_1, \dots, w_n$ , be n measurable bounded functions on  $\mathbb{R}^n$ , with compact supports, and  $w_i$  independent of  $x_i$ .

Then

(2.10) 
$$\int_{\mathbb{R}^n} \left( \frac{n}{1+1} w_i(\hat{x}_i) \right) dx \leq \frac{n}{1+1} \left\{ \int_{\mathbb{R}^{n-1}} |w_i(\hat{x}_i)|^{n-1} d\hat{x}_i \right\}^{\frac{1}{n-1}}.$$

This is a particular case of an inequality of E. Gagliardo [1]; see also Lions [1], page 31.

## Remark 2.1.

For p = n, if the support of  $u_h$  is included in a bounded set  $\Omega$ , then

(2.11) 
$$|u_h|_{L^q(\mathbb{R}^n)} \leq c(n,q,\Omega) \sum_{i=1}^n |\delta_{ih}u_h|_{L^p(\mathbb{R}^n)}$$

for each  $u_h$  of type (2.4), and for any q,  $1 \le q < +\infty$ . Actually any such qgreater than p can be written as  $\frac{p_1 n}{n - p_1}$  with  $1 \le p_1 < n$ . The inequality (2.3) is then applicable, with  $c = c(n, p_1) = c'(n, q)$ :

$$|\mathbf{u}_{h}|_{\mathbf{L}^{q}(\mathbf{R}^{n})} \leq \mathbf{c}'(\mathbf{n},q) \sum_{i=1}^{n} |\delta_{ih}\mathbf{u}_{h}|_{\mathbf{L}^{p_{1}}(\mathbf{R}^{n})}$$

The Hölder inequality shows us that

$$\left| \delta_{ih} u_{h} \right|_{L^{p_{1}}(\mathbb{R}^{n})} \leq (\text{meas } \Omega')^{\frac{1}{p_{1}} - \frac{1}{p}} \left| \delta_{ih} u_{h} \right|_{L^{p}(\mathbb{R}^{n})}$$

where  $\Omega'$  contains the support of  $\delta_{ih}u_h$ . If we suppose that |h| is bounded by 1 (or by some constant d), (meas  $\Omega'$ ) is bounded by (meas  $\Omega$ )×Const; then combining the last two inequalities, we obtain (2.11).

In the two and three dimensional cases, we prove another related inequality which will be useful.

#### Proof.

We use the inequality (2.7) with n = 2 and  $p = \frac{4}{3}$ ; a more precise analysis of the proof of (2.7) shows that  $c_1(n,p) = 2$  in the present case; actually  $g(s) = s^2$  and for (2.5) we have clearly

 $|g(s_1)-g(s_2)| \leq 2|s_1-s_2|\{|s_1| + |s_2|\}.$ 

We then have, for any  $x \in \sigma_{b}(M)$ , and  $M \in \mathcal{R}_{b}$ ,

$$(2.14) |u_{h}(\mathbf{x})|^{2} \leq 2 \int_{-\infty}^{+\infty} |\delta_{ih}u_{h}(\hat{\mathbf{x}}_{i},\xi_{i})| \{\sum_{\alpha=-1}^{+1} |u_{h}(\hat{\mathbf{x}}_{i},\xi_{i}+\frac{\alpha h_{i}}{2})| \} d\xi_{i}.$$

Since the right-hand side of (2.14) is independent of  $x_i$ , we obtain (2.15)  $\sup_{x_i} |u_h(x)|^2 \leq 2 \int_{-\infty}^{+\infty} |\delta_{ih}u_h(\hat{x}_i,\xi_i)| \left\{ \sum_{\alpha=-1}^{+1} |u_h(\hat{x}_i,\xi_i + \frac{\alpha h_i}{2})| \right\} d\xi_i.$ 

Now we may write, for the two dimensional case,

$$(2.16) \int_{\mathbb{R}^{2}} |u_{h}(x)|^{4} dx \leq \int_{\mathbb{R}^{2}} [\sup_{x_{1}} |u_{h}(x)|^{2}] [\sup_{x_{2}} |u_{h}(x)|^{2}] dx$$
$$\leq \{\int_{-\infty}^{+\infty} [\sup_{x_{1}} |u_{h}(x)|^{2}] dx_{2}\} \{\int_{-\infty}^{+\infty} [\sup_{x_{2}} |u_{h}(x)|^{2}] dx_{1}\}$$

$$\leq (\text{because of } (2.15)) \\ \leq 4\{\int_{\mathbb{R}^2} |\delta_{1h} u_h(\xi_1, x_2)| [\sum_{\alpha=-1}^{+1} |u_h(\xi_1 + \frac{\alpha h_1}{2}, x_2)|] d\xi_1 dx_2\} \\ \cdot \{\int_{\mathbb{R}^2} |\delta_{2h} u_h(x_1, \xi_2)| [\sum_{\alpha=-1}^{+1} |u_h(x_1, \xi_2 + \frac{\alpha h_2}{2})|] dx_1 d\xi_2\}.$$

By the Schwarz inequality and since

(2.17) 
$$\int_{\mathbb{R}^2} |u_h(\xi_1 + \frac{\alpha h_1}{2}, x_2)|^2 d\xi_1 dx_2 = \int_{\mathbb{R}^2} |u_h(x)|^2 dx_1 dx_2 = |u_h|^2_{L^2(\mathbb{R}^2)}$$

the last expression is majorized by

$$36\{|\delta_{1h}u_{h}|_{L^{2}(\mathbb{R}^{2})}|u_{h}|_{L^{2}(\mathbb{R}^{2})}\}\{|\delta_{2h}u_{h}|_{L^{2}(\mathbb{R}^{2})}|u_{h}|_{L^{2}(\mathbb{R}^{2})}\}$$

$$\leq 18|u_{h}|_{L^{2}(\mathbb{R}^{2})}^{2}\{\sum_{i=1}^{2}|\delta_{ih}u_{h}|_{L^{2}(\mathbb{R}^{2})}^{2}\}.$$

Hence (2.12) is proved.

In the three dimensional case, using (2.12) and (2.15), we write

$$(2.18) \int_{\mathbb{R}^{3}} |u_{h}(x)|^{4} dx \leq 18 \int \{ [\int |u_{h}|^{2} dx_{1} dx_{2}] [\sum_{i=1}^{2} \int |\delta_{ih} u_{h}|^{2} dx_{1} dx_{2}] \} dx_{3}$$
$$\leq 18 \{ \sup_{x_{3}} \int |u_{h}|^{2} dx_{1} dx_{2} \} \{ \sum_{i=1}^{2} |\delta_{ih} u_{h}|^{2} |z_{3}|^{2} \}$$

$$\leq (\text{because of } (2.15))$$

$$\leq 36\{\sum_{i=1}^{2} |\delta_{ih}u_{h}|_{L^{2}(\mathbb{R}^{3})}^{2} \}\{\int_{\mathbb{R}^{3}} |\delta_{3h}u_{h}(\hat{x}_{3}, x_{3})| \left[\sum_{\alpha=-1}^{+1} |u_{h}(\hat{x}_{3}, x_{3} + \frac{\alpha h_{3}}{2})|\right]dx\}$$

$$\leq (\text{by Schwarz's inequality})$$

$$\leq 3^{3}2^{2} |u_{h}|_{L^{2}(\mathbb{R}^{3})} |\delta_{3h}u_{h}|_{L^{2}(\mathbb{R}^{3})} \{\sum_{i=1}^{2} |\delta_{ih}u_{h}|_{L^{2}(\mathbb{R}^{3})}^{2}\}$$

$$\leq 3^{3}2^{2} |u_{h}|_{L^{2}(\mathbb{R}^{3})} \{\sum_{i=1}^{3} |\delta_{ih}u_{h}|_{L^{2}(\mathbb{R}^{3})}^{2}\},$$

and (2.13) is proved.

#### Remark 2.2.

The inequalities (2.3), (2.11), and (2.12) can be extended by continuity to classes of step functions with unbounded support.

#### 2.2 A Discrete Compactness Theorem.

We give here a discrete analog of Theorem 1.1, more precisely of the fact that the injection of  $\hat{W}^{1,p}(\Omega)$  into  $L^{q_1}(\Omega)$  is compact if

(2.19) 
$$1 \le p \le n$$
 and  $q_1$  is any number, such that  
 $1 \le q_1 \le q$ , where  $\frac{1}{q} = \frac{1}{p} - \frac{1}{n}$ ,

(2.20) 
$$p = n$$
 and  $q_1$  is any number,  $1 \le q_1 < +\infty$ .

Theorem 2.2.

(2.21) Let  $\mathcal{E}_{h}$  be a family, maybe empty, of step functions of type (2.4) and let  $\mathcal{E}_{h} = \bigcup_{|h| \leq c_{0}} \mathcal{E}_{h}$ .

Let us suppose that

(2.22) the functions  $u_h$  of  $\xi$  have their supports included in some fixed bounded measurable subset of  $\mathbb{R}^n$ , say  $\Omega$ ,

(2.23) 
$$\sup_{\mathbf{u}_{h} \in \mathcal{E}} \{ |\mathbf{u}_{h}|_{L^{p}(\mathbb{R}^{n})} + \sum_{i=1}^{n} |\delta_{ih}\mathbf{u}_{h}|_{L^{p}(\mathbb{R}^{n})} \} < +\infty.$$

Then if p and  $q_1$  satisfy conditions (2.19)-(2.20), the family  $\xi$  is relatively compact in  $L^{q_1}(\mathbb{R}^n)$  (or  $L^{q_1}(\Omega)$ ).

Proof.

According to a theorem of M. Riesz [1], we must prove the following two properties:

i) For each  $\varepsilon > 0$ , there exists a compact set  $K \subseteq \Omega$ , such that

(2.24) 
$$\int_{\Omega-K} |u_h|^{q_1} dx \leq \varepsilon, \quad \forall u_h \in \xi.$$

ii) For each  $\varepsilon > 0$ , there exists  $\eta > 0$  such that,

(2.25) 
$$|\tau_{\ell} u_h - u_h|_{L^{q_1}(\mathbb{R}^n)} \leq \varepsilon$$

for any  $u_h \in \mathcal{C}$  and any  $\ell = (\ell_1, \cdots, \ell_n)$ , with  $|\ell| \leq \eta; \tau_{\ell}$  denotes the translation operator

(2.26) 
$$(\tau_{0}\phi)(x) = \phi(x + \ell).$$

### Proof of (i).

Because of the Sobolev inequalities (2.3) and (2.11), the family  $\mathcal{E}$  is bounded in  $L^{q}(\Omega)$  where q is given by (2.19) if p < n, and q is some fixed number, q > q, otherwise (p = n).

By the Hölder inequality, we then get

(2.27) 
$$\int_{\Omega-K} |u_{h}|^{q_{1}} dx \leq \left(\int_{\Omega-K} dx\right)^{1-\frac{q_{1}}{q}} \left(\int_{\Omega-K} |u_{h}|^{q} dx\right)^{\frac{q_{1}}{q}} dx$$
$$\int_{\Omega-K} |u_{h}|^{q_{1}} dx \leq c \left(\max\left(\Omega-K\right)\right)^{1-\frac{q_{1}}{q}}, \quad \forall u_{h} \in \mathcal{E}.$$

The right-hand side of (2.27) (and hence the left-hand side) can be made less than  $\varepsilon$ , by choosing the compact K sufficiently large; (i) is proved.

Proof of (ii).

First, we show that 2.25 may be replaced by a similar condition on  $|\tau_{l}u_{h} - u_{h}|_{L^{p}}$  (condition 2.30 below).

<u>Case (a)</u>:  $q_1 \leq p$ . For any  $f \in L^q(\Omega)$  we have  $f \in L^{q_1}(\Omega)$  as well as  $f \in L^p(\Omega)$  since  $\Omega$  is bounded and  $q_1 \leq p < q$ . Also,  $0 \leq \frac{1}{q_1} - \frac{1}{p} < 1$ . By the Hölder inequality,

$$|f|_{L^{q_1}(\Omega)} \leq (\text{meas } \Omega)^{q_1} - \frac{1}{p} \cdot |f|_{L^{p}(\Omega)} = \text{Const.} \cdot |f|_{L^{p}(\Omega)}.$$

<u>Case (b)</u>:  $q_1 > p$ . For any function  $f \in L^q(\Omega)$  we can write, using the Hölder inequality,

$$\int |\mathbf{f}|^{q_1} d\mathbf{x} = \int |\mathbf{f}|^{\theta q_1} |\mathbf{f}|^{(1-\theta)q_1} d\mathbf{x}$$

$$\leq \left( \int |\mathbf{f}|^{q_1\theta\rho} d\mathbf{x} \right)^{\frac{1}{\rho}} \left( \int |\mathbf{f}|^{q_1(1-\theta)\rho'} d\mathbf{x} \right)^{\frac{1}{\rho'}}$$

where  $\theta \in (0,1)$ ,  $\rho > 1$ , and as usual  $\frac{1}{\rho'} + \frac{1}{\rho} = 1$ . We can chose  $\theta$  and  $\rho$  so that

$$q_1 \theta \rho = p$$
,  $q_1 (1-\theta) \rho' = q$ ;

this defines  $\rho$  and  $\theta$  uniquely, and these numbers belong to the specified intervals,  $(\theta q_1(q-p) = p(q-q_1), \text{ and } \rho(q-q_1) = q-p).$ 

Then

(2.28) 
$$\int |\mathbf{f}|^{q_1} d\mathbf{x} \leq \left( \int |\mathbf{f}|^p d\mathbf{x} \right)^{\frac{1}{p}} \left( \int |\mathbf{f}|^q d\mathbf{x} \right)^{\frac{1}{p'}}.$$

In particular, for any  $\ell$  and  $u_{\rm h}$ ,

$$\begin{aligned} |\tau_{\ell}u_{h}-u_{h}|_{L^{q_{1}}(\mathbb{R}^{n})} &\leq |\tau_{\ell}u_{h}-u_{h}|_{L^{q}(\mathbb{R}^{n})}^{1-\theta} \cdot |\tau_{\ell}u_{h}-u_{h}|_{L^{p}(\mathbb{R}^{n})}^{\theta} \\ &\leq (|\tau_{\ell}u_{h}|_{L^{q}(\mathbb{R}^{n})} + |u_{h}|_{L^{q}(\mathbb{R}^{n})})^{1-\theta} |\tau_{\ell}u_{h}-u_{h}|_{L^{p}(\mathbb{R}^{n})}^{\theta} \end{aligned}$$

$$= 2^{1-\theta} |u_h|_{L^q(\mathbb{R}^n)}^{1-\theta} |\tau_{\ell} u_h - u_h|_{L^p(\mathbb{R}^n)}^{\theta}.$$

Since the family  $\mathcal{E}$  is bounded in  $L^q(\mathbb{R}^n)$ ,

(2.29) 
$$|\tau_{\ell} u_{h} - u_{h}|_{L^{q_{1}}(\mathbb{R}^{n})} \leq c|\tau_{\ell} u_{h} - u_{h}|_{L^{p}(\mathbb{R}^{n})}^{\theta}.$$

Inequality (2.29) shows us that it suffices to prove condition (ii) with  $\ensuremath{\,q_1}$  replaced by p:

(2.30) 
$$\begin{cases} \forall \varepsilon > 0, \exists \eta, \text{ such that} \\ |\tau_{\ell} u_{h} - u_{h}| \underset{L^{p}(\mathbb{R}^{n})}{\overset{\leq}{} \varepsilon} \\ \text{for } |\ell| \leq \eta \text{ and } u_{h} \in \mathcal{E}. \end{cases}$$

The proof of (2.30) follows easily from (2.23) and the next two lemmas. Lemma 2.2.

(2.31) 
$$|\tau_{\ell} u_{h} - u_{h}|_{L^{p}(\mathbb{R}^{n})} \leq \sum_{i=1}^{n} |\tau_{\ell} u_{h} - u_{h}|_{L^{p}(\mathbb{R}^{n})}$$

where  $\vec{k}_i$  denotes the vector  $k_i \vec{h}_i$ .

Proof.

Denoting by I the identity operator, one can check easily the identity

(2.32) 
$$\tau_{\ell} - I = \sum_{i=1}^{n} \tau_{\ell_{1}} \cdots \tau_{\ell_{i-1}} (\tau_{\ell_{i-1}} - I).$$

This identity allows us to majorize the norm  $|\tau_{l}u_{h}-u_{h}|_{L^{p}(\mathbb{R}^{n})}$  by

$$\sum_{i=1}^{n} |\tau_{\vec{k}_{1}} \cdots \tau_{\vec{k}_{i-1}} (\tau_{\vec{k}_{i}} u_{h} - u_{h})|_{L^{p}(\mathbb{R}^{n})}$$

We obtain (2.31) recalling that

(2.33) 
$$|\tau_{\alpha}f|_{L^{p}(\mathbb{R}^{n})} = |f|_{L^{p}(\mathbb{R}^{n})}$$

for any  $\alpha \in \mathbb{R}^n$  and any function  $f \in L^p(\mathbb{R}^n)$ .

$$\underbrace{\underline{\text{Lemma}}}_{(2.34)} \underbrace{\frac{2.3}{|\tau_{k_{i}}^{+}u_{h}^{-}u_{h}^{+}|}_{L^{p}(\mathbb{R}^{n})} \leq c(|\ell_{i}| + |\ell_{i}|^{p}) |\delta_{ih}u_{h}|_{L^{p}(\mathbb{R}^{n})}, \quad 1 \leq i \leq n.$$

## Proof.

Since

$$\left| \tau_{\lambda_{i}}^{\tau} u_{h} - u_{h} \right|_{L^{p}(\mathbb{R}^{n})} = \left| \tau_{-\lambda_{i}}^{\tau} u_{h} - u_{h} \right|_{L^{p}(\mathbb{R}^{n})},$$

we can suppose that  $l_i \ge 0$  and we then set

(2.35) 
$$\begin{cases} \ell_{i} = (\alpha_{i} + \beta_{i})h_{i}, \text{ where } \alpha_{i} \text{ is an integer } \geq 0,\\ \text{and } 0 \leq \beta_{i} \leq 1, \quad 1 \leq i \leq n. \end{cases}$$

We write

170

(2.36) 
$$\tau_{\mathcal{A}_{1}}^{+}u_{h}^{-}u_{h}^{+} = \sum_{j=0}^{\alpha_{1}-1} \tau_{jh_{1}}^{+}(\tau_{h_{1}}^{+}u_{h}^{-}u_{h}^{-}) + \tau_{\alpha_{1}h_{1}}^{-}(\tau_{\beta_{1}}^{+}h_{1}^{+}u_{h}^{-}u_{h}^{-}).$$

From (2.36) and (2.33), we get the majoration  $\alpha_{1}-1$ (2.37)  $|\tau_{\ell_{1}} u_{h}-u_{h}|_{L^{p}(\mathbb{R}^{n})} \leq \sum_{j=0}^{\infty} |\tau_{h_{1}} u_{h}-u_{h}|_{L^{p}(\mathbb{R}^{n})} + |\tau_{\beta_{1}} t_{h_{1}} u_{h}-u_{h}|_{L^{p}(\mathbb{R}^{n})}.$ 

But

$$\tau \overrightarrow{h}_{1} - I = h_{1} \tau \overrightarrow{h}_{1} \delta_{1h}$$

and the sum on the right-hand side of (2.37) is equal to

$$\alpha_{1}h_{1} |\delta_{1h}u_{h}|_{L^{p}(\mathbb{R}^{n})};$$

since  $\alpha_1 h_1 \leq \ell_1$ , we obtain

(2.38) 
$$|\tau_{\lambda_1}^{\dagger} u_h - u_h|_{L^p(\mathbb{R}^n)} \leq \ell_1 |\delta_{1h} u_h|_{L^p(\mathbb{R}^n)} + |\tau_{\beta_1 \dot{h}_1} u_h - u_h|_{L^p(\mathbb{R}^n)}$$

Let us now majorize the norm of  $\tau_{\beta_1 \overset{\rightarrow}{h}_1 \overset{\rightarrow}{h}_1 \overset{u}{h}} \overset{-u}{h}$ . For  $x \in \sigma_h(M)$ ,  $x = (x_1, \cdots, x_n)$ , and  $M \in \mathcal{R}_h$ ,  $M = (m_1 h_1, \cdots, m_n h_n)$ , we have

$$\tau_{\beta_{1}\overset{\rightarrow}{h}_{1}}u_{h}(x)-u_{h}(x) = \begin{cases} 0 \quad \text{if} \quad (m_{1}-\frac{1}{2})h_{1} < x_{1} < (m_{1}-\beta_{1}+\frac{1}{2})h_{1} \\ h_{1}\delta_{1h}u_{h}(M+\frac{\dot{h}}{2}) \quad \text{if} \quad (m_{1}-\beta_{1}+\frac{1}{2})h_{1} < x_{1} < (m+\frac{1}{2})h_{1}. \end{cases}$$

Hence

$$\int_{\sigma_{h}(M)} |\tau_{\beta_{1}\vec{h}_{1}}u_{h}-u_{h}|^{p} dx = \beta_{1}h_{1}^{p} \int_{\sigma_{h}(M)} |\delta_{1h}u_{h}(x+\frac{\vec{h}_{1}}{2})|^{p} dx,$$

and summing these equations for the different points M of  $\mathcal{R}_h$ , we obtain

$$|\tau_{\beta_1 \overset{\rightarrow}{h}_1} u_h - u_h|_{L^p(\mathbb{R}^n)} = \beta_1^{\overset{\rightarrow}{p}} h_1 |\tau_{\overset{\rightarrow}{h}_1} \delta_{1h} u_h|_{L^p(\mathbb{R}^n)}$$

Since h is bounded and  $\beta_1 h_1 \leq \ell_1$ , this is less than 1

$$c^{L_{1}^{\overline{p}}} |\delta_{1h}^{u_{h}}|_{L^{p}(\mathbb{R}^{n})}$$

and (2.34) follows for i = 1; the proof is the same for  $i = 2, \dots, n$ .

#### Remark 2.3.

In the most common applications of Theorem 2.2, the family  $\mathcal{E}$  is a sequence of elements  $u_{h_m}^{}$ , where  $h_m^{}$  is converging to zero. Hence  $\mathcal{E}_{h_m^{}} = \{u_h^{}\}$  and  $\mathcal{E}_{h_m^{}}$  is empty if h is not an  $h_m^{}$ .

#### Remark 2.4.

Let us suppose that  $\Omega$  is bounded and that

(2.39) 
$$\mathcal{E} = \bigcup_{|h| < c_0} \mathcal{E}_h, \quad \mathcal{E}_h = \{u_h \in W_h, \quad ||u_h||_h \le 1\}$$

where  $W_h$  is the approximation of  $\mathbb{H}^1_0(\Omega)$  by finite differences, (APX 1). We infer from Theorem 2.2, that  $\mathcal{E}$  is a relatively compact set in  $\mathbb{L}^2(\Omega)$ . The following set  $\mathcal{E}$ , which is a subset of  $\mathcal{E}$ , is also relatively compact in  $\mathbb{L}^2(\Omega)$ :

(2.40) 
$$\mathcal{E}' = \bigcup_{|h| \le c_0} \mathcal{E}'_h, \quad \mathcal{E}'_h = \{u_h \in V_h, \|u_h\|_h \le 1\};$$

 $V^{}_{\rm h}$  corresponds to the approximation (APX 1) of V.
#### \$3. APPROXIMATION OF THE STATIONARY NAVIER-STOKES EQUATIONS.

We discuss here the approximation of the stationary Navier-Stokes equations by numerical schemes of the same type as those used for the linear Stokes problem. We give in Section 3.1 a general convergence theorem; we then apply it in Section 3.2 to numerical schemes based on the approximations (APX 1),...,(APX 4) of the space V. In Section 3.3 we extend to the nonlinear case the numerical algorithms discussed in Section 5, Chapter I.

All of this section appears to be an extension to the nonlinear case of the results obtained in the linear case, in Sections 3,4 and 5. Nevertheless, the results are not as strong here as in the linear case because, in particular, of the nonuniqueness of solutions of the exact problem. Moreover, the convergence of numerical schemes based on the approximation (APX 5) of V is not known, because of the lack in this case of a discrete compactness theorem of the type contained in Section 2, for finite differences.

#### 3.1 A General Convergence Theorem.

Let  $\Omega$  be a bounded lipschitzian open set in  $\mathbb{R}^n$  and let f be given in  $\mathbb{L}^2(\Omega)$ . By Theorem 1.2 there exists at least one u in V such that

(3.1)  $\forall ((u,v)) + b(u,u,v) = (f,v), \quad \forall v \in \widetilde{V}.$ 

Because of Theorem 1.3, this u is unique if  $n \leq 4$  and if v is sufficiently large.

Our purpose here is to discuss the approximation of Problem 3.1 .

Let there be given first an external, stable and convergent Hilbert approximation of the space V, say  $(\overline{\omega},F)$ ,  $(V_h,p_h,r_h)_{h\in\mathcal{H}}$ , where the  $V_h$  are finite dimensional; at this point this approximation could be, in particular, any of the approximations (APX 1),..., (APX 5), described in Chapter I.

Let us suppose that we are given some consistent approximation of the bilinear form v((u,v)), and the linear form (f,v), satisfying the same hypotheses as in Section 3, Chapter I:

(i) for each  $h \in \mathcal{H}$ ,  $a_h(u_h, v_h)$  is a bilinear continuous form on  $V_h \times V_h$ , uniformly coercive in the sense

(3.2) 
$$\exists \alpha_0 > 0$$
 independent of h, such that  
 $a_h(u_h, u_h) \ge \alpha_0 \|u_h\|_h^2, \ \forall u_h \in V_h,$ 

where  $\|\cdot\|_{h}$  stands for the norm in  $V_{h}$ .

(ii) for each 
$$h \in \mathcal{H}$$
,  $\ell_h$  is a linear continuous form on  $V_h$ , such that  
(3.3)  $\| \ell_h \|_{V, \frac{1}{h}} \leq \beta.$ 

The required consistency hypotheses are:

(3.4) If the family  $v_h$  converges weakly to v as  $h \rightarrow 0$  and if the family  $w_h$  converges strongly to w as  $h \rightarrow 0$ <sup>(1)</sup>, then  $\lim_{h \rightarrow 0} a_h(v_h, w_h) = v((v, w)),$   $\lim_{h \rightarrow 0} a_h(w_h, v_h) = v((w, v)).$ 

(3.5) If the family  $v_{\rm b}$  converges weakly to v as  $h \rightarrow 0$ , then

$$\lim_{h \to 0} \langle \ell_h, v_h \rangle = (f, v).$$

For the approximation of the form b, we suppose that we are given a trilinear continuous form  $b_h(u_h, v_h, w_h)$ , on  $V_h$ , such that:

$$(3.6) \qquad b_h(u_h, v_h, v_h) = 0, \quad \forall u_h, v_h \in V_h.$$

(3.7) if the family  $v_h$  converges weakly to v, as  $h \rightarrow 0$ , and if w belongs to  $\mathcal{V}$ , then

$$\lim_{h \to 0} b_h(v_h, v_h, r_h w) = b(v, v, w),$$

Sometimes it will be useful to be more precise about the continuity of  $b_h$  and we will require

(3.8) 
$$|\mathbf{b}_{h}(\mathbf{u}_{h},\mathbf{v}_{h},\mathbf{w}_{h})| \leq c(n,\Omega) \|\mathbf{u}_{h}\|_{h} \|\mathbf{v}_{h}\|_{h} \|\mathbf{w}_{h}\|_{h} \forall \mathbf{u}_{h},\mathbf{v}_{h},\mathbf{w}_{h} \in \mathbf{V}_{h},$$

where the constant  $c = c(n, \Omega)$  depends on n and  $\Omega$  but not on h: an inequality

(1) We recall that this means

 $p_h v_h \longrightarrow \overline{\omega} v$  in F weakly,  $p_h w_h \longrightarrow \overline{\omega} w$  in F strongly. such as (3.8) with c depending on h is obvious, since such an inequality is equivalent to the continuity of the trilinear form  $b_{h}$ .

We can now define an approximate problem for (3.1):

(3.9) 
$$\begin{cases} \underline{\text{To find}} & u_h \in V_h, \underline{\text{such that}} \\ a_h(u_h, v_h) + b_h(u_h, u_h, v_h) = \langle l_h, v_h \rangle. \end{cases}$$

We then have

Proposition 3.1.

For each h, there exists at least one  $u_h \stackrel{\text{in } V}{=} h$ , which is a solution of (3.9).

If (3.8) holds and if

$$(3.10) \qquad \qquad \alpha_0^2 > c(n,\Omega)\beta,$$

then, u is unique.

Proof.

The existence of  $u_h$  follows from Lemma 1.4. We apply this lemma with  $X = V_h$  which is a finite dimensional Hilbert space for the scalar product  $((\cdot, \cdot))_h$ . We define the operator P from  $V_h$  into  $V_h$  by,

$$(3.11) \quad ((P(u_h), v_h))_h = a_h(u_h, v_h) + b_h(u_h, u_h, v_h) - \langle \ell_h, v_h \rangle, \quad \forall u_h, v_h \in V_h.$$

The operator P is continuous and there remains only to check (1.29); but, with (3.6).

$$((P(u_{h}), u_{h}))_{h} = a_{h}(u_{h}, u_{h}) - \langle \ell_{h}, u_{h} \rangle$$

$$\geq (by (3.2) - (3.3))$$

$$\geq \alpha_{0} \|u_{h}\|_{h}^{2} - \|\ell_{h}\|_{V, h} \|u_{h}\|_{h}$$

$$\geq (\alpha_{0} \|u_{h}\|_{h} - \beta) \|u_{h}\|_{h}.$$

Therefore

$$((P(u_h), u_h))_h > 0,$$

provided

$$\|u_h\|_h = k$$
, and  $k > \frac{\beta}{\alpha_0}$ .

Lemma 1.4 gives the existence of at least one  $u_{h}$  such that

$$P(u_h) = 0$$

or

(3.12) 
$$((P(u_h), v_h))_h = 0, \ \forall v_h \in V_h,$$

which is exactly equation (3.9).

Let us suppose that (3.8) and (3.10) hold and let us show that  $u_h$  is unique. If  $u_h^*$  and  $u_h^{**}$  are two solutions of (3.9) and if  $u_h = u_h^* - u_h^{**}$ , then

$$a_{h}(u_{h},v_{h}) + b_{h}(u_{h}^{*},u_{h}^{*},v_{h}) - b_{h}(u_{h}^{*},u_{h}^{*},v_{h}) = 0, \forall v_{h} \in V_{h};$$

taking  $v_h = u_h$  and using (3.6) we find

$$a_h(u_h, u_h) = b_h(u_h, u_h^*, u_h)$$

Because of (3.2) and (3.8), we have

(3.13) 
$$\alpha_0 \| u_h \|_h^2 \le c \| u_h^* \|_h \| u_h \|_h^2$$

If we set  $v_h = u_h^*$  in equation (3.9) satisfied by  $u_h^*$ , we find

$$a_{h}(u_{h}^{*},u_{h}^{*}) = {}^{$$

and with (3.2) and (3.3),

(3.14) 
$$\begin{aligned} \alpha_0 \| \mathbf{u}^*_{\mathbf{h}} \|_{\mathbf{h}}^2 &\leq \beta \| \mathbf{u}^*_{\mathbf{h}} \|_{\mathbf{h}} \\ \| \mathbf{u}^*_{\mathbf{h}} \|_{\mathbf{h}} &\leq \frac{\beta}{\alpha_0} \end{aligned}$$

Using this majoration in (3.13) gives,

(3.15) 
$$(\alpha_0 - \frac{c\beta}{\alpha_0}) \|u_h\|_h^2 \le 0$$

if (3.10) holds, this shows that  $u_h = 0$ .

Theorem 3.1.

We assume that conditions (3.2) to (3.7) are satisfied;  $u_h$  is some solution of (3.9).

If  $n \le 4$ , the family  $\{p_h u_h\}$  contains subsequences which are strongly

convergent in F. Any such subsequence converges to  $\overline{\omega}u$ , where u is some solution of (3.1). If the solution of (3.1) is unique, the whole family  $\{p_h u_h\}$  converges to  $\overline{\omega}u$ .

If  $n \ge 5$ , we have the same conclusions, with only weak convergences in F. Proof.

We suppose that n is arbitrary.

Putting  $v_h = u_h$  in (3.9), and using (3.2), (3.3) and (3.6) we find

(3.16) 
$$a_{h}(u_{h},u_{h}) = \langle l_{h},u_{h} \rangle ,$$
$$\alpha_{0} \| u_{h} \|_{h} \leq \beta.$$

Since the  $p_h$  are stable, the family  $p_h u_h$  is bounded in F; therefore there exists some subsequence  $h' \rightarrow 0$ , and some  $\phi \in F$  such that

$$p_{h}, u_{h}, \longrightarrow \phi$$
 in F weakly.

The condition (C2) for the approximation of a space shows that, necessarily  $\phi \in \overline{\omega}V$ , or  $\phi = \overline{\omega}u$ ,  $u \in V$ :

$$(3.17) \qquad p_{h'} u_{h'} \xrightarrow{\omega} u \quad \text{in F weakly, } h' \longrightarrow 0.$$

Let v be an element of V and let us write (3.9) with  $v_h = r_h v$ :

(3.18) 
$$a_h(u_h, r_h v) + b_h(u_h, u_h, r_h v) = \langle \ell_h, r_h v \rangle.$$

As  $h' \rightarrow 0$ , according to (3.4), (3.5), (3.7),

$$a_{h}, (u_{h}, r_{h}, v) \longrightarrow v((u, v)),$$
  

$$b_{h}, (u_{h}, u_{h}, r_{h}, v) \longrightarrow b(u, u, v),$$
  

$$< \ell_{h}, r_{h}, v > \longrightarrow (f, v).$$

Hence u belongs to V and satisfies

(3.19) 
$$\qquad \qquad \forall ((u,v)) + b(u,u,v) = (f,v), \quad \forall v \in \mathcal{V}.$$

If  $n \le 4$ , equation (3.19) holds, by continuity, for each  $v \in V$ ; if  $n \ge 5$ , we find, by continuity, that (3.19) is satisfied for each  $v \in \tilde{V}$ . In both cases, u is a solution of the stationary Navier-Stokes equations.

It can be proved by exactly the same method, that any convergent subsequence of  $p_h u_h$ , converges to  $\overline{\omega}u$ , where u is some solution of (3.1). If this solution is unique, the whole family  $p_h u_h$  converges to  $\overline{\omega}u$  in F weakly.

Let us show the strong convergence when  $n \leq 4$ .

In order to prove this strong convergence, we consider as in the linear case (see Theorem I.3.1) the expression

$$X_h = a_h(u_h - r_h u, u_h - r_h u)$$
.

Expanding this expression and using (3.9) with  $v_{\rm h} = u_{\rm h}$  we find

$$X_{h} = \langle \ell_{h}, u_{h} \rangle - a_{h}(u_{h}, r_{h}u) - a_{h}(r_{h}u, u_{h}) + a(r_{h}u, r_{h}u).$$

Because of (2.4), (3.5), and (3.17),

$$X_{h'} \longrightarrow \langle f, u \rangle - a(u, u), \text{ as } h' \longrightarrow 0.$$

We take v = u in (3.1) and use (1.22); this gives

hence

$$(3.20) X_{h'} \longrightarrow 0 \text{ as } h' \longrightarrow 0.$$

We then finish the proof as in the linear case; the inequality (3.2) shows that

$$\|\mathbf{u}_{\mathbf{h}'} - \mathbf{r}_{\mathbf{h}'}\mathbf{u}\|_{\mathbf{h}'} \longrightarrow 0.$$

Since the  $p_h$  are stable, this implies

$$\|\mathbf{p}_{h},\mathbf{u}_{h},-\mathbf{p}_{h},\mathbf{r}_{h},\mathbf{u}\|_{F} \leq \|\mathbf{p}_{h},\|_{\mathcal{L}}(\mathbf{v}_{h},\mathbf{F}) \|\mathbf{u}_{h},-\mathbf{r}_{h},\mathbf{u}\|_{h},\longrightarrow 0.$$

Then we write

$$\left\|\mathbf{p}_{\mathbf{h}},\mathbf{u}_{\mathbf{h}},-\overline{\boldsymbol{\omega}}\mathbf{u}\right\|_{\mathbf{F}}\leq\left\|\mathbf{p}_{\mathbf{h}},\mathbf{u}_{\mathbf{h}},-\mathbf{p}_{\mathbf{h}},\mathbf{r}_{\mathbf{h}},\mathbf{u}\right\|_{\mathbf{F}}+\left\|\mathbf{p}_{\mathbf{h}},\mathbf{r}_{\mathbf{h}},\mathbf{u}-\overline{\boldsymbol{\omega}}\mathbf{u}\right\|_{\mathbf{F}},$$

<sup>&</sup>lt;sup>(1)</sup>This equation does not hold for  $n \ge 5$ , and this is the reason the proof cannot be extended to these cases.

and the two terms on the right-hand side of this inequality converge to 0 as  $h' \rightarrow 0$ . (1)

## 3.2 Applications.

In this section we apply the general convergence theorem to approximation schemes corresponding to the approximations (APX 1), ..., (APX 4) of V.

# Approximation (APX 1)

We choose  $a_h$  and  $l_h$  as in the linear case (see I. (3.62) and (3.63)),

(3.21)  
(3.22) 
$$a_h(u_h,v_h) = v((u_h,v_h))_h$$
  
 $< \ell_h, v_h> = (f,v_h).$ 

Before defining  $b_h$ , we introduce the trilinear form  $\hat{b}(u,v,w)$ ,

(3.23) 
$$\hat{b}(u,v,w) = b'(u,v,w) + b''(u,v,w)$$

(3.24) 
$$b'(u,v,w) = \frac{1}{2} \sum_{i,j=1}^{n} \int_{\Omega} u_{i}(D_{i}v_{j})w_{j}dx$$

(3.25) 
$$b''(u,v,w) = -\frac{1}{2} \sum_{i,j=1}^{n} \int_{\Omega} u_{i}v_{j}(D_{i}w_{j})dx.$$

It is not difficult to see that

(3.26) 
$$\hat{b}(u,u,v) = b(u,u,v), \quad \forall u \in V, \quad \forall v \in \mathcal{V},$$

but  $\hat{b}$  and b are otherwise different.

We now define b<sub>h</sub> as,

(3.27) 
$$b_h(u_h, v_h, w_h) = b_h'(u_h, v_h, w_h) + b_h''(u_h, v_h, w_h)$$

(3.28) 
$$b'_{h}(u_{h}, v_{h}, w_{h}) = \frac{1}{2} \sum_{i,j=1}^{n} \int_{\Omega} u_{ih}(\delta_{ih}v_{jh}) w_{jh} dx$$

(3.29) 
$$b_{h}^{"}(u_{h},v_{h},w_{h}) = -\frac{1}{2} \sum_{i,j=1}^{n} \int_{\Omega} u_{ih} v_{jh}(\delta_{ih} w_{jh}) dx.$$

It is clear that  $b'_h$ ,  $b''_h$ , and hence  $b_h$  are trilinear forms on  $V_h$ ; since

<sup>(1)</sup>We recall that for each  $u \in V$ , and each  $h \in \mathcal{H}$ , there exists  $r_h u \in V_h$ , such that  $p_h r_h u \rightarrow \overline{\omega} u$ , in F strongly, as  $h \rightarrow 0$ (see Proposition 3.1).

.

 $V_{\rm b}$  has a finite dimension, these forms are continuous.

We have to check (3.6) and (3.7); (3.6) is obvious with our choice of the form  $b_{\rm h}$ , and (3.7) is the purpose of the next lemma.

Proof.

Saying that  $p_h u_h$  converges weakly to  $\overline{\omega} u$  means that

(3.31) 
$$u_h \longrightarrow u \text{ in } \mathbb{L}^2(\Omega)$$
 weakly

and

(3.32) 
$$\delta_{ih} u_{h} \longrightarrow D_{i} u$$
 in  $\mathbb{L}^{2}(\Omega)$  weakly,  $1 \leq i \leq n$ .

The Compactness Theorem 2.2 is applicable and shows that

(3.33) 
$$u_h \longrightarrow u \text{ in } \mathbb{L}^2(\Omega) \text{ strongly.}$$

We know that if  $v \in \mathcal{V}$ ,  $p_h r_h v$  converges to  $\overline{\omega}v$  in F strongly; but the proofs of Lemma I.3.1 and of Proposition I.3.5 show actually that

(3.34) 
$$r_h v \longrightarrow v$$
 in the norm of  $\mathbb{L}^{\infty}(\Omega)$ ,

(3.35) 
$$\delta_{ih}r_{h}v \longrightarrow D_{i}v$$
 in the norm of  $\mathbb{L}^{\infty}(\Omega)$ .

If we prove that

$$(3.36) b'_h(u_h, u_h, r_h v) \longrightarrow b'(u, u, v)$$

$$(3.37) b''_h(u_h, u_h, r_h v) \longrightarrow b''(u, u, v),$$

then, according to (3.27), the proof of (3.7) will be complete.

For (3.36) we write

$$\begin{aligned} |\mathbf{b}_{h}^{\prime}(\mathbf{u}_{h},\mathbf{u}_{h},\mathbf{r}_{h}\mathbf{v})-\mathbf{b}^{\prime}(\mathbf{u},\mathbf{u},\mathbf{v})| &\leq c_{0} \sum_{\mathbf{i},\mathbf{j}=1}^{n} \left| \int_{\Omega} (\mathbf{u}_{\mathbf{i}h}\mathbf{v}_{\mathbf{j}h}-\mathbf{u}_{\mathbf{i}}\mathbf{v}_{\mathbf{j}})\delta_{\mathbf{i}h}\mathbf{u}_{\mathbf{j}h}d\mathbf{x} \right| \\ &+ c_{0} \sum_{\mathbf{i},\mathbf{j}=1}^{n} \left| \int_{\Omega} \mathbf{u}_{\mathbf{i}}\mathbf{v}_{\mathbf{j}}(\delta_{\mathbf{i}h}\mathbf{u}_{\mathbf{j}h}-\mathbf{D}_{\mathbf{i}}\mathbf{u}_{\mathbf{j}})d\mathbf{x} \right|. \end{aligned}$$

All the preceding integrals converge to 0, and (3.36) is proved; the proof of (3.37) is similar.

The convergence result given by Theorem 3.1 is the following one:

$$(3.38) u_h, \longrightarrow u in L^2(\Omega) weakly,$$

(3.39) 
$$\delta_{ih}, u_{h}, \longrightarrow D_{i}u$$
 in  $\mathbb{L}^{2}(\Omega)$  weakly,  $1 \leq i \leq n$ .

Exactly as in the linear case it can be shown that there exists some step function

(3.40) 
$$\pi_{h} = \sum_{M \in \hat{\Omega}^{1} h} \pi_{h}^{(M)w}_{hM}$$

such that

therefore a solution  $u_h$  of (3.9) is a step function

(3.42) 
$$\mathbf{u}_{h} = \sum_{M \in \Omega_{h}^{1}} \mathbf{u}_{h}^{(M)} \mathbf{w}_{hM}$$

such that

(3.43) 
$$\sum_{i=1}^{n} (\nabla_{ih} u_{ih}) (M) = 0, \quad \forall M \in \mathbb{A}_{h}^{1}$$

and

$$(3.44) \quad -\nu \sum_{i=1}^{n} \delta_{ih}^{2} u_{h}(M) + \frac{1}{2} \sum_{i=1}^{n} u_{ih}(M) \delta_{ih} u_{h}(M) \\ - \frac{1}{2} \sum_{i=1}^{n} \delta_{ih}(u_{ih} u_{h})(M) + \overline{\nu}_{h} \pi_{h}(M) = f_{h}(M), \quad \forall M \in \widehat{\Omega}_{h}^{1}$$

where

(3.45) 
$$f_h(M) = \frac{1}{h_1 \cdots h_n} \int_{\sigma_h(M)} f(x) dx.$$

When condition (3.8) and some condition similar to (3.10) are satisfied,  $u_h$ and u are unique and the error between u and  $u_h$  can be estimated as in the linear case, if moreover  $u \in \mathbb{C}^3(\overline{\Omega})$  and  $p \in \mathbb{C}^2(\overline{\Omega})$ .

Using the Taylor formula we can write,

(3.46) 
$$-\nu \sum_{i=1}^{n} (\delta_{ih} r_{h} u) (M) + \frac{1}{2} \sum_{i=1}^{n} (r_{h} u)_{i} (M) \delta_{ih} (r_{h} u) (M)$$
$$- \frac{1}{2} \sum_{i=1}^{n} \delta_{ih} (r_{h} u)_{i} r_{h} u) (M) + (\overline{\nu}_{h} p) (M)$$

$$f(M) + \varepsilon_h(M), \quad \forall M \in \mathring{\Omega}_h^1,$$

with

$$|\varepsilon_{h}(M)| \leq c(u,p)|h|,$$

where c(u,p) depends only on the maximum norms of the second and third derivatives of u, and of the second derivatives of p. Equations (3.46) show that

(3.48) 
$$\nu((\mathbf{r}_h \mathbf{u}, \mathbf{v}_h))_h + b_h(\mathbf{r}_h \mathbf{u}, \mathbf{r}_h \mathbf{u}, \mathbf{v}_h) - (\overline{\nabla}_h \pi'_h, \mathbf{v}_h) = (\mathbf{f} + \varepsilon_h, \mathbf{v}_h),$$

for each  $v_h \in W_h$  where  $\pi'_h$  is the step function

(3.49) 
$$\pi'_{h} = \sum_{M \in \hat{\Omega}_{h}^{1}} p(M) w_{hM}$$

Subtracting (3.48) from (3.9) gives

$$\nu((\mathbf{u}_{h} - \mathbf{r}_{h}\mathbf{u}, \mathbf{v}_{h}))_{h} = -\mathbf{b}_{h}(\mathbf{u}_{h}, \mathbf{u}_{h}, \mathbf{v}_{h}) + \mathbf{b}_{h}(\mathbf{r}_{h}\mathbf{u}, \mathbf{r}_{h}\mathbf{u}, \mathbf{v}_{h}) + (\varepsilon_{h}, \mathbf{v}_{h}), \quad \forall \mathbf{v}_{h} \in \mathbb{W}_{h}$$

We take  $v_h = u_h - r_h u$  and use (3.6) to get:

$$\forall \| \mathbf{u}_{h} - \mathbf{r}_{h} \mathbf{u} \|_{h}^{2} = -\mathbf{b}_{h} (\mathbf{u}_{h} - \mathbf{r}_{h} \mathbf{u}, \mathbf{u}_{h}, \mathbf{u}_{h} - \mathbf{r}_{h} \mathbf{u}) + (\varepsilon_{h}, \mathbf{u}_{h} - \mathbf{r}_{h} \mathbf{u}),$$

and with (3.8) and (3.47),

$$\begin{array}{l} v \| u_{h} - r_{h} u \|_{h}^{2} \leq c(n, \Omega) \| u_{h} \|_{h} \| u_{h} - r_{h} u \|_{h}^{2} + c(u, p) \| u_{h} - r_{h} u \|_{h} \| h \| \\ v \| u_{h} - r_{h} u \|_{h} \leq c(n, \Omega) \| u_{h} \|_{h} \| u_{h} - r_{h} u \|_{h} + c(u, p) \| h \| \\ \leq (by \ (3.16)) \\ \leq \frac{\beta}{\alpha_{b}} c(n, \Omega) \| u_{h} - r_{h} u \|_{h} + c(u, p) \| h \|. \end{array}$$

Finally

(3.50) 
$$(v - \frac{\beta}{\alpha_0} c(n, \Omega)) \| u_h - r_h u \|_h \le c(u, p) |h|.$$

181

If

$$(3.51) \qquad \qquad \forall \alpha_n > \beta c(n, \Omega),$$

(the constant  $c(n,\Omega)$  in (3.8)), this gives the following majoration of the error

(3.52) 
$$\| u_h - r_h u \|_h \leq \frac{c(u,p)}{\left[ v - \frac{\beta}{\alpha_0} c(n,\Omega) \right]} \| h \|.$$

## Approximation (APX 2)

In the two dimensional case we can associate to the approximation (APX 2) of V described in Section I.4.2, a new discretization scheme for the Navier-Stokes equations.

We recall that  $V_h$  is a subspace of  $\mathbb{H}^1_0(\Omega)$  for this approximation, and we take as in the linear case

(3.53) 
$$a_h(u_h, v_h) = v((u_h, v_h)),$$

(3.54) 
$$\langle l_h, v_h \rangle = (f, v_h),$$

where  $((\cdot, \cdot))$  is the scalar product in  $\mathbb{V}_{h}$  and in  $\mathbb{H}_{0}^{1}(\Omega)$ . We define the form  $b_{h}$  by

(3.55) 
$$b_h(u_h, v_h, w_h) = \hat{b}(u_h, v_h, w_h), \quad \forall u_h, v_h, w_h \in V_h,$$

 $\hat{b}$  defined by (3.23) - (3.25); since  $V_h$  is a space of bounded vector functions, the forms  $\hat{b}$  are defined on  $V_h$ ; they are trilinear, hence continuous.

Condition (3.6) is obviously satisfied with our choice of  $b_h$ ; condition (3.7) is the purpose of the next lemma.

$$\frac{\text{Lemma } 3.2}{\text{If } p_h u_h = u_h} \xrightarrow{\text{converges weakly to}} \overline{\omega} u = u, \quad \underline{\text{then}}$$
(3.56)  $b_h(u_h, u_h, r_h v) \rightarrow b(u, u, v), \quad \forall v \in \mathcal{V}.$ 

## Proof.

We recall that  $F = \mathbb{H}_0^1(\Omega)$  and  $\overline{\omega}$  and  $p_h$  are the identity. Saying that  $p_h^u$  converges weakly in F to  $\overline{\omega}u$ , amounts to saying that

(3.57) 
$$u_h \longrightarrow u \text{ in } \mathbb{H}^1_0(\Omega) \text{ weakly.}$$

The Compactness Theorem 1.1 shows then that

(3.58)  $u_h \longrightarrow u \text{ in } \mathbb{L}^2(\Omega) \text{ strongly.}$ 

The proof of Lemma 3.2 will be the same as that of Lemma 3.1, if we observe that

(3.59) 
$$r_{b}v \rightarrow v$$
 in the norm of  $\mathbb{L}^{\infty}(\Omega)$ 

(3.60) 
$$D_i r_h v \longrightarrow D_i v$$
 in the norm of  $\mathbb{L}^{(\Omega)}$ ,  $1 \le i \le n$ ,

which was actually proved in Proposition I.4.3 (see (4.63)).

The weak (or strong) convergence result given by Theorem 3.1 is the following one:

If  $\rho(h) \longrightarrow 0$ , with  $\sigma(h) \le \alpha$  (i.e.  $h \in \mathcal{H}_{\alpha}$ ), then

(3.61) 
$$u_h^{} \longrightarrow u_h^{} \stackrel{\text{in}}{\longrightarrow} \mathbb{H}^1_0(\Omega) \stackrel{\text{weakly}}{\longrightarrow} (\underline{\text{or strongly}}).$$

Exactly as in the linear case (see Section I.4.2), we can show that there exists a step function  $\pi_{\rm b}^{},$ 

(3.62) 
$$\pi_{h} = \sum_{\substack{b \in \mathfrak{T}_{h}}} \pi_{h}(b) \chi_{h,b},$$

 $(\chi_{h \downarrow} = \text{the characteristic function of the simplex }), such that$ 

$$(3.63) \quad \vee((u_{h},v_{h})) + b_{h}(u_{h},u_{h},v_{h}) - (\pi_{h}, \operatorname{div} v_{h}) = (f,v_{h}), \quad \forall v_{h} \in W_{h}.$$

This equation is the discrete analog of

$$(3.64) \ \nu((u,v)) + b(u,u,v) - (p, \operatorname{div} v) = (f,v), \ \forall v \in \mathbb{H}_0^1(\Omega) \cap \mathbb{L}^n(\Omega).$$

Since the dimension of the space is n = 2, the form  $\hat{b}$  is trilinear continuous on  $\mathbb{H}^1_0(\Omega)$  and there exists some constant  $\hat{c}$  such that

$$(3.65) \qquad |\hat{\mathbf{b}}(\mathbf{u},\mathbf{v},\mathbf{w})| \leq \hat{\mathbf{c}} \|\mathbf{u}\| \|\mathbf{v}\| \|\mathbf{w}\|, \quad \forall \mathbf{u},\mathbf{v},\mathbf{w} \in \mathbb{H}_0^1(\Omega).$$

This can be shown by the same method as (1.18); hence (3.8) holds with

$$c(n,\Omega) = \hat{c}.$$

We will get an estimation of the error assuming that  $u \in \mathbb{C}^3(\overline{\Omega})$ ,  $p \in \mathbb{C}^1(\overline{\Omega})$ , and an hypothesis similar to (3.51). We take  $v_h = r_h u - u_h$  in (3.63) and  $v = r_h u - u_h$  in (3.64); subtracting these equations we find 184

(3.66) 
$$\vee ((u-u_h, r_h u-u_h)) + \hat{b}(u, u, r_h u-u_h) - \hat{b}(u_h, u_h, r_h u-u_h)$$
  
-  $(p-\pi_h, \operatorname{div}(r_h u-u_h)) = 0.$ 

Since div( $r_h u - u_h$ ) is a step function which is constant on each simplex  $\mathcal{A} \in \mathbb{N}$ , we have

$$(p-\pi_h, \operatorname{div}(r_h u-u_h)) = (p-\pi'_h, \operatorname{div}(r_h u-u_h))$$

where  $\pi_{h}^{i}$  is defined by

(3.67) 
$$\pi'_{h} = \sum_{\mathcal{A} \in \mathfrak{T}_{h}} \frac{1}{(\text{meas } \mathcal{A})} \left( \int_{\mathcal{A}} p(\mathbf{x}) d\mathbf{x} \right) \chi_{h \mathcal{A}}.$$

Hence

(3.68) 
$$|(p-\pi_{h}, div(r_{h}u-u_{h})| \leq |\pi'_{h}-p| |div(r_{h}u-u_{h})| \leq \sqrt{2} |\pi'_{h}-p| ||u_{h}-r_{h}u||.$$

We then estimate the difference

$$\hat{b}(u_{h}, u_{h}, r_{h}u-u) - \hat{b}(u, u, r_{h}u-u)$$

$$= \hat{b}(u_{h}-r_{h}u, u_{h}, r_{h}u-u_{h}) + \hat{b}(r_{h}u, u_{h}, r_{h}u-u_{h}) - \hat{b}(u, u, r_{h}u-u)$$

$$= \hat{b}(u_{h}-r_{h}u, u_{h}, r_{h}u-u_{h}) + \hat{b}(r_{h}u, r_{h}u-u, r_{h}u-u_{h}) + \hat{b}(r_{h}u-u, u, r_{h}u-u_{h})$$

The absolute value of this sum can be majorized because of (3.65) by

(3.69) 
$$\hat{c} \| u_h \| \| r_h u - u_h \|^2 + \hat{c} \{ \| r_h u \| + \| u \| \} \| r_h u - u \| \| r_h u - u_h \|.$$

We recall that

$$v \| \mathbf{u}_h \|^2 = \langle \mathbf{f}, \mathbf{u}_h \rangle$$

and therefore

$$\|\mathbf{u}_{\mathbf{h}}\| \leq \frac{1}{\nu} \|\mathbf{f}\|.$$

Hence the sum (3.69) is less than or equal to

(3.70) 
$$\frac{\hat{c}}{v} |f| ||u_h - r_h u||^2 + \hat{c} \{ ||r_h u|| + ||u|| \} ||r_h u - u|| ||r_h u - u_h||.$$

With the majorations (3.68) and (3.70), we get from (3.66)

+ 
$$\hat{c}\{\|\mathbf{r}_{h}\mathbf{u}\| + \|\mathbf{u}\|\}\|\|\mathbf{u}_{h}-\mathbf{r}_{h}\mathbf{u}\|\|\|\mathbf{u}-\mathbf{r}_{h}\mathbf{u}\|$$
,

and finally

$$(3.71) (v - \frac{\hat{c}}{v} |f|) ||u_h - r_h u|| \le v ||u - r_h u|| + \sqrt{2} |\pi'_h - p| + \hat{c} \{||r_h u|| + ||u||\} ||u - r_h u||.$$

Under the assumption

(3.72) 
$$v^2 > \hat{c} |f|,$$

the inequality (3.71) gives a majoration of the error between  $u_h$  and  $r_h u$  and hence between u and  $u_h$ .

## Approximation (APX 3).

As in the linear case, the method here is very similar to the method used for the approximation (APX 2).

#### Approximation (APX 4).

We recall that  $\Omega$  must be a simply connected open set in  $\mathbb{R}^2$ .

Since (APX 4) is an internal approximation of V, the simplest scheme (3.9) associated with this approximation is

(3.73) To find  $u_h \in V_h$  such that  $v((u_h, v_h)) + b(u_h, u_h, v_h) = \langle f, v_h \rangle, \forall v_h \in V_h.$ 

Theorem 3.1 is applicable and shows that

(3.74) 
$$u_h \longrightarrow u \text{ in } V, \text{ as } \rho(h) \longrightarrow 0,$$

provided  $\sigma(h) \leq \alpha$  (i.e.  $h \in \mathcal{H}_{\alpha}$ ).

The error between u and  $u_h$  can be estimated as follows (if we have uniqueness): we take  $v = u-u_h$  in the variational equation satisfied by u:

$$v((u,v)) + b(u,u,v) = (f,v), \quad \forall v \in V$$

We then take  $v = r_h u - u_h$  in (3.73); we subtract these equations and find

(3.75) 
$$\|u_h - u\|^2 = v((u - u_h, u - r_h u)) + b(u_h, u_h, r_h u - u_h) - b(u, u, r_h u - u_h).$$

The difference

$$b(u_h, u_h, r_h u - u_h) - b(u, u, r_h u - u_h)$$

is equal to

$$b(u_{h}-u,u_{h},r_{h}u-u_{h}) + b(u,u_{h}-u,r_{h}u-u_{h})$$

$$= b(u_{h}-u,u_{h},r_{h}u-u) + b(u_{h}-u,u_{h},u-u_{h}) + b(u,u_{h}-u,r_{h}u-u)$$

$$= b(u_{h}-u,u_{h},r_{h}u-u) + b(u_{h}-u,u,u-u_{h}) + b(u,u_{h}-u,r_{h}u-u).$$

Because of (1.18) this is majorized by

$$c\{\|u\| + \|u_h\|\} \|u_h - u\| \|r_h u - u\| + c\|u\| \|u - u_h\|^2$$
.

We recall that

$$v \| u \|^{2} = \langle f, u \rangle,$$
  
 $\| u \| \leq \frac{1}{v} \| f \|_{V},$ 

and similarly

$$\begin{aligned} & \left\| \mathbf{u}_{h} \right\|^{2} = \langle \mathbf{f}, \mathbf{u}_{h} \rangle \\ & \left\| \mathbf{u}_{h} \right\| \leq \frac{1}{\nu} \left\| \mathbf{f} \right\|_{V}, \end{aligned}$$

Therefore the last expression is majorized by

$$\frac{2c}{\nu} |\mathbf{f}| \|\mathbf{u}_{\mathbf{h}} - \mathbf{u}\| \|\mathbf{r}_{\mathbf{h}} - \mathbf{u}\| + \frac{c}{\nu} |\mathbf{f}| \|\mathbf{u} - \mathbf{u}_{\mathbf{h}}\|^{2}.$$

We deduce then from (3.75)

(3.76) 
$$(\nu - \frac{c}{\nu} |f|) ||u_{h} - u||^{2} \leq (\nu + \frac{2c}{\nu} |f|) ||u_{h} - u|| ||r_{h} u - u||,$$

$$(\nu - \frac{c}{\nu} |f|) ||u_{h} - u|| \leq (\nu + \frac{2c}{\nu} |f|) ||r_{h} u - u||.$$

If

(3.77) 
$$v^2 > c \|f\|_{v}, ,$$

inequality (3.76) shows that the error  $\|\mathbf{u}_{h}-\mathbf{u}\|$  has the same order as  $\|\mathbf{r}_{h}\mathbf{u}-\mathbf{u}\|$ . Since c is the constant c(n), n = 2, in (1.18), the inequality (3.77) is exactly inequality (1.37) which ensures the uniqueness of the solution u of the exact problem.

# Approximation (APX 5).

If we consider the approximation (APX 5) of V, we can set

(3.78)  $a_h(u_h, v_h) = v((u_h, v_h))_h$ 

(3.79) 
$$< l_h, v_h > = (f, v_h)$$

$$(3.80) \quad b_{h}(u_{h},v_{h},w_{h}) = \frac{1}{2} \sum_{i,j=1}^{n} \int_{\Omega} u_{ih}[(D_{ih}v_{jh})w_{jh}-v_{jh}(D_{ih}w_{jh})]dx$$

Proposition 3.1 is applicable and shows the existence of at least one  $u_{\rm h} \in V_{\rm h}$  such that

(3.81) 
$$\nu((u_h, v_h))_h + b_h(u_h, u_h, v_h) = (f_i, v_h), \quad \forall v_h \in v_h.$$

Theorem 3.1 cannot be applied since we do not know if condition (3.7) is satisfied, because of the lack of discrete compactness theorems for this type of approximation.

#### 3.3 Numerical Algorithms.

The following analysis is restricted to the dimensions  $n \leq 4$ . We wish to extend to the nonlinear case the numerical algorithms described, for the linear case, in Section 5, Chapter I.

We observe that the stationary Navier-Stokes equations are not the Euler equations of an optimization problem like the Stokes equations. The following algorithms are then some extension of the Uzawa and Arrow-Hurwicz algorithms classically related to optimization problems.

In the sequel of this section we will always use the trilinear form  $\hat{b}(u,v,w)$ defined by (3.23) - (3.25). This is a trilinear continuous form on  $\mathbb{H}_0^1(\Omega) \times \mathbb{H}_0^1(\Omega) \times \mathbb{H}_0^1(\Omega)$  and there exists some constant  $\hat{c} = \hat{c}(n)$  such that

 $(3.82) |\hat{b}(u,v,w)| \leq \hat{c}(n) ||u|| ||v|| ||w||, \quad \forall u,v,w \in \mathbb{H}_0^1(\Omega), \ (n \leq 4).$ 

We already noticed that

(3.83) 
$$\hat{b}(u,v,w) = b(u,v,w) \text{ if } u \in \mathbb{V}, v, w \in \mathbb{H}_0^1(\Omega)$$

(3.84) 
$$\hat{b}(u,v,v) = 0, \quad u,v \in \mathbb{H}_0^1(\Omega).$$

# Uzawa Algorithm.

In order to approximate the solutions of (1.8) - (1.11) we will construct as in Section I.5, two sequences of elements

(3.85)  $u^{m} \in \mathbb{H}_{0}^{1}(\Omega), p^{m} \in \mathbb{L}^{2}(\Omega).$ 

This construction is relatively easy because the condition div = 0 will disappear in the approximate equations.

We start the algorithm with an arbitrary element p°:

$$(3.86) p0 \in L2(\Omega).$$

When  $p^{m}$  is known (m  $\geq$  0), we define  $u^{m+1}$  as some solution of

(3.87) 
$$\begin{cases} u^{m+1} \in \mathbb{H}_0^1(\Omega), \text{ and} \\ v((u^{m+1}, v)) + \hat{b}(u^{m+1}, u^{m+1}, v) = (p^m, \operatorname{div} v) + (f, v), \forall v \in \mathbb{H}_0^1(\Omega). \end{cases}$$

We then define p by

(3.88) 
$$\begin{cases} p^{m+1} \in L^{2}(\Omega), \text{ and} \\ (p^{m+1}-p^{m},q) + \rho(\operatorname{div} u^{m+1},q) = 0, \forall q \in L^{2}(\Omega). \end{cases}$$

Later we will give the conditions that the number  $\rho > 0$  must satisfy.

The existence of  $u^{m+1}$  satisfying (3.87) is not obvious, but can be proved using the Galerkin method, exactly as in Theorem 1.2. Therefore we will skip the proof. It is not difficult to see that  $u^{m+1}$  is the solution of the following nonlinear Dirichlet problem:

(3.89) 
$$\begin{cases} u^{m+1} \in \mathbb{H}_{0}^{1}(\Omega) \\ -\nu\Delta u^{m+1} + \sum_{i=1}^{n} u^{m+1}_{i} D_{i} u^{m+1} + \frac{1}{2} (\operatorname{div} u^{m+1}) u^{m+1} \\ & = -\operatorname{grad} p^{m} + f \in \mathbb{H}^{-1}(\Omega). \end{cases}$$

The solution of (3.87)-(3.89) is not unique, in general. When  $u^{m+1}$  is known,  $p^{m+1}$  is explicitly given by (3.88) which is equivalent to

(3.90) 
$$p^{m+1} = p^m - \rho \operatorname{div} u^{m+1} \in L^2(\Omega).$$

To investigate convergence we will assume that

(3.91) 
$$v - \frac{c(n)}{v} \|f\|_{V}^{-} = \overline{v} > 0;$$

with (3.82), (3.83) and Theorem 1.3, the condition (3.91) implies the uniqueness of the solution of (1.8) - (1.11); p is unique up to an additive constant; we fix this constant by asking that

(3.92) 
$$\int_{\Omega} p(x) dx = 0.$$

Proposition 3.2.

We assume that  $n \le 4$  and that condition (3.91) holds. We suppose also that the number  $\rho$  satisfies

(3.93) 
$$0 < \rho < \frac{2\nu}{n}$$
.

Then, as  $m \rightarrow \infty$ ,

(3.94) 
$$u^{m} \rightarrow u \quad \underline{\text{in the norm of }} \mathbb{H}^{1}_{0}(\Omega)$$
,  
(3.95)  $p^{m} \rightarrow p \quad \underline{\text{in }} \mathbb{L}^{2}(\Omega), \quad \underline{\text{weakly}},$ 

where {u,p} is the unique solution of (1.8) - (1.11) which satisfies (3.92). Proof.

We set

(3.96) 
$$\begin{cases} v^{m+1} = u^{m+1} - u \\ q^{m+1} = p^{m+1} - p \end{cases}$$

and we proceed as in the proof of Theorem 1.5.1.

We subtract equation (3.87) from the equation

(3.97) 
$$V((u,v)) + \hat{b}(u,u,v) - (p,divv) = \langle f,v \rangle, \forall v \in \mathbb{H}^{1}_{0}(\Omega)$$

and we take  $v = v^{m+1}$  to obtain

$$v \| v^{m+1} \|^{2} = (q^{m}, \text{ div } v^{m+1}) + \hat{b}(u, u, v^{m+1}) - \hat{b}(u^{m+1}, u^{m+1}, v^{m+1})$$
$$= -(q^{m+1}-q^{m}, \text{ div } v^{m+1}) - (q^{m+1}, \text{ div } v^{m+1})$$
$$+ \hat{b}(v^{m+1}, u, v^{m+1})$$

$$\leq (by (3.82) \text{ and } (1.39))$$

$$\leq -(q^{m+1}-q^{m}, div v^{m+1}) - (q^{m+1}, div v^{m+1})$$

$$+ \frac{\hat{c}}{v} \|f\|_{V}, \|v^{m+1}\|^{2}.$$

We take  $q = q^{m+1}$  in (3.88) and we find, (3.99)  $|q^{m+1}|^2 - |q^m|^2 + |q^{m+1} - q^m|^2 = -2\rho(\text{div } v^{m+1}, q^{m+1}).$  We multiply the last inequality in (3.98) by  $2\rho$  and add it to (3.99); this gives

$$(3.100) |q^{m+1}|^{2} - |q^{m}|^{2} + |q^{m+1}-q^{m}|^{2} + (v - \frac{\hat{c}}{v} ||f||_{V}) ||v^{m+1}||^{2} \\ \leq -2\rho(q^{m+1}-q^{m}, div v^{m+1}).$$

This inequality is similar to equation (5.12) in the proof of Theorem I.5.1 with  $\nu$  replaced by  $\overline{\nu}$  (see (3.91)). The proof can be completed exactly as in Theorem I.5.1.

#### Remark 3.1.

In the general case, when uniqueness is not assumed, we can prove weak convergence results for the average values

$$\frac{1}{N}\sum_{m=1}^{N}u_{m}, \qquad \frac{1}{N}\sum_{m=1}^{N}p^{m}.$$

These sequences are bounded in  $\mathbb{H}_0^1(\Omega)$  and  $L^2(\Omega)$ , and every weakly convergent subsequence converges to a couple  $\{u,p\}$  which is a solution of (1.8) - (1.11).

#### Arrow-Hurwicz Algorithm.

L

We construct a sequence of couples  $\{u^m, p^m\}$  defined as follows. We start the algorithm with arbitrary elements

(3.101) 
$$u^0 \in H^1_0(\Omega)$$
,  $p^0 \in L^2(\Omega)$ .  
When  $p^m, u^m$  are known, we define  $p^{m+1}, u^{m+1}$ , as solutions of

(3.102) 
$$\begin{cases} u^{m+1} \in \mathbb{H}_0^1(\Omega) & \text{and} \\ ((u^{m+1}-u^m,v)) + \rho v((u^m,v)) + b(u^m,u^{m+1},v) - \rho(p^m, \operatorname{div} v) = \rho(f,v), \end{cases}$$

(3.103) 
$$\begin{cases} p^{m+1} \in L^{2}(\Omega) \text{ and} \\ \alpha(p^{m+1}-p^{m},q) + \rho(\operatorname{div} u^{m+1},q) = 0, \quad \forall q \in L^{2}(\Omega). \end{cases}$$

We suppose that  $\rho$  and  $\alpha$  are two strictly positive numbers; conditions on  $\rho$  and  $\alpha$  will be given later.

The existence and uniqueness of  $u^{m+1} \in H_0^1(\Omega)$  satisfying (3.102) is easy with

 $\forall v \in \mathbb{H}_0^1(\Omega)$ 

the projection theorem; (3.102) is a linear variational equation equivalent to the Dirichlet problem

$$(3.104) \begin{cases} u^{m+1} \in \mathbb{H}_0^1(\Omega) \\ -\Delta u^{m+1} + \rho \sum_{i=1}^n u_i^m D_i u^{m+1} + \frac{\rho}{2} (\operatorname{div} u^m) u^{m+1} = -\Delta u^m + \rho \vee \Delta u^m + \rho \operatorname{grad} p^m + f. \end{cases}$$

Then  $p^{m+1}$  is explicitly given by (3.103) which is equivalent to

(3.105) 
$$p^{m+1} = p^m - \frac{\rho}{\alpha} \operatorname{div} u^{m+1} \in L^2(\Omega).$$

Convergence can be proved under stronger conditions than those used in Proposition 3.2.

# Proposition 3.3.

We assume that n < 4, that

(3.106) 
$$v - \frac{2\hat{c}}{v} \| f \|_{V}, - \frac{4\hat{c}^2}{v^2} \| f \|_{V}^2, = v^* > 0$$

and that

(3.107) 
$$0 < \rho < \frac{\alpha \nu^*}{2(n+\nu^2 \alpha)}$$
.

<u>Then</u>, as  $m \rightarrow \infty$ ,

(3.108) 
$$u^{m} \longrightarrow u \quad in \ the \ norm \ of \ \mathbb{H}_{0}^{1}(\Omega) ,$$

$$(3.109) pm \longrightarrow p in L2(\Omega) weakly,$$

where {u,p} is the unique solution of (1.8) - (1.11) which satisfies (3.92).
Proof.

We use again the notation (3.96). We take  $v = 2v^{m+1}$  in (3.97) and (3.102) and subtract these equations; this gives

$$\|v^{m+1}\|^{2} - \|v^{m}\|^{2} + \|v^{m+1}-v^{m}\|^{2} + 2\rho v\|v^{m+1}\|^{2}$$

$$= 2\rho v ((v^{m+1}, v^{m+1}-v^{m})) + 2\rho(q^{m}, div v^{m+1})$$

$$+ 2b(u^{m}, u^{m+1}, v^{m+1}) - 2b(u, u, v^{m+1})$$

$$\leq \frac{1}{4} \|v^{m+1}-v^{m}\|^{2} + 4\rho^{2}v^{2}\|v^{m+1}\|^{2} + 2\rho(q^{m}, div v^{m+1})$$

$$+ 2b(v^{m}-v^{m+1}, u, v^{m+1}) + 2b(v^{m+1}, u, v^{m+1})$$

$$+ 2b(v^{m}-v^{m+1}, u, v^{m+1}) + 2b(v^{m+1}, u, v^{m+1})$$

$$(3.110) \leq (because of (3.82) and (1.39)$$

$$\leq \frac{1}{4} \|v^{m+1}-v^{m}\|^{2} + 4\rho^{2}v^{2}\|v^{m+1}\|^{2} + 2\rho(q^{m}, div v^{m+1})$$

$$+ \frac{2\hat{c}}{v} \|f\|_{V}, \|v^{m+1}-v^{m}\| \|v^{m+1}\| + \frac{2\hat{c}}{v} \|f\|_{V} \|v^{m+1}\|^{2}$$

$$< \frac{1}{v} \|v^{m+1}-v^{m}\|^{2} + 4\rho^{2}v^{2} \|v^{m+1}\|^{2} + 2\rho(q^{m}, div v^{m+1})$$

$$+ \frac{2\hat{c}}{v} \|f\|_{V}, \|v^{m+1}-v^{m}\| \|v^{m+1}\| + \frac{2\hat{c}}{v} \|f\|_{V} \|v^{m+1}\|^{2}$$

$$+ \frac{2c}{\nu} \|f\|_{V}, \|v^{m+1} - v^{m}\| \|v^{m+1}\| + \frac{2c}{\nu} \|f\|_{V}, \|v^{m+1}\| + \frac{2c}{\nu} \|f\|_{V}, \|v^{m+1}\| + \frac{2c}{\nu} \|f\|_{V}, \|v^{m+1}\| + \frac{2c}{\nu} \|f\|_{V}, \|v^{m+1}\| + 2\rho(q^{m}, \text{ div } v^{m+1}).$$

We take 
$$q = 2q^{m+1}$$
 in (3.103):  
 $\alpha |q^{m+1}|^2 - \alpha |q^m|^2 = -2\rho(q^{m+1}, \operatorname{div} v^{m+1})$   
 $= -2\rho(q^m, \operatorname{div} v^{m+1}) - 2\rho(q^{m+1}-q^m, \operatorname{div} v^{m+1})$   
(3.111)  
 $\leq -2\rho(q^m, \operatorname{div} v^{m+1}) + 2\rho\sqrt{n}|q^{m+1}-q^m| ||v^{m+1}||$   
 $\leq \frac{\alpha}{2} |q^{m+1}-q^m|^2 + \frac{2\rho^2 n}{\alpha} ||v^{m+1}||^2 - 2\rho(q^m, \operatorname{div} v^{m+1})$ 

(see (5.13), (5.25), (5.26), Chapter I). We add the last inequality in (3.110) to the last inequality in (3.111) and obtain

$$\alpha |q^{m+1}|^{2} - \alpha |q^{m}|^{2} + \frac{1}{2} |q^{m+1} - q^{m}|^{2} + ||v^{m+1}||^{2} - ||v^{m}||^{2} + \frac{1}{2} ||v^{m+1} - v^{m}||^{2}$$

$$+ 2\rho(v - \frac{2\hat{c}}{v} ||f||_{V}, - \frac{4\hat{c}^{2}}{v^{2}} ||f||_{V}^{2}, - 2\rho v^{2} - \frac{2\rho n}{\alpha} )||v^{m+1}||^{2} \leq 0.$$

The conditions (3.106)-(3.107) ensure that the coefficient of  $\|v^{m+1}\|^2$  in (3.112) is strictly positive; this inequality is then similar to the inequality (5.27) in the proof of Theorem I.5.2; we finish the proof as in that theorem.

# CHAPTER III The Evolution Navier-Stokes Equations

#### INTRODUCTION

This final chapter deals with the full Navier-Stokes Equations; i.e., the evolution nonlinear case. First we describe a few basic results concerning existence and uniqueness of solutions, and then study the approximation of these equations by several methods.

In Section 1 we rapidly study the linear evolution equations (evolution Stokes equations). This section contains some technical lemmas appropriate for the study of evolution equations. Section 2 gives compactness theorems which will enable us to obtain strong convergence results in the evolution case, and to pass to the limit in the nonlinear terms. Section 3 contains the variational formulation of the problem (weak or turbulent solutions, according to J. Leray [1], [2], [3]; E. Höpf [2]) and the main results of existence and uniqueness of solution (the dimension of the space is n = 2 or 3); the existence is based on the construction of an approximate solution by the Galerkin method. In Section 4 further existence and uniqueness results are presented; there existence is obtained by semi-discretization in time, and is valid for any dimension of the space.

In the final section we study the approximation of the evolution Navier-Stokes equations, in the two and three dimensional cases. Several schemes are considered corresponding to a classical discretization in the time variable (implicit, Cranck-Nicholson, explicit) associated with any of the discretizations in the space variables introduced in Chapter I (finite differences, finite elements). We conclude with a study of the nonlinear stability of these schemes, establishing sufficient conditions for stability and proving the convergence of all these schemes when they are stable.

## \$1. THE LINEAR CASE.

In this section we develop some results of existence, uniqueness, and regularity of the solutions of the linearized Navier-Stokes equations. After introducing some notation useful in the linear as well as in the nonlinear case (Section 1.1), we give the classical and variational formulations of the problem and the statement of the main existence and uniqueness result (Section 1.2); the proofs of the existence and of the uniqueness are then given in Sections 1.3 and 1.4.

# 1.1 Notation.

Let  $\Omega$  be an open lipschitzian set in  $\mathbb{R}^n$ ; for simplicity we suppose  $\Omega$  bounded, and we refer to the remarks in Section 1.5 for the unbounded case. We recall the definition of the spaces  $\mathcal{V}$ , V,H, used in the previous chapters and which will be the basic spaces in this chapter too:

(1.1)  $\mathcal{V} = \{ u \in \mathfrak{D}(\Omega), \text{ div } u = 0 \}$ 

(1.2) 
$$V = \text{the closure of } \mathcal{V} \text{ in } \mathbb{H}_0^1(\Omega),$$

(1.3)  $H = \text{the closure of } \mathcal{V} \text{ in } \mathbb{L}^2(\Omega).$ 

The space H is equipped with the scalar product  $(\cdot, \cdot)$  induced by  $\mathbb{L}^2(\Omega)$ ; the space V is a Hilbert space with the scalar product

(1.4) 
$$((u,v)) = \sum_{i=1}^{n} (D_{i}u, D_{i}v),$$

since  $\Omega$  is bounded.

The space V is contained in H, is dense in H and the injection is continuous. Let H' and V' denote the dual spaces of H and V, and let i denote the injection mapping from V into H. The adjoint operator i' is linear continuous from H' into V', is one to one since i(V) = V is dense in H and i'(H') is dense in V' since i is one to one; therefore H' can be identified with a dense subspace of V'. Moreover, by the Riesz representation theorem, we can identify H and H', and we arrive at the inclusions

$$(1.5) V \subset H \equiv H' \subset V',$$

where each space is dense in the following one and the injections are continuous.

As a consequence of the previous identifications, the scalar product in H of  $f \in H$  and  $u \in V$  is the same as the scalar product of f and u in the duality between V' and V:

(1.6) 
$$\langle f, u \rangle = (f, u), \forall f \in H, \forall u \in V.$$

For each u in V, the form

(1.7)  $v \in V \mapsto ((u,v)) \in \mathbb{R}$ 

is linear and continuous on V; therefore, there exists an element of V' which we denote by Au such that

(1.8)  $\langle Au, v \rangle = ((u,v)), \quad \forall v \in V.$ 

It is easy to see that the mapping  $u \mapsto Au$  is linear and continuous, and by Theorem I.2.2 and Remark I.2.2, is an isomorphism from V onto V'.

If  $\Omega$  is unbounded, the space V is equipped with the scalar product

(1.9) 
$$[[u,v]] = ((u,v)) + (u,v);$$

the inclusions (1.5) hold. The operator A is linear continuous from V into V' but is not in general an isomorphism; for every  $\varepsilon > 0$ , A +  $\varepsilon$ I is an isomorphism from V onto V'.

Let a,b be two extended real numbers,  $-\infty \le a \le b \le \infty$ , and let X be a Banach space. For given  $\alpha$ ,  $1 \le \alpha \le +\infty$ ,  $L^{\alpha}(a,b;X)$  denotes the space of  $L^{\alpha}$ -integrable functions from [a,b] into X, which is a Banach space with the norm

(1.10) 
$$\left\{\int_{a}^{b} \|f(t)\|_{X}^{\alpha} dt\right\}^{\frac{1}{\alpha}}.$$

The space  $L^{\infty}(a,b;X)$  is the space of essentially bounded functions from [a,b] into X, and is equipped with the Banach norm

(1.11) Ess Sup  $||f(t)||_X$ . [a,b]

The space  $\mathcal{C}([a,b];X)$  is the space of continuous functions from [a,b]into X and if  $-\infty < a < b < \infty$  is equipped with the Banach norm

(1.12) 
$$\sup_{t \in [a,b]} \|f(t)\|_{X}$$
.

Most often the interval [a,b] will be the interval [0,T], T > 0 fixed; when no confusion can arise, we will use the following more condensed notations,

- (1.13)  $\alpha(X) = L^{\alpha}(0,T;X), \quad 1 \leq \alpha \leq +\infty$
- (1.14)  $\mathscr{C}(X) = \mathscr{C}([0,T];X).$

The remainder of this Section 1.1 is devoted to the proof of the following technical lemma concerning the derivatives of functions with values in a Banach space.

<u>Lemma 1.1</u>.

Let X be a given Banach space with dual X' and let u and g be two functions belonging to  $L^1(a,b;X)$ . Then, the following three conditions are equivalent

(i) u is absolutely continuous and u' = g,  
(1.15) 
$$u(t_1) - u(t_0) = \int_{t_0}^{t_1} g(s)ds, \quad \forall t_0, t_1 \in [a,b];$$
  
(ii) For each test function  $\phi \in \mathcal{D}((a,b)),$   
(1.16)  $\int_a^b u(t)\phi'(t)dt = -\int_a^b g(t)\phi(t)dt \quad (\phi' = \frac{d\phi}{dt});$   
(iii) For each  $\eta \in X',$   
(1.17)  $\frac{d}{dt} < u, \eta > = < g, \eta >,$ 

in the scalar distribution sense, on (a,b).

Proof.

We suppose for simplicity that the interval [a,b] is the interval [0,T]. A legitimate integration by parts shows that (i) implies (ii) and (iii); it remains to check that the property (iii) implies the property (ii) and that (ii) implies (i).

If (iii) is satisfied and  $\phi \in \mathcal{D}((0,T))$ , then by definition,

(1.18) 
$$\int_{0}^{T} \langle u(t), n \rangle \phi'(t) dt = - \int_{0}^{T} \langle g(t), n \rangle \phi(t) dt$$

or

$$< \int_0^T u(t)\phi'(t)dt + \int_0^T g(t)\phi(t)dt, \eta > = 0, \forall \eta \in X',$$

so that (1.16) is satisfied.

Let us now prove that (ii) implies (i). We can reduce the general case to the case g = 0. To see this, we set  $v = u-u_0$  with

(1.19) 
$$u_0(t) = \int_0^t g(s) ds;$$

it is clear that  $u_0$  is an absolutely continuous function and that  $u'_0 = g$ ; hence (1.16) holds with u replaced by  $u_0$  and

(1.20) 
$$\int_0^T v(t)\phi'(t)dt = 0, \quad \forall \phi \in \mathcal{D}((0,T)).$$

The proof of (i) will be achieved if we show that (1.20) implies that v is a constant element of X.

Let  $\phi_0$  be some function in  $\mathscr{D}((0,T))$ , such that

 $\int_{0}^{T} \phi_{0}(t) dt = 1.$ 

Any function  $\phi$  in  $\mathscr{D}((0,T))$  can be written as

(1.21) 
$$\phi = \lambda \phi_0 + \psi', \quad \lambda = \int_0^T \phi(t) dt, \quad \psi \in \mathscr{D}((0,T));$$

indeed since

$$\int_0^T (\phi(t) - \lambda \phi_0(t) dt = 0,$$

the primitive function of  $\phi - \lambda \phi_0$  vanishing at 0, belongs to  $\mathscr{D}((0,T))$ , and  $\psi$  is precisely this primitive function. According to (1.20) and (1.21),

(1.21a) 
$$\int_0^T (v(t)-\xi)\phi(t)dt = 0, \quad \forall \phi \in \mathscr{D}((0,T))$$

where

$$\xi = \int_0^T v(s)\phi_0(s)ds.$$

To achieve the proof, it remains to show that (1.21a) implies that

 $v(t) = \xi$  a.e..

i.e., that a function  $w \in L^{1}(X)$  such that

(1.22) 
$$\int_0^T w(t)\phi(t)dt = 0, \quad \forall \phi \in \mathscr{D}((0,T)),$$

is zero almost everywhere. This well-known result is proved by regularization: if  $\widetilde{w}$  is the function equal to w on [0,T] and to 0 outside this interval, and if  $\rho_{\epsilon}$  is an even regularizing function, then for  $\epsilon$  small enough,  $\rho_{\epsilon} \star \phi$  belongs to  $\mathcal{D}((0,T))$ ,  $\forall \phi \in \mathcal{D}((0,T))$ , and

$$\int_{0}^{T} w(t) (\rho_{\varepsilon} \star \phi)(t) dt = \int_{-\infty}^{+\infty} \widetilde{w}(t) (\rho_{\varepsilon} \star \phi)(t) dt = \int_{-\infty}^{+\infty} (\rho_{\varepsilon} \star \widetilde{w})(t) \phi(t) dt = 0.$$

Hence, for any  $\eta > 0$  fixed,  $\rho_{s} \star \widetilde{w}$  is equal to 0 on the interval  $[\eta, T-\eta]$ , for  $\varepsilon$  small enough; as  $\varepsilon \longrightarrow 0$ ,  $\rho_{\varepsilon} * \tilde{w}$  converges to  $\tilde{w}$  in  $L^{1}(-\infty, +\infty; X)$ . Thus w is zero on [n, T-n]; since n > 0 is arbitrarily small, w is zero on the whole interval [0,T].

1.2 The Existence and Uniqueness Theorem.

Let  $\Omega$  be a lipschitzian open bounded set in  $\mathbb{R}^n$  and let T > 0 be fixed. We denote by Q the cylinder  $\Omega \times (0,T)$ . The linearized Navier-Stokes equations, are the evolution equations corresponding to the Stokes problem:

To find a vector function

 $u: \Omega \times [0,T] \longmapsto \mathbb{R}^n$ 

and a scalar function

 $p: \Omega \times [0,T] \mapsto \mathbb{R}$ 

respectively equal to the velocity of the fluid and to its pressure, such that

(1.23) 
$$\frac{\partial u}{\partial t} - v\Delta u + \text{grad } p = f \text{ in } Q = \Omega \times (0,T),$$
  
(1.24) 
$$div \ u = 0 \text{ in } Q,$$
  
(1.25) 
$$u = 0 \text{ on } \partial\Omega \times [0,T],$$
  
(1.26) 
$$u(x,0) = u_0(x), \text{ in } \Omega,$$

where the vector functions f and  $u_0$  are given, f defined on  $\Omega \times [0,T]$ ,  $u_0$ defined on  $\Omega$ ; the equations (1.25) and (1.26) give respectively the boundary and initial conditions.

Let us suppose that u and p are classical solutions of (1.23) - (1.26), say  $u \in \mathcal{C}^2(\overline{Q})$ ,  $p \in \mathcal{C}^1(\overline{Q})$ . If v denotes any element of  $\mathcal{N}$ , it is easily seen that

(1.27) 
$$(\frac{\partial u}{\partial t}, v) + v((u, v)) = (f, v).$$

By continuity, the equality (1.27) holds also for each  $v \in V$ ; we observe also that

$$(\frac{\partial u}{\partial t}, v) = \frac{d}{dt} (u, v).$$

This leads to the following weak formulation of the problem (1.23) - (1.26):

For f and 
$$u_0$$
 given,  
(1.28)  $f \in L^2(0,T;V')$   
(1.29)  $u_0 \in H$ ,

(1.29)

to find u, satisfying

(1.30) 
$$u \in L^2(0,T;V)$$

and

(1.31) 
$$\frac{d}{dt}(u,v) + v((u,v)) = \langle f,v \rangle, \quad \forall v \in V.$$

(1.32)

If u belongs to  $L^2(0,T;V)$  the condition (1.32) does not make sense in general; its meaning will be explained after the following two remarks:

 $u(0) = u_0$ .

(i) The spaces in (1.28) (1.29) (1.30) are the spaces for which existence and uniqueness will be proved; it is clear at least that a smooth solution u of (1.23) - (1.26) satisfies (1.30).

(ii) We cannot check right now that a solution of (1.30) - (1.32) is solution, in some weak sense, of (1.23) - (1.26); hence we postpone the investigation of this point until Section 1.5.

By (1.6) and (1.8), we can write (1.30) as

(1.33) 
$$\frac{d}{dt} \langle u, v \rangle = \langle f - v A u, v \rangle, \quad \forall v \in V.$$

Since A is linear and continuous from V into V' and  $u \in L^2(V)$ , the function Au belongs to  $L^2(V')$ ; hence  $f-vAu \in L^2(V')$  and (1.33) and Lemma 1.1 show that

(1.34) 
$$u' \in L^2(0,T;V')$$

and that u is a.e. equal to an absolutely continuous function from [0,T] into V'. Any function satisfying (1.30) and (1.31) is, after modification on a set of measure zero, a continuous function from [0,T] into V', and therefore the condition (1.32) makes sense.

Let us suppose again that f is given in  $L^2(V')$  as in (1.28). If u satisfies (1.30) and (1.31), then as observed before u satisfies (1.34) and (1.33). According to Lemma 1.1 the equality (1.33) is itself equivalent to

(1.35) u' + vAu = f.

Conversely if u satisfies (1.30), (1.34), and (1.35), then u clearly satisfies  $(1.31), \forall v \in V$ .

## An alternate formulation of the weak problem is the following:

Given f and 
$$u_0$$
 satisfying (1.28)-(1.29), to find u satisfying

- (1.36)  $u \in L^{2}(0,T;V), u' \in L^{2}(0,T;V'),$
- (1.37) u' + vAu = f, on (0,T),
- (1.38)  $u(0) = u_0$ .

Any solution of (1.36) - (1.38) is a solution of (1.30) - (1.32) and conversely Concerning the existence and uniqueness of solution of these problems, we will prove the following result.

Theorem 1.1.

For given f and u<sub>0</sub> which satisfy (1.28) and (1.29), there exists a unique function u which satisfies (1.36) - (1.38). Moreover

(1.39)  $u \in \mathcal{C}([0,T];H).$ 

The proof of the existence is given in Section 1.3, that of the uniqueness and of (1.39) in Section 1.4.

1.3 Proof of the Existence in Theorem 1.1.

We use the Faedo-Galerkin method. Since V is separable there exists a sequence  $w_1, \dots, w_m, \dots$ , which is free and total in V. For each m we define an approximate solution  $u_m$  of (1.37) or (1.31) as follows:

(1.40) 
$$u_{m} = \sum_{i=1}^{m} g_{im}(t)w_{i},$$

and

(1.41)  $(u'_{m},w_{j}) + v((u_{m},w_{j})) = \langle f,w_{j} \rangle, \quad j = 1, \cdots, m,$ 

(1.42)  $u_m(0)=u_{om}$ ,

where  $u_{0m}$  is, for example, the orthogonal projection in H of  $u_0$  on the space spanned by  $w_1, \dots, w_m$  (1).

The functions  $g_{im}$ ,  $1 \le i \le m$ , are scalar functions defined on [0,T], and (1.41) is a linear differential system for these functions; indeed we have

(1.43) 
$$\sum_{i=1}^{m} (w_i, w_j) g'_{im}(t) + v \sum_{i=1}^{m} ((w_i, w_j)) g_{im}(t) = \langle f(t), w_j \rangle; \quad j = 1, \cdots, m$$

since the elements  $w_1, \dots, w_m$ , are linearly independent, it is well-known that the matrix with elements  $(w_i, w_j)$   $(1 \le i, j \le m)$  is non-singular; hence by inverting this matrix we reduce (1.43) to a linear system with constant coefficients

(1) 
$$u_{0m}$$
 can be any element of the space spanned by  $w_1, \dots, w_m$  such that  $u_{0m} \rightarrow u_0$  in the norm of H, as  $m \rightarrow \infty$ .

(1.44) 
$$g'_{im}(t) + \sum_{j=1}^{m} \alpha_{ij} g_{jm}(t) = \sum_{j=1}^{m} \beta_{ij} \langle f(t), w_{j} \rangle, \quad 1 \leq i \leq m,$$

where  $\alpha_{ij}, \beta_{ij} \in \mathbb{R}$ .

The condition (1.42) is equivalent to m equations

(1.45) 
$$g_{im}(0) = \text{the } i^{\underline{th}} \text{ component of } u_{om}$$

The linear differential system (1.44) together with the initial conditions (1.45) defines uniquely the g<sub>im</sub> on the whole interval [0,T].

Since the scalar functions  $t \mapsto \langle f(t), w \rangle_j$  are square integrable, so are the functions  $g_{im}$  and therefore, for each m

(1.46) 
$$u_{m} \in L^{2}(0,T;V), \quad u'_{m} \in L^{2}(0,T;V).$$

We will obtain a priori estimates independent of m for the functions  $u_m$  and then pass to the limit.

## A Priori Estimates.

We multiply equation (1.41) by  $g_{jm}(t)$  and add these equations for  $j = 1, \dots, m$ . We get

$$(u'_{m}(t),u_{m}(t)) + v \|u_{m}(t)\|^{2} = \langle f(t),u_{m}(t) \rangle$$

But, because of (1.46),

$$2(u'_{m}(t), u_{m}(t)) = \frac{d}{dt} |u_{m}(t)|^{2},$$

and this gives

(1.47) 
$$\frac{d}{dt} \|u_{m}(t)\|^{2} + 2\nu \|u_{m}(t)\|^{2} = 2 \langle f(t), u_{m}(t) \rangle$$

The right-hand side of (1.47) is majorized by

$$2\|f(t)\|_{V}, \|u_{m}(t)\| \leq v\|u_{m}(t)\|^{2} + \frac{1}{v}\|f(t)\|^{2}_{V},$$

Therefore

(1.48) 
$$\frac{\mathrm{d}}{\mathrm{d}t} \|u_{\mathrm{m}}(t)\|^{2} + \sqrt{\|u_{\mathrm{m}}(t)\|^{2}} \leq \frac{1}{\nu} \|f(t)\|_{\mathrm{V}}^{2}.$$

Integrating (1.48) from 0 to s, s > 0, we obtain in particular

(1.49) 
$$|u_{m}(s)|^{2} \leq |u_{0m}|^{2} + \frac{1}{\nu} \int_{0}^{s} ||f(t)||_{V}^{2} dt \leq |u_{0}|^{2} + \frac{1}{\nu} \int_{0}^{T} ||f(t)||_{V}^{2} dt$$

Hence:

(1.50) 
$$\sup_{s \in [0,T]} |u_{m}(s)|^{2} \leq |u_{0}|^{2} + \frac{1}{\nu} \int_{0}^{T} ||f(t)||_{V}^{2} dt.$$

The right-hand side of (1.50) is finite and independent of m; therefore (1.51) The sequence  $u_{m}$  remains in a bounded set of  $L^{\infty}(0,T;H)$ . We then integrate (1.48) from 0 to T and get

(1.52) 
$$|u_{m}(T)|^{2} + \nu \int_{0}^{T} ||u_{m}(t)||^{2} dt \leq |u_{0m}|^{2} + \frac{1}{\nu} \int_{0}^{T} ||f(t)||_{V}^{2}, dt$$
  
 $\leq |u_{0}|^{2} + \frac{1}{\nu} \int_{0}^{T} ||f(t)||_{V}^{2}, dt.$ 

This shows that

(1.53) The sequence 
$$u_{m}$$
 remains in a bounded set of  $L^{2}(0,T;V)$ .

# Passage to the Limit.

(1.54) 
$$u_{m}$$
, converges to u, for the weak-star topology of L (0,T;H);  
(1.54) means that for each  $v \in L^{1}(0,T;H)$ ,

ω

(1.55) 
$$\int_0^T (u_m, (t) - u(t), v(t)) dt \longrightarrow 0, \quad m' \longrightarrow \infty.$$

By (1.53) the subsequence  $u_m$ , belongs to a bounded set of  $L^2(0,T;V)$ ; therefore another passage to a subsequence shows the existence of some  $u_{\star}$  in  $L^2(0,T;V)$ and some subsequence (still denoted  $u_m$ ,) such that

(1.56)  $u_m$ , converges to  $u_*$ , for the weak topology of  $L^2(0,T;V)$ .

The convergence (1.56) means

$$\int_0^T \langle u_m'(t) - u_k(t), v(t) \rangle dt \longrightarrow 0, \quad \forall v \in L^2(0,T;V').$$

In particular, by (1.6),

(1.57) 
$$\int_0^T (u_m, (t), v(t)) dt \longrightarrow \int_0^T (u_k(t), v(t)) dt,$$

for each v in  $L^2(0,T;H)$ . By comparison with (1.55) we see that

(1.58) 
$$\int_{0}^{T} (u(t)-u_{*}(t),v(t))dt = 0,$$

for each v in  $L^2(0,T;H)$ ; hence

(1.59) 
$$u = u_* \in L^2(0,T;V) \cap L^{\infty}(0,T;H).$$

In order to pass to the limit in equations (1.41) and (1.42), let us consider scalar functions  $\Psi$  continuously differentiable on [0,T] and such that

(1.60) 
$$\psi(T) = 0$$

For such a function  $\psi$  we multiply (1.41) by  $\psi(t)$ , integrate in t and integrate by parts:

$$\int_{0}^{T} (u'_{m}(t), w_{j}) \psi(t) dt = - \int_{0}^{T} (u'_{m}(t) \psi'(t), w_{j}) dt - (u_{m}(0), w_{j}) \psi(0).$$

Hence we find,

(1.61) 
$$- \int_{0}^{T} (u_{m}(t), \psi'(t)w_{j}) dt + \nu \int_{0}^{T} ((u_{m}(t), \psi(t)w_{j})) dt$$
$$= (u_{0m}, w_{j})\psi(0) + \int_{0}^{T} \langle f(t), w_{j} \rangle dt.$$

The passage to the limit for  $m = m' \longrightarrow \infty$  in the integrals on the left-hand side is easy using (1.54), (1.57), and (1.59); we observe also that

(1.62) 
$$u_{0m} \rightarrow u_0$$
, in H, strongly.

Hence we find in the limit

(1.63) 
$$- \int_{0}^{T} (u(t), \psi'(t)w_{j}) dt + v \int_{0}^{T} ((u(t), \psi(t)w_{j})) dt$$
$$= (u_{0}, w_{j}) \psi(0) + \int_{0}^{T} \langle f(t), w_{j} \rangle \psi(t) dt.$$

The equality (1.63) which holds for each j, allows us to write by a linearity argument that

(1.64) 
$$-\int_{0}^{T} (u(t),v)\psi'(t)dt + v \int_{0}^{T} ((u(t),v))\psi(t)dt$$
$$= (u_{0},v)\psi(0) + \int_{0}^{T} \langle f(t),v \rangle \psi(t)dt,$$

for each v which is a finite linear combination of the  $w_j$ . Since each term of (1.64) depends linearly and continuously on v, for the norm of V, the equality (1.64) is still valid, by continuity, for each v in V.

Now, writing in particular (1.64) with  $\psi = \phi \in \mathscr{F}((0,T))$ , we find the following equality which is valid in the distribution sense on (0,T):

(1.65) 
$$\frac{d}{dt}(u,v) + v((u,v)) = \langle f,v \rangle, \quad \forall v \in V;$$

this is exactly (1.31). As proved before the statement of the Theorem 1.1, this equality and (1.59) imply that u' belongs to  $L^2(0,T;V')$  and

(1.66) 
$$u' + vAu = f.$$

Finally, it remains to check that  $u(0) = u_0$  (the continuity of u is proved in Section 1.4). For this, we multiply (1.65) by  $\psi(t)$ , (the same  $\psi$  as before), integrate in t, and integrate by parts:

$$\int_0^T \frac{d}{dt} (u(t), v) \psi(t) dt = - \int_0^T (u(t), v) \psi'(t) dt + (u(0), v) \psi(0).$$

We get

$$(1.67) - \int_{0}^{T} (u(t), v) \psi'(t) dt + v \int_{0}^{T} ((u(t), v)) \psi(t) dt$$
$$= (u(0), v) \psi(0) + \int_{0}^{T} \langle f(t), v \rangle \psi(t) dt.$$

By comparison with (1.64), we see that

$$(u_0-u(0),v)\psi(0) = 0,$$

for each  $v \in V$ , and for each function  $\psi$  of the type considered. We can choose such that  $\psi(0) \neq 0$ , and therefore

$$(u(0)-u_n,v) = 0, \quad \forall v \in V.$$

This equality implies that

$$u(0) = u_0$$

and achieves the proof of the existence.

1.4 Proof of the Continuity and Uniqueness.

This proof is based on the following lemma which is a particular case of a general theorem of interpolation of Lions-Magenes [1]:

Lemma 1.2.

Let V,H,V' be three Hilbert spaces, each space included in the following one as in (1.5), V' being the dual of V. If a function u belongs to  $L^2(0,T;V)$ and its derivative u' belongs to  $L^2(0,T;V')$ , then u is almost everywhere equal to a function continuous from [0,T] into H and we have the following equality, which holds in the scalar distribution sense on (0,T)

(1.68) 
$$\frac{d}{dt} |u|^2 = 2 \langle u', u \rangle$$

The equality (1.68) is meaningful since the functions

$$t \mapsto |u(t)|^2$$
,  $t \mapsto \langle u'(t), u(t) \rangle$ 

are both integrable on [0,T].

An alternate elementary proof of the lemma is given below.

If we assume this lemma, (1.39) becomes obvious and it only remains to check the uniqueness. Let us assume that u and v are two solutions of (1.36) -(1.38) and let w = u-v. Then w belongs to the same spaces as u and v, and

$$(1.69) w' + vAw = 0, w(0) = 0$$

Taking the scalar product of the first equality (1.69) with w(t), we find

$$\langle w'(t), w(t) \rangle + v w(t) \|^2 = 0$$
 a.e.

Using then (1.68) with u replaced by w, we obtain

$$\frac{d}{dt} |w(t)|^{2} + 2v ||w(t)||^{2} = 0$$
$$|w(t)|^{2} \le |w(0)|^{2} = 0, \quad t \in [0,T],$$

and hence u(t) = v(t) for each t.

Proof of Lemma 1.2.

The elementary proof of Lemma 1.2 which was announced before, is now given in the two following lemmas.

Lemma 1.3. Under the assumptions of Lemma 1.2, the equality (1.68) is satisfied. Proof.

By regularizing the function  $\tilde{u}$  from  $\mathbb{R}$  into V, which is equal to u on [0,T] and to 0 outside this interval, we easily obtain a sequence of functions  $u_m$  such that

(1.70)  $\forall m, u_m$  is infinitely differentiable from [0,T] into V, (1.71) as  $m \rightarrow \infty$ ,

$$u_{m} \longrightarrow u \quad \text{in} \quad L^{2}_{\text{loc}}(0,T;V)$$
$$u'_{m} \longrightarrow u' \quad \text{in} \quad L^{2}_{\text{loc}}(0,T;V').$$

Because of (1.6) and (1.70), the equality (1.68) for  $u_m$  is obvious:

(1.72) 
$$\frac{d}{dt} |u_m(t)|^2 = 2(u'_m(t), u_m(t)) = 2 \langle u'_m(t), u_m(t) \rangle, \quad \forall m.$$

As  $m \rightarrow \infty$ , it follows from (1.71) that

$$|u_{m}|^{2} \longrightarrow |u|^{2} \text{ in } L^{1}_{\text{loc}}(0,T)$$
  
$$\langle u'_{m}, u_{m}^{} \rangle \longrightarrow \langle u', u \rangle \text{ in } L^{1}_{\text{loc}}(0,T).$$

These convergences also hold in the distribution sense; therefore we are allowed to pass to the limit in (1.72) in the distribution sense; in the limit we find precisely (1.68).

Since the function

$$t \mapsto \langle u'(t), u(t) \rangle$$

is integrable on [0,T], the equality (1.68) shows us that the function u of Lemma 1.3 satisfies

(1.73) 
$$u \in L^{\infty}(0,T;H)$$
.

In the particular case of the function u satisfying (1.36) - (1.38), this was proved directly in Section 1.3.

According to Lemma 1.1, u is continuous from [0,T] into V'. Therefore, with this and (1.73), the following Lemma 1.4 shows us that u is weakly continuous from [0,T] into H, i.e.,

(1.74)  $\forall v \in H$ , the function  $t \mapsto (u(t), v)$  is continuous.

Admitting temporarily this point we can achieve the proof of Lemma 1.2. We must prove that for each  $t_0 \in [0,T]$ ,

(1.75) 
$$|u(t) - u(t_0)|^2 \longrightarrow 0$$
, as  $t \longrightarrow t_0$ .

Expanding this term, we find

 $|u(t)|^{2} + |u(t_{0})|^{2} - 2(u(t), u(t_{0})).$ When  $t \rightarrow t_{0}$ ,  $|u(t)|^{2} \rightarrow |u(t_{0})|^{2}$  since by (1.68),  $|u(t)|^{2} = |u(t_{0})|^{2} + 2 \int_{t_{0}}^{t} \langle u'(s), u(s) \rangle ds;$ 

because of (1.74)

$$(u(t), u(t_0)) \longrightarrow |u(t_0)|^2,$$

and (1.75) is proved.

The proof of Lemma 1.2 is achieved as soon as we prove the next lemma. This lemma is stated in a slightly more general form.

Lemma 1.4. Let X and Y be two Banach spaces, such that (1.76)  $X \subset Y$ 

with a continuous injection.

If a function  $\phi$  belongs to  $L^{\infty}(0,T;X)$  and is weakly continuous with values in Y, then  $\phi$  is weakly continuous with values in X.

#### Proof.

Perhaps replacing Y by the closure of X in Y, we can suppose that X is dense in Y. Hence the dense continuous imbedding of X into Y gives by duality a dense continuous imbedding of Y' (dual of Y) into X' (dual of X):

(1.77)  $Y' \subset X'$ .

By assumption, for each  $\eta \in Y'$ ,

(1.78)  $\langle \phi(t), \eta \rangle \longrightarrow \langle \phi(t_0), \eta \rangle$ , as  $t \longrightarrow t_0$ ,  $\forall t_0$ ,

and we must prove that (1.78) is also true for each  $\eta \in X'$ .

We first prove that  $\phi(t) \in X$  for each  $\tau$  and that
(1.79) 
$$\|\phi(t)\|_{X} \leq \|\phi\|_{L^{\infty}(0,T;X)}, \forall t \in [0,T].$$

Indeed, by regularizing the function  $\tilde{\phi}$  equal to  $\phi$  on [0,T] and to 0 outside this interval, we find a sequence of smooth functions  $\phi_m$  from [0,T] into X such that

$$\|\phi_{\mathbf{m}}(t)\|_{\mathbf{X}} \leq \|\phi\|_{\mathbf{L}^{\infty}(\mathbf{X})}, \quad \forall \mathbf{m}, \forall t \in [0,T]$$

and

$$\langle \phi_{\mathfrak{m}}(t), \mathfrak{n} \rangle \longrightarrow \langle \phi(t), \mathfrak{n} \rangle, \quad \mathfrak{m} \longrightarrow, \quad \forall \mathfrak{n} \in \mathfrak{Y}'.$$

Since

$$|\langle \phi_{\mathbf{m}}(t), \eta \rangle| \leq ||\phi||_{L^{\infty}(\mathbf{X})} ||\eta||_{\mathbf{X}}, \forall \mathbf{m}, \forall t,$$

we obtain in the limit

$$|\langle \phi(t), \eta \rangle| \leq ||\phi||_{L^{\infty}(X)} ||\eta||_{X'}, \forall t \in [0,T], \forall \eta \in Y'.$$

This inequality shows that  $\phi(t) \in X$  and that (1.79) holds.

Finally let us prove (1.78) for  $\eta$  in X'. Since Y' is dense in X', there exists, for each  $\varepsilon > 0$ , some  $\eta_{\varepsilon} \in Y'$  such that

$$\|n-n_{\varepsilon}\|_{\mathbf{X}}, \leq \varepsilon.$$

We then write

$$\langle \phi(t) - \phi(t_0), \eta \rangle = \langle \phi(t) - \phi(t_0), \eta - \eta_{\varepsilon} \rangle + \langle \phi(t) - \phi(t_0), \eta_{\varepsilon} \rangle$$

$$\left| \langle \phi(t) - \phi(t_0), \eta \rangle \right| \leq 2\varepsilon \|\phi\|_{L^{\infty}(X)} + \left| \langle \phi(t) - \phi(t_0), \eta_{\varepsilon} \rangle .$$

As  $t \rightarrow t_0$ , since  $\eta_e \in Y'$ , the continuity assumption implies that

$$\langle \phi(t) - \phi(t_n), \eta_c \rangle \longrightarrow 0$$

and hence

$$\overline{\lim} | \langle \phi(t) - \phi(t_0), \eta \rangle | \leq 2\varepsilon \| \phi \|_{L^{\infty}(X)}$$

Since  $\varepsilon > 0$  is arbitrarily small, the preceding upper limit is zero, and (1.78) is proved.

#### 1.5 Miscellaneous Remarks.

We give in this section some remarks and complements to Theorem 1.1.

An Extension of Theorem 1.1.

Theorem 1.1 is a particular case of an abstract theorem, involving abstract spaces V,H, and an abstract operator A; see Lions-Magenes [1].

If instead of (1.28) we assume that

(1.80) 
$$f = f_1 + f_2, f_1 \in L^2(0,T;V'), f_2 \in L^1(0,T;H),$$

then all the conclusions of Theorem 1.1 are true with only one modification:

(1.81) 
$$u' \in L^2(0,T;V') + L^1(0,T;H).$$

In the proof of the existence, we write after (1.47):

$$\frac{d}{dt} |u_{m}(t)|^{2} + 2v ||u_{m}(t)||^{2} \leq 2 ||f_{1}(t)||_{V}, ||u_{m}(t)|| + 2 |f_{2}(t)| ||u_{m}(t)||$$

$$(1.82) \leq v ||u_{m}(t)||^{2} + \frac{1}{V} ||f_{1}(t)||_{V}^{2}, + ||f_{2}(t)|| \{1 + |u_{m}(t)||^{2}\}.$$

Hence, in particular

(1.83) 
$$\frac{d}{dt} \{1 + |u_m(t)|^2\} \leq \frac{1}{\nu} ||f_1(t)||_V^2, + |f_2(t)|\{1 + |u_m(t)|^2\}.$$

Multiplying this by  $\exp\{-\int_0^t |f_2(\sigma)| d\sigma\}$ , we obtain

$$\frac{d}{dt} \{ \exp(-\int_0^t |f_2(\sigma)| d\sigma) \cdot (1 + |u_m(t)|^2) \} \le \frac{1}{\nu} ||f_1(t)||_V^2, \exp(-\int_0^t |f_2(\sigma)| d\sigma).$$

Integrating this inequality from 0 to s, s > 0, we obtain a majoration similar to (1.50) which implies (1.51). Then integrating (1.82) from 0 to T we obtain (1.53).

The proof of the existence is then conducted exactly as in Section 1.3. Concerning the derivative u', we have

(1.84) 
$$u' = -vAu + f_1 + f_2 \in L^2(0,T;V') + L^1(0,T;H).$$

It is easy to see that Lemma 1.2 is also valid if

(1.85) 
$$\begin{cases} u \in L^{2}(0,T;V) \cap L^{\infty}(0,T;H), \\ u' \in L^{2}(0,T;V') + L^{1}(0,T;H). \end{cases}$$

After noticing that, we can prove the uniqueness and the continuity of u,  $u \in C([0,T];H)$ , exactly as in Section 1.4.

#### The Case $\Omega$ Unbounded.

For the evolution problem, when  $\Omega$  is unbounded, <u>the introduction of the space</u> Y <u>considered in the stationary unbounded case</u> (Chapter I, Section 2.3) is no longer necessary. All the previous results hold if  $\Omega$  is unbounded and V is equipped with the norm (1.9). Let us assume, in the most general case, that f satisfies (1.80). We have exactly the same results as in Theorem 1.1, if f satisfies (1.28), and the same results as in Remark 1.1 if f satisfies (1.80). The only difference is that we must replace (1.82) by

(1.86)  
$$\frac{d}{dt} |u_{m}(t)|^{2} + 2v||u_{m}(t)||^{2} \leq 2||f_{1}(t)||_{V}, [|u_{m}(t)]| + 2|f_{2}(t)||u_{m}(t)||$$
$$\leq v||u_{m}(t)||^{2} + v|u_{m}(t)|^{2} + \frac{1}{v}||f_{1}(t)||_{V}^{2}, + |f_{2}(t)||(1 + |u_{m}(t)|^{2})$$

Hence

$$(1.87) \quad \frac{\mathrm{d}}{\mathrm{dt}} \left\{ 1 + \left| u_{\mathrm{m}}(t) \right|^{2} \right\} \leq \left( \left| f_{2}(t) \right| + v \right) \left\{ 1 + \left| u_{\mathrm{m}}(t) \right|^{2} \right\} + \frac{1}{v} \left\| f_{1}(t) \right\|_{\mathrm{V}}^{2},$$

This inequality is then treated exactly as (1.83), to obtain (1.51). After that, integrating (1.86) from 0 to T, we obtain

$$\int_0^T \|u_m(t)\|^2 dt \leq \text{Const.}$$

This majoration, together with (1.51), gives (1.53).

The proofs of the existence, the uniqueness, and the continuity are then exactly the same as before.

#### Interpretation of the Variational Problem.

We wish to make precise in what sense the function u defined by Theorem 1.1 is solution of the initial problem (1.23) - (1.26).

Proposition 1.1.

Under the assumptions of Theorem 1.1, there exists a distribution p on  $Q = \Omega \times (0,T)$ , such that the function u defined by Theorem 1.1 and p satisfy (1.23) in the distribution sense in Q; (1.24) is satisfied in the distribution sense too and (1.26) is satisfied in the sense

(1.88) 
$$u(t) \xrightarrow{} u_n \neq \text{ in } L^2(\Omega), \text{ as } t \xrightarrow{} 0.$$

Proof.

The equality (1.24) is an easy consequence of  $u \in L^2(0,T;V)$ ; (1.26) and (1.88) follow also immediately from Theorem 1.1; (1.25) is satisfied in a sense which

depends on the trace theorems available for  $\Omega$  since u is in  $L^2(0,T;H^1_0(\Omega))$ .

To introduce the pressure, let us set

(1.89) 
$$U(t) = \int_0^t u(s) ds, \quad F(t) = \int_0^t f(s) ds.$$

It is clear that, at least,

UEC([0,T];V), FEC([0,T];V').

Integrating (1.31), we see that

$$(1.90) \quad (u(t)-u_0,v) + v((U(t),v)) = \langle F(t),v \rangle, \quad \forall t \in [0,T], \quad \forall v \in V,$$

or

$$\langle u(t)-u_{0} - v \Delta U(t)-F(t), v \rangle = 0, \forall t \in [0,T], \forall v \in V.$$

By an application of Proposition I.1.3, we find, for each  $t \in [0,T]$ , the existence of some function P(t),

 $P(t) \in L^2(\Omega)$ ,

such that

(1.91) 
$$u(t)-u_0 - v\Delta U(t) + grad P(t) = F(t)$$

By Proposition I.1.2, the gradient operator is an isomorphism from  $L^2(\Omega)/\mathbb{R}$  into  $\mathbb{H}^{-1}(\Omega)$ . Observing that

(1.92) 
$$\operatorname{grad} P = F + v\Delta U - u + u_0,$$

we conclude that grad P belongs to  $\mathcal{C}([0,T];\mathbb{H}^{-1}(\Omega))$  as the right-hand side of (1.92) does; therefore

(1.93) 
$$\mathbf{P} \in \mathcal{C}([0,T]; \mathbf{L}^2(\Omega)).$$

This enables us to differentiate (1.91) in the t variable, in the distribution sense in  $Q = \Omega \times (0,T)$ ; setting

$$(1.94) p = \frac{\partial P}{\partial t} ,$$

we obtain precisely (1.23).

We do not have in general any information on p better than (1.93)-(1.94). In the next proposition, we will get more regularity on p after assuming more regularity on the data f and  $u_0$ .

#### Some Results of Regularity.

Assuming that the data  $\Omega$ , f, u<sub>0</sub>, are sufficiently smooth, we can obtain as much regularity as desired for u and p. We will only prove a simple result of this type:

Proposition 1.2.

Let us assume that  $\Omega$  is of class  $\mathcal{C}^2$ , that

(1.95)  $f \in L^2(0,T;H)$ 

and

(1.96)	u <sub>o</sub>	$\in$	V
--------	----------------	-------	---

Then	
(1.97)	$u \in L^2(0,T; \mathbb{H}^2(\Omega))$ ,
(1.98)	$u' \in L^2(0,T;H)$ , i.e., $u' \in L^2(Q)$ ,
(1.99)	$p \in L^2(0,T;H^1(\Omega)).$

Proof.

The first point is to obtain (1.98); this is proved by getting another a priori estimate for the approximate solution  $u_m$  constructed by the Galerkin method.

Using the notation of Section 1.3, we multiply (1.41) by  $g'_{jm}(t)$ , and add these equalities for  $j = 1, \dots, m$ ; this gives

$$|u'_{m}(t)|^{2} + v((u_{m}(t),u'_{m}(t))) = (f(t),u'_{m}(t))$$

or

(1.100) 
$$2|u'_{m}(t)|^{2} + v \frac{d}{dt} ||u_{m}(t)||^{2} = 2(f(t), u'_{m}(t)).$$

We then integrate (1.100) from 0 to T, and use the Schwarz inequality; we obtain:

$$2 \int_{0}^{T} |u'_{m}(t)|^{2} dt + v ||u_{m}(T)||^{2} = v ||u_{0m}||^{2} + 2 \int_{0}^{T} (f(t), u'_{m}(t)) dt$$

$$\leq v ||u_{0m}||^{2} + \int_{0}^{T} |f(t)|^{2} dt + \int_{0}^{T} |u'_{m}(t)|^{2} dt,$$
(1.101)
$$\int_{0}^{T} |u'_{m}(t)|^{2} dt \leq v ||u_{0m}||^{2} + \int_{0}^{T} |f(t)|^{2} dt.$$

The basis w used for the Galerkin method can be chosen so that  $w_j \in V$  for each j and we can take

 $u_{0m} = the projection in V of <math>u_0$  on

the space spanned by 
$$w_1, \dots, w_m$$
.

Therefore

(1.102)  $u_{0m} \rightarrow u_{0}$  in V strongly as  $m \rightarrow \infty$ ,

and

 $(1.103) \| u_{0m} \| \le \| u_0 \|.$ 

With these choices of the  $w_j$  and of  $u_{0m}$ , (1.101) shows that

(1.104) the 
$$u'_{m}$$
 belong to a bounded set of  $L^{2}(0,T;H)$ ,

and (1.98) is proved.

We then come back to the equalities (1.23), (1.24), (1.25) and we apply the regularity theorem of the stationary case (Theorem I.2.4): For almost every t in [0,T],

(1.105) 
$$\begin{cases} -\nu\Delta u(t) + \text{grad } p(t) = f - u' \in L^2(0,T; \mathbb{L}^2(\Omega)) \\ \text{div } u(t) = 0 \quad \text{in } \Omega \\ u(t) = 0 \quad \text{on } \partial\Omega \end{cases}$$

so that u(t) belongs to  $\mathbb{H}^2(\Omega)$  and p(t) belongs to  $\mathbb{H}^1(\Omega)$ . Moreover, since the mapping

$$f(t)-u'(t) \longrightarrow {u(t),p(t)}$$

is linear continuous from  $\mathbb{L}^2(\Omega)$  into  $\mathbb{H}^2(\Omega) \times \mathbb{H}^1(\Omega)$ , due to I(2.40), and since  $f-u^* \in L^2(0,T;\mathbb{L}^2(\Omega))$ ,

it is clear that (1.97) and (1.99) are satisfied,

#### \$2. COMPACTNESS THEOREMS.

The compactness theorems presented in Chapter II are not sufficient for the nonlinear evolution problems. Our goal now is to prove some compactness theorems which are appropriate for the nonlinear problems of the remainder of this chapter.

After a preliminary result given in Section 2.1, we prove in Section 2.2 a compactness theorem in the frame of Banach spaces. In Section 2.3 we prove two other compactness theorems in the frame of Hilbert spaces; one of them involves fractional derivatives in time of the functions.

Some related discrete form of these theorems will be studied later on.

2.1 A Preliminary Result.

The proof of the compactness theorems of the next two sections will be based on the following lemma.

Lemma 2.1.

Let  $X_0$ , X, and  $X_1$  be three Banach spaces such that

$$(2.1) X_0 \subset X \subset X_1$$

the injection of X into X, being continuous and:

(2.2) <u>the injection of  $X_0$  into X is compact</u>.

Then for each  $\eta > 0$ , there exists some constant  $c_{\eta}$  depending on  $\eta$  (and on the spaces  $X_0, X, X_1$ ) such that:

(2.3)  $\|v\|_{X} \leq \eta \|v\|_{X_{0}} + c_{\eta} \|v\|_{X_{1}}, \quad \forall v \in X_{0}.$ 

#### Proof.

The proof is by contradiciton. Saying that (2.3) is not true amounts to saying that there exists some  $\eta > 0$  such that for each c in  $\mathbb{R}$ ,

$$\|v\|_{X} \ge \eta \|v\|_{X_{0}} + c \|v\|_{X_{1}},$$

for at least one v. Taking c = m, we obtain a sequence of elements  $v_m$  satisfying

$$|\mathbf{v}_{\mathbf{m}}|_{\mathbf{X}} \geq \eta |\mathbf{v}_{\mathbf{m}}|_{\mathbf{X}_{0}} + \mathbf{m} |\mathbf{v}_{\mathbf{m}}|_{\mathbf{X}_{1}}, \quad \forall \mathbf{m}.$$

We consider then the normalized sequence

$$w_{m} = \frac{v_{m}}{\|v_{m}\|_{X_{0}}},$$

which satisfies

(2.4) 
$$\|w_{m}\|_{X} \ge \eta + m \|w_{m}\|_{X_{1}}, \quad \forall m.$$

Since  $\|w_m\|_{X_0} = 1$ , the sequence  $w_m$  is bounded in X and (2.4) shows that

(2.5) 
$$\|w_{m}\|_{X_{1}} \longrightarrow 0$$
, as  $m \longrightarrow \infty$ .

In addition, by (2.2) the sequence  $w_m$  is relatively compact in X; hence we can extract from  $w_m$  a subsequence  $w_\mu$  strongly convergent in X; because of (2.5) the limit of  $w_\mu$  must be 0, but this contradicts (2.4) as:

 $\|\mathbf{w}_{\mu}\|_{\mathbf{X}} \geq \eta > 0, \quad \forall \mu.$ 

2.2 <u>A Compactness Theorem in Banach Space</u>.

Let  $X_0$ , X,  $X_1$ , be three Banach spaces such that

 $(2.6) X_0 \subset X \subset X_1,$ 

where the injections are continuous and:

(2.7)  $X_{i}$  is reflexive, i = 0, 1,

(2.8) the injection 
$$X_n \longmapsto X$$
 is compact.

Let T > 0 be a fixed finite number, and let  $\alpha_0$ ,  $\alpha_1$ , be two finite numbers such that  $\alpha_i > 1$ , i = 0, 1.

We consider the space

(2.9)  

$$\begin{aligned}
\psi &= \psi(0,T;\alpha_0,\alpha_1;X_0,X_1) \\
\end{aligned}$$
(2.10)  

$$\begin{aligned}
\psi &= \{v \in L^{\alpha_0}(0,T;X_0), v' = \frac{dv}{dt} \in L^{\alpha_1}(0,T;X_1)\}
\end{aligned}$$

The space  $\,\mathcal{U}$  is provided with the norm

(2.11) 
$$\|v\|_{\mathcal{Y}} = \|v\|_{L^{\alpha_0}(0,T;X_0)} + \|v'\|_{L^{\alpha_1}(0,T;X_1)}$$

which makes it a Banach space. It is evident that

$$\mathcal{Y} \subset L^{\alpha_0}(0,T;X),$$

with a continuous injection. Actually we will prove that this injection is compact.

Theorem 2.1.

Under the assumptions (2.6) to (2.9) the injection of  $\mathcal{Y}$  into  $L^{\alpha}(0,T;X)$  is compact.

Proof.

(i) Let  $u_m$  be some sequence which is bounded in  $\mathcal{Y}$ . We must prove that this sequence contains a subsequence  $u_\mu$  strongly convergent in  $L^{\alpha_0}(0,T;X)$ . Since the spaces  $X_i$  are reflexive and  $1 < \alpha_i < +\infty$ , the spaces  $L^{\alpha_i}(0,T;X_i)$ , i = 0,1, are likewise reflexive, and hence  $\mathcal{Y}$  is reflexive. Therefore, there exists some u in  $\mathcal{Y}$  and some subsequence  $u_\mu$  with

(2.12) 
$$u_{\mu} \longrightarrow u$$
 in  $\mathcal{Y}$  weakly, as  $\mu \longrightarrow \infty$ ,

which means

(2.13) 
$$\begin{cases} u_{\mu} \longrightarrow u \text{ in } L^{\alpha_{0}}(0,T;X_{0}) \text{ weakly} \\ u'_{\mu} \longrightarrow u' \text{ in } L^{\alpha_{1}}(0,T;X_{1}) \text{ weakly.} \end{cases}$$

It suffices to prove that

(2.14) 
$$v_{\mu} = u_{\mu} - u$$
 converges to 0 in  $L^{\alpha_0}(0,T;X)$  strongly.

(ii) The theorem will be proved if we show that

(2.15) 
$$v_{\mu} \longrightarrow 0$$
 in  $L^{\alpha_0}(0,T;X_1)$  strongly.

In fact, due to Lemma 2.1, we have

$$\|v_{\mu}\|_{L^{\alpha_{0}}(0,T;X)} \leq \eta \|v_{\mu}\|_{L^{\alpha_{0}}(0,T;X_{0})} + c_{\eta} \|v_{\mu}\|_{L^{\alpha_{0}}(0,T;X_{1})}$$

and since the sequence  $v_{\mu}$  is bounded in  $\mathcal{Y}$ :

(2.16) 
$$\|v_{\mu}\|_{L^{\alpha_0}(0,T;X)} \leq cn + c_{\eta}\|v_{\mu}\|_{L^{\alpha_0}(0,T;X_1)}$$

Passing to the limit in (2.16) we see by (2.15) that

(2.17) 
$$\frac{\lim_{\mu \to \infty} \|v_{\mu}\|_{\alpha_0}}{\mu \to \infty} L^{\alpha_0}(0,T;X) \leq cn;$$

since  $\eta > 0$  is arbitrarily small in Lemma 2.1, this upper limit is 0 and thus (2.14) is proved.

(iii) To prove (2.15) we observe that

(2.18) 
$$\mathcal{Y} \subset \mathcal{C}([0,T];X_1),$$

with a continuous injection; the inclusion (2.18) results from Lemma 1.1, and the continuity of the injection is very easy to check.

We infer from this, the majoration

(2.19) 
$$\|v_{\mu}(t)\|_{X_1} \leq c, \quad \forall t \in [0,T], \forall \mu.$$

According to Lebesgue's Theorem, (2.15) is now proved if we show that, for almost every t in [0,T],

(2.20) 
$$v_{\mu}(t) \rightarrow 0$$
 in  $X_1$  strongly, as  $\mu \rightarrow \infty$ .

We will prove (2.20) for t = 0; the proof would be similar for any other t. We write

$$v_{\mu}(0) = v_{\mu}(t) - \int_{0}^{t} v_{\mu}^{*}(\tau) d\tau$$

and by integration

$$\mathbf{v}_{\mu}(0) = \frac{1}{s} \left\{ \int_{0}^{s} \mathbf{v}_{\mu}(t) dt - \int_{0}^{s} \int_{0}^{t} \mathbf{v}_{\mu}^{\tau}(\tau) d\tau dt \right\}$$

#### Hence

(2.21) 
$$v_{\mu}(0) = a_{\mu} + b_{\mu}$$

with

(2.22) 
$$a_{\mu} = \frac{1}{s} \int_{0}^{s} v_{\mu}(t) dt, \quad b_{\mu} = -\frac{1}{s} \int_{0}^{s} (s-t) v'_{\mu}(t) dt.$$

For a given  $\varepsilon > 0$ , we choose s so that

$$\|b_{\mu}\|_{X_{1}} \leq \int_{0}^{s} \|v'_{\mu}(t)\|_{X_{1}} dt \leq \frac{\varepsilon}{2}$$

Then, for this fixed s, we observe that, as  $\mu \rightarrow \infty$ ,  $a_{\mu} \rightarrow 0$  in  $X_0$  weakly and thus in  $X_1$  strongly; for  $\mu$  sufficiently large

$$\|a_{\mu}\|_{X_{1}} \leq \frac{\varepsilon}{2},$$

and (2.20), for t = 0, follows.

# 2.3 <u>A Compactness Theorem Involving Fractional Derivatives.</u>

The next compactness theorem is in the frame of Hilbert spaces and is based on the notion of fractional derivatives of a function.

Let us assume that  $X_0$ , X,  $X_1$ , are Hilbert spaces with

 $(2.23) x_0 \subset x \subset x_1,$ 

the injections being continuous and

(2.24) the injection of  $X_0$  into X is compact.

218

If v is a function from  $\mathbb R$  into  $X_1$ , we denote by  $\hat{v}$  its Fourier transform

(2.25) 
$$\hat{\mathbf{v}}(\tau) = \int_{-\infty}^{+\infty} e^{-2i\pi t\tau} \mathbf{v}(t) dt.$$

The derivative in t of order Y of v is the inverse Fourier transform of (2iπτ)<sup>Ŷ</sup>ŷ or

(2.26) 
$$\widehat{D_{t}^{\gamma}v(\tau)} = (2i\pi\tau)^{\gamma} \hat{v}(\tau).$$
 (1)

For given  $\gamma > 0$  <sup>(2)</sup>, we define the space

(2.27) 
$$\mathcal{H}^{\gamma}(\mathbb{R}; X_0, X_1) = \{ v \in L^2(\mathbb{R}; X_0), D_t^{\gamma} v \in L^2(\mathbb{R}; X_1) \}.$$

This is a Hilbert space for the norm,

$$\| v \|_{\mathcal{A}^{\gamma}(\mathbb{R}, X_{0}, X_{1})} = \{ \| v \|_{L^{2}(\mathbb{R}; X_{0})}^{2} + \| |\tau|^{\gamma} \hat{v} \|_{L^{2}(\mathbb{R}; X_{1})}^{2} \}^{\frac{1}{2}}$$

We associate to any set KCR, the subspace  $\mathscr{H}^\gamma_K$  of  $\mathscr{H}^\gamma$  defined as the set of functions u in  $\mathscr{H}$  with support contained in K:

(2.28) 
$$\mathcal{H}_{K}^{\gamma}(\mathbb{R}; X_{0}, X_{1}) = \{ u \in \mathcal{H}^{\gamma}(\mathbb{R}; X_{0}, X_{1}), \text{ support } u \subset K \}.$$

The compactness theorem can now be stated:

Theorem 2.2.

Let us assume that  $X_0, X, X_1$  are Hilbert spaces which satisfy (2.23) and (2.24). Then for any bounded set K and any  $\gamma > 0$ , the injection of  $\mathcal{H}_{K}^{\gamma}(\mathbb{R}; X_{0}, X_{1})$ into L<sup>2</sup>(R;X) is compact.

Proof.

(i) Let  $\gamma$  and K be fixed, and let  $u_m$  be a bounded sequence in  $\mathcal{H}_K^{\gamma}(\mathbb{R};X_0,X_1)$ . We must show that  $u_m$  contains a subsequence strongly convergent in  $L^2(\mathbb{R};X)$ .

Since  $\mathcal{H}^{\gamma}(\mathbb{R}; X_0, X_1)$  is a Hilbert space, the sequence  $u_m$  contains a subsequence  $u_{_{11}}$  weakly convergent in this space to some element u. It is clear that u must belong to  $\mathcal{H}_{K}^{\gamma}$  too; therefore, setting

$$v_{\mu} = u_{\mu} - u_{\mu}$$

the sequence  $v_{\mu}$  appears as a bounded sequence of  $\mathscr{H}_{K}^{\gamma}(\mathbb{R}; X_{0}, X_{1})$ , which converges weakly to 0 in  $\mathcal{H}^{\gamma}$ ; this means

(2.29) 
$$v_{\mu} \longrightarrow 0$$
 in  $L^{2}(\mathbb{R}; X_{0})$  weakly

(1) The definition (2.26) is consistent with the usual definition for  $\gamma$  an integer. <sup>(2)</sup> In the applications,  $0 < \gamma \leq 1$  in general.

(2.30) 
$$|\tau|^{\gamma} \hat{v}_{\mu} \longrightarrow 0 \text{ in } L^2(\mathbb{R}; X_1) \text{ weakly.}$$

The theorem is proved if we show that  $\,u_{\mu}\,$  converges strongly to  $\,u\,$  in L^2(R;X), that is to say

(2.31) 
$$v_{\mu} \longrightarrow 0$$
 in  $L^{2}(\mathbb{R}; X)$  strongly.

(ii) The second point of the proof is to show that (2.31) is proved if we prove that

(2.32) 
$$v_{\mu} \longrightarrow 0$$
 in  $L^{2}(\mathbb{R}; X_{1})$  strongly.

Due to Lemma 2.1,

(2.33) 
$$\|v_{\mu}\|_{L^{2}(\mathbb{R};X)} \leq \eta \|v_{\mu}\|_{L^{2}(\mathbb{R};X_{0})} + c_{\eta}\|v_{\mu}\|_{L^{2}(\mathbb{R};X_{1})}$$

and since  $v_{U}$  is bounded in  $L^{2}(\mathbb{R}; X_{0})$ ,

(2.34) 
$$\|v_{\mu}\|_{L^{2}(\mathbb{R};X)} \leq c\eta + c_{\eta}\|v_{\mu}\|_{L^{2}(\mathbb{R};X_{1})}$$

If we assume (2.32), then, letting  $\mu \rightarrow \infty$  in (2.34), we obtain

$$\frac{\lim_{\mu \to \infty} \|\mathbf{v}_{\mu}\|_{L^{2}(\mathbb{R};X)} \leq c\eta.$$

Since  $\eta$  is arbitrarily small in Lemma 2.1, this upper limit must be 0 and (2.31) follows.

(iii,) Finally let us prove (2.32). According to the Parseval theorem,

(2.35) 
$$I_{\mu} = \int_{-\infty}^{+\infty} \|v_{\mu}(t)\|_{X_{1}}^{2} dt = \int_{-\infty}^{+\infty} \|\hat{v}_{\mu}(\tau)\|_{X_{1}}^{2} d\tau,$$

where  $\hat{v}_{_{\scriptstyle U}}$  denotes the Fourier transform of  $v_{_{\scriptstyle U}}.$  We must show that

(2.36) 
$$I_{\mu} \longrightarrow 0 \text{ as } \mu \longrightarrow \infty.$$

For this, we write

$$I_{\mu} = \int_{|\tau| \le M} \|\hat{v}_{\mu}(\tau)\|_{X_{1}}^{2} d\tau + \int_{|\tau| > M} (1 + |\tau|^{2\gamma}) \|\hat{v}_{\mu}(\tau)\|_{X_{1}}^{2} \cdot \frac{d\tau}{(1 + |\tau|^{2\gamma})}$$
$$\leq \frac{c}{1 + M^{2\gamma}} + \int_{|\tau| \le M} \|\hat{v}_{\mu}(\tau)\|_{X_{1}}^{2} d\tau,$$

since  $v_{\mu}$  is bounded in  $\mathcal{Y}$ .

For  $\varepsilon > 0$  given, we choose M so that

$$\frac{c}{1+M^{2\gamma}} \leq \frac{\varepsilon}{2}$$

Hence

$$I_{\mu} \leq \int_{|\tau| \leq M} \|\hat{v}_{\mu}(\tau)\|_{X_{1}}^{2} d\tau + \frac{\varepsilon}{2},$$

and (2.36) is proved if we show that, for this fixed M,

(2.37) 
$$J_{\mu} = \int_{|\tau| \leq M} \|\hat{v}_{\mu}(\tau)\|_{X_{1}}^{2} d\tau \rightarrow 0, \text{ as } \mu \rightarrow \infty.$$

This is proved via the Lebesgue theorem. If  $\chi$  denotes the characteristic function of K, then  $v_{_{\rm U}}\chi$  =  $v_{_{\rm U}}$  and

$$\hat{v}_{\mu}(\tau) = \int_{-\infty}^{+\infty} e^{-2i\pi t\tau} \chi(t) v_{\mu}(t) dt.$$

Thus

(2.38) 
$$\| \hat{v}_{\mu}(\tau) \|_{X_{1}} \leq \| v_{\mu} \|_{L^{2}(\mathbb{R}; X_{1})} \| e^{-2i\pi t\tau} \chi \|_{L^{2}(\mathbb{R})} ,$$
$$\| \hat{v}_{\mu}(\tau) \|_{X_{1}} \leq \text{Const.}$$

On the other hand for each  $\sigma$  in  $X_{n}$ , and each fixed  $\tau$ ,

$$((\hat{v}_{\mu}(\tau),\sigma))_{X_{0}} = \int_{-\infty}^{+\infty} ((v_{\mu}(t),e^{-2i\pi t\tau}\chi(t)\sigma))_{X_{0}}dt,$$

and this goes to 0, as  $\mu \rightarrow \infty$ , because of (2.29). The sequence  $\hat{v}_{\mu}(\tau)$  converges to 0 weakly in  $X_0$  and therefore strongly in X and  $X_1$ .

With this last remark and (2.38), the Lebesgue theorem implies (2.37).

Using the methods of the last theorem, we can prove another compactness theorem similar to Theorem 2.1. Nevertheless, this theorem is not contained in and does not contain Theorem 2.2.

Theorem 2.3.

Under the hypotheses (2.23) and (2.24), the injection of  $\mathcal{Y}(0,T;2,1;X_0,X_1)$  (1) into  $L^2(0,T;X)$  is compact.

Proof.

Let  $u_m$  be a bounded sequence in this space  $\mathcal{Y}$ ; we denote by  $\widetilde{u}_m$  the function defined on the whole line  $\mathbb{R}$ , which is equal to  $u_m$  on [0,T], and

<sup>(1)</sup> For the definition of this space see (2.9)-(2.10).

Because of Lemma 1.1, each function  $u_m$  is, after modification on a set of measure 0, continuous from [0,T] into  $X_1$ , and more precisely the injection of  $\mathcal{Y}$  into  $\mathcal{C}([0,T];X_1)$  is continuous.

It is classical that since  $\tilde{u}_m$  has two discontinuities, at 0 and T, the distribution derivative of  $\tilde{u}_m$  is given by

(2.39) 
$$\frac{\mathrm{d}}{\mathrm{d}t} \tilde{u}_{\mathrm{m}} = \tilde{g}_{\mathrm{m}} + u_{\mathrm{m}}(0)\delta_{0} - u_{\mathrm{m}}(\mathrm{T})\delta_{\mathrm{T}},$$

where  $\delta_{\mu}$  and  $\delta_{\pi}$  are the Dirac distributions at 0 and T, and

(2.40) 
$$g_m = u'_m = \text{the derivative of } u_m \text{ on } [0,T].$$

After a Fourier transformation, (2.39) gives

(2.41) 
$$2i\pi \tau \hat{u}_{m}(\tau) = \hat{g}_{m}(\tau) + u_{m}(0) - u_{m}(\tau) \exp(-2i\pi \tau \tau),$$

where  $\hat{g}_{m}$  and  $\hat{u}_{m}$  denote the Fourier transform of  $\tilde{g}_{m}$  and  $\tilde{u}_{m}$  respectively.

Since the functions  $g_m$  remain bounded in  $L^1(0,T;X_1)$ , the functions  $\tilde{g}_m$  remain bounded in  $L^1(\mathbb{R};X_1)$  and the functions  $\hat{g}_m$  are bounded in  $L^\infty(\mathbb{R};X_1)$ :

(2.42) 
$$\|\hat{g}_{m}(\tau)\|_{X_{1}} \leq \text{Const.}, \forall m, \forall \tau \in \mathbb{R}.$$

We already pointed out that the injection of  $\mathcal{V}$  into  $\mathcal{C}([0,T];X_1)$  is continuous; thus

$$\| u_{\mathbf{m}}(0) \|_{\mathbf{X}_{1}} \leq \text{Const.}, \| u_{\mathbf{m}}(\mathbf{T}) \|_{\mathbf{X}_{1}} \leq \text{Const.},$$

and (2.41) shows us that

(2.43) 
$$|\tau|^2 \|\hat{u}_m(\tau)\|_{X_1}^2 \leq c, \forall m, \forall \tau \in \mathbb{R}.$$
  
For  $\gamma$  fixed,  $\gamma < \frac{1}{2}$ , we observe that

$$|\tau|^{2\gamma} \leq c_0(\gamma) \frac{1+\tau^2}{1+|\tau|^{2}(1-\gamma)}, \quad \forall \tau \in \mathbb{R}.$$

Therefore

 $\gamma > 0.$ 

$$\int_{-\infty}^{+\infty} |\tau|^{2\gamma} \|\hat{u}_{m}(\tau)\|_{X_{1}}^{2} d\tau \leq c_{0}(\gamma) \int_{-\infty}^{+\infty} \frac{1+\tau^{2}}{1+|\tau|^{2}(1-\gamma)} \|\hat{u}_{m}(\tau)\|_{X_{1}}^{2} d\tau$$
$$\leq c_{1} \int_{-\infty}^{+\infty} \frac{d\tau}{1+|\tau|^{2}(1-\gamma)} + c_{0}(\gamma) \int_{-\infty}^{+\infty} \|\hat{u}_{m}(\tau)\|_{X_{1}}^{2} d\tau$$

by (2.43).

Since  $\gamma < \frac{1}{2}$ , the integral

$$\int_{-\infty}^{+\infty} \frac{d\tau}{1+|\tau|^2(1-\gamma)}$$

is convergent; on the other hand, by the Parseval equality, we see that

$$\int_{-\infty}^{+\infty} \|\dot{u}_{m}(\tau)\|_{X_{1}}^{2} d\tau = \int_{0}^{T} \|u_{m}(t)\|_{X_{1}}^{2} dt,$$

and these integrals are bounded.

We conclude that

(2.44) 
$$\int_{-\infty}^{+\infty} |\tau|^{2\gamma} ||\hat{u}_{m}(\tau)||_{X_{1}}^{2} d\tau \leq c_{2},$$

where  $c_2$  depends on  $\gamma$ .

It is clear now that the sequence  $u_m$  is bounded in  $\mathbb{W}^{\gamma}(\mathbb{R};X_0,X_1)$  and the proof is achieved.

Remark 2.1.

Assuming only that  $X_1$  is a Hilbert space,  $X_0$ , X being Banach spaces, it can be proved in a similar way that the injection of

$$\mathcal{Y}(0,T;\alpha_0,1;X_0,X_1),$$

into  $L^{\alpha_0}(0,T;X)$  is compact,  $\alpha_0$  any finite number,  $\alpha_0 > 1$ .

#### §3. EXISTENCE AND UNIQUENESS THEOREMS (n < 4).

This section is concerned with existence and uniqueness theorems for weak solutions of the full Navier-Stokes equations  $(n \le 4)$ . In Section 3.1 we give the variational formulation of these equations, following J. Leray, and we state an existence theorem for such solutions when the dimension  $n \le 4$ . The proof of this theorem, due to J.L. Lions, is given in Section 3.2. It is based on the construction of an approximate solution by the Galerkin method; then a passage to the limit using, in particular, an a priori estimate on a fractional derivative in time of the approximate solution, and a compactness theorem contained in Section 2. An alternate proof based on a semi-discretization in time and valid in all dimensions is discussed in Section 4.

In Section 3.3 we develop the uniqueness theorem of weak solutions (n = 2). In the three-dimensional case there is a gap between the class of functions where existence is known, and the smaller classes where uniqueness is proved; an example of such a uniqueness theorem is developed in Section 3.4 (n = 3). In Section 3.5 we show in the two dimensional case the existence of more regular solutions, assuming more regularity on the data; a similar result holds in the three dimensional case for local solutions, that is to say solutions which are defined on some "small" interval of time, assuming that the data is sufficiently small.

3.1 An Existence Theorem in  $\mathbb{R}^n$  (n < 4).

The notations are the usual ones, in particular those recalled at the beginning of Section 1.1;  $\Omega$  is an open lipschitzian set which we suppose bounded for simplicity; the unbounded case is discussed in Remarks 3.1 and 3.2.

We recall <sup>(1)</sup> that since the dimension is less than or equal to 4, one can define on  $\mathbb{H}^1_0(\Omega)$ , and in particular on V, a trilinear continuous form b by setting

(3.1) 
$$b(u,v,w) = \sum_{i,j=1}^{n} \int_{\Omega} u_{i}(D_{i}v_{j})w_{j}dx.$$

If  $u \in V$ , then

(3.2) 
$$b(u,v,v) = 0, \forall v \in \mathbb{H}_0^1(\Omega).$$

For u, v in V, we denote by B(u, v) the element of V' defined by

<sup>(1)</sup> Cf. Section 1.1, Chapter II.

$$(3.3) \qquad \langle B(u,v),w\rangle = b(u,v,w), \quad \forall w \in V,$$

and we set

$$(3.4) \qquad B(u) = B(u,u) \in V', \quad \forall u \in V.$$

In its classical formulation, the initial boundary value problem of the full Navier-Stokes equations is the following:

To find a vector function

u: 
$$\Omega \times [0,T] \longrightarrow \mathbb{R}^n$$

and a scalar function

$$p: \Omega \times [0,T] \longrightarrow \mathbb{R},$$

such that

(3.5)  $\frac{\partial u}{\partial t} - v\Delta u + \sum_{i=1}^{n} u_i D_i u + \text{grad } p = f \text{ in } Q = \Omega \times (0,T),$ (3.6) div u = 0 in Q,(3.7)  $u = 0 \text{ on } \partial\Omega \times (0,T),$ (3.8)  $u(x,0) = u_0(x), \text{ in } \Omega.$ 

As before, the functions f and  $u_0$  are given, respectively defined on  $\Omega \times [0,T]$  and  $\Omega$ .

Let us assume that u and p are classical solutions of (3.5) - (3.8), say  $u \in \mathbb{C}^2(\overline{Q})$ ,  $p \in \mathcal{C}^1(\overline{Q})$ . Obviously  $u \in L^2(0,T;V)$ , and if v is an element of  $\mathcal{V}$ ; one can check easily that

(3.9) 
$$\frac{d}{dt}(u,v) + v((u,v)) + b(u,u,v) = \langle f,v \rangle$$

By continuity, equation (3.9) will hold for each  $v \in V$ .

This suggests the following weak formulation of the problem (3.5) - (3.9) (cf. J. Leray [1], [2], [3]):

Problem 3.1.						
For	f	and	u	given with		
(3.10)					$f \in L^2(0,T;V')$	
(3.1	1)				u₀∈H,	

to find u satisfying

(3.12) 
$$u \in L^2(0,T;V)$$

and

(3.13) 
$$\frac{u}{dt}(u,v) + v((u,v)) + b(u,u,v) = \langle f,v \rangle, \quad \forall v \in V$$

(3.14)  $u(0) = u_{0}$ .

If u merely belongs to  $L^2(0,T;V)$ , the condition (3.14) need not make sense. But if u belongs to  $L^2(0,T;V)$  and satisfies (3.13), then we will show as in the linear case (using Lemma 1.1) that u is almost everywhere equal to some continuous function, so that (3.14) is meaningful.

Before showing this, we recall that we are considering the case  $n \leq 4$ ; we will modify slightly the preceding formulation in higher dimensions (see Section 4.1).

<u>Lemma 3.1</u>.

We assume that the dimension of the space is  $n \le 4$  and that u belongs to  $L^2(0,T;V)$ .

Then the fucntion Bu defined by

$$\langle Bu(t), v \rangle = b(u(t), u(t), v), \quad \forall v \in V, \text{ a.e. in } t \in [0,T],$$

belongs to L<sup>1</sup>(0,T;V').

Proof.

For almost all t, Bu(t) is an element of V', and the measurability of the function

 $t \in [0,T] \mapsto Bu(t) \in V'$ 

is easy to check. Moreover, since b is trilinear continuous on V,

 $(3.15) Bw |_{V}, \leq c ||w||^2, \quad \forall w \in V,$ 

so that

$$\int_{0}^{T} \|Bu(t)\|_{V} dt \leq c \int_{0}^{T} \|u(t)\|^{2} dt < +\infty,$$

and the lemma is proved.

Now if u satisfies (3.12) - (3.13), then according to (1.6), (1.8), and the above lemma, one can write (3.13) as

$$\frac{d}{dt} < u, v > = < f - v A u - B u, v >, \quad \forall v \in v.$$

Since Au belongs to  $L^2(0,T;V')$ , as in the linear case, the function f-vAu-Bu belongs to  $L^1(0,T;V')$ . Lemma 1.1 implies then that

(3.16) 
$$\begin{cases} u' \in L^{1}(0,T;V') \\ u' = f - \forall Au - Bu, \end{cases}$$

and that u is almost everywhere equal to a function continuous from [0,T] into V'. This remark gives a sense to (3.14).

An alternate formulation of the problem (3.12)- (3.14) is the following one:

 Problem 3.2.

 Given f and u, satisfying (3.10) - (3.11), to find u satisfying

 (3.17)
  $u \in L^2(0,T;V)$ ,  $u' \in L^1(0,T;V')$ ,

 (3.18)
 u' + vAu + Bu = f on (0,T),

 (3.19)
  $u(0) = u_0$ .

We showed that any solution of Problem 3.1 is a solution of Problem 3.2; the converse is very easily checked and these problems are thus equivalent.

The existence of solutions of these problems is ensured by the following theorem (cf. J.L. Lions [1]).

#### Theorem 3.1.

<u>The dimension is  $n \le 4$ . Let there be given f and  $u_0$  which satisfy (3.10) - (3.11). Then, there exists at least one function u which satisfies (3.17) - (3.19). <u>Moreover</u>,</u>

(3.20)

$$u \in L^{\circ}(0,T;H)$$

and u is weakly continuous from [0,T] into H<sup>(1)</sup>.

The proof of the existence of a u satisfying (3.20) is developed in Section 3.2; the continuity result is a direct consequence of (3.20), the continuity of u in V', and Lemma 1.4.

Remark 3.1.

(i) Theorem 3.1 also holds if we assume that

 $f = f_1 + f_2; f_1 \in L^2(0,T;V'), f_2 \in L^1(0,T;H).$ 

For the corresponding modifications of the proof of the theorem, the reader is referred to Section 1.5.

(ii) Theorem 3.1 is also valid if  $\Omega$  is unbounded; for the details, cf. Remark 3.2.

(1) i.e.,  $\forall v \in H$ ,  $t \mapsto (u(t), v)$  is a continuous scalar function.

### 3.2 Proof of Theorem 3.1.

(i) We apply the Galerkin procedure as in the linear case. Since V is separable and  $\mathcal{V}$  is dense in V, there exists a sequence  $w_1, \cdots, w_m, \cdots$  of elements of  $\mathcal{V}$ ; which is free and total in V <sup>(1)</sup>. For each m we define an approximate solution  $u_m$  of (3.13) as follows:

(3.21) 
$$u_{m} = \sum_{i=1}^{m} g_{im}(t) w_{i}$$

and

(3.22) 
$$(u'_{m}(t), w_{j}) + v((u_{m}(t), w_{j})) + b(u_{m}(t), u_{m}(t), w_{j})$$
  
=  $\langle f(t), w_{j} \rangle$ ,  $t \in [0, T]$ ,  $j = 1, \cdots, m$ ,  
(3.23)  $u_{m}(0) = u_{0m}$ ,

where  $u_{0m}$  is the orthogonal projection in H of  $u_0$  onto the space spanned by  $w_1, \cdots, w_m$  (2).

The equations (3.22) form a nonlinear differential system for the functions g<sub>1m</sub>,...,g<sub>mm</sub>;

(3.24) 
$$\sum_{i=1}^{m} (w_{i}, w_{j}) g'_{im}(t) + v \sum_{i=1}^{m} ((w_{i}, w_{j})) g_{im}(t) + \sum_{i, l=1}^{m} b(w_{i}, w_{l}, w_{j}) g_{im}(t) g_{lm}(t) = \langle f(t), w_{j} \rangle$$

Inverting the nonsingular matrix with elements  $(w_i, w_j)$ ,  $1 \le i, j \le m$ , we can write the differential equations in the usual form

(3.25) 
$$g'_{im}(t) + \sum_{j=1}^{m} \alpha_{ij}g_{jm}(t) + \sum_{j,k=1}^{m} \alpha_{ijk}g_{jm}(t)g_{km}(t) = \sum_{j=1}^{m} \beta_{ij}f(t),w_{j}$$

where  $\alpha_{ij}, \alpha_{ijk}, \beta_{ij} \in \mathbb{R}$ .

The condition (3.23) is equivalent to the m scalar initial conditions  $g_{im}(0) = the i^{th} component of u_{om}$ . (3.26)

(1) The w, are chosen in  $\mathcal{V}$  for simplicity. With some technical modifications, we could take the  $w_{j}$  in V. (2) We could take for  $u_{0m}$  any element of that space such that

$$u_{0m} \rightarrow u_{0}$$
, in H, as  $m \rightarrow \infty$ .

The nonlinear differential system (3.25) with the initial condition (3.26) has a maximal solution defined on some interval  $[0,t_m)$ . If  $t_m < T$ , then  $|u_m(t)|$  must tend to  $+\infty$  as  $t \longrightarrow t_m$ ; the a priori estimates we will prove later show that this does not happen and therefore  $t_m = T$ .

(ii) The first a priori estimates are obtained as in the linear case. We multiply (3.22) by  $g_{jm}(t)$  and add these equations for  $j = 1, \dots, m$ . Taking (3.2) into account, we get

(3.27) 
$$(u'_{m}(t), u_{m}(t)) + v \|u_{m}(t)\|^{2} = \langle f(t), u_{m}(t) \rangle.$$

Then we write

$$\frac{d}{dt} \|u_{m}(t)\|^{2} + 2v\|u_{m}(t)\|^{2} = 2\langle f(t), u_{m}(t) \rangle \leq 2\|f(t)\|_{V}, \|u_{m}(t)\|$$
$$\leq v\|u_{m}(t)\|^{2} + \frac{1}{v}\|f(t)\|_{V}^{2},$$

so that

(3.28) 
$$\frac{d}{dt} \|u_{m}(t)\|^{2} + v\|u_{m}(t)\|^{2} \leq \frac{1}{v} \|f(t)\|_{V}^{2},$$

Integrating (3.28) from 0 to s we obtain, in particular,

$$|u_{m}(s)|^{2} \leq |u_{0m}|^{2} + \frac{1}{\nu} \int_{0}^{s} ||f(t)||_{V}^{2} dt \leq |u_{0}|^{2} + \frac{1}{\nu} \int_{0}^{T} ||f(t)||_{V}^{2} dt.$$

Hence

(3.29) 
$$\sup_{s \in [0,T]} |u_{m}(s)|^{2} \leq |u_{0}|^{2} + \frac{1}{\nu} \int_{0}^{T} ||f(t)||^{2} v^{1} dt$$

which implies that

(3.30) The sequence  $u_m$  remains in a bounded set of  $L^{\infty}(0,T;H)$ . We then integrate (3.28) from 0 to T to get

$$\begin{aligned} |u_{m}(T)|^{2} + v \int_{0}^{T} ||u_{m}(t)||^{2} dt &\leq |u_{0m}|^{2} + \frac{1}{v} \int_{0}^{T} ||f(t)||_{V}^{2} dt \\ &\leq |u_{0}|^{2} + \frac{1}{v} \int_{0}^{T} ||f(t)||_{V}^{2} dt. \end{aligned}$$

This estimate enables us to say that

# (3.31) The sequence $u_m$ remains in a bounded set of $L^2(0,T;V)$ .

(iii) Let  $\tilde{u}_m$  denote the function from IR into V, which is equal to  $u_m$  on [0,T] and to 0 on the complement of this interval. The Fourier

transform of  $\tilde{u}_m$  is denoted by  $\hat{u}_m$ .

In addition to the previous inequalities, which are similar to the estimates in the linear case, we want to show that

(3.32) 
$$\int_{-\infty}^{+\infty} |\tau|^{2\gamma} |\hat{u}_{m}(\tau)|^{2} dt \leq \text{Const., for some } \gamma > 0.$$

Along with (3.31), this will imply that

 $\tilde{u}_{m}$  belongs to a bounded set of  $\mathcal{H}(\mathbb{R}; \mathbb{V}, \mathbb{H})$ (3.33)

and will enable us to apply the compactness result of Theorem 2.2.

In order to prove (3.32) we observe that (3.22) can be written (1)

$$(3.34) \quad \frac{\mathrm{d}}{\mathrm{dt}} (\tilde{\mathbf{u}}_{\mathrm{m}}, \mathbf{w}_{\mathrm{j}}) = \langle \tilde{\mathbf{f}}_{\mathrm{m}}, \mathbf{w}_{\mathrm{j}} \rangle + (\mathbf{u}_{0\mathrm{m}}, \mathbf{w}_{\mathrm{j}}) \delta_{0} - (\mathbf{u}_{\mathrm{m}}(\mathrm{T}), \mathbf{w}_{\mathrm{j}}) \delta_{\mathrm{T}}, \quad \mathrm{j} = 1, \cdots, \mathrm{m}$$

where  $\boldsymbol{\delta}_{0}^{}$  ,  $\boldsymbol{\delta}_{T}^{}$  are the Dirac distributions at 0 and T and

(3.35) 
$$\begin{cases} f_m = f - vAu_m - Bu_m \\ \tilde{f}_m = f_m \text{ on } [0,T], 0 \text{ outside this interval.} \end{cases}$$
By the Fourier transform, (3.34) gives

(3.36) 
$$2i\pi\tau(\hat{u}_{m},w_{j}) = \langle \hat{f}_{m},w_{j} \rangle + (u_{0m},w_{j}) - (u_{m}(T),w_{j}) \exp(-2i\pi T\tau),$$

 $\hat{u}_{m}(\text{resp.}\hat{f}_{m})$  denoting the Fourier transform of  $\tilde{u}_{m}$  (resp.  $\tilde{f}_{m}$ ). We multiply (3.35) by  $\hat{g}_{jm}(\tau)$  (= Fourier transform of  $\tilde{g}_{jm}$ ) and add the resulting equations for  $j = 1, \dots, m$ ; we get:

(3.37) 
$$2i\pi\tau |\hat{u}_{m}(\tau)|^{2} = \langle \hat{f}_{m}(\tau), \hat{u}_{m}(\tau) \rangle + (u_{0m}, \hat{u}_{m}(\tau)) - (u_{m}(T), \hat{u}_{m}(\tau)) \exp(-2i\pi T\tau).$$

Because of inequality (3.15),

$$\int_{0}^{T} \|f_{m}(t)\|_{V} dt \leq \int_{0}^{T} (\|f(t)\|_{V} + v\|u_{m}(t)\| + c\|u_{m}(t)\|^{2}) dt,$$

and this remains bounded according to (3.31). Therefore

$$\sup_{\tau \in \mathbb{R}} \|\hat{f}_{m}(\tau)\|_{V} \leq \text{Const., } \forall m.$$

Due to (3.29),

(1) Compare this with the proof of Theorem 2.3.

 $|u_{m}(0)| \leq Const., |u_{m}(T)| \leq Const.,$ 

and we deduce from (3.37) that

$$|\tau| |\hat{u}_{m}(\tau)|^{2} \leq c_{2} ||\hat{u}_{m}(\tau)|| + c_{3} |\hat{u}_{m}(\tau)|$$

or

(3.38) 
$$|\tau| |\hat{u}_{m}(\tau)|^{2} \leq c_{4} \|\hat{u}_{m}(\tau)\|.$$

For  $\gamma$  fixed,  $\gamma < \frac{1}{4}$ , we observe that

$$|\tau|^{2\gamma} \leq c_{5}(\gamma) \frac{1+|\tau|}{1+|\tau|^{1-2\gamma}}, \quad \forall \tau \in \mathbb{R}.$$

Thus

$$\int_{-\infty}^{+\infty} |\tau|^{2\gamma} |\hat{u}_{m}(\tau)|^{2} d\tau \leq c_{5}(\gamma) \int_{-\infty}^{+\infty} \frac{1+|\tau|}{1+|\tau|^{1-2\gamma}} |\hat{u}_{m}(\tau)|^{2} d\tau$$

$$\leq (by (3.38))$$

$$\leq c_{6} \int_{-\infty}^{+\infty} \frac{\|\hat{u}_{m}(\tau)\| d\tau}{1+|\tau|^{1-2\gamma}} + c_{7} \int_{-\infty}^{+\infty} \|\hat{u}_{m}(\tau)\|^{2} d\tau$$

Because of the Parseval equality and (3.31), the last integral is bounded as  $m \longrightarrow \infty$ ; thus (3.32) will be proved if we show that

$$\int_{-\infty}^{+\infty} \frac{\|\hat{u}_{m}(\tau)\|}{1+|\tau|^{1-2\gamma}} d\tau \leq \text{Const.}$$

By the Schwarz inequality and the Parseval equality we can estimate these integrals by

$$\left(\int_{-\infty}^{+\infty} \frac{d\tau}{(1+|\tau|)^{1-2\gamma}}\right)^{\frac{1}{2}} \left(\int_{0}^{T} \|u_{m}(t)\|^{2} dt\right)^{\frac{1}{2}},$$

which is finite since  $\gamma < \frac{1}{4}$ , and bounded as  $m \rightarrow \infty$  by (3.31). The proof of (3.32) and (3.33) is achieved.

(iv) The estimates (3.30) and (3.31) enable us to assert the existence of an element  $u \in L^2(0,T;V) \cap L^{\infty}(0,T;H)$  and a subsequence  $u_m$ , such that

(3.40) 
$$\begin{cases} u_{m}, \longrightarrow u \text{ in } L^{2}(0,T;V) \text{ weakly, and in} \\ L^{\infty}(0,T;H) \text{ weak-star, as } m' \longrightarrow \infty. \end{cases}$$

Due to (3.33) and Theorem 2.2, we also have

(3.41)  $u_m \rightarrow u$  in  $L^2(0,T;H)$  strongly.

The convergence results (3.40) and (3.41) enable us to pass to the limit. We proceed essentially as in the linear case.

Let  $\psi$  be a continuously differentiable function on [0,T] with  $\psi(T) = 0$ . We multiply (3.22) by  $\psi(t)$ , integrate in t, and then integrate by parts. This leads to the equation

$$-\int_{0}^{T} (u_{m}(t), \psi'(t)w_{j})dt + v \int_{0}^{T} ((u_{m}(t), w_{j}\psi(t))) dt$$

$$(3.42) + \int_{0}^{T} b(u_{m}(t), u_{m}(t), w_{j}\psi(t))dt = (u_{0m}, w_{j})\psi(0) + \int_{0}^{T} \langle f(t), w_{j}\psi(t) \rangle dt.$$

Passing to the limit with the sequence m' is easy for the linear terms; for the nonlinear term we apply the next lemma, Lemma 3.2. In the limit we find that the equation

$$(3.43) - \int_{0}^{T} (u(t), v\psi'(t))dt + v \int_{0}^{T} ((u(t), v\psi(t)))dt + \int_{0}^{T} b(u(t), u(t), v\psi(t))dt = (u_{0}, v)\psi(0) + \int_{0}^{T} \langle f(t), v\psi(t) \rangle dt,$$

holds for  $v = w_1, w_2, \cdots$ ; by linearity this equation holds for v = any finite linear combination of the  $w_j$ , and by a continuity argument (3.43) is still true for any  $v \in V$ .

Now writing, in particular, (3.43) with  $\psi = \phi \in \mathcal{D}((0,T))$ , we see that u satisfies (3.13) in the distribution sense.

Finally, it remains to prove that u satisfies (3.14). For this we multiply (3.13) by  $\psi$ , and integrate. After integrating by parts the first term, we get

$$-\int_{0}^{T} (u(t),v\psi(t))dt + v \int_{0}^{T} (\dot{u}(t),v\psi(t)))dt$$
(3.44)
$$+\int_{0}^{T} b(u(t),u(t),v\psi(t))dt = (u(0),v)\psi(0) + \int_{0}^{T} \langle f(t),v\psi(t) \rangle dt.$$

By comparison with (3.43),

$$(u(0)-u_0,v)\psi(0) = 0.$$

We can choose  $\psi$  with  $\psi(0) = 1$ ; thus

$$(u(0)-u_n,v) = 0, \forall v \in V,$$

and (3.14) follows.

The proof of Theorem 3.1 will be complete once we prove the following lemma.

Lemma 3.2.

If  $u_{\mu}$  converges to  $u_{\mu}$  in L (0,T;V) weakly and L<sup>2</sup>(0,T;H) strongly, then for any vector function w with components in  $\mathcal{C}^{1}(\overline{Q})$ ,

(3.45) 
$$\int_0^T b(u_{\mu}(t), u_{\mu}(t), w(t)) dt \longrightarrow \int_0^T b(u(t), u(t), w(t)) dt.$$

Proof.

We write

$$\int_{0}^{T} b(u_{\mu}, u_{\mu}, w) dt = - \int_{0}^{T} b(u_{\mu}, w, u_{\mu}) dt = - \sum_{i,j=1}^{n} \int_{0}^{T} \int_{\Omega} (u_{\mu})_{i} (D_{i}w_{j}) (u_{\mu})_{j} dx dt.$$

These integrals converge to

$$-\sum_{i,j=1}^{n}\int_{0}^{T}\int_{\Omega}u_{i}(D_{i}w_{j})u_{j}dxdt = -\int_{0}^{T}b(u,w,u)dt = \int_{0}^{T}b(u,u,w)dt,$$

and the lemma is proved.

Remark 3.2.

When  $\Omega$  is unbounded, we prove (3.30) and (3.31) as we did in Section 1.5 for the linear case. Then (3.32) and (3.33) follow in the same way as before. The main difference lies in the fact that the injection of V into H is no longer compact.

Nevertheless we can extract a subsequence  $u_{m'}$  which satisfies (3.40). Then, for any ball  $\vartheta$  included in  $\Omega$ , the injection of  $H^1(\vartheta)$  into  $L^2(\vartheta)$  is compact and (3.33) shows that:

(3.46)  $u_{m}|_{\mathcal{O}}$  belongs to a bounded set of  $\mathscr{W}(\mathbb{R};\mathbb{H}^{1}(\mathcal{O}),\mathbb{L}^{2}(\mathcal{O})), \forall \mathcal{O}$ . Then Theorem 2.2 implies that

 $u_{m'} | \xrightarrow{\longrightarrow} u |$  in  $\mathbb{L}^{2}(\mathcal{O})$  strongly,  $\forall \mathcal{O}$ ,

which means

In particular, for a fixed j,

$$u_{m'} | \xrightarrow{\Omega'} u |$$
 in  $\mathbb{L}^{2}(\Omega')$  strongly,

where  $\Omega'$  denotes the support of  $w_j$ , and this suffices to pass to the limit in (3.42).

3.3 Regularity and Uniqueness (n = 2).

When the dimension of the space is n = 2, the solution of (3.17) - (3.19)whose existence is ensured by Theorem 3.1 satisfies some further regularity property and is actually unique.

The proof of these results is based on the following lemmas.

$$\frac{\text{Lemma } \underline{3.3}}{\text{If } n = 2, \quad \underline{\text{for any open set}} \quad \Omega}$$

$$(3.48) \qquad \left\| v \right\|_{L^{4}(\Omega)} \leq 2^{\frac{1}{4}} \left\| v \right\|_{L^{2}(\Omega)}^{\frac{1}{2}} \left\| \text{grad } v \right\|_{L^{2}(\Omega)}^{\frac{1}{2}}, \quad \forall v \in H_{0}^{1}(\Omega).$$

Proof.

It suffices to prove (3.48) for  $v \in \mathscr{D}(\Omega)$ . For such a v, we write

$$v^{2}(x) = 2 \int_{-\infty}^{x_{1}} v(\xi_{1}, x_{2}) D_{1} v(\xi_{1}, x_{2}) d\xi_{1},$$

and therefore

(3.49) 
$$v^2(x) \leq 2v_1(x_2),$$

where

(3.50) 
$$v_1(x_2) = \int_{-\infty}^{+\infty} |v(\xi_1, x_2)| |D_1 v(\xi_1, x_2)| d\xi_1$$

Similarly

(3.51) 
$$v^{2}(x) \leq 2v_{2}(x_{1}),$$

where

(3.52) 
$$v_2(x_1) = \int_{-\infty}^{+\infty} |v(x_1,\xi_2)| |D_2v(x_1,\xi_2)| d\xi_2$$

and thus

$$\begin{split} \int_{\mathbb{R}^{2}} v^{4}(x) dx &\leq 4 \int_{\mathbb{R}^{2}} v_{1}(x_{2}) v_{2}(x_{1}) dx \leq 4 \left( \int_{-\infty}^{+\infty} v_{1}(x_{2}) dx_{2} \right) \left( \int_{-\infty}^{+\infty} v_{2}(x_{1}) dx_{1} \right) \\ & \leq 4 \| v \|_{L^{2}(\mathbb{R}^{2})}^{2} \| D_{1} v \|_{L^{2}(\mathbb{R}^{2})}^{2} \| D_{2} v \|_{L^{2}(\mathbb{R}^{2})}^{2} \\ & \leq 2 \| v \|_{L^{2}(\mathbb{R}^{2})}^{2} \| \operatorname{grad} v \|_{L^{2}(\mathbb{R}^{2})}^{2} \end{split}$$

$$\frac{\text{Lemma } 3.4}{\text{If } n = 2},$$

(3.53) 
$$|b(u,v,w)| \leq 2^{\frac{1}{2}} |u|^{\frac{1}{2}} ||u|^{\frac{1}{2}} ||v|| ||w|^{\frac{1}{2}} ||w||^{\frac{1}{2}}, \forall u,v,w \in \mathbb{H}_{0}^{1}(\Omega).$$

If u belongs to  $L^2(0,T;V) \cap L^{\infty}(0,T;H)$ , then Bu belongs to  $L^2(0,T;V')$ and

(3.54) 
$$\|Bu\|_{L^{2}(0,T;V')} \leq 2^{\frac{1}{2}} \|u\|_{L^{\infty}(0,T;H)} \|u\|_{L^{2}(0,T;V)}$$

Proof.

By repeated application of the Schwarz and Hölder inequalities we find:

$$\begin{aligned} |b(u,v,w)| &\leq \sum_{i,j=1}^{2} \int_{\Omega} |u_{i}(D_{i}v_{j})w_{j}| dx \\ &\leq \sum_{i,j=1}^{2} ||u_{i}||_{L^{4}(\Omega)} ||D_{i}v_{j}||_{L^{2}(\Omega)} ||w_{j}||_{L^{4}(\Omega)} \\ &\leq \left(\sum_{i,j=1}^{2} ||D_{i}v_{j}||_{L^{2}(\Omega)}^{2} \left(\sum_{i=1}^{2} ||u_{i}||_{L^{4}(\Omega)}^{2} \left(\sum_{j=1}^{2} ||w_{j}||_{L^{4}(\Omega)}^{2} \right)^{\frac{1}{2}} \end{aligned}$$

Due to (3.48),

$$\sum_{i=1}^{2} \|u_{i}\|_{L^{4}(\Omega)}^{2} \leq 2^{\frac{1}{2}} \sum_{i=1}^{2} \left( \|u_{i}\|_{L^{2}(\Omega)} \|\text{grad } u_{i}\|_{L^{2}(\Omega)} \right) \\
\leq 2^{\frac{1}{2}} \|u\| \|u\|.$$

With a similar inequality for w, we finally get (3.53).

If u,v,w belong to V, the relation

$$b(u,v,w) = -b(u,w,v)$$

gives another estimate of b:

(3.55) 
$$|b(u,v,w)| \leq 2^{\frac{1}{2}} |u|^{\frac{1}{2}} ||u|^{\frac{1}{2}} ||v|^{\frac{1}{2}} ||v|^{\frac{1}{2}} ||w||, \quad \forall u,v,w \in V.$$

In particular,

(3.56) 
$$|b(u,u,v)| \leq 2^{\frac{1}{2}} |u| ||u|| ||v||, \forall u,v \in V,$$

and hence

(3.57) 
$$||Bu||_{V}, \leq 2^{\frac{1}{2}} |u| ||u||, \forall u \in V.$$

If now  $u \in L^2(0,T;V) \cap L^{\infty}(0,T;H)$ , Bu(t) belongs to V' for almost every t and the estimate

(3.58) 
$$\|Bu(t)\|_{V'} \leq 2^{\frac{1}{2}} \|u(t)\| \|u(t)\|$$

shows that Bu belongs to  $L^2(0,T;V')$  and implies (3.54).

We can now state and prove the main result (cf. J.L. Lions and G. Prodi [1]).

#### Theorem 3.2.

<u>In the two-dimensional case, the solution</u> u of Problems 3.1-3.2 given by <u>Theorem 3.1 is unique.</u> Moreover u is almost everywhere equal to a function <u>continuous from [0,T] into H and</u>

(3.59)  $u(t) \rightarrow u_0$ , in H, as  $t \rightarrow 0$ .

Proof.

(i) We first prove the result of regularity.

According to (3.18) and Lemma 3.4,

$$u' = f - vAu - Bu$$
,

and since each term in the right-hand side of this equation belongs to  $L^2(0,T;V')$ , u' also belongs to  $L^2(0,T;V')$ ; this remark improves (3.17):

(3.60)  $u' \in L^2(0,T;V').$ 

This improvement of (3.17) enables us to apply Lemma 1.2, which states exactly that u is almost everywhere equal to a function continuous from [0,T]into H. Thus

(3.61) 
$$u \in \mathcal{C}([0,T];H)$$

and (3.59) follows easily.

We recall also that Lemma 1.2 asserts that for any function u in  $L^2(0,T;V)$  which satisfies (3.60), the equation below holds:

(3.62) 
$$\frac{d}{dt} |u(t)|^2 = 2\langle u'(t), u(t) \rangle.$$

This result will be used in the proof of uniqueness which we will start now.

(ii) Let us assume that  $u_1$  and  $u_2$  are two solutions of (3.17) - (3.19), and let  $u = u_1 - u_2$ . As shown before,  $u_1, u_2$ , and thus  $u_1$ , satisfy (3.60).

The difference  $u = u_1 - u_2$  satisfies

(3.63) 
$$u' + vAu = -Bu_1 + Bu_2$$

$$(3.64)$$
  $u(0) = 0$ .

We take a.e. in t the scalar product of (3.63) with u(t) in the duality between V and V'. Using (3.62), we get

$$(3.65) \quad \frac{d}{dt} \|u(t)\|^2 + 2\nu \|u(t)\|^2 = 2b(u_2(t), u_2(t), u(t)) - 2b(u_1(t), u_1(t), u(t)).$$

Because of (3.2) the right-hand side of this equality is equal to

 $-2b(u(t),u_{2}(t),u(t)).$ 

With (3.53) we can majorize this expression by

$$2^{\frac{3}{2}} |u(t)| ||u(t)|| ||u_{2}(t)|| \leq 2\nu ||u(t)||^{2} + \frac{1}{\nu} ||u(t)||^{2} ||u_{2}(t)||^{2}.$$

Putting this in (3.65) we find

$$\frac{d}{dt} |u(t)|^2 \leq \frac{1}{v} |u(t)|^2 ||u_2(t)||^2.$$

Since the function  $t \mapsto \|u_2(t)\|^2$  is integrable, this shows that

$$\frac{d}{dt} \{ \exp(-\frac{1}{\nu} \int_0^t ||u_2(s)||^2 ds) \cdot |u(t)|^2 \} \le 0.$$

By integration and (3.64), we have

$$|u(t)|^2 \leq 0, \quad \forall t \in [0,T].$$

Thus

 $u_1 = u_2$ ,

and the solution is unique.

## Remark 3.3.

As a consequence of (3.48) the (unique) solution of the Navier-Stokes equations satisfies

(3.66) 
$$u \in \mathbb{L}^{4}(Q)$$
 (n = 2).

Theorem 3.2 covers both the bounded and unbounded cases; there are no differences in the proofs of the two cases.

3.4 About Regularity and Uniqueness (n = 3).

The results of Section 3.3 cannot be extended to higher dimensions due to lack of information concerning the regularity of the weak solutions given by Theorem 3.1.

Nevertheless, we will prove some further regularity properties of a solution, which are weaker than those of the two-dimensional case. We then give an uniqueness theorem in a class of functions for which the existence is not known; this result shows, nevertheless, how an improvement in the information concerning the regularity of weak solutions leads to uniqueness.

The result similar to Lemma 3.3 is

Proof.

We only have to prove (3.67) for  $v \in \mathcal{D}(\Omega)$ . For such a v, by application of (3.48), we write

$$(3.68) \int_{\mathbb{R}^{3}} v^{4}(x) dx \leq 2 \int_{-\infty}^{+\infty} \left\{ \left( \int_{\mathbb{R}^{2}} v^{2} dx_{1} dx_{2} \right) \left( \int_{\mathbb{R}^{2}} \sum_{i=1}^{2} (D_{i}v)^{2} dx_{1} dx_{2} \right) \right\} dx_{3}$$
$$\leq 2 \left( \sup_{x_{3}} \cdot \int_{\mathbb{R}^{2}} v^{2} dx_{1} dx_{2} \right) \left( \sum_{i=1}^{2} \|D_{i}v\|_{L^{2}(\mathbb{R}^{3})}^{2} \right).$$

But

$$v^{2}(x) = 2 \int_{-\infty}^{x_{3}} v(x_{1}, x_{2}, \xi_{3}) D_{3} v(x_{1}, x_{2}, \xi_{3}) d\xi_{3}$$
  
$$\leq 2 \int_{-\infty}^{+\infty} |v(x_{1}, x_{2}, \xi_{3})| |D_{3} v(x_{1}, x_{2}, \xi_{3})| d\xi_{3}$$

and hence

$$\sup_{\mathbf{x}_{3}} \int_{\mathbb{R}^{2}} v^{2} dx_{1} dx_{2} \leq 2 \int_{\mathbb{R}^{3}} |v| |D_{3}v| dx \leq 2 ||v||_{L^{2}(\mathbb{R}^{3})} ||D_{3}v||_{L^{2}(\mathbb{R}^{3})}.$$

With this inequality we deduce from (3.68) that

$$\int_{\mathbb{R}^{3}} v^{4}(x) dx \leq 4 \|v\|_{L^{2}(\mathbb{R}^{3})} \|D_{3}v\|_{L^{2}(\mathbb{R}^{3})} \left( \sum_{i=1}^{2} \|D_{i}v\|_{L^{2}(\mathbb{R}^{3})} \right)$$
$$\leq 4 \|v\|_{L^{2}(\mathbb{R}^{3})} \left( \sum_{i=1}^{3} \|D_{i}v\|_{L^{2}(\mathbb{R}^{3})} \right)^{\frac{3}{2}}$$

and (3.67) follows.

Theorem 3.3. <u>If</u> n = 3, the solution u of (3.17) - (3.19) given by Theorem 3.1 satisfies (3.69)  $u \in L^{\frac{8}{3}}(0,T;L^{4}(\Omega))$ (3.70)  $\cdot u' \in L^{\frac{4}{3}}(0,T;V')$ .

Proof.

For almost every t, according to (3.67),

$$(3.71) \|u(t)\|_{\mathbb{L}^{4}(\Omega)} \leq c_{\mathfrak{g}} |u(t)|^{\frac{1}{4}} \|u(t)\|^{\frac{3}{4}}.$$

The function on the right-hand side belongs to  $L^{\frac{8}{3}}(0,T)$ , and thus also the function on the left-hand side.

By use of the Hölder inequality, we derived in Chapter II the bound

$$(3.72) |b(u,u,v)| = |b(u,v,u)| \le c_1 ||u||_{\mathbf{L}^4(\Omega)}^2 ||v||, \forall u,v \in V^{(1)}.$$

Therefore, if  $u \in L^2(0,T;V) \cap L^{\infty}(0,T;H)$ , Bu belongs to  $L^{\frac{4}{3}}(0,T;V')$  since

(3.73)  $\|Bu(t)\|_{V} \leq c_1 \|u(t)\|_{L^{4}(\Omega)}^{2}$ 

(3.74) 
$$\|Bu(t)\|_{V^{1}} \leq c_{2} |u(t)|^{\frac{1}{2}} |u(t)|^{\frac{3}{2}}, a.e.$$

In the two dimensional case we established that any solution of (3.17) - (3.19) satisfies (3.60) and (3.66) and this was the property which essentially enabled us to prove uniqueness. For n = 3, (3.60) and (3.66) are replaced by the weaker results (3.69) - (3.70).

We show now that there is at most one solution in a smaller class of functions than the class in which we obtained existence.

# <u>Theorem 3.4</u>. <u>If n = 3, there is at most one solution of Problem 3.2 such that</u>

(1) This inequality with c depending on n holds for any dimension of space.

(3.75) 
$$u \in L^{2}(0,T;V) \cap L^{\infty}(0,T;H)$$

(3.76)  $u \in L^{8}(0,T; L^{4}(\Omega)).$ 

Such a solution would be continuous from [0,T] into H.

Proof.

(i) The inequalities 
$$(3.72) - (3.73)$$
 imply that if u satisfies  $(3.76)$  then  
(3.77) Bu  $\in L^2(0,T;V^4)$  (at least).

Therefore if u satisfies (3.75) - (3.76) and (3.18), then

$$(3.78) u' \in L^2(0,T;V')$$

and according to Lemma 1.2, u is almost everywhere equal to a continuous function from [0,T] into H.

(ii) By the Hölder inequality and (3.67),

$$|b(u,u,v)| \leq c_0 ||u||_{L^4(\Omega)} ||v||_{L^4(\Omega)} ||u||,$$

$$(3.79) ||b(u,u,v)| \leq c_1 ||u||^{\frac{1}{4}} ||u||^{\frac{7}{4}} ||v||_{L^4(\Omega)}.$$

(iii) Let us assume that  $u_1$  and  $u_2$  are two solutions of (3.17) - (3.19) which satisfy (3.75) - (3.76), and let  $u = u_1 - u_2$ .

As in the proof of Theorem 3.2 one can show that

(3.80) 
$$\frac{d}{dt} |u(t)|^2 + 2v ||u(t)||^2 = 2b(u(t), u(t), u_2(t)).$$

We then bound the right-hand side, according to (3.79), by

$$2c_{1}|u(t)|^{\frac{1}{4}}|u(t)|^{\frac{7}{4}}|u_{2}(t)||_{\mathbb{L}^{4}(\Omega)} \leq v||u(t)|^{2} + c_{2}|u(t)|^{2}||u_{2}(t)||_{\mathbb{L}^{4}(\Omega)}^{8}.$$

We get

$$\frac{d}{dt} |u(t)|^2 \leq c_2 ||u_2(t)||_{\mathbb{L}^4(\Omega)}^8 |u(t)|^2.$$

Since the function  $t \mapsto |u_2(t)|_{\mathbb{L}^4(\Omega)}^8$  is integrable, we can complete the proof as done for Theorem 3.2.

The preceding proof is valid for  $\Omega$  bounded or unbounded.

Remark 3.6.

Remark 3.5.

There are many similar results of uniqueness which can be proved by assuming

徽

(3.75) 
$$u \in L^{2}(0,T;V) \cap L^{\infty}(0,T;H)$$

(3.76)  $u \in L^{8}(0,T; L^{4}(\Omega)).$ 

Such a solution would be continuous from [0,T] into H.

Proof.

(i) The inequalities 
$$(3.72) - (3.73)$$
 imply that if u satisfies  $(3.76)$  then  
(3.77) Bu  $\in L^2(0,T;V^4)$  (at least).

Therefore if u satisfies (3.75) - (3.76) and (3.18), then

$$(3.78) u' \in L^2(0,T;V')$$

and according to Lemma 1.2, u is almost everywhere equal to a continuous function from [0,T] into H.

(ii) By the Hölder inequality and (3.67),

$$|b(u,u,v)| \leq c_0 ||u||_{L^4(\Omega)} ||v||_{L^4(\Omega)} ||u||,$$

$$(3.79) ||b(u,u,v)| \leq c_1 ||u||^{\frac{1}{4}} ||u||^{\frac{7}{4}} ||v||_{L^4(\Omega)}.$$

(iii) Let us assume that  $u_1$  and  $u_2$  are two solutions of (3.17) - (3.19) which satisfy (3.75) - (3.76), and let  $u = u_1 - u_2$ .

As in the proof of Theorem 3.2 one can show that

(3.80) 
$$\frac{d}{dt} |u(t)|^2 + 2v ||u(t)||^2 = 2b(u(t), u(t), u_2(t)).$$

We then bound the right-hand side, according to (3.79), by

$$2c_{1}|u(t)|^{\frac{1}{4}}|u(t)|^{\frac{7}{4}}|u_{2}(t)||_{\mathbb{L}^{4}(\Omega)} \leq v||u(t)|^{2} + c_{2}|u(t)|^{2}||u_{2}(t)||_{\mathbb{L}^{4}(\Omega)}^{8}.$$

We get

$$\frac{d}{dt} |u(t)|^2 \leq c_2 ||u_2(t)||_{\mathbb{L}^4(\Omega)}^8 |u(t)|^2.$$

Since the function  $t \mapsto |u_2(t)|_{\mathbb{L}^4(\Omega)}^8$  is integrable, we can complete the proof as done for Theorem 3.2.

The preceding proof is valid for  $\Omega$  bounded or unbounded.

Remark 3.6.

Remark 3.5.

There are many similar results of uniqueness which can be proved by assuming

徽

some other properties of regularity. For example (cf. Lions [2] p. 84), there is uniqueness in any dimension if, in place of (3.76), u satisfies

$$(3.81) u \in L^{S}(0,T;L^{r}(\Omega))$$

with

(3.82) 
$$\begin{cases} \frac{2}{s} + \frac{n}{r} \leq 1 \text{ if } \Omega \text{ is bounded,} \\ \frac{2}{s} + \frac{n}{r} = 1 \text{ if } \Omega \text{ is unbounded.} \end{cases}$$

3.5 More Regular Solutions.

C

Our purpose in this section is to prove that by assuming more regularity on the data, we can obtain more regular solutions in the two dimensional case. In the three dimensional case the existence of such more regular solutions is only proved if we assume that the given data  $u_0$ , f, are "small enough" or that  $\nu$  is large enough.

3.5.1 The <u>Two-Dimensional</u> <u>Case</u>.

We assume that n = 2 and that

(3.83) f and  $f' \in L^2(0,T;V')$ ,  $f(0) \in H$ 

 $(3.84) u_0 \in \mathbb{H}^2(\Omega) \cap \mathbb{V}.$ 

Then the unique solution of Problem 3.2 given by Theorems 3.1 and 3.2 satisfies

(3.85)  $u' \in L^2(0,T;V) \cap L^{\infty}(0,T;H).$ 

Proof.

Theorem 3.5.

(i) We return to the Galerkin approximation used in the proof of Theorem3.1. We need only show that this approximate solution also satisfies the two a priori estimates:

(3.86)  $\begin{cases} u'_{\mathfrak{m}} & \underline{\operatorname{remains in } a \text{ bounded set of}} \\ L^2(0,T;V) \cap L^{\infty}(0,T;H). \end{cases}$ 

In the limit (3.86) implies (3.85).

Since  $u_0 \in V \cap \mathbb{H}^2(\Omega)$ , we can choose  $u_{0m}$  as the orthogonal projection in  $V \cap \mathbb{H}^2(\Omega)$  of  $u_0$  onto the space spanned by  $w_1, \cdots, w_m$ ; then

(3.87) 
$$\begin{cases} u_{0m} \rightarrow u_{0} \text{ in } \mathbb{H}^{2}(\Omega), \text{ as } m \rightarrow \infty, \\ \|u_{0m}\|_{\mathbb{H}^{2}(\Omega)} \leq \|u_{0}\|_{\mathbb{H}^{2}(\Omega)}. \end{cases}$$

(ii) We multiply (3.22) by  $g'_{jm}(t)$  and add the resulting equations for  $j = 1, \dots, m$ ; this gives

(3.88)  $|u_{m}'(t)|^{2} + v((u_{m}(t), u_{m}'(t))) + b(u_{m}(t), u_{m}(t), u_{m}'(t)) = \langle f(t), u_{m}'(t) \rangle$ . In particular, at time t = 0,

(3.89) 
$$|u_{m}^{\dagger}(0)|^{2} = (f(0), u_{m}^{\dagger}(0)) + \nu(\Delta u_{0m}, u_{m}^{\dagger}(0)) - b(u_{0m}, u_{0m}, u_{m}^{\dagger}(0))$$

so that

(3.90) 
$$|u_{m}^{\dagger}(0)| \leq |f(0)| + \nu |\Delta u_{0m}| + |Bu_{0m}|$$

It is clear from (3.87) that

$$|\Delta u_{0m}| \leq c_0 \|u_{0m}\|_{\mathbf{H}^2(\Omega)} \leq c_0 \|u_0\|_{\mathbf{H}^2(\Omega)}.$$

u we have, by the Hölder inequality,

$$\begin{aligned} |b(u,u,v)| &\leq c_1 \|u\|_{\mathbb{L}^4(\Omega)} |\operatorname{grad} u|_{\mathbb{L}^4(\Omega)} |v| \\ &\leq (by \ (3.48) \text{ and the Sobolev inequality}) \\ &\leq c_2 \|u\| \|u\|_{\mathbb{H}^2(\Omega)} |v|, \ \forall u \in \mathbb{H}^2(\Omega), \ \forall v \in \mathbb{L}^2(\Omega) \end{aligned}$$

and hence

(3.91) 
$$|Bu_{0m}| \leq c_2 ||u_{0m}|| ||u_{0m}||_{H^2(\Omega)} \leq (by (3.87)) \leq c_2 ||u_0||_{H^2(\Omega)}^2$$

Finally (3.90) and the above estimates show that

(3.92)  $u'_{m}(0)$  belongs to a bounded set of H.

(iii) We are allowed to differentiate (3.22) in the t variable, and since f satisfies (3.83), we get

$$(3.93) \quad (u_{m}'', w_{j}) + \nu((u_{m}', w_{j})) + b(u_{m}', u_{m}, w_{j}) + b(u_{m}, u_{m}', w_{j}) = \langle f', w_{j} \rangle, \quad j = 1, \cdots, m$$

We multiply (3.93) by  $g'_{jm}(t)$  and add the resulting equations for  $j = 1, \dots, m$ ; we find (taking (3.2) into account):

$$(3.94) \quad \frac{d}{dt} \left\| u'_{m}(t) \right\|^{2} + 2\nu \| u'_{m}(t) \|^{2} + 2b(u'_{m}(t), u'_{m}(t), u''_{m}(t)) = 2 \langle f'(t), u''_{m}(t) \rangle.$$

By Lemma 3.4,

$$2b(u'_{m}(t), u_{m}(t), u'_{m}(t) \leq 2^{\frac{3}{2}} |u'_{m}(t)| ||u'_{m}(t)|| ||u_{m}(t)||$$
  
$$\leq v ||u'_{m}(t)||^{2} + \frac{2}{v} ||u_{m}(t)||^{2} |u'_{m}(t)|^{2}.$$

Thus, we deduce from (3.94) that

(3.95) 
$$\frac{d}{dt} |u_{m}'(t)|^{2} + \frac{v}{2} ||u_{m}'(t)||^{2} \leq \frac{2}{v} |f'(t)|_{V}^{2} + \phi_{m}(t) |u_{m}'(t)|^{2}$$

where

$$\phi_{\rm m}(t) = \frac{2}{\nu} \| u_{\rm m}(t) \|^2.$$

Then, by the usual method of the Gronwall inequality,

$$\frac{\mathrm{d}}{\mathrm{dt}} \left\{ \left| \mathbf{u}_{\mathrm{m}}^{\prime}(t) \right|^{2} \exp\left(-\int_{0}^{t} \phi_{\mathrm{m}}^{\prime}(s) \mathrm{ds}\right) \right\} \leq \frac{2}{\nu} \left| \mathbf{f}^{\prime}(t) \right|_{\mathrm{V}}^{2},$$

whence

(3.96) 
$$|u_{m}^{\prime}(t)|^{2} \leq \{|u_{m}^{\prime}(0)|^{2} + \frac{2}{\nu} \int_{0}^{t} |f^{\prime}(s)|_{V}^{2} ds\} \exp \int_{0}^{t} \phi_{m}^{\prime}(s) ds.$$

Since the functions  $u_m$  remain in a bounded set of  $L^2(0,T;V)$  (cf. (3.31)) and because of (3.92), the right-hand side of (3.96) is uniformly bounded in  $s \in [0,T]$  and m:

(3.97)  $u''_m$  belongs to a bounded set of  $L^{\infty}(0,T;H)$ .

With (3.97) we infer easily from (3.95) that the  $u'_m$  remain in a bounded set of  $L^2(0,T;V)$ .

The proof is achieved.

Theorem 3.6.

The assumptions are those of Theorem 3.5 and we assume moreover that  $\,\Omega\,$  is a bounded set of class  $\,\mathcal{C}^2\,$  and that

(3.98)  $f \in L^{\infty}(0,T;H)$ .

Then the function u satisfies

 $(3.99) u \in L^{\infty}(0,T; \mathbb{H}^{2}(\Omega)).$ 

Proof.

(i) We write (3.18) in the form

(3.100)  $\forall ((u(t),v)) = (g(t),v), \forall v \in V,$ 

where

18 18
(3.101) 
$$g(t) = f(t)-u'(t)-Bu(t).$$

The proof is now based on two successive applications of Proposition I.2.2. (ii) Since  $u \in L^{\infty}(0,T;V)$  and

(3.102) 
$$|b(u(t), u(t), v)| \leq c_0 ||u(t)||_{L^4(\Omega)} ||u(t)|| ||v||_{L^4(\Omega)}$$
  
 $\leq c_1 ||u(t)||^2 ||v||_{L^4(\Omega)},$ 

we have

Bu∈L<sup>∞</sup>(0,T;L<sup>$$\frac{4}{3}$$</sup>(Ω)).

Thus  $(f-u' \in L^{\infty}(0,T;H))$ ,

(3.103)  $g \in L^{\infty}(0,T; \mathbb{L}^{\frac{4}{3}}(\Omega)).$ 

Proposition I.2.2 then implies that

$$u \in L^{\infty}(0,T;W^{2,\frac{4}{3}}(\Omega)).$$

By the Sobolev theorem,  $W^{2,\frac{4}{3}}(\Omega) \subset L^{\infty}(\Omega)$ , and hence  $u \in L^{\infty}(Q)$ .

(iii) We can now improve (3.103). We replace (3.102) by the inequality

$$\left| b(u(t),u(t),v) \right| \leq c_2 \|u\|_{\operatorname{L}^{\infty}(Q)} \|u(t)\| \|v\|$$

which shows that

 $Bu \in L^{\infty}(0,T;H).$ 

This implies that

 $g \in L^{\infty}(0,T;H)$ 

and another application of Proposition I.2.2 gives

$$\mu \in L^{\sim}(0,T; \mathbb{H}^{2}(\Omega)).$$

Remark 3.7.

By repeated application of Proposition I.2.2 it is now easy to prove, exactly as in Proposition II.1.1, that if  $\Omega$  is of class  $\mathcal{C}^{\infty}$ ,  $u_{0}$  is given in  $\mathcal{C}^{\infty}(\Omega)$ , and f is given in  $\mathcal{C}^{\infty}(\overline{Q})$ , then the solution u is in  $\mathcal{C}^{\infty}(\overline{Q})$ . By the same methods, intermediate regularity properties can be obtained with suitable hypotheses on the data. We will prove for n = 3 some regularity properties similar to those obtained for n = 2, but in the present case this will be done only by assuming that the data are "small."

In the next theorem we denote by c some constant such that

 $(3.104) |b(u,v,w)| \leq c ||u|| ||v|| ||w||, \forall u,v,w \in V.$ 

Theorem 3.7.

We assume that n = 3 and that there are given f and  $u_0$  satisfying (3.105)  $u_0 \in \mathbb{H}^2(\Omega) \cap V$ 

(3.106)  $f \in L^{\infty}(0,T;H), f' \in L^{1}(0,T;H)$ 

and a further condition given in the course of the proof which is satisfied if v is large enough or if f and  $u_0$  are "small enough" (1).

Then there exists a unique solution of Problem 3.2 which satisfies moreover (3.107)  $u' \in L^2(0,T;V) \cap L^{\infty}(0,T;H).$ 

Proof.

(i) To begin with, we observe that uniqueness is merely a consequence of Theorem 3.4, because such a solution will satisfy

(3.108)  $u \in L^{\infty}(0,T;V)$ 

and then  $V \subseteq L^{4}(\Omega)$  implies (see (3.76)) that

(3.109) 
$$u \in L^{\infty}(0,T; L^{4}(\Omega)).$$

(ii) Some of the steps of the existence proof are the same as in Theorem
3.4: we use the Galerkin method of Theorem 3.1, and we choose the basis and u m
so that (3.87) holds. The estimates (3.90), (3.91), and thus (3.92) still hold:

$$(3.110) |u'_{m}(0)| \leq d_{1} = |f(0)| + vc_{0} ||u_{0}||_{H^{2}(\Omega)}^{2} + c_{1} ||u_{0}||_{H^{2}(\Omega)}^{2}.$$

We derive in the same fashion equation (3.94) and using (3.104) we deduce now from it:

$$(3.111) \quad \frac{d}{dt} |u'_{m}(t)|^{2} + 2(v-c||u_{m}(t)||) ||u'_{m}(t)||^{2} \leq 2|f'(t)| ||u'_{m}(t)||.$$

(iii) There results from (3.28) and (3.29) that

<sup>(1)</sup>Like (3.115).

(3.112) 
$$\begin{aligned} & v \| u_{m}(t) \|^{2} \leq \frac{1}{v} \| f(t) \|_{V}^{2}, -2(u_{m}(t), u_{m}^{\prime}(t)) \\ & \leq \frac{1}{v} \| f(t) \|_{V}^{2}, +2 | u_{m}(t) | | | u_{m}^{\prime}(t) | \\ & \leq \frac{d_{2}}{v} + 2(| u_{0} |^{2} + \frac{Td_{2}}{v} )^{\frac{1}{2}} | u_{m}^{\prime}(t) | \end{aligned}$$

where

(3.113) 
$$d_{2} = \|f\|^{2} L^{\infty}(0,T;V')$$

Using (3.110), we infer from (3.112) that, at time t = 0,

The hypothesis mentioned in the statement of the theorem is that

$$(3.115) \quad d_{4} = \frac{d_{2}}{\nu} + (1+d_{1}^{2}) \left( \left| u_{0} \right|^{2} + \frac{Td_{2}}{\nu} \right)^{\frac{1}{2}} \exp\left( \int_{0}^{T} \left| f'(s) \right| ds \right) < \frac{\nu^{3}}{c^{2}} .$$
  
Since  $d_{3} \leq d_{4}$ , we get as a consequence of (3.114) - (3.115)  
 $\nu \left\| u_{m}(0) \right\|^{2} \leq d_{3} \leq d_{4} < \frac{\nu^{3}}{c^{2}}$ 

and then

$$v-c \|u_{m}(0)\| > 0.$$

We deduce from this inequality that  $v-c \|u_m(t)\|$  remains positive on some interval with origin 0. We denote by  $T_m$  the first time  $t \leq T$  such that

$$v - c \| u_m(T_m) \| = 0$$

or, if this does not happen,  $T_m = T$ . Then

(3.116) 
$$v_{-c} \|u_{m}(t)\| \ge 0, \quad 0 \le t \le T_{m}.$$

(iv) With (3.116) we deduce from (3.111) that

$$\frac{d}{dt} |u'_{m}(t)|^{2} \leq 2|f'(t)| |u'_{m}(t)|,$$

$$\frac{d}{dt} (1+|u'_{m}(t)|^{2}) \leq |f'(t)| (1+|u'_{m}(t)|^{2}).$$

Thus

$$\frac{d}{dt} \{ (1+|u'_{m}(t)|^{2}) \exp(-\int_{0}^{t} |f'(s)|ds) \} \leq 0,$$
  
$$1 + |u'_{m}(t)|^{2} \leq (1+|u'_{m}(0)|^{2}) \exp(\int_{0}^{t} |f'(s)|ds),$$

and, by (3.110),

$$(3.117) \quad 1+|u_{m}'(t)|^{2} \leq (1+d_{1}^{2}) \exp(\int_{0}^{T} |f'(s)|ds), \quad 0 \leq t \leq T_{m}.$$

From (3.112), (3.115), and (3.117) we get

(3.118) 
$$\begin{aligned} & v \| u_{m}(t) \|^{2} \leq d_{4}, \quad 0 \leq t \leq T_{m}, \\ & v - c \| u_{m}(t) \| \geq v - c \sqrt{\frac{d}{b}} > 0, \quad 0 \leq t \leq T_{m} \end{aligned}$$

Then  $T_m = T$ , and (3.111) implies

$$\frac{d}{dt} |u'_{m}(t)|^{2} + 2(v - c\sqrt{\frac{d_{4}}{v}}) ||u'_{m}(t)||^{2} \le 2|f'(t)| |u'_{m}(t)|, \quad 0 \le t \le T,$$

and we easily deduce from this relation that

(3.119) 
$$u'_{m}$$
 remains in a bounded set of  $L^{2}(0,T;V) \cap L^{\infty}(0,T;H)$ .

The existence is proved.

As in the two-dimensional case, we also have

Theorem 3.8.

With the assumptions of Theorem 3.7, and if we assume moreover that  $\Omega$  is of class  $\mathcal{C}^2$ , the function u satisfies

(3.120) 
$$u \in L^{\sim}(0,T; \mathbb{H}^{2}(\Omega))$$

Proof.

We write (3.18) in the form,

$$v((u(t),v)) = (g(t),v), \forall v \in V,$$

with

$$g(t) = f(t)-u'(t)-Bu(t).$$

Since  $f-u' \in L^{\infty}(0,T;H)$ , Proposition I.2.2 gives (3.120) provided we show that (3.121)  $Bu \in L^{\infty}(0,T;H)$  (and hence  $g \in L^{\infty}(0,T;H)$ ).

This result is also obtained by repeated application of Proposition I.2.2 and various estimates on the form b.

By the Hölder inequality, we have:

$$|b(u(t),u(t),v)| \leq c_0 ||u(t)||_{\mathbb{L}^6(\Omega)} ||u(t)|| ||v||_{\mathbb{L}^3(\Omega)}$$

(3.122) 
$$|b(u(t), u(t), v)| \leq c_1 ||u(t)||^2 ||v||_{\mathbb{L}^3(\Omega)}.$$

We deduce from (3.122) (and  $u \in L^{\infty}(0,T;V)$ ), that  $Bu \in L^{\infty}(0,T;L^{\frac{3}{2}}(\Omega))$ ,  $g \in L^{\infty}(0,T;L^{\frac{3}{2}}(\Omega))$ .

Proposition I.2.2 implies that

$$u \in L^{\infty}(0,T;W^{2,\frac{3}{2}}(\Omega))$$

and, in particular (since  $W^{2,\frac{3}{2}}(\Omega) \subset L^{8}(\Omega)$  for example, since n = 3),  $u \in L^{\infty}(0,T; L^{8}(\Omega))$ .

Using again the Hölder inequality we estimate b by

$$b(u(t),u(t),v) \leq c_0 \|u(t)\|_{\mathbb{L}^8(\Omega)} \|u(t)\| \|v\|_{\mathbb{L}^3(\Omega)}^8.$$

Hence

and by Proposition 1.2.2,

(3.123) 
$$u \in L^{\infty}(0,T; \mathbb{W}^{2,\frac{8}{5}}(\Omega)) \subset \mathbb{L}^{\infty}(\Omega \times [0,T]).$$

With (3.123) and  $u \in L^{\infty}(0,T;V)$ , the proof of (3.121) is easy, and thus Theorem 3.8 is proved.

Remark 3.8.

The same remark about regularity as in Remark 3.7 holds.

Introduction of the Pressure (n < 4).

For introducing the pressure, let us set

$$U(t) = \int_0^t u(s)ds, \quad \beta(t) = \int_0^t Bu(s)ds, \quad F(t) = \int_0^t f(s)ds.$$

If u is a solution of (3.17) - (3.19) then, for any  $n \le 4$ ,

(3.124) 
$$U,\beta,F\in\mathcal{C}([0,T];V').$$

Integrating (3.18), we see that

20

with

$$g(t) = F(t)-\beta(t)-u(t)+u_{0}, g \in \mathcal{C}([0,T];V').$$

By application of Proposition I.1.3, we get for each  $t \in [0,T]$ , the existence of some function P(t),

 $P(t) \in L^2(\Omega)$ ,

such that

$$-v\Delta U(t) + \text{grad } P(t) = g(t)$$

or

$$u(t)-u_{\lambda}-v\Delta U(t) + \beta(t) + \text{grad } P(t) = F(t)$$

According to Proposition I.1.2, the gradient operator is an isomorphism from  $L^2(\Omega)/\mathbb{R}$  into  $\mathbb{H}^{-1}(\Omega)$ . Observing that

grad  $P = g - v \Delta U$ ,

we conclude that grad P belongs to  $\mathcal{C}([0,T];\mathbb{H}^{-1}(\Omega))$  and therefore

(3.127) 
$$P \in \mathcal{C}([0,T]; L^2(\Omega)).$$

This enables us to differentiate (3.126) in the distribution sense in  $Q = \Omega \times (0,T)$ ; setting

$$(3.128) p = \frac{\partial P}{\partial t},$$

ω

we obtain

(3.129) 
$$\frac{\partial u}{\partial t} - v\Delta u + \sum_{i=1}^{n} u_i D_i u + \text{grad } p = f, \text{ in } Q.$$

The pressure appears in general as a distribution on Q defined by (3.127) - (3.128). Under the assumptions of Theorems 3.6 (n = 2) or 3.8 (n = 3), the application of Proposition I.2.2 shows also that

(3.130) 
$$p \in L(0,T;H^{1}(\Omega)).$$

#### \$4. ALTERNATE PROOF OF EXISTENCE BY SEMI-DISCRETIZATION.

Our goal now is to give an alternate proof of the existence of weak solutions of the Navier-Stokes equations which will be valid in any number of space dimensions. An approximate solution is constructed by semi-discretization in t, and we then pass to the limit using compactness arguments.

In Section 4.1 we reformulate the problem in a way which is appropriate in any dimension and we state the existence results; Section 4.2 describes the construction of the approximate solution; Sections 4.3 and 4.4 deal with the a priori estimates and the passage to the limit.

#### 4.1 Statement of the Problem.

Before giving the existence theorem in the higher dimensions we must reformulate the problem of weak solutions. As in the stationary case, if n > 4, the form b is not trilinear continuous on V and a statement such as (3.10)-(3.14) does not make sense since the b(u(t),u(t),v) term in (3.13) is perhaps not defined.

For this purpose we introduce again (see Chapter II, Section 1.2) the spaces  ${\rm V}_{\rm S}$  :

(4.1) 
$$V_s = \text{the closure of } \mathcal{V} \text{ in } \mathbb{H}^1_0(\Omega) \cap \mathbb{H}^S(\Omega), s \ge 1.$$

The spaces  $\mathbb{H}^1_0(\Omega) \cap \mathbb{H}^{\mathbf{S}}(\Omega)$  and  $\mathbb{V}$  are endowed with the usual Hilbert norm of  $\mathbb{H}^{\mathbf{S}}(\Omega)$ :

(4.2) 
$$\|u\| = \{\sum_{\substack{j \leq s}} |D^{j}u|^{2}\}^{\frac{1}{2}}$$
 (s integer).

Obviously (s > 1),

(4.3)

with a continuous injection and  $V_s$  is dense in V.

The form b is defined on  $V \times V \times V_s$ , provided  $s \ge \frac{n}{s}$ ; more precisely:

v ⊂ v

Lemma 4.1.

 $\frac{1}{1} \frac{1}{1} \frac{1}$ 

Proof.

For  $u, v, w \in \mathcal{V}$ , the Hölder inequality gives

(1) Any dimension,  $\Omega$  bounded or not.

250

If  $w \in V$  then  $D_i w_j$  belongs to  $H^{s-1}(\Omega)$  and to  $L^q(\Omega)$ ;  $D_i w_j$  belonging to  $L^q(\Omega) \cap L^2(\Omega)$  implies that  $D_i w_j \in L^n(\Omega)$  too and

$$\left\| \mathbf{D}_{\mathbf{j}} \mathbf{w}_{\mathbf{j}} \right\|_{\mathbf{L}^{n}(\Omega)} \leq \mathbf{c}_{\mathbf{j}} \left\| \mathbf{w} \right\|_{\mathbf{V}_{\mathbf{s}}}$$

so that

(4.6) 
$$|b(u,v,w)| \leq c_2 |u| ||v|| ||w||_V$$

This estimate shows that we can extend by continuity the form b from  $\mathcal{V} \times \mathcal{V} \times \mathcal{V}$  onto  $\mathcal{V} \times \mathcal{V} \times \mathcal{V}_s$ , and even  $\mathcal{H} \times \mathcal{V} \times \mathcal{V}_s$ , by (4.6).

Lemma 4.2.

 $\frac{\text{If } u \text{ belongs to } L^2(0,T;V) \cap L^{\infty}(0,T;H) \text{ then } Bu \text{ belongs to } L^2(0,T;V'_s)}{\text{for } s \geq \frac{n}{2}}.$ 

Proof.

By the definition of B and because of (4.4),

$$|\langle Bu(t), v \rangle| = |b(u(t), u(t), v)| \leq c|u(t)| ||u(t)|| ||v||_{V_s}, \forall v \in V_s;$$

hence

(4.7) 
$$\|Bu(t)\|_{V'_{c}} \leq c |u(t)| \|u(t)\|$$
 for a.a.  $t \in [0,T]$ 

and the lemma is proved.

In all dimensions of space, we can give the following weak formulation of the Navier-Stokes problem:

Problem 4.1.

For f and u given such that

(4.7)  $f \in L^2(0,T;V'),$ 

(4.8)

to find u satisfying

(4.9)  $u \in L^{2}(0,T;V) \cap L^{\infty}(0,T;H)$ (4.10)  $\frac{d}{dt}(u,v) + v((u,v)) + b(u,u,v) = \langle f,v \rangle, \quad \forall v \in V_{s} \quad (s \geq \frac{n}{2})$ (4.11)  $u(0) = u_{0}.$ 

u ∈ H,

If u satisfies (4.9) and (4.10), then

# $\frac{d}{dt} <_{u,v} = <_{g,v}, \forall_v \in V_s$

with

### g = f - Bu - vAu.

Due to Lemma 4.2, Bu belongs to  $L^2(0,T;V_S^{\prime})$  and since f-vAu belongs to  $L^2(0,T;V^{\prime})$ ,

(4.12) 
$$g \in L^2(0,T;V'_c).$$

Lemma 1.1 then implies that

(4.13) 
$$\begin{cases} u' \in L^{2}(0,T;V_{s}^{1}) \\ u' = f - vAu - Bu; \end{cases}$$

therefore u is almost everywhere equal to a continuous function from [0,T] into V' and (4.11) makes sense.

An alternate formulation of Problem 4.1 is the following one:

 $\frac{\text{Problem } 4.2}{\text{Given f and } u_0, \quad \underline{\text{satisfying } (4.7) - (4.8), \quad \underline{\text{to find } u \quad \underline{\text{satisfying }}}{u \in L^2(0,T;V) \cap L^\infty(0,T;H), \quad u' \in L^2(0,T;V_s') \quad (s \geq \frac{n}{2}),$   $(4.15) \qquad u' + \forall Au + Bu = f \quad \text{on } (0,T),$ 

(4.16) u(0) = u.

The formulations (4.9) - (4.11) and (4.14) - (4.16) are equivalent.

The existence of solutions of these problems is given by the following theorem which implies Theorem 3.1:

Theorem 4.1.

Let there be given f and u, which satisfy (4.7)-(4.8). Then, there

exists at least one solution u of Problem 4.2. Moreover u is weakly continuous from [0,T] into H.

This theorem is proved in Sections 4.2 and 4.3; the weak continuity in H is a direct consequence of (4.14) and Lemma 1.4.

4.2 The Approximate Solutions.

Let N be a integer which will later go to infinity and set

(4.17) k = T/N.

We will define recursively a family of elements of V, say  $u^0, u^1, \dots, u^N$ , where  $u^m$  will be in some sense an approximation of the function u we are looking for, on the interval mk < t < (m+1)k.

We define first the elements f<sup>1</sup>,...,f<sup>N</sup> of V':

(4.18) 
$$f^{m} = \frac{1}{k} \int_{(m-1)k}^{mk} f(t) dt, m = 1, \dots, N; f^{m} \in V'.$$

We begin with

(4.19)  $u^0 = u_0$ , the given initial data;

then when  $u^0, \dots, u^{m-1}$  are known, we define  $u^m$  as an element of V which satisfies

(4.20) 
$$\frac{u^{m}-u^{m-1}}{k} + vAu^{m} + Bu^{m} = f^{m};$$

 $u^m$  depends on k; for simplification we denote them  $u^m$  in lieu of  $u^m_{\nu}$ .

The existence of such a  $u^m$  is asserted by Lemma 4.3, whose proof is postponed to the end of this section.

#### Lemma 4.3.

For each fixed k and each m > 1, there exists at least one  $u^m$  satisfying (4.20) and moreover

$$(4.21) |u^{m}|^{2} - |u^{m-1}|^{2} + |u^{m} - u^{m-1}|^{2} + 2kv||u^{m}|^{2} \le 2k \le f^{m}, u^{m} > 0$$

For each fixed k (or N), we associate to the elements  $u^1, \dots, u^N$ , the following approximate functions:

(4.22) 
$$u_k: [0,T) \mapsto V, u_k(t) = u^m, t \in [(m-1)k, mk), m = 1, \dots, N$$

$$w_k: [0,T] \mapsto H, w_k \text{ is continuous, linear on}$$

each interval [(m-1)k,mk] and 
$$w_k(mk) = u^m$$
,  $m = 0, \dots, N$ .

In Section 4.3 we will give a priori estimates of these functions; we will then pass to the limit  $k \rightarrow 0$  (Section 4.3).

#### Proof of Lemma 4.3.

The equation (4.20) must be understood in a space larger than V', for example in a space  $V'_s$ ,  $s \ge \frac{n}{2}$ . It is equivalent to

(4.24) 
$$(u^{m},v) + kv((u^{m},v)) + kb(u^{m},u^{m},v) = \langle u^{m-1} + kf^{m},v \rangle, \forall v \in V_{s}.$$

We proceed by the Galerkin method, essentially as for Theorem II.1.2.

We choose a sequence of elements  $w_1, \dots, w_j, \dots, of V_s$  which is free and total in  $V_s$  and thus in V. For each r, by application of Lemma 1.4, we prove the existence of an element  $\phi_r$  (depending on r,k,m):

(4.25) 
$$\phi_{r} = \sum_{i=1}^{r} \xi_{i,r} w_{i},$$

(4.26) 
$$(\phi_r, v) + kv((\phi_r, v)) + kb(\phi_r, \phi_r, v) = \langle u^{m-1} + kf^m, v \rangle, \forall v \in Sp(w_1, \cdots, w_r)$$
 (1)

We must then get an a priori estimate independent of r, and pass to the limit  $r \rightarrow \infty$ . (k and m are fixed in this proof).

Taking  $v = \phi_r$  in (4.26) we get

(4.27) 
$$(\phi_{\mathbf{r}} - u^{m-1}, \phi_{\mathbf{r}}) + kv \|\phi_{\mathbf{r}}\|^2 = k < f^m, \phi^r > .$$

Now

(4.28) 
$$2(a-b,a) = |a|^2 - |b|^2 + |a-b|^2, \forall a,b \in H,$$

so that (4.27) gives

(4.29) 
$$|\phi_{\mathbf{r}}|^{2} + |\phi_{\mathbf{r}} - u^{m-1}|^{2} + 2kv \|\phi_{\mathbf{r}}\|^{2} = |u^{m-1}|^{2} + 2k \langle \mathbf{f}^{m}, \phi_{\mathbf{r}} \rangle$$
  

$$\leq |u^{m-1}|^{2} + 2k \|\mathbf{f}^{m}\|_{V'} |\phi_{\mathbf{r}}| \leq |u^{m-1}|^{2} + vk \|\phi_{\mathbf{r}}\|^{2} + \frac{k}{v} \|\mathbf{f}^{m}\|_{V'}^{2}, \quad .$$

Hence

(4.30) 
$$|\phi_{\mathbf{r}}|^{2} + |\phi_{\mathbf{r}} - u^{m-1}|^{2} + kv \|\phi_{\mathbf{r}}\|^{2} \leq |u^{m-1}|^{2} + \frac{k}{v} \|f^{m}\|_{V}^{2},$$

(1)  $Sp(w_1, \dots, w_r)$  = the space spanned by  $w_1, \dots, w_r$ .

The inequality (4.30) shows that the sequence  $\phi_r$  remains bounded in V as  $r \rightarrow \infty$ . Therefore we can extract from  $\phi_r$  a subsequence  $\phi_r$ , such that

(4.31) 
$$\phi_r \xrightarrow{\longrightarrow} \phi$$
 in V weakly, as  $r' \xrightarrow{\longrightarrow} \infty$ .

By standard arguments we then pass to the limit in (4.26) and prove that  $\phi = u^m$  satisfies (4.24).

It remains to establish (4.21). This would be obvious if we could take  $v = u^m$  in (4.24); since  $u^m \notin V_s$  in general we proceed instead by passage to the limit. We pass to the lower limit in (4.29), noting that the norm is lower semicontinuous for the weak topology:

$$|\phi|^2 \leq \lim_{r \to \infty} |\phi_r|^2, \|\phi\|^2 \leq \lim_{r \to \infty} \|\phi_r\|^2.$$

The proof is complete. 4.3 A Priori Estimates.

where d<sub>1</sub> depends only on the data:

(4.35) 
$$d_{1} = \left| u_{0} \right|^{2} + \frac{1}{\nu} \int_{0}^{T} \left\| f(s) \right\|_{V}^{2} ds.$$

#### Proof.

As mentioned in the proof of Lemma 4.3, we cannot take the scalar product of (4.20) by  $u^m$ , at least for n > 4 (Bu<sup>m</sup>  $\notin V'$ ). But (4.21) will play the same role as the equation we would obtain by this prodedure.

We majorize the right-hand side of (4.21) by

$$2k \| f^{m} \|_{V}, \| u^{m} \| \leq k v \| u^{m} \|^{2} + \frac{k}{v} \| f^{m} \|_{V}^{2},.$$

and we obtain

瀫

(4.36) 
$$|u^{m}|^{2} - |u^{m-1}|^{2} + |u^{m} - u^{m-1}|^{2} + kv ||u^{m}||^{2} \le \frac{k}{v} ||f^{m}||_{V}^{2}$$
,  $m = 1, \dots, N$ .  
Adding the equalities (4.36) for  $m = 1, \dots, N$ , we find

$$(4.37) |u^{N}|^{2} + \sum_{m=1}^{N} |u^{m}-u^{m-1}|^{2} + \nu k \sum_{m=1}^{N} ||u^{m}||^{2} \le |u_{0}|^{2} + \frac{k}{\nu} \sum_{m=1}^{N} ||f^{m}||_{V}^{2},$$

Adding the equalities (4.36) for  $m = 1, \dots, r$ , and dropping the terms  $\|u^{m}\|^{2}$ , we get

$$(4.38) |u^{r}|^{2} \leq |u_{0}|^{2} + \frac{k}{v} \sum_{m=1}^{r} ||f^{m}||_{V}^{2}, \leq |u_{0}|^{2} + \frac{k}{v} \sum_{m=1}^{N} ||f^{m}||_{V}^{2}, \quad r = 1, \cdots, N.$$

The lemma is now a consequence of (4.37)-(4.38) and of a majoration of the right-hand side of these inequalities given in the next lemma.

Proof.

Due to the Schwarz inequality,

$$\|f^{m}\|_{V}^{2} = \frac{1}{k^{2}} \|\int_{(m-1)k}^{mk} f(t)dt\|_{V}^{2} \leq \frac{1}{k} \int_{(m-1)k}^{mk} \|f(t)\|^{2} dt$$

Then (4.39) follows by summation of these inequalities for m = 1, •••, N.

 $\frac{\text{Lemma } 4.6}{\text{The sum } k} \sum_{m=1}^{N} \left\| \frac{u^m - u^m - 1}{k} \right\|^2 \qquad \text{is bounded independently of } k.$ 

#### Proof.

Taking the norm in (4.20) we obtain

$$\begin{split} \|\frac{u^{m}-u^{m-1}}{k}\|_{V_{s}^{\prime}} &\leq \|f^{m}\|_{V_{s}^{\prime}} + v\|Au^{m}\|_{V_{s}^{\prime}} + \|Bu^{m}\|_{V_{s}^{\prime}} \leq c_{1}\{\|f^{m}\|_{V^{\prime}} + \|u^{m}\|_{V}\} + \|Bu^{m}\|_{V_{s}^{\prime}}, \\ & \|\frac{u^{m}-u^{m-1}}{k}\|_{V_{s}^{\prime}}^{2} \leq c_{2}\{\|f^{m}\|_{V^{\prime}}^{2}, + \|u^{m}\|_{V}^{2} + \|Bu^{m}\|_{V_{s}^{\prime}}^{2}\}. \end{split}$$

From (4.4) and (4.32) we get

$$\| Bu^{m} \|_{V_{s}^{1}}^{2} \leq c_{3} \| u^{m} \|^{2} \| u^{m} \|^{2} \leq c_{4} \| u^{m} \|^{2}.$$

We finally have

$$k \sum_{m=1}^{N} \left\| \frac{u^{m} - u^{m-1}}{k} \right\|_{S}^{2} \leq c_{5} k \sum_{m=1}^{N} \left( \left\| f^{m} \right\|_{V}^{2}, + \left\| u^{m} \right\|^{2} \right),$$

and we finish the proof using (4.33) and Lemma 4.5.

It is interesting new to interpret the above in terms of the approximate functions:

<u>Lemma 4.7</u>.

Proof.

The estimations on  $u_k$  and  $w_k$  are just interpretations of (4.32)-(4.33) and Lemma 4.6; (4.40) is a consequence of (4.34) and the next lemma.

<u>Lemma 4.8</u>.

(4.41) 
$$|u_{k}^{-w_{k}}|_{L^{2}(0,T;H)} = \sqrt{\frac{k}{3}} \left(\sum_{m=1}^{N} |u^{m}-u^{m-1}|^{2}\right)^{\frac{1}{2}}.$$

Proof.

(4.42)

$$w_{k}(t)-u_{k}(t) = \frac{(t-mk)}{k} (u^{m}-u^{m-1}) \text{ for } (m-1)k \leq t \leq mk,$$

$$\int_{(m-1)k}^{mk} |w_{k}(t)-u_{k}(t)|^{2} dt = \frac{k}{3} |u^{m}-u^{m-1}|^{2},$$

and we find (4.41) by summation.

4.4 Passage to the Limit.

Due to Lemma 4.7, we can extract from  $u_{\rm b}$  a subsequence  $u_{\rm b}$ , such that

$$u_k$$
,  $\rightarrow u$  in  $L^2(0,T;V)$  weakly,  
in  $L^{\infty}(0,T;H)$  weak-star

(1) Strictly speaking, if  $u_0 \notin V$ ,  $w_k(t) \notin V$  for  $0 \le t \le k$ ; in this case we simply replace  $L^2(0,T;V)$  by  $L^2_{loc}((0,T];V)$  whenever we are considering the functions  $w_k$ .

functions  $w_k$  will play, for this, a useful auxiliary role.

We can choose the subsequence k', so that

 $w_k$ ,  $\rightarrow u_*$  in  $L^2(0,T;V)$  weakly, (4.43)in  $L^{\infty}(0,T;H)$  weak-star,  $\frac{dw_{k'}}{dt} \longrightarrow u'_{*} \text{ in } L^{2}(0,T;V'_{s}) \text{ weakly.}$ (4.44)

Because of (4.40),  $u = u_{\star}$ .

Theorem 2.1 shows us that

(4.45) 
$$w_{t}, \rightarrow u \text{ in } L^{2}(0,T;H);$$

thus by (4.40),

 $u_k$ ,  $\rightarrow u$  in  $L^2(0,T;H)$ . (4.46)

The equations (4.20) can be interpreted as

(4.47) 
$$\frac{dw_k}{dt} + vAu_k + Bu_k = f_k,$$

with  $f_{t}$  defined by

$$f_k(t) = f^m$$
,  $(m-1)k \le t \le mk$ ,  $m = 1, \dots, N$ .

Because of (4.42), (4.46), and Lemmas 3.2 and 4.2,

$$Bu_{k'} \longrightarrow Bu$$
 in  $L^{2}(0,T;V')$  weakly.

By Lemma 4.9 below,

$$f_k \rightarrow f$$
 in  $L^2(0,T;V');$ 

therefore we can pass to the limit in (4.47), and we find

$$u^{\dagger} + vAu + Bu = f.$$

Due to (4.43), (4.44) and Lemma 4.1,

$$\langle w_{k}, (t), \sigma \rangle \rightarrow \langle u(t), \sigma \rangle, \forall \sigma \in V_{s}^{r}, \forall t \in [0,T];$$

since  $w_{k'}(0) = u_0$ , we get

$$u(0) = u_0.$$

We have proved that u satisfies (4.14) - (4.16); the proof of Theorem 4.1 will be complete once we prove

The

Lemma 4.9.

 $f_k \longrightarrow f \quad in \quad L^2(0,T;V'), \quad as \quad k \longrightarrow 0.$ 

Proof.

(4.48)

We observe that the transformation

is a linear averaging mapping in L<sup>2</sup>(0,T;V'); this mapping is continuous by Lemma 4.5 which enables us to assert

(4.49) 
$$\|f_k\|_{L^2(0,T;V')} \leq \|f\|_{L^2(0,T;V)}$$

Therefore, instead of proving (4.48) for any f in  $L^2(0,T;V')$  we need only to prove it for f in a dense subspace of  $L^2(0,T;V')$ ; for an f in  $\mathcal{C}([0,T];V')$ the result is elementary and we skip its proof.

Remark 4.1.

Summing the equations (4.21) for  $m = 1, \dots, r$ , and dropping the terms  $|u^m - u^{m-1}|^2$ , we get

(4.50) 
$$|u^{r}|^{2} + 2kv \sum_{m=1}^{r} ||u_{\cdot}^{m}||^{2} \leq |u_{0}|^{2} + 2k \sum_{m=1}^{r} \langle f^{m}, u^{m} \rangle$$

The relation (4.50) can be interpreted as

(4.51) 
$$|u_{k}(t)|^{2} + 2\nu \int_{0}^{t_{k}} ||u_{k}(s)||^{2} ds \leq |u_{0}|^{2} + \int_{0}^{t_{k}} ||u_{k}(s)||^{2} ds$$
,

where

(4.52)  $t_k = (m+1)k$ , for  $mk \le t \le (m+1)k$ .

For each fixed t,  $u_k(t)$  is bounded in H independently of k and t; as k' $\longrightarrow 0$ ,  $u_{k'}(t)$  converges to u(t) in V' weakly; therefore (1)

(4.53)  $u_k$ ,  $(t) \rightarrow u(t)$  in H weakly, as  $k' \rightarrow 0$ ,  $\forall t \in [0,T]$ .

We then pass to the lower limit in (4.51) (t fixed,  $k! \rightarrow 0$ ), using (4.42) and (4.53). This leads to the <u>energy inequality</u>:

$$(4.54) |u(t)|^{2} + 2\nu \int_{0}^{t} ||u(s)||^{2} ds \leq |u_{0}|^{2} + 2 \int_{0}^{t} \langle f(s), u(s) \rangle ds, \quad \forall t \in [0,T].$$
If n = 2, using (3.62), it is easy to prove directly the energy equality:
$$(4.55) ||u(t)|^{2} + 2\nu \int_{0}^{t} ||u(s)||^{2} ds = |u_{0}|^{2} + 2 \int_{0}^{t} \langle f(s), u(s) \rangle ds, \quad \forall t \in [0,T].$$

(1) Proof by contradiction.

#### \$5. DISCRETIZATION OF THE NAVIER-STOKES EQUATIONS:

#### I. General Stability and Convergence Theorems.

This section is concerned with a general discussion of the discretization of the evolution Navier-Stokes equations. We study here a full discretization of the equations, both in the space and time variables:

1) The discretization in the space variables appears through the introduction of an external approximation of the space V; for example, one of the approximations (APX 1) to (APX 4), corresponding either to finite differences or finite elements. Actually these particular examples will be discussed in more detail in reference [9].

2) For the discretization in the time variables, we propose, among many natural and classical schemes, four schemes with two levels in time (fully implicit scheme, Cranck-Nicholson scheme, a scheme implicit in the linear part and explicit in its nonlinear part, and a scheme of explicit type).

After the description of the scheme under consideration we proceed to study the stability of these schemes. The problem of stability is the terminology in Numerical Analysis for the problem of getting <u>a priori</u> estimates on the approximate solutions. It is classical that the discretization in both space and time of evolution equations can lead to unstable or conditionally stable schemes: the approximate solutions are unbounded unless the discretization parameters satisfy some restriction. We discuss in full detail the numerical stability of the four schemes considered. To our knowledge the methods used here are nonclassical methods for studying the stability of nonlinear equations. The study of nonlinear instability is a difficult problem; our study here, based on the energy method, leads only to sufficient conditions for stability; the stability conditions which are obtained seem close to being necessary, but the problem of necessary conditions of stability is not studied at all in the text.

The last subject treated in this section is the convergence of the schemes. Two general convergence theorems in suitable spaces are proved for the different schemes. The proof of convergence depends on discrete compactness methods. Owing to the lack of uniqueness of weak solutions in the three dimensional case, the convergence results obtained in the two and three dimensional cases are different, and better, of course, if n = 2.

The repartition of this material throughout subsequent subsections is the following: In Subsection 5.1 we describe the general type of discretization and the numerical schemes which will be studied. In Subsections 5.2, 5.3, and 5.4, we successively study the stability of Schemes 5.1 and 5.2 (Subsection 5.2),

5.3 (Subsection 5.3) and 5.4 (Subsection 5.4). Subsection 5.5 deals with auxiliary a priori estimates of a rather technical character (involving fractional derivatives in time of the approximate functions). Subsection 5.6 contains the description of the consistency hypotheses, the statement of the general convergence theorems, and the proofs of these theorems.

The application of these results to specific approximations of the space V will be treated in reference [9]. There we will also study practical methods for the resolution of the discrete problems. In fact, a more complete account of the numerical study of the nonlinear evolution Navier-Stokes equations is given therein and other methods of approximation include the fractional step or projection method and the artificial compressibility method.

From now on we restrict ourselves to the "concrete" dimensions of space, n = 2 and n = 3.

溪

5.1 Description of the Approximation Schemes.

From now on we will be concerned with the approximation of the solutions of the Navier-Stokes equations in the two and three dimensional cases exclusively,  $\Omega$  being bounded. For simplicity we suppose that the given data, u, f, satisfy

(5.1)  $f \in L^2(0,T;H),$ 

and, as before,

 $(5.2) u_0 \in H.$ 

Theorems 3.1 and 3.2 ensure us that there exists a unique solution of Problem 3.2 if n = 2, and that there exists at least one such solution if n = 3.

Let there be given a stable and convergent external approximation of the space V, say  $\{(V_h, p_h, r_h)_{h \in \mathcal{H}}, (\overline{\omega}, F)\}$ ; the  $V_h$  are assumed to be finite dimensional spaces. This approximation could be any of the approximations (APX 1),..., (APX 4), that were described in Chapter I. For simplicity we assume that

(5.3)  $V_h \subset \mathbb{L}^2(\Omega), \forall h \in \mathcal{H},$ 

a condition which is realized by all the previous approximations. The space  $V_h$  is therefore equipped with two norms: the norm  $|\cdot|$  induced by  $\mathbb{L}^2(\Omega)$  and its own norm  $\|\cdot\|_h$ . Since  $V_h$  is finite dimensional, these norms must be equivalent; the quotient of the two norms is bounded by a constant which may depend on h. Therefore we assume more precisely that

$$(5.4) |u_h| \leq d_0 ||u_h||_h, \quad \forall u_h \in V_h$$

d, independent of h, and

(5.5) 
$$\| u_h \|_h \leq S(h) \| u_h \|, \quad \forall u_h \in V_h$$

The constant S(h), which usually depends on h, plays an important role in the study of the stability of the numerical approximation; for this reason S(h) is sometimes called the stability constant. Usually  $S(h) \rightarrow +\infty$ , as  $h \rightarrow 0$ .

Let there be given a trilinear continuous form on  $V_h$ , say  $b_h(u_h, v_h, w_h)$ , which satisfies

(5.6) 
$$b_h(u_h, v_h, v_h) = 0 \quad \forall u_h, v_h \in V_h,$$

(5.7) 
$$|b_{h}(u_{h}, v_{h}, w_{h})| \leq d_{1} ||u_{h}||_{h} ||v_{h}||_{h} ||w_{h}||_{h},$$

 $\forall u_h, v_h, w_h \in V_h$ , (d<sub>1</sub> independent of h),

and some further properties which will be announced when needed (i.e., when discussing the stability and the convergence of the schemes).

$$(5.8) k = T/N.$$

As in Section 4, we associate with k and the functions f, the elements  $f^1, \dots, f^N$ :

(5.9) 
$$f^{m} = \frac{1}{k} \int_{(m-1)k}^{mk} f(t) dt, \quad m = 1, \dots, N; \quad f^{m} \in \mathbb{L}^{2}(\Omega).$$

We will describe and study four basic schemes chosen from among a large class of interesting and sometimes classical schemes which have been proposed for the Navier-Stokes equations.

For all the four schemes we define recursively for each h and k a family of elements  $u_h^0, \dots, u_h^N$ , of V. Actually these elements depend on h,k (and the data), and should be denoted  $u_{hk}^m$ ; nevertheless, for simiplicity we do not emphasize this double dependence.

In each of the four schemes, we start the recurrence with

(5.10) 
$$u_h^0 = \underline{the orthogonal projection of } u_0 \underline{onto } V_h, \underline{in } \mathbb{L}^2(\Omega);$$
  
this definition makes sense by (5.3) and we immediately observe that

$$(5.11) |u_h^0| \leq |u_0|, \ \forall h.$$

Scheme 5.1. When  $u_h^0, \dots, u_h^{m-1}$ , are known,  $u_h^m$  is the solution in  $V_h$  of  $(5.12) \quad \frac{1}{k} (u_{h}^{m} - u_{h}^{m-1}, v_{h}) + v((u_{h}^{m}, v_{h}))_{h} + b_{h}(u_{h}^{m-1}, u_{h}^{m}, v_{h}) = (f^{m}, v_{h}), \quad \forall v_{h} \in V_{h}.$ Scheme 5.2. When  $u_h^0, \dots, u_h^{m-1}$ , are known,  $u_h^m$  is the solution in  $V_h$  of (5.13)  $\frac{1}{k} (u_{b}^{m} - u_{b}^{m-1}, v_{b}) + \frac{v}{2} ((u_{b}^{m-1} + u_{b}^{m}, v_{b}))_{b} + \frac{1}{2} b_{b} (u_{b}^{m-1}, u_{b}^{m-1} + u_{b}^{m}, v_{b})$ =  $(f^m, v_h), \forall v_h \in V_h.$ <u>Scheme 5.3</u>. <u>When</u>  $u_h^c, \dots, u_h^{m-1}$ , are known,  $u_h^m$  is the solution in  $V_h$  of  $(5.14) \frac{1}{k} (u_{h}^{m} - u_{h}^{m-1}, v_{h}) + v((u_{h}^{m}, v_{h}))_{h} + b_{h}(u_{h}^{m-1}, u_{h}^{m-1}, v_{h}) = (f^{m}, v_{h}), \quad \forall v_{h} \in V_{h}.$ 

<u>Scheme 5.4</u>. <u>When</u>  $u_h^0, \dots, u_h^{m-1}$ , are known,  $u_h^m$  is the solution in  $V_h$  of  $(5.15) \quad \frac{1}{\nu} (u_{h}^{m} - u_{h}^{m-1}, v_{h}) + \nu((u_{h}^{m-1}, v_{h}))_{h} + b_{h}(u_{h}^{m-1}, u_{h}^{m-1}, v_{h}^{*}) = (f_{h}^{m}, v_{h}), \forall v_{h} \in V_{h}.$ 

For all the schemes, the equation defining  $u_h^m$  is equivalent to a linear equation of the form

(5.16) 
$$a_h(u_h^m, v_h) = L_h(v_h), \forall v_h \in V_h,$$

L<sub>b</sub> depends on m, a<sub>b</sub> depends on m for Schemes 5.1 and 5.2, but not in the case of Schemes 5.3 and 5.4.

We observe, in all cases, that

(5.17) 
$$a_h(v_h, v_h) \ge \frac{1}{k} |v_h|^2,$$

and therefore the existence and uniqueness of the solution of (5.16) is a consequence of the Projection Theorem (Theorem I.2.2).

#### Remark 5.1.

(i) The computation of  $u_h^m$  requires the inversion of a matrix;

- the matrix is positive definite, nonsymmetric, and depends on m for Schemes 5.1 and 5.2,

- the matrix is positive definite, symmetric, and does not depend on m for Schemes 5.3 and 5.4.

(ii) Scheme 5.1 is the standard fully implicit scheme; Scheme 5.2 is an interpretation of the classical Cranck-Nicholson scheme. Scheme 5.3 is a partly implicit scheme, implicit only in the linear part of the operator.

(iii) Scheme 5.4 is an explicit scheme or more precisely an interpretation of the so-called explicit schemes; this terminology is justified by the fact that this type of scheme usually gives  $u_h^m$  explicitly, that is to say without inverting any matrix. In the present case, due to the discrete condition div u = 0 built in the space  $V_h$ , the determination of  $u_h^m$  necessitates an inversion of a matrix. This restricts considerably the interest of this scheme, but we considered it of interest nevertheless.

(iv) Besides this discussion on the type of scheme, the reader is referred to reference [9] for practical methods of computation of the  $u_{L}^{m}$ .

Remark 5.2: Related Schemes.

(i) A related form of Schemes 5.1 and 5.2 is a nonlinear form of these schemes:

$$(5.18) \quad \frac{1}{k} (u_{h}^{m} - u_{h}^{m-1}, v_{h}) + \vee ((u_{h}^{m}, v_{h}))_{h} + b_{h} (u_{h}^{m}, u_{h}^{m}, v_{h}) = (f^{m}, v_{h}), \quad \forall v_{h} \in V_{h}.$$

Scheme 5.2'.

$$\frac{1}{k} (u_{h}^{m} - u_{h}^{m-1}, v_{h}) + \frac{\nu}{2} ((u_{h}^{m-1} + u_{h}^{m}, v_{h}))_{h} + \frac{1}{4} b_{h} (u_{h}^{m-1} + u_{h}^{m}, u_{h}^{m-1} + u_{h}^{m}, v_{h})$$
(5.19)
$$= (f^{m}, v_{h}), \quad \forall v_{h} \in V_{h}.$$

(ii) A related form of Scheme 5.3 is a Cranck-Nicholson scheme implicit in its linear part:<u>Scheme 5.3</u>'.

(5.20) 
$$\frac{1}{k} (u_{h}^{m} - u_{h}^{m-1}, v_{h}) + \frac{\nu}{2} ((u_{h}^{m-1} + u_{h}^{m}, v_{h}))_{h} + b_{h} (u_{h}^{m-1}, u_{h}^{m-1}, v_{h}) \\ = (f^{m}, v_{h}), \quad \forall v_{h} \in V_{h}.$$

(iii) These Schemes could be studied by exactly the same methods as Schemes 5.1 - 5.4.

5.2 <u>Stability of Schemes 5.1 and 5.2</u>.

The problem is to prove some a priori estimates on the approximate solution. 5.2.1 <u>Scheme 5.1</u>.

- Lemma 5.1. The solutions  $u_h^m$  of (5.12) remain bounded in the following sense:
  - (5.21)  $|u_{h}^{m}|^{2} \leq d_{2}, m = 0, \dots, N,$ (5.22)  $\sum_{m=1}^{N} |u_{h}^{m} - u_{h}^{m-1}|^{2} \leq d_{2},$

(5.23) 
$$k \cdot \sum_{m=1}^{N} \|u_{h}^{m}\|^{2} \leq \frac{1}{\nu} d_{2},$$

where

(5.24) 
$$d_{2} = |u_{0}|^{2} + \frac{d_{0}^{2}}{\nu} \int_{0}^{T} |f(s)|^{2} ds.$$

Proof.  
We take 
$$v_h = u_h^m$$
 in (5.12). Due to (5.6) and the identity  
(5.25) 2(a-b,a) =  $|a|^2 - |b|^2 + |a-b|^2$ , ∀a,b ∈ L<sup>2</sup>(Ω),

we obtain

$$|u_{h}^{m}|^{2} - |u_{h}^{m-1}|^{2} + |u_{h}^{m} - u_{h}^{m-1}|^{2} + 2kv||u_{h}^{m}||_{h}^{2}$$

$$= 2k(f^{m}, u_{h}^{m})$$

$$\leq 2k|f^{m}| |u_{h}^{m}| \leq (by (5.4))$$

$$\leq 2kd_{0}|f^{m}| ||u_{h}^{m}||_{h}$$

$$\leq kv||u_{h}^{m}||_{h}^{2} + \frac{kd_{0}^{2}}{\nu}|f^{m}|^{2}.$$

Hence

$$(5.27) |u_{h}^{m}|^{2} - |u_{h}^{m-1}|^{2} + |u_{h}^{m} - u_{h}^{m-1}|^{2} + kv ||u_{h}^{m}||_{h}^{2} \leq \frac{kd_{0}^{2}}{v} |f^{m}|^{2}, m = 1, \cdots, N.$$

Adding these inequalities for  $m = 1, \dots, N$ , we get

265

$$(5.28) |u_{h}^{N}|^{2} + \sum_{m=1}^{N} |u_{h}^{m} - u_{h}^{m-1}|^{2} + k\nu \sum_{m=1}^{N} ||u_{h}^{m}||_{h}^{2} \le |u_{h}^{0}|^{2} + \frac{kd^{2}}{\nu} \sum_{m=1}^{N} ||f^{m}|^{2}$$

One can check as in Lemma 4.5 that

(5.29) 
$$k \sum_{m=1}^{N} |f^{m}|^{2} \leq \int_{0}^{T} |f(s)|^{2} ds;$$

thus, by (5.11) and (5.29), it follows that the right-hand side of (5.28) is bounded by

(5.30) 
$$d_{2} = |u_{0}|^{2} + \frac{d_{0}^{2}}{\nu} \int_{0}^{T} |f(s)|^{2} ds.$$

This proves (5.22) and (5.23).

We then add the inequalities (5.27) for  $m = 1, \dots, r$ , dropping some positive terms, we get

$$\begin{aligned} |\mathbf{u}_{h}^{\mathbf{r}}|^{2} &\leq |\mathbf{u}_{h}^{0}|^{2} + \frac{\mathrm{kd}_{0}^{2}}{\nu} \sum_{m=1}^{\mathbf{r}} |\mathbf{f}^{m}|^{2} \\ &\leq (\mathrm{due \ to \ the \ above}) \leq \mathrm{d}_{2}; \end{aligned}$$

(5.21) is proved too.

5.2.2 <u>Scheme</u> 5.2.

 $\underline{\text{Lemma } 5.2}.$   $\underline{\text{The solutions } u_{h}^{m}} \quad \underline{\text{of } (5.13) \text{ remain bounded in the following sense:}}$   $(5.31) \quad |u_{h}^{m}|^{2} \leq d_{2}, \quad m = 1, \cdots, N,$   $(5.32) \quad k \sum_{m=1}^{N} \|\frac{u_{h}^{m} + u_{h}^{m-1}}{2}\|_{h}^{2} \leq \frac{d_{2}}{\nu},$ 

with the same  $d_2$  as (5.24).

$$\frac{Proof}{We take } v_{h} = u_{h}^{m} + u_{h}^{m-1} \text{ in (5.13). Due to (5.6) we find} \\ |u_{h}^{m}|^{2} - |u_{h}^{m-1}|^{2} + 2kv \| \frac{u_{h}^{m} + u_{h}^{m-1}}{2} \|_{h}^{2} \\ = k(f^{m}, u_{h}^{m} + u_{h}^{m-1}) \\ (5.33) \\ \leq 2kd_{0} |f^{m}| \| \frac{u_{h}^{m} + u_{h}^{m-1}}{2} \| \\ \leq kv \| \frac{u_{h}^{m} + u_{h}^{m-1}}{2} \|_{h}^{2} + k \frac{d_{0}^{2}}{v} |f^{m}|^{2}. \end{cases}$$

Therefore

(5.34) 
$$|u_{h}^{m}|^{2} + kv \| \frac{u_{h}^{m+u_{h}^{m-1}}}{2} \|_{h}^{2} \leq |u_{h}^{m-1}|^{2} + \frac{kd_{0}^{2}}{v} |f^{m}|^{2}.$$

We add these relations for  $m = 1, \dots, N$  and get

$$\begin{split} |u_{h}^{N}|^{2} + kv \sum_{m=1}^{N} \|\frac{u_{h}^{m} + u_{h}^{m-1}}{2}\|_{h}^{2} \\ &\leq |u_{h}^{0}|^{2} + \frac{kd_{0}^{2}}{v} \sum_{m=1}^{N} |f^{m}|^{2} \\ &\leq (\text{as before}) \leq d_{2}. \end{split}$$

This proves (5.32); adding then the relations (5.34) for  $m = 1, \dots, r$ , and dropping the unnecessary terms, we find

$$|u_{h}^{r}|^{2} \leq |u_{h}^{0}|^{2} + \frac{kd_{0}^{2}}{v} \sum_{m=1}^{r} |f^{m}|^{2} \leq d_{2};$$

this implies (5.31).

5.2.3 Stability Theorems.

We recall first a definition:

Definition 5.1.

An infinite set of functions  $\mathcal{E}$  is called  $L^{p}(0,T;X)$  stable if and only if  $\mathcal{E}$  is a bounded subset of  $L^{p}(0,T;X)$ .

It is interesting to deduce from the previous estimations some stability results.

In order to state these results, we introduce the approximate functions  $u_h$ 

(5.35)  $u_{h}: [0,T) \mapsto V_{h},$ (5.36)  $\begin{cases} u_{h}(t) = u_{h}^{m}, \quad (m-1)k \leq t \leq mk \quad (\text{Scheme 5.1}) \\ u_{h}(t) = \frac{u_{h}^{m} + u_{h}^{m-1}}{2}, \quad (m-1)k \leq t \leq mk \quad (\text{Scheme 5.2}), \quad m = 1, \dots, N. \end{cases}$ 

Due to Lemmas 5.1 and 5.2,

(5.37)  

$$\sup_{t \in [0,T]} |u_{h}(t)| \leq \sqrt{d_{2}},$$
(5.38)  

$$\int_{0}^{T} ||u_{h}(t)||_{h}^{2} dt \leq \frac{d_{2}}{\nu}.$$

Since the prolongation operators  $p_h \in \mathcal{L}^{(V_h,F)}$  are stable, we have

(5.39)  $\| \mathbf{p}_{h} \mathbf{u}_{h} \|_{F} \leq d_{3} \| \mathbf{u}_{h} \|_{h}, \forall \mathbf{u}_{h} \in V_{h}, (d_{3} \text{ independent of } h).$ 

We infer from (5.38) that

$$\int_{0}^{T} \|p_{h}u_{h}(t)\|_{F}^{2} dt \leq \frac{d_{3}^{2}d_{2}}{\nu}$$

These remarks enable us to state the stability theorem:

#### Theorem 5.1.

The functions  $u_h$ ,  $h \in \mathcal{H}$ , corresponding to Schemes 5.1 and 5.2 are unconditionally  $L^{\infty}(0,T;\mathbb{L}^2(\Omega))$  stable; the functions  $p_h u_h$  are unconditionally  $L^2(0,T;F)$  stable.

#### Remark 5.2.

The majoration (5.22), and similar majorations for the other schemes which we will give later on, does not correspond to stability results but will be technically useful for the proof of the convergence of the scheme.

For the same majoration for Scheme 5.2, see Subsection 5.4.3.

5.3 <u>Stability of Scheme 5.3</u>. We infer from (5.5) and (5.7) that

$$(5.40) | \mathbf{b}_{h}(\mathbf{u}_{h},\mathbf{u}_{h},\mathbf{v}_{h})| \leq d_{1} ||\mathbf{u}_{h}||_{h}^{2} ||\mathbf{v}_{h}||_{h} \leq d_{1} S^{2}(h) ||\mathbf{u}_{h}|| ||\mathbf{u}_{h}||_{h} ||\mathbf{v}_{h}||, \forall \mathbf{u}_{h},\mathbf{v}_{h} \in \mathbf{V}_{h}.$$

Sometimes this relation can be improved and this means an important improvement of some restrictive conditions of stability which will appear later on in this section; for this reason we will assume that

(5.41) 
$$|b_{h}(u_{h}, u_{h}, v_{h})| \leq S_{1}(h) |u_{h}| ||u_{h}||_{h} ||v_{h}|, \forall u_{h}, v_{h} \in V_{h},$$

where at least

(5.42) 
$$S_1(h) \le d_1 S^2(h)$$
.

5.3.1 <u>A Priori Estimates</u>.

<u>Lemma 5.3</u>.

We assume that k and h satisfy

(5.43)  $kS_1^2(h) \le d', kS^2(h) \le d''$  (1)

<sup>(1)</sup>In practice, one of these relations should be a consequence of the other (this depends on the explicit values of S and  $S_1$ ).

where d' and d" are some constants depending on the data and are estimated in the course of the proof.

Then, the $u_h^m$ given by(5.14)remainbounded inthefollowing sense:(5.44) $|u_h^m| \leq d_4$ , $m = 0, \dots, N$ (5.45) $\sum_{m=1}^{N} |u_h^m - u_h^{m-1}|^2 \leq d_4$ ,(5.46) $k \sum_{m=1}^{N} ||u_h^m||_h^2 \leq d_4$ ,

where d<sub>4</sub> is some constant depending only on the data, d', and d".

Proof.

We write (5.14) with 
$$v_h = u_h^m$$
. Using again (5.25), we obtain the relation  
(5.47)  $|u_h^m|^2 - |u_h^{m-1}|^2 + |u_h^m - u_h^{m-1}|^2 + 2kv ||u_h^m||_h^2$   
 $= -2kb_h(u_h^{m-1}, u_h^{m-1}, u_h^m) + 2k(f^m, u_h^m).$ 

Due to (5.6) the right-hand side of (5.47) is equal to

$$-2kb_{h}(u_{h}^{m-1}, u_{h}^{m-1}, u_{h}^{m}-u_{h}^{m-1}) + 2k(f^{m}, u_{h}^{m});$$

this expression is less than (cf. (5.4) and (5.41)):

$$2kS_{1}(h) |u_{h}^{m-1}| ||u_{h}^{m-1}||_{h} ||u_{h}^{m}-u_{h}^{m-1}| + 2kd_{0}|f^{m}| ||u_{h}^{m}||_{h} ,$$
  
$$kv ||u_{h}^{m}||_{h}^{2} + 2k^{2}S_{1}^{2}(h) ||u_{h}^{m-1}||^{2} ||u_{h}^{m-1}||_{h}^{2} + \frac{1}{2} ||u_{h}^{m}-u_{h}^{m-1}||^{2} + \frac{kd_{0}^{2}}{\nu} ||f^{m}|^{2}.$$

Therefore

(5.48) 
$$|u_{h}^{m}|^{2} - |u_{h}^{m-1}|^{2} + \frac{1}{2} |u_{h}^{m} - u_{h}^{m-1}|^{2} + kv ||u_{h}^{m}||_{h}^{2}$$
$$-2k^{2}S_{1}^{2}(h) |u_{h}^{m-1}|^{2} ||u_{h}^{m-1}||_{h}^{2} \leq \frac{kd_{0}^{2}}{\nu} |f^{m}|^{2}.$$

We add these inequalities for m = 1, ..., r;

(5.49) 
$$|u_{h}^{r}|^{2} + \frac{1}{2} \sum_{m=1}^{r} |u_{h}^{m} - u_{h}^{m-1}|^{2} + kv \sum_{m=1}^{r} ||u_{h}^{m}||_{h}^{2}$$
$$-2k^{2}S_{1}^{2}(h) \sum_{m=2}^{r} |u_{h}^{m-1}|^{2} ||u_{h}^{m-1}||_{h}^{2} \leq \lambda_{m},$$

269

(5.50) 
$$\lambda_{m} = |u_{h}^{0}|^{2} + \frac{kd_{0}^{2}}{v} \sum_{m=1}^{1} |f^{m}|^{2} + 2k^{2}S_{1}^{2}(h)|u_{h}^{0}|^{2} ||u_{h}^{0}||_{h}^{2}$$

Let us assume that

(5.51) 
$$2kS_1^2(h)\lambda_N \leq \nu - \delta$$
, for some fixed  $\delta$ ,  $0 < \delta < \nu$ .

If this inequality holds, it is easy to show recursively that

(5.52) 
$$|u_{h}^{r}|^{2} + \frac{1}{2} \sum_{m=1}^{r} |u_{h}^{m} - u_{h}^{m-1}|^{2} + k\delta \sum_{m=1}^{r} ||u_{h}^{m}||_{h}^{2} \leq \lambda_{r}, r = 1, \cdots, N.$$

Indeed the relation (5.48) written with m = 1, shows us that (5.52) is true for r = 1. Let us assume then that (5.52) is valid up to the order r-1, and let us show this relation for the integer r.

We observe that, by assumption,

(5.53) 
$$|u_h^m|^2 \leq \lambda_m \leq \lambda_N, m = 1, \cdots, r-1;$$

therefore, by (5.51),

$$2k^{2}S_{1}^{2}(h) \sum_{m=2}^{r} |u_{h}^{m-1}|^{2} ||u_{h}^{m-1}||_{h}^{2}$$

$$\leq 2k^{2}S_{1}^{2}(h)\lambda_{N} \sum_{m=2}^{r} ||u_{h}^{m-1}||_{h}^{2}$$

$$\leq 2k^{2}S_{1}^{2}(h)\lambda_{N} \sum_{m=1}^{r} ||u_{h}^{m}||_{h}^{2}$$

$$\leq k(\nu-\delta) \sum_{m=1}^{r} ||u_{h}^{m}||_{h}^{2}.$$

Putting this majoration into (5.49), we get (5.52) for the integer r.

The proof is complete if we show that a condition of the type (5.43) ensures (5.51).

According to a majoration used in Lemmas 5.1 and 5.2 (see (5.11), (5.29))

$$\begin{split} \lambda_{\rm N} &\leq \|u_0\|^2 + \frac{d_0^2}{\nu} \int_0^{\rm T} \|f(s)\|^2 ds + 2k^2 S_1^2(h) \|u_0\|^2 \|u_h^0\|_h^2 \\ &\leq (\text{see (5.5) and (5.24)}) \\ &\leq d_2 + 2k^2 S_1^2(h) S^2(h) \|u_0\|^2 \,. \end{split}$$

Hence, if (5.43) is satisfied,

$$2kS_{1}^{2}(h)\lambda_{N} \leq 2d'(d_{2}+2d'd''|u_{0}|^{2})$$

and this is certainly bounded by  $\nu-\delta$  if d' and d" are sufficiently small:

(5.54) 
$$2d'(d_2+2d'd''|u_0|^2) \leq v-\delta.$$

The proof is complete.

5.3.2 The Stability Theorem.

We define for the Scheme 5.3 the approximate functions  $u_h$  by:

$$(5.55) u_h: [0,T) \longrightarrow V_h$$

$$u_{h}(t) = u_{h}^{m}, \quad (m-1)k \le t \le mk, \quad m = 1, \dots, N.$$

We infer from (5.39), (5.44), (5.45) that if (5.43) holds then

$$\begin{aligned} \sup_{t \in [0,T]} & |u_{h}(t)| \leq \sqrt{d}_{4}, \\ & t \in [0,T] \end{aligned}$$

and thus

Theorem 5.2.

The functions  $u_h$  and  $p_h u_h$ ,  $h \in \mathcal{H}$ , corresponding to the Scheme 5.3 are respectively  $L^{\infty}(0,T;\mathbb{L}^2(\Omega))$  and  $L^2(0,T;F)$  stable, provided k and h remain connected by (5.43).

Definition 5.2.

<u>Conditions such as (5.43) are called stability conditions.</u> They are sufficient conditions ensuring the stability of the scheme. A scheme is called conditionally or unconditionally stable according to whether such a condition occurs or not in proving stability.

5.4 Stability of Scheme 5.4.

5.4.1 <u>A Priori Estimates</u>.

<u>Lemma 5.4</u>.

We assume that k and h satisfy

(5.56)  $kS^{2}(h) \leq \frac{1-\delta}{4\nu}$ , for some  $\delta$ ,  $0 < \delta < 1$ ,

and

- 2	к	53	Ł.
- 5	80	15	
	۰.	14	Ε.

where

(5.58) 
$$d_{5} = |u_{0}|^{2} + (\frac{d_{0}^{2}}{v} + 4T) \int_{0}^{T} |f(s)|^{2} ds.$$

Then the  $u_h^m$  given by (5.15) remain bounded in the following sense: (5.59)  $|u_h^m|^2 \leq d_5, m = 1, \dots, N$ 

(5.60) 
$$k \sum_{m=1}^{N} \|u_{h}^{m}\|_{h}^{2} \leq \frac{2d_{5}}{\delta v}$$

(5.61) 
$$\sum_{m=1}^{N} |u_{h}^{m} - u_{h}^{m-1}|^{2} \leq \frac{d_{5}}{\delta} (2-\delta) + 4T \int_{0}^{T} |f(s)|^{2} ds.$$

Proof.

We replace 
$$v_h$$
 by  $u_h^{m-1}$  in (5.15); due to the identity  
(5.62)  $2(a-b,b) = |a|^2 - |b|^2 - |a-b|^2$ ,  $\forall a,b \in \mathbb{L}^2(\Omega)$ ,

we find

$$\begin{aligned} |u_{h}^{m}|^{2} - |u_{h}^{m-1}|^{2} &- |u_{h}^{m} - u_{h}^{m-1}|^{2} + 2kv \|u_{h}^{m-1}\|_{h}^{2} \\ &= 2k(f^{m}, u_{h}^{m}) \\ &\leq 2kd_{0} \|f^{m}\| \|u_{h}^{m}\|_{h} \\ &\leq \nu k \|u_{h}^{m}\|_{h}^{2} + \frac{kd_{0}^{2}}{\nu} \|f^{m}\|^{2} , \end{aligned}$$

$$(5.63) \qquad |u_{h}^{m}|^{2} - |u_{h}^{m-1}|^{2} - |u_{h}^{m} - u_{h}^{m-1}|^{2} + kv \|u_{h}^{m-1}\|_{h}^{2} \leq \frac{kd_{0}^{2}}{\nu} \|f^{m}\|. \end{aligned}$$

The difference with the preceding is that the term  $|u_h^m - u_h^{m-1}|^2$  on the lefthand side is affected with a minus sign and so, we must majorize it.

hand side is affected with a minus sign and so, we must majorize it. In order to majorize  $|u_{h}^{m}-u_{h}^{m-1}|^{2}$ , we write (5.15) with  $v = u_{h}^{m}-u_{h}^{m-1}$ . This gives

(5.64)  
$$2|u_{h}^{m}-u_{h}^{m-1}|^{2} = -2kv((u_{h}^{m-1},u_{h}^{m}-u_{h}^{m-1}))_{h}$$
$$-2kb_{h}(u_{h}^{m-1},u_{h}^{m-1},u_{h}^{m}-u_{h}^{m-1}) + 2k(f^{m},u_{h}^{m}-u_{h}^{m-1}).$$

We successively majorize all the terms on the right-hand side, using repeatedly (5.5), (5.41), and the Schwarz inequality:

$$\begin{aligned} -2kv((u_{h}^{m-1}, u_{h}^{m} - u_{h}^{m-1}))_{h} &\leq 2kv \| u_{h}^{m-1} \|_{h} \| u_{h}^{m} - u_{h}^{m-1} \|_{h} \\ &\leq 2kvS(h) \| u_{h}^{m-1} \|_{h} \| |u_{h}^{m} - u_{h}^{m-1}|_{h} \\ &\leq \frac{1}{4} | |u_{h}^{m} - u_{h}^{m-1}|_{h}^{2} + 4k^{2}v^{2}S^{2}(h) \| |u_{h}^{m-1} \|_{h}^{2}, \\ -2kb_{h}(u_{h}^{m-1}, u_{h}^{m-1}, u_{h}^{m} - u_{h}^{m-1}) \cdot \\ &\leq 2kS_{1}(h) | |u_{h}^{m-1}| \| ||u_{h}^{m-1} \|_{h} \| ||u_{h}^{m} - u_{h}^{m-1}| \\ &\leq \frac{1}{4} | |u_{h}^{m} - u_{h}^{m-1}|^{2} + 4k^{2}S_{1}^{2}(h) | |u_{h}^{m-1}|^{2} ||u_{h}^{m-1} \|_{h}^{2} \\ &\leq 2k(f^{m}, u_{h}^{m} - u_{h}^{m-1}) \leq 2k|f^{m}| | ||u_{h}^{m} - u_{h}^{m-1}| \\ &\leq \frac{1}{4} | |u_{h}^{m} - u_{h}^{m-1}|^{2} + 4k^{2}|f^{m}|^{2}. \end{aligned}$$

,

Therefore (5.64) becomes

$$\|u_{h}^{m}-u_{h}^{m-1}\|^{2} \leq 4k^{2}v^{2}S^{2}(h)\|u_{h}^{m-1}\|_{h} + 4k^{2}S_{1}^{2}(h)\|u_{h}^{m-1}\|^{2}\|u_{h}^{m-1}\|_{h}^{2} + 4k^{2}\|f^{m}\|^{2}$$

(5.65)

$$\leq (by (5.56))$$

$$\leq kv(1-\delta) \|u_{h}^{m-1}\|_{h}^{2} + 4k^{2}S_{1}^{2}(h) |u_{h}^{m-1}|^{2} \|u_{h}^{m-1}\|_{h}^{2} + 4k^{2} |f^{m}|^{2};$$

for (5.63) we then have

(5.66)  
$$\begin{aligned} |u_{h}^{m}|^{2} - |u_{h}^{m-1}|^{2} + k(v\delta - 4kS_{1}^{2}(h)|u_{h}^{m-1}|^{2})||u_{h}^{m-1}||_{h}^{2} \\ & \leq k(\frac{d_{0}^{2}}{v} + 4k)|f^{m}|^{2} \\ & \leq (since \ k \leq T) \\ & \leq k(\frac{d_{0}^{2}}{v} + 4T)|f^{m}|^{2}. \end{aligned}$$

Summing these inequalities for  $m = 1, \dots, r$ , we arrive at

(5.67) 
$$|u_{h}^{r}|^{2} + k \sum_{m=1}^{r} (v\delta - 4kS_{1}^{2}(h)|u_{h}^{m-1}|^{2})||u_{h}^{m-1}||_{h}^{2} \leq \mu_{r},$$

where

(5.68) 
$$\mu_{\mathbf{r}} = |\mathbf{u}_{\mathbf{h}}^{0}|^{2} + k(\frac{d_{0}^{2}}{\nu} + 4\mathbf{T}) \sum_{m=1}^{\mathbf{r}} |\mathbf{f}^{m}|^{2}.$$

273

Using (5.57) we will now prove recursively that

(5.69) 
$$|u_{h}^{r}|^{2} + \frac{k\nu\delta}{2}\sum_{m=1}^{r} ||u_{h}^{m-1}||_{h}^{2} \leq \mu_{r}, r = 1, \cdots, N.$$

We observe first that

(5.70) 
$$\mu_{r} \leq \mu_{N} = |u_{h}^{0}|^{2} + k(\frac{d_{0}^{2}}{\nu} + 4T) \sum_{m=1}^{N} |f^{m}|^{2} \\ \leq |u_{0}|^{2} + (\frac{d_{0}^{2}}{\nu} + 4T) \int_{0}^{T} |f(s)|^{2} ds = d_{5}.$$

The relation (5.69) is obvious for r = 1; writing (5.66) for m = 1 and using (5.57) we get

$$\|\mathbf{u}_{h}^{1}\|^{2} + k v \delta \|\mathbf{u}_{h}^{0}\|_{h}^{2} \leq \|\mathbf{u}_{h}^{0}\|^{2} + k (\frac{d_{0}^{2}}{v} + 4T) \|\mathbf{f}^{1}\|^{2} + 4k^{2} S_{1}^{2}(h) \|\mathbf{u}_{h}^{0}\|^{2} \|\mathbf{u}_{h}^{0}\|_{h}^{2} \leq \mu_{1} + \frac{v \delta}{2} \|\mathbf{u}_{h}^{0}\|_{h}^{2},$$

which is (5.69) for r = 1.

Assuming now that the relation (5.69) holds up to the order r-1, we will prove it at the order r. In fact by the recurrence hypothesis,

(5.71)  $|u_{h}^{r-1}|^{2} \leq \mu_{r-1} \leq \mu_{N} \leq (by (5.70)) \leq d_{5}.$ 

Hence (5.67) gives

(5.72) 
$$\|u_{h}^{r}\|^{2} + kv\delta \sum_{m=1}^{r} \|u_{h}^{m-1}\|_{h}^{2} \leq \mu_{r} + 4k^{2}S_{1}^{2}(h)d_{5} \sum_{m=1}^{r} \|u_{h}^{m-1}\|_{h}^{2}$$
  
$$\leq \mu_{r} + \frac{kv\delta}{2} \sum_{m=1}^{r} \|u_{h}^{m-1}\|_{h}^{2},$$

and (5.69) at the order r follows.

It remains to prove (5.61). For this we return to (5.65); using (5.56),(5.57), we get

(5.73) 
$$|u_{h}^{m}-u_{h}^{m-1}|^{2} \leq k v (1 - \frac{\delta}{2}) ||u_{h}^{m-1}||_{h}^{2} + 4kT |f^{m}|^{2}.$$

By summation and using (5.29), we find (5.61).

5.4.2 The Stability Theorem.  
We now set  
(5.74) 
$$u_h: [0,T) \mapsto V_h$$

(5.75) 
$$u_h(t) = u_h^{m-1}, (m-1)k \le t \le mk, m = 1, \dots, N_n$$

and we have

<u>The functions</u>  $u_h$  and  $p_h u_h$ ,  $h \in \mathcal{H}$ ; corresponding to Scheme 5.4 are respectively  $L^{\infty}(0,T;L^2(\Omega))$  and  $L^2(0,T;F)$  stable, provided k and h remain connected by (5.56) - (5.58).

## 5.5 <u>A</u> <u>Complementary</u> <u>Estimate</u> for <u>Scheme</u> <u>5.2</u>.

Using the techniques extensively applied in Sections 5.3 and 5.4, we can complete Section 5.2 by giving, in the case of Scheme 5.2, an estimation similar to the estimation

(5.76) 
$$\sum_{m=1}^{N} |u_{h}^{m} - u_{h}^{m-1}|^{2} \leq \text{Const.}$$

that we proved for Schemes 5.1, 5.3, and 5.4. As mentioned in Remark 5.2, these estimations will be useful for the proof of the convergence.

$$\frac{\text{Lemma } 5.5}{\text{The } u_{h}^{m}} \stackrel{\text{defined by } (5.13)(\underline{\text{Scheme } 5.2}) \underline{\text{satisfy}};$$
(5.77)
$$\sum_{m=1}^{N} |u_{h}^{m} - u_{h}^{m-1}|^{2} \leq d(1 + kS^{4}(h)),$$

where d denotes a constant depending only on the data.

$$\begin{split} \frac{\text{Proof.}}{\text{We take }} & \text{We take } v_h = 2k(u_h^m - u_h^{m-1}) \quad \text{in (5.13)} \quad \text{and obtain} \\ 2|u_h^m - u_h^{m-1}|^2 = -kv||u_h^m||_h^2 + kv||u_h^{m-1}|_h^2 - kb_h(u_h^{m-1}, u_h^m + u_h^{m-1}, u_h^m - u_h^{m-1}) + 2k(f^m, u_h^m - u_h^{m-1}) \\ & \leq (by (5.40) \text{ and } (5.5)) \\ & \leq -kv||u_h^m||_h^2 + kv||u_h^{m-1}||_h^2 + kd_1s^2(h)|u_h^{m-1}| ||u_h^m + u_h^{m-1}||_h ||u_h^m - u_h^{m-1}| \\ & \quad + 2k|f^m| ||u_h^m - u_h^{m-1}| \\ & \leq -kv||u_h^m||_h^2 + kv||u_h^{m-1}||_h^2 + \frac{1}{2}||u_h^m - u_h^{m-1}||^2 + \frac{k^2}{2}d_1^2S^4(h)|u_h^{m-1}||u_h^m + u_h^{m-1}||_h^2 \\ & \quad + \frac{1}{2}||u_h^m - u_h^{m-1}||^2 + 2kT||f^m||^2. \end{split}$$

Thus

$$\begin{split} \sum_{m=1}^{N} \|u_{h}^{m} - u_{h}^{m-1}\|^{2} &\leq k \nu \|u_{h}^{0}\|_{h}^{2} + 2kT \sum_{m=1}^{N} \|f^{m}\|^{2} + \frac{k^{2}}{2} d_{1}^{2} S^{4}(h) \sum_{m=1}^{N} \|u_{h}^{m-1}\|^{2} \|u_{h}^{m} + u_{h}^{m-1}\|_{h}^{2} \\ &\leq (by \ (5.5), (5.11), (5.29), (5.31), (5.32)) \\ &\leq k \nu S^{2}(h) \|u_{0}\|^{2} + 2T \int_{0}^{T} \|f(s)\|^{2} ds + \frac{2}{\nu} d_{1}^{2} d_{2}^{2} k S^{4}(h). \end{split}$$

The proof is complete.

5.6 Other A Priori Estimates.

In order to prove strong convergence results we will establish some further a priori estimates concerning the fractional derivatives in t of approximate functions. This Section 5.6 is essentially a technical section which is used in Section 5.7 where the convergence of the schemes is proved.

For all the four schemes we define  $w_h$ , a function from  $\mathbb{R}$  into  $V_h$ , by:

(5.78) 
$$w_h \underline{is a \text{ continuous function from } \mathbb{R} \text{ into } V_h, \underline{linear \text{ on each}}$$
  
 $(5.78) \underline{interval} [mk, (m+1)k], \underline{and} w_h(mk) = u_h^m, m = 0, \dots, N-1;$   
 $w_h = 0 \underline{outside the interval} [0,T].$ 

Lemma 5.6.

Assuming the same stability conditions as in Theorems 5.1, 5.2, 5.3, <sup>(1)</sup> the Fourier transform  $\hat{w}_h$  of  $w_h$  satisfies

(5.79) 
$$\int_{-\infty}^{+\infty} |\tau|^{2\gamma} |\hat{w}_{h}(\tau)|^{2} d\tau \leq \text{Const., for } 0 < \gamma < \frac{1}{4}$$

where the constant depends on  $\gamma$  and on the data.

#### Proof.

The four equations (5.12) - (5.15) can be interpreted as

(5.80) 
$$\frac{d}{dt} (w_h(t), v_h) = ((g_h(t), v_h))_h, \forall v_h \in V_h, t \in (0, T),$$

where the function g<sub>h</sub> satisfies

(5.81) 
$$\int_0^T \|g_h(t)\|_h dt \leq \text{Const.}$$

For example, for Scheme 5.1,  $g_h$  is defined by

<sup>(1)</sup> No condition for Schemes 5.1, 5.2; conditions (5.43) for Scheme 5.3; conditions (5.56)-(5.57) for Scheme 5.4.

$$((g_{h}(t),v_{h}))_{h} = (f^{m},v_{h}) - b_{h}(u_{h}^{m-1},u_{h}^{m},v_{h}) - \nu((u_{h}^{m},v_{h}))_{h},$$
$$\forall v_{h} \in V_{h}, \quad (m-1)k \leq t < mk.$$

Inequality (5.81) follows from (5.4), (5.7), and the previous a priori estimates:

$$\|g_{h}(t)\|_{h} \leq d_{0} \|f^{m}\| + d_{1} \|u_{h}^{m-1}\|_{h} \|u_{h}^{m}\|_{h} + v \|u_{h}^{m}\|_{h} ,$$

$$\int_{0}^{T} \|g_{h}(t)\|_{h} dt \leq k \sum_{m=1}^{N} (d_{0} \|f^{m}\| + d_{1} \|u_{h}^{m-1}\|_{h} \|u_{h}^{m}\|_{h} + v \|u_{h}^{m}\|_{h} );$$

the right-hand side of this relation is bounded according to Lemma 5.1.

Let us infer (5.79) from (5,80)-(5.81). Extending  $g_h$  by 0 outside [0,T] we get a function  $\tilde{g}_h$  such that the following equality holds on the whole t line:

$$(5.82) \quad \frac{\mathrm{d}}{\mathrm{dt}} \left( w_{\mathrm{h}}(t), v_{\mathrm{h}} \right) = \left( \left( \tilde{g}_{\mathrm{h}}(t), v_{\mathrm{h}} \right) \right)_{\mathrm{h}} + \left( u_{\mathrm{h}}^{0}, v_{\mathrm{h}} \right) \delta_{\mathrm{g}} - \left( u_{\mathrm{h}}^{\mathrm{N}}, v_{\mathrm{h}} \right) \delta_{\mathrm{T}}, \quad \forall v_{\mathrm{h}} \in V_{\mathrm{h}},$$

where  $\boldsymbol{\delta}_0\,,\boldsymbol{\delta}_T$  denote the Dirac distribution at 0 and T.

By taking the Fourier transform, we then have

$$-2i\pi\tau(\hat{w}_{h}(\tau),v_{h}) = ((\hat{g}_{h}(\tau),v_{h}))_{h} + (u_{h}^{0},v_{h}) - (u_{h}^{N},v_{h}) \exp(-2i\pi\tau T);$$

 $(\hat{g}_h = Fourier transform of \tilde{g}_h).$ 

Putting  $v_h = \hat{w}_h(\tau)$  and then taking absolute values we get

$$2\pi |\tau| | \hat{w}_{h}(\tau)|^{2} \leq \|\hat{g}_{h}(\tau)\|_{h} \|\hat{w}_{h}(\tau)\|_{h} + c_{1} |\hat{w}_{h}(\tau)|,$$

since  $u_h^0$  and  $u_h^N$  remain bounded. Due to (5.81) we also have

(T

$$\left\|\hat{g}_{h}(\tau)\right\|_{h} \leq \int_{0}^{1} \left\|g_{h}(t)\right\|_{h} dt \leq \text{Const.} = c_{2},$$

(5.83) 
$$|\tau| |\hat{w}_{h}(\tau)|^{2} \leq c_{3} ||\hat{w}_{h}(\tau)||_{h}^{2}$$

For fixed  $\gamma$ ,  $\gamma < \frac{1}{4}$ , we observe that

$$|\tau|^{2\gamma} \leq c_{4}(\gamma) \frac{1+|\tau|}{1+|\tau|^{1-2\gamma}}, \quad \forall \tau \in \mathbb{R}.$$

Hence

$$\begin{split} \left| \tau \right|^{2\gamma} \left| \hat{w}_{h}^{}(\tau) \right|^{2} d\tau &\leq c_{4}^{}(\gamma) \int_{-\infty}^{+\infty} \frac{1 + |\tau|}{1 + |\tau|^{1 - 2\gamma}} \left| \hat{w}_{h}^{}(\tau) \right|^{2} d\tau \\ &\leq (by \ (5.83)) \\ &\leq c_{4}^{}(\gamma) \int_{-\infty}^{+\infty} \left| \hat{w}_{h}^{}(\tau) \right|^{2} d\tau + c_{5}^{} \int_{-\infty}^{+\infty} \frac{\left\| \hat{w}_{h}^{}(\tau) \right\|_{h}}{1 + |\tau|^{1 - 2\gamma}} \ d\tau \\ &\leq (by \ the \ Schwarz \ inequality) \\ &\leq c_{4}^{} \int_{-\infty}^{+\infty} \left| \hat{w}_{h}^{}(\tau) \right|^{2} d\tau + c_{5}^{} \left( \int_{-\infty}^{+\infty} \frac{d\tau}{(1 + |\tau|^{1 - 2\gamma})^{2}} \right)^{\frac{1}{2}} \\ &\cdot \left( \int_{-\infty}^{+\infty} \left\| \hat{w}_{h}^{}(\tau) \right\|_{h}^{2} \ d\tau \right)^{\frac{1}{2}} . \end{split}$$

The integral

(5.84)

$$\int_{-\infty}^{+\infty} \frac{d\tau}{\left(1+\left|\tau\right|^{1-2\gamma}\right)^{2}}$$

is finite for  $\gamma < \frac{1}{4}$ . Therefore the right-hand side of the last inequality is finite and bounded due to the Parseval relation and the previous estimations:

(5.85) 
$$\int_{-\infty}^{+\infty} |\hat{w}_{h}(\tau)|^{2} d\tau \leq d_{0}^{2} \int_{-\infty}^{+\infty} ||\hat{w}_{h}(\tau)||_{h}^{2} d\tau = d_{0}^{2} \int_{0}^{T} ||w_{h}(t)||_{h}^{2} dt \leq \text{Const.}$$

The lemma follows.

5.7 Convergence of the Numerical Schemes.

Our aim is to prove the convergence of Schemes 5.1 to 5.4, in some sense which will be described later on. We first state the consistency and compactness properties on the discretized data which are required to ensure the convergence. We then state and prove the convergence results.

5.7.1 Consistency and Compactness Hypotheses.

The subsequent hypotheses will be easier to state after this lemma:

Lemma 5.7.

Let  $\{(V_h, p_h, r_h)_{h \in \{\!\!\!\ \ \!\!\!\}}, (\overline{\omega}, F)\}$  denote a stable and convergent external approximation of V. Let us assume that for some sequence  $h' \longrightarrow 0$ , a family of functions

 $u_{h}: [0,T] \mapsto V_{h},$ 

#### satisfies

(5.86)  $p_{h}, u_{h}, \rightarrow \phi \quad in \quad L^{2}(0,T;F) \quad weakly, as \quad h' \rightarrow 0.$ Then for almost every t,  $\phi(t) = \overline{\omega}u(t)$ , and (5.87)  $u = \overline{\omega}^{-1}\phi \in L^{2}(0,T;V).$ 

#### Proof.

Let us denote by  $\theta$  some function in  $\mathcal{D}((0,T))$ . It is easily checked that

$$\int_0^T p_h, u_h, (t)\theta(t)dt \longrightarrow \int_0^T \phi(t)\theta(t)dt,$$

as h' goes to 0. But condition (C2) of Definition I.3.6 <sup>(1)</sup> shows us that, under these circumstances,

$$\int_0^T \phi(t)\theta(t)dt \in \overline{\omega} V.$$

Since by definition,  $\overline{\omega}V$  is isomorphic to V,  $\overline{\omega}V$  is a closed subspace of F; taking now a sequence of functions  $\theta_{\varepsilon}$  converging to the Dirac distribution at the point s, s  $\in$  (0,T), we see that for almost every s in [0,T],

$$\int_0^T \phi(t)\theta_{\varepsilon}(t)dt \longrightarrow \phi(s) \quad \text{in } F,$$

and hence

 $\phi(s) \in \overline{\omega}V$  a.e.

Then, as  $\overline{\omega}$  is an isomorphism,  $\overline{\omega}^{-1}\phi$  is defined and belongs to  $L^2(0,T;V)$ .

The preceding lemma was quite general, but, in the present situation we assumed that

(5.88) 
$$V_{h} \subset \mathbb{L}^{2}(\Omega), \forall h;$$

therefore it can happen that for some sequence  $h' \rightarrow 0$ ,

$$u_{h} \xrightarrow{} u$$
 in  $L^{2}(0,T;L^{2}(\Omega))$  weakly,  
 $p_{h}, u_{h} \xrightarrow{} \phi$  in  $L^{2}(0,T;F)$  weakly.

By Lemma 5.6,  $\phi = \overline{\omega}u_{\star}, u_{\star} \in L^2(0,T;V)$ . Without further information we cannot assert that  $u = u_{\star}$ . But, actually this will be proved for each approximation considered:

(1) Definition of the approximation of a normed space in the general frame.
Let  $u_h$ , be a sequence of functions from [0,T] into  $V_h$ , (5.89)such that, as  $h' \longrightarrow 0$ ,  $u_h, \rightarrow u \quad in \quad L^2(0,T; L^2(\Omega)) \quad weakly,$  $p_h, u_h, \longrightarrow \phi \quad in \quad L^2(0,T;F) \quad weakly.$ Then  $u \in L^2(0,T;V)$  and  $\phi = \overline{\omega}u$ . Besides (5.89), the consistency hypotheses are now the following (1): Let  $v_{h'}, w_{h'}$ , be two sequences of functions from [0,T] into (5.90) $V_{h'}$ , such that, as  $h' \rightarrow 0$ ,  $p_{h}, v_{h}, \longrightarrow \overline{w}v \quad \underline{in} \quad L^{2}(0,T;F) \quad \underline{weakly},$  $p_h, w_h, \rightarrow \overline{\omega} w \quad \underline{in} \quad L^2(0,T;F) \quad (\underline{strongly}).$ <u>Then</u>, as  $h' \rightarrow 0$ ,  $\int_{0}^{T} ((v_{h},(t),w_{h},(t)))_{h} dt \longrightarrow \int_{0}^{T} ((v(t),w(t))) dt.$ Let  $u_{h'}, v_{h'}$ , be two sequences of functions from [0,T] (5.91)<u>into</u>  $V_h$ , such that, as  $h' \rightarrow 0$ ,  $p_{h}, u_{h}, \rightarrow \overline{w}u \quad \underline{in} \quad L^{2}(0,T;F) \quad \underline{weakly},$  $u_{h} \rightarrow u \quad in \quad \mathbb{L}^{2}(Q) \quad strongly, \quad Q = \Omega \times (0,T),$ and  $p_{h}, v_{h}, \longrightarrow \overline{w}v \quad in \quad L^{2}(0,T;F) \quad weakly.$ Then as  $h' \longrightarrow 0$ ,  $\int_{0}^{T} b_{h'}(u_{h'}(t), v_{h'}(t), \psi(t)r_{h'}w_{h'})dt \longrightarrow \int_{0}^{T} b(u(t), v(t), \psi(t)w)dt,$ for each scalar valued function  $\psi \in L^{\infty}(0,T)$  and each  $w \in \mathcal{V}$ . If moreover a sequence of functions  $\psi_k$ , is given with  $\psi_{k}, \longrightarrow \psi \quad \underline{in} \quad L^{\infty}(0,T), \quad \underline{as} \quad k' \longrightarrow 0,$ 

<sup>(1)</sup>Compare with the stationary case, (3.7), (3.8), Chapter I; (3.4), (3.5), (3.7), Chapter II.

$$\frac{\text{then, as } h' \longrightarrow 0, \quad k' \longrightarrow 0,}{\int_0^T b_{h'}(u_{h'}(t), v_{h'}(t), \psi_{k'}(t)r_{h'}w_{h'})dt} \longrightarrow \int_0^T b(u(t), v(t), \psi(t)w)dt.$$

**3** 

In order to prove strong convergence results as those needed in (5.91)  $(u_h, \rightarrow u$  in  $\mathbb{L}^2(\mathbb{Q})$  strongly) we will assume the following:

$$(5.92) \quad \underline{\text{Let } v_h}, \quad \underline{\text{denote a sequence of functions from } \mathbb{R}}$$

$$\underline{\text{into } V_h}, \quad \underline{\text{with support in } [0,T] \quad \underline{\text{and such that}}}$$

$$\int_{0}^{T} \|v_h,(t)\|_{h}^{2}, dt \leq \text{Const.},$$

$$\int_{-\infty}^{+\infty} |\tau|^{2\gamma} |\hat{v}_h,(\tau)|^2 d\tau \leq \text{Const.}, \quad \underline{\text{for some } 0 < \gamma},$$

$$\underline{\text{where } \hat{v}_h \quad \underline{\text{is the Fourier transform of } v_h}.$$

$$\underline{\text{Then such a sequence } v_h, \quad \underline{\text{is relatively compact in } \mathbb{L}^2(Q).}$$

$$\underline{\text{In particular, one can extract from } v_h, \quad \underline{a \text{ subsequence }} (\underline{\text{still denoted } v_h}, \underline{) \quad \underline{\text{with}}}$$

$$p_h, v_h, \rightarrow \overline{\text{wv}} \quad \underline{\text{in } \mathbb{L}^2(0, T; F) \quad \underline{\text{weakly}},}$$

$$v_h, \rightarrow v \quad \underline{\text{in } \mathbb{L}^2(Q) \quad \underline{\text{strongly}}.}$$

5.7.2 The Convergence Theorems.

The convergence theorems are stated differently according to the dimension of the space (n = 2 or 3) and to the scheme considered.

We recall that we associated with the elements  $u_h^m$  a function  $u_h$ 

$$u_h: [0,T] \longrightarrow V_h,$$

defined slightly differently for the four schemes (see (5.36), (5.55), (5.75)) (1)

(5.93) for  $(m-1)k \le t \le mk$   $(m = 1, \dots, N)$ ,

 $u_{h}(t) = \begin{cases} u_{h}^{m} & (\text{Schemes 5.1 and 5.3}) \\ \frac{1}{2} (u_{h}^{m} + u_{h}^{m-1}) & (\text{Scheme 5.2}) \\ u_{h}^{m-1} & (\text{Scheme 5.4}). \end{cases}$ 

(1) We emphasize that u<sub>h</sub> depends on h and k; only for reasons of simplicity have we denoted this function by u<sub>h</sub> instead of u<sub>hk</sub>. We have:

Theorem 5.4.

The dimension of the space is n = 2 and the assumptions are (5.1) to (5.7), (5.9), (5.10), (5.41), and (5.89) to (5.92). We denote by u the unique solution of Problem 3.1.

The following convergence results hold, as h and  $k \rightarrow 0$ ,

(5.94)  $u_{h} \longrightarrow u \quad in \quad \mathbb{L}^{2}(Q) \quad strongly, \quad L^{\infty}(0,T;\mathbb{L}^{2}(\Omega)) \quad weak-star,$ 

(5.95)  $p_h u_h \rightarrow \overline{\omega} u_h \underline{in} L^2(0,T;F) \underline{weakly},$ 

provided:

(i) Scheme 5.1: no condition,

(ii) <u>Scheme</u> 5.2,

 $(5.96) kS<sup>2</sup>(h) \longrightarrow 0,$ 

(iii) Scheme 5.3: (5.43) is satisfied,

(iv) <u>Scheme</u> 5.4: (5.56)-(5.57) <u>are satisfied</u>.

Remark 5.3.

(i) For Schemes 5.1 and 5.2 it can be proved, without any further hypotheses, that

(5.97)  $p_{\mu}u_{\mu} \rightarrow \overline{\omega}u$  in  $L^{2}(0,T;F)$  strongly, as  $h,k \rightarrow 0$ .

(ii) The same results hold for the other schemes provided we assume moreover that

(5.98)  $kS_1^2(h) \longrightarrow 0$  and  $kS^2(h) \longrightarrow 0$  (Schemes 5.3 and 5.4).

(iii) The hypothesis (5.96) used in the proof of the convergence of Scheme 5.2 is probably unnecessary since the scheme is unconditionally  $L^2(0,T;F)$  and  $L^{\infty}(0,T;L^2(\Omega))$  stable.

Theorem 5.5.

The dimension of the space is n = 3 and, otherwise, the assumptions are the same as in Theorem 5.4.

Then, there exists some sequence  $h', k' \rightarrow 0$ , such that

(5.99)  $u_h \rightarrow u \quad \underline{in} \quad \mathbb{L}^2(Q) \quad \underline{strongly},$ 

(5.100)  $u_h, \rightarrow u \quad \underline{in} \quad L^{\infty}(0,T; \mathbb{L}^2(\Omega)) \quad \underline{weak-star},$ 

(5.101)  $p_h, u_h, \rightarrow \overline{u}u \quad \underline{in} \quad L^2(0,T;F) \quad \underline{weakly},$ 

where u is some solution of Problem 3.1.

For any other sequence h',  $k' \rightarrow 0$ , such that the convergences (5.99) to (5.101) hold, u must be some solution of Problem 3.1.

Remark 5.4.

We are not able to prove that the whole sequence converges due to lack of uniqueness of solution for Problem 3.1.

We also cannot prove strong convergence in  $L^2(0,T;F)$  due to lack of an energy equality for the exact problem (Problem 3.1) (for n = 3 we only have an energy inequality; see Remark 4.1).

The two theorems are proved in the remainder of this Section 5.7; we will prove Theorem 5.4 with full details for Scheme 5.1 (including (5.97)) and in the other cases we will only sketch the proofs which are actually very similar.

5.7.3 Proof of Theorem 5.4 (Scheme 5.1).

According to the stability theorem (Theorem 5.1), and to (5.89), there exists a sequence  $h',k' \rightarrow 0$ , such that

(5.102)  $u_h' \xrightarrow{\longrightarrow} u$  in  $L^{\infty}(0,T;L^2(\Omega))$  weak-star,  $p_h'u_h' \xrightarrow{\longrightarrow} \omega u$  in  $L^2(0,T;F)$  weakly,

for some u in  $L^2(0,T;V) \cap L^{\infty}(0,T;H)$ .

Let us consider the piecewise linear function  $w_h$  introduced in Section 5.6 (see (5.78)). By Lemma 5.6 and the estimations on the  $u_h^m$ , we have

$$\begin{split} \left\| \mathbf{w}_{h} \right\|_{L^{\infty}(0,T;\mathbf{L}^{2}(\Omega))} &\leq \text{Const.}, \\ \left\| \mathbf{p}_{h} \mathbf{w}_{h} \right\|_{L^{2}(0,T;F)} &\leq \text{Const.}, \\ \\ \int_{-\infty}^{+\infty} |\tau|^{2\gamma} |\hat{\mathbf{w}}_{h}(\tau)|^{2} d\tau &\leq \text{Const.}. \end{split}$$

Hence, according to (5.92), the sequence  $h',k' \rightarrow 0$  can be chosen so that

(5.103) 
$$w_h, \longrightarrow w$$
 in  $L^{2}(0,T; L^{2}(\Omega))$  weak-star,  
 $w_h, \longrightarrow w$  in  $L^{2}(0,T; L^{2}(\Omega))$  strongly,  
 $p_h, w_h, \longrightarrow w$  in  $L^{2}(0,T;F)$  weakly,

where  $w \in L^2(0,T;V) \cap L^{\infty}(0,T;H)$ .

We now observe that:

1

Lemma 5.8.

(5.104)  $u_h \rightarrow 0$  in  $L^2(0,T;\mathbb{L}^2(\Omega))$  strongly as h and  $k \rightarrow 0$ .

Thus

$$(5.105)$$
 w = u

and

(5.106)  $u_h, \rightarrow u \quad \underline{in} \quad L^2(0,T; \mathbb{L}^2(\Omega)) \quad \underline{strongly},$ 

as h' and  $k' \rightarrow 0$ .

Proof.

Exactly as in Lemma 4.8, we check that

(5.107) 
$$|u_{h}-w_{h}|_{L^{2}(0,T;\mathbb{L}^{2}(\Omega))} = \sqrt{\frac{k}{3}} \left(\sum_{m=1}^{N} |u_{h}^{m}-u_{h}^{m-1}|^{2}\right)^{\frac{1}{2}}.$$

Then (5.104) follows from the majoration (5.22); (5.105), (5.106) are obvious consequences of (5.102), (5.103), and (5.104).

The next point is to prove that u is a solution of Problem 3.1.

Lemma 5.9.

The function u appearing in (5.102), (5.103), (5.105) is a solution of Problem 3.1.

Proof.

We easily interpret (5.12) in the following way:

(5.108) 
$$\frac{d}{dt} (w_h(t), v_h) + v((u_h(t), v_h))_h + b_h(u_h(t-k), u_h(t), v_h)$$
  
=  $(f_k(t), v_h), \forall t \in [0, 1], \forall v_h \in v_h,$ 

where

(5.109)  $f_k(t) = f^m, (m-1)k \le t \le mk.$ 

Let v be any element in  $\mathcal{V}$  and let us take  $v_h = r_h v$  in (5.108). Let  $\psi$  be a continuously differentiable scalar function on [0,T], with

(5.110)  $\psi(T) = 0.$ 

We multiply (5.108) (where  $v_h = r_h v$ ) by  $\psi(t)$ , integrate in t, and integrate the first term by parts to get:

$$(5.111) - \int_{0}^{T} (w_{h}(t), \psi'(t)r_{h}v)dt + v \int_{0}^{T} ((u_{h}(t), \psi(t)r_{h}v))_{h}dt + \int_{0}^{T} b_{h}(u_{h}(t-k), u_{h}(t), \psi(t)r_{h}v)dt = (u_{h}^{0}, r_{h}v)\psi(0) + \int_{0}^{T} (f_{k}(t), \psi(t)r_{h}v)dt.$$

We now pass to the limit in (5.111) with the sequence  $h',k' \rightarrow 0$  using essentially (5.90), (5.91), (5.102), (5.103), and Lemma 5.8; we recall also that

(5.112)  $r_h v \rightarrow \overline{\omega} v$  in F (strongly),

5.113) 
$$u_h^0 \longrightarrow u_0$$
 in  $\mathbb{L}^2(\Omega)$  (strongly) (1),

(5.114) 
$$f_k \longrightarrow f$$
 in  $L^2(0,T; \mathbb{L}^2(\Omega))$  (see Lemma 4.9).

We find in the limit

(

$$-\int_{0}^{T} (u(t),\psi'(t)v)dt + v \int_{0}^{T} ((u(t),\psi(t)v))dt + \int_{0}^{T} b(u(t),u(t),\psi v)dt$$
(5.115)
$$= (u_{0},v)\psi(0) + \int_{0}^{T} (f(t),v)\psi(t)dt.$$

We infer from this equality that u is a solution of Problem 3.1, exactly as we did in the proof of Theorem 3.1 after (3.43).

Since the solution of Problem 3.1 is unique (see Theorem 3.2), a contradiction argument that we have already used very often shows that

(5.116) The convergences (5.102), (5.103) hold for

the whole family  $h, k \rightarrow 0$ .

This completes the proof of Theorem 5.4.

5.7.4 Proof of (5.97).

For the sake of completeness we will also prove (5.97). In order to prove this point we need a preliminary result which is quite general and interesting by itself.

3

<sup>(1)</sup>We recall the proof of (5.113); due to (5.11) it suffices to prove this for  $u_{n} \in \mathcal{V}$  and in this case

 $|\mathbf{u}_{h}^{0}-\mathbf{u}_{0}| \leq |\mathbf{r}_{h}\mathbf{u}_{0}-\mathbf{u}_{0}| \leq \|\mathbf{p}_{h}\mathbf{r}_{h}\mathbf{u}_{0}-\overline{\omega}\mathbf{u}_{0}\| \xrightarrow{F} 0.$ 

Let  $\{(V_h, p_h, r_h)_{\mathcal{H}}, (\overline{\omega}, F)\}$  be a stable and convergent external approximation of V. For a given element v of  $L^2(0,T;V)$ , one can define for each  $h \in \mathcal{H}$ a function  $v_h^+ \in L^2(0,T;V_h)$  such that

$$p_h v_h^+ \rightarrow \overline{\omega} v \quad \underline{in} \quad L^2(0,T;F) \quad \underline{as} \quad h \rightarrow 0.$$

# Proof.

The proof is essentially the same as that of Proposition I.3.1.

The result is obvious if v is a step function; since the step functions are dense in  $L^2(0,T;V)$ , the result follows in the general case by an argument of double passage to the limit as in Proposition I.3.1.

Lemma 5.11.

The dimension of the space is n = 2; then for Scheme 5.1

(5.117)  $p_h u_h \longrightarrow \overline{\omega} u \quad in \quad L^2(0,T;F) \quad (strongly),$ 

as h and  $k \rightarrow 0$ .

Proof.

We consider the expression

$$X_{h} = \left| u_{h}^{N} - u(T) \right|^{2} + \sum_{m=1}^{N} \left| u_{h}^{m} - u_{h}^{m-1} \right|^{2} + 2\nu \int_{0}^{T} \left\| u_{h}(t) - u_{h}^{+}(t) \right\|_{h}^{2} dt,$$

with  $u_h^+$  defined as in Lemma 5.10.

According to (5.21) (Lemma 5.1),

$$|u_{h}^{N}| \leq Const.;$$

hence there exists a sequence  $h', k' \rightarrow 0$ , with

$$u_h^N \longrightarrow \chi$$
 in  $\mathbb{L}^2(\Omega)$  weakly.

We temporarily assume that

(5.119) 
$$(\chi-u(T),v) = 0, \forall v \in H.$$

We then prove that

$$x_h \to 0$$
.

Actually, we consider

$$x_{h} = x_{h}^{1} + x_{h}^{2} + x_{h}^{3},$$

where

$$X_{h}^{1} = |u(T)|^{2} + 2\nu \int_{0}^{T} ||u_{h}^{+}(t)||_{h}^{2} dt \rightarrow |u(T)|^{2} + 2\nu \int_{0}^{T} ||u(t)||^{2} dt$$

(by Lemma 5.10 and (5.90)),

$$x_{h}^{2} = -2(u_{h}^{N}, u(T)) - 4\nu \int_{0}^{T} ((u_{h}(t), u_{h}^{+}(t))_{h} \rightarrow -2|u(T)|^{2} - 4\nu \int_{0}^{T} ||u(t)||^{2} dt$$

\_

(by Lemma 5.10, (5.90) and (5.118)–(5.119), we recall that  $u(T) \in H$ ), and

(5.120)  
$$X_{h}^{3} = |u_{h}^{N}|^{2} + \sum_{m=1}^{N} |u_{h}^{m} - u_{h}^{m-1}|^{2} + 2\nu \int_{0}^{T} ||u_{h}(t)||_{h}^{2} dt$$
$$= |u_{h}^{N}|^{2} + \sum_{m=1}^{N} ||u_{h}^{m} - u_{h}^{m-1}|^{2} + 2k \sum_{m=1}^{N} ||u_{h}^{m}||_{h}^{2}.$$

---

By summation of the equalities (5.26) for  $m = 1, \dots, N$ , we get

(5.121) 
$$X_{h}^{3} = |u_{h}^{0}|^{2} + 2k \sum_{m=1}^{N} (f^{m}, u_{h}^{m})$$
$$= |u_{h}^{0}|^{2} + 2 \int_{0}^{T} (f_{k}(t), u_{h}(t) dt.$$

It is then clear that

$$X_h^3 \longrightarrow |u_0|^2 + 2 \int_0^T (f(t), u(t)) dt$$
, as  $h, k \longrightarrow 0$ .

Hence

(5.122) 
$$X_{h}^{T} \rightarrow |u_{0}^{2}|^{2} + 2 \int_{0}^{T} (f(t), u(t)) dt - |u(T)|^{2} - 2\nu \int_{0}^{T} ||u(t)||^{2} dt,$$

and this limit is 0 due to (4.55).

By a contradiction argument we show as well that the whole family  $\overset{X}{\overset{}_{h}}$  converges to 0:

$$x_{h} \rightarrow 0$$
, as  $h, k \rightarrow 0$ .

In particular

$$\int_0^T \|u_h(t) - u_h^+(t)\|_h^2 dt \longrightarrow 0,$$

and

$$\int_{0}^{T} \| \mathbf{p}_{h} \mathbf{u}_{h}(t) - \overline{\omega} \mathbf{u}(t) \|_{F}^{2} dt \leq c \{ \int_{0}^{T} \| \mathbf{u}_{h}(t) - \mathbf{u}_{h}^{+}(t) \|_{h}^{2} dt + \int_{0}^{T} \| \mathbf{p}_{h} \mathbf{u}_{h}^{+}(t) - \overline{\omega} \mathbf{u}(t) \|_{F}^{2} dt \} \longrightarrow 0$$

and (5.117) follows.

- It remains to prove (5.119).
- By summation of (5.12) for  $m = 1, \dots, N$ , we get

$$(u_{h}^{N}-u_{h}^{0},v_{h}) + kv \sum_{m=1}^{N} ((u_{h}^{m},v_{h}))_{h} + k \sum_{m=1}^{N} b_{h}(u_{h}^{m-1},u_{h}^{m},v_{h}) = k \sum_{m=1}^{N} (f^{m},v_{h}).$$

Taking  $v_h = r_h v$ ,  $v \in \mathcal{V}$ , we easily pass to the limit and get

$$(\chi-u_0,v)+\nu\int_0^T ((u(t),v))dt+\int_0^T b(u(t),u(t),v)dt=\int_0^T (f(t),v)dt, \quad \forall v \in \mathcal{V}.$$

But since we deduce by integration of (3.13) a similar equation with  $\chi$  replaced by u(T), we conclude that

$$(\chi-u(T),v) = 0, \forall v \in \mathcal{V},$$

which implies (5.119) by density.

The proof of Lemma 5.11 is complete.

5.7.5 Proof of Theorems 5.4 and 5.5 (other cases).

For Schemes 5.2, 5.3, 5.4 and in the case n = 2, the proof is very similar to the above one, using the corresponding a priori estimates.

For Scheme 5.2, we introduced the condition (5.96) as a sufficient condition to prove (5.104); more precisely, in this case, the analog of (5.104) is a consequence of (5.77), (5.107) and (5.96). For Schemes 5.3, 5.4, the stability conditions (5.43), (5.56), (5.57) merely ensure that  $u_h$  and  $p_h u_h$  remain bounded in the suitable spaces.

For the proof of (5.97), the condition (5.98) appears as follows: - For Scheme 5.4 the "natural" expression similar to  $X_h$  in Lemma 5.11 is

$$Y_{h} = |u_{h}^{N} - u(T)|^{2} - \sum_{m=1}^{N} |u_{h}^{m} - u_{h}^{m-1}|^{2} + 2\nu \int_{0}^{T} ||u_{h}(t) - u_{h}^{+}(t)||_{h}^{2} dt;$$

in order to deduce (5.97) from the fact that  $Y_{h} \rightarrow 0$ , it suffices that

$$\sum_{m=1}^{N} |u_h^m - u_h^{m-1}|^2 \longrightarrow 0,$$

and this is a consequence of (5.98).

- For Scheme 5.3, we consider the same expression  $X_h$ , but due to some terms involving  $b_h$  (see (5.47)), the expression  $X_h^3$  is not as simple as (5.121); (5.98) shows that the supplementary terms of  $X_h^3$  involving  $b_h$  converge to 0.

If n = 3, we observe first that the stability conditions are not explicitly mentioned in the statement of Theorem 5.5. Actually it is with the help of these conditions, and the a priori estimates that they imply, that we prove the existence of a subsequence  $h',k' \rightarrow 0$ , such that (5.99) - (5.101) hold; that u is a solution of Problem 3.1 is proved exactly as before. COMMENTS AND BIBLIOGRAPHY

# Chapter I

Section 1 contains a preliminary study of the basic spaces V and H; the trace theorem is proved by the methods of JL. Lions-E. Magenes, see ref. [1]. The characterization of  $H^{\perp}$  was given by O.A. Ladyzhenskaya [1] for n = 3; the proof presented here is slightly simplified and extended to all dimensions.

We have not given any systematic study nor review concerning the Sobolev spaces. We restrict ourselves to recalling properties of these spaces when needed (Sections 1.1 of Chapters I and II in particular). As mentioned in the text, the reader is referred for proofs and further material to J.L. Lions [1], J.L. Lions and E. Magenes [1], J. Nečas [1], L. Sobolev [1], among other references.

The variational formulation of Stokes equations was first introduced (in the general frame of the nonlinear evolution case) by J. Leray [1], [2], [3], for the study of weak or turbulent solutions of the Navier-Stokes equations. The existence of a solution of the Stokes variational problem is easily obtained by the Classical Projection Theorem, whose proof is recalled for the sake of completeness. The study of the nonvariational Stokes problem, and the regularity of solutions is based on the paper of Cattabriga [1] (if n = 3) and on the paper of Agmon-Douglis-Nirenberg [1] on elliptic systems (any dimension); the results are recalled without proofs.

The concept of approximation of a normed space and variational problem was studied in particular by J.P. Aubin [1], J. Céa [1]; the presentation followed here is that of R. Temam [8]. The discrete Poincaré Inequality (Section 3.3) and the approximation of V by finite differences are in J. Céa [1]. The approximation of V by conforming finite elements was first studied and used by M. Fortin [2]; our description of the approximations (APX 2), (APX 3), (APX 4) (conforming finite elements), follows essentially M. Fortin [2]. In this reference one can also find many results of computations using this type of discretization; see also J.P. Thomasset [1], Borsenberger [1]. The material related to the nonconforming finite elements for the approximation of divergence free vector functions is due to Crouzeix, R. Glowinsky, P.A. Raviart, and the author. Other aspects of the subject (nonconforming finite elements of higher degree and more refined error estimates) can be found in Crouzeix and P.A. Raviart [1]; for numerical experiments, see Borsenberger [1]. For other applications of finite elements in fluid mechanics, see J.E. Hirsh [1]. Concerning the general theory of finite elements, let us mention the synthesis works of I. Babushka and A.K. Aziz [1], P.A. Raviart [2], G. Strang [1], and the proceedings edited by I. Babushka [1]. For more references on finite elements (in general situations) the reader is referred to the bibliography of these works. The description of finite element methods given here is almost completely self-contained; we only assume a few specific results whose proofs would necessitate an introduction of tools quite remote from our previous work.

After discretization of the Stokes problem, we have to solve a finite dimensional linear problem where the unknown is an element  $u_h$  of a finite dimensional space  $V_h$ . There are then two possibilities:

a) either the space  $V_h$  possesses a natural and simple basis, such that the problem reduces to a linear system with a sparse matrix for the components of  $u_h$  in this basis; in this case we solve the problem by resolution of this linear system.

b) or, if not, the finite dimensional problem is not so simple to solve (ill-conditioned or non-sparse matrix), even if it possesses a unique solution. In this case, appropriate algorithms must be introduced in order to solve these problems; this is the purpose of Section 5.

The algorithms described in Section 5 were introduced in the frame of optimization theory and economics in Arrow-Hurwicz-Uzawa [1]; the application of these procedures to problems of hydrodynamics is studied in J. Céa and R. Glowinsky [1], M. Fortin [2], M. Fortin, R. Peyret, and R. Temam [1]. See in D. Bégis [1], M. Fortin [2], an experimental investigation of the optimal choice of the parameter  $\rho$  (or  $\rho$  and  $\alpha$ ); a theoretical resolution of this problem in a very particular case is given in Crouzeix [2].

The approximation of incompressible fluids by slightly compressible fluids, as in Section 6, has been studied by J.L. Lions [4] and R. Temam [2]. The full asymptotic development of  $u_{\varepsilon}$  given here is new; for further investigations of this point see M.C. Pélissier [1].

# Chapter II

Section 1 develops a few standard results concerning existence and uniqueness of solutions of the nonlinear stationary Navier-Stokes equations. We follow essentially 0.A. Ladyzhenskaya [1] and J.L. Lions [2]. A more complete discussion of the regularity of solutions and the theory of hydrodynamical potentials can be found in 0.A. Ladyzhenskaya [1]; for regularity, see also H. Fujita [1]. The stationary Navier-Stokes equations in an unbounded domain has been studied by R. Finn [1]-[5]; R. Finn and D.R. Smith [1], [2].

Section 2 gives discrete Sobolev inequalities and compactness theorems. The principle of the proofs in the case of finite differences parallels the corresponding proofs in the continuous case (see, for instance, J.L. Lions [1], J.L. Lions-E. Magenes [1]). The proof of discrete Sobolev inequalities has never been published before, the proof of the discrete compactness theorem can be found in P.A. Raviart [1]. For conforming finite elements the proofs are much simpler: in particular, for the discrete compactness theorem, the problem is reduced by a simple device to the continuous case; these results are new.

The discussion of the discretization of the stationary Navier-Stokes equations follows the principles developed in Chapter I. The general convergence theorem is similar to that of Chapter I and the same type of discretizations of V are considered; the differences lie in the lack of uniqueness of solutions of the exact problem. The numerical algorithms of Subsection 3.3 has been introduced and tested in Fortin-Peyret-Temam [1].

We did not establish here nonuniqueness results for the stationary Navier-Stokes equations <sup>(1)</sup>. The nonuniqueness of stationary solutions of the Navier-Stokes and related equations has been proved with the aid of bifurcation theory and topological degree theory. The main results in this direction are due to P.H. Rabinowitz [2] and W. Velte [1], [2]. In [2], W. Velte proves the nonuniqueness of solutions of the Taylor problem and the situation is very similar to the problems for which existence is proved in Section 1, although not identical. For other applications of bifurcation theory see in particular, J.B. Keller and S. Antman [1], P.H. Rabinowitz [1], [3].

(1) This proof will be included in R. Temam [9].

## Chapter III

The existence and uniqueness results for the linearized Navier-Stokes equations (Section 1) are a special case of general results of existence and uniqueness of solutions of linear variational equations (see for instance, J.L. Lions-E. Magenes [1], vol. 2). In order to be self-contained, we have given an elementary proof of some technical results which usually are established as easy consequences of deeper results {Lemma 1.1 which is more natural in the frame of vector valued distribution theory (L. Schwartz [2]) or Lemma 1.2 which can be proved by interpolation methods (J.L. Lions-E. Magenes [1])}.

Theorem 2.1 is one of the standard compactness theorems used in the theory of nonlinear evolution equations. Other compactness theorems are proved and used in J.L. Lions [2].

The existence and uniqueness results related to the nonlinear Navier-Stokes equations and given in Sections 3 and 4 are now classical and are in the line of the early works of J. Leray [1], [2], [3]; see E. Höpf [1], [2], O.A. Ladyzhenskaya [1], J.L. Lions [2], [3], J.L. Lions and G. Prodi [1], Serrin [3]. Further results on the regularity of solutions and the study of existence of classically differentiable solutions of the Navier-Stokes equations can be found in the 2<sup>nd</sup> edition of O.A. Ladyzhenskaya [1]. For the analyticity of solutions, see C. Foias and G. Prodi [1], H. Fujita and K. Masuda [1], C. Kahane [1], K. Masuda [1], J. Serrin [3].

Let us mention also two recent and completely different approaches to the existence and uniqueness theory that we did not treat here. The first one is that of E.B. Fabes, B.F. Jones, and N.M. Riviere [1] based on singular integral operator methods and giving existence and uniqueness results in  $L^p$  spaces. The other one is the method of D.G. Ebin and J. Marsden [1] connecting the Navier-Stokes initial value problem with the geodesics of a Riemannian manifold and thus using the methods of global analysis.

The material of Section 5 containing a discussion of the stability and convergence of simple discretization schemes for the Navier-Stokes equations is essentially new; a similar study for different equations or different schemes was done in R. Temam [2], [3], [4]. Stability and convergence of some unconditionally stable one step schemes are given in O.A. Ladyzhenskaya [5]. A more complete treatment of the numerical analysis of the Navier-Stokes equations is included in R. Temam [9] (in particular, the practical application of the schemes in Section 5, the fractional step or projection method, and the artificial compressibility method). Among many other references on the subject, see A.J. Chorin [1], [2], [3], C.K. Chu and G. Johansson [1], C.K. Chu, K.W. Morton, and K.V. Roberts [1], M. Fortin, R. Peyret, and R. Temam [1], M. Fortin [1], M. Fortin and R. Temem [1], O.A. Ladyzhenskaya and V.I. Rivkind [1], G. Marshall [1], [2], C.S. Peskin [1], [2].

#### REFERENCES

# S. AGMON, A. DOUGLIS, L. NIRENBERG [1] Estimates near the boundary for solutions of elliptic partial differential equations satisfying general boundary conditions I, Comm. Pure Appl. Math., 12, 1959, p. 623-727. [2] Estimates near the boundary for solutions of elliptic partial differential equations satisfying general boundary conditions II, Comm. Pure Appl. Math., 17, 1964, p. 35-92. B. ALDER and S. FERBACH [1] Editors, Methods in Computational Physics, Academic Press, New York, 1965. S. ANTMAN See J.B. KELLER and S. ANTMAN. A. ARAKAWA [1] D.G. ARONSON [1] Regularity property of flows through porous media, SIAM J. Appl. Math., 17, 1969, p. 461-467. [2] Regularity properties of flows through porous media: counterexample, SIAM J. Appl. Math., 19, 1970, p. 299-307. [3] <u>Regularity properties of flows through porous media:</u> interface, Arch. Rat. Mech. Anal., <u>37</u>, 1970, p. 1-10. the K. ARROW, L. HURWICZ, H. UZAWA [1] Studies in nonlinear programming, Stanford University Press, 1968. J.P. AUBIN

[1] <u>Approximation of Elliptic Boundary Value Problems</u>, Wiley Interscience, New York 1972.

#### A.K. AZIZ

See I. BABUSHKA and A.K. AZIZ.

I. BABUSHKA

[1] Editor, Proceedings of the Symposium on Mathematical Foundations of the Finite Element Method with Applications to Partial Differential Equations, University of Maryland, Baltimore County (June 1972), To appear.

I. BABUSHKA and A.K. AZIZ

[1] Lectures on Mathematical Foundations of the Finite Element Method, Technical Report B.N. 748, Institute for Fluid Dynamics and Applied Mathematics, University of Maryland, 1972. I. BABUSHKA and M. ZLAMAL

[1] <u>Nonconforming Elements in the Finite Element Method</u>, Technical Note B.N. 729, Institute for Fluid Dynamics and Applied Mathematics, University of Maryland, 1972.

C. BARDOS

[1] <u>Existence et unicité de la solution de l'équation d'Euler</u> <u>en dimension deux</u>, J. Math. Anal. Appl., <u>40</u>, 1972, p. 769-790.

C. BARDOS and L. TARTAR [1] About backward uniqueness for evolution equations and some related questions, To appear.

D. BEGIS

[1] <u>Analyse numérique de l'écoulement d'un fluide de Bingham</u>, Thèse de 3<sup>ème</sup> cycle, Université de Paris, 1972.

O.M. BELOTSERKOVSKII

[1] Editor, Proceedings of the First International Conference on Numerical Methods in Fluid Dynamics, Novossibirsk, August 1969; Publication of the Academy of Science of U.R.S.S., 1970.

BORSENBERGER

[1] Thèse de 3<sup>ème</sup> cycle, Université de Paris-Sud, à paraître.

J.P. BOUJOT, J.L. SOULE, and R. TEMAM [1]<u>Traitement Numérique d'un Problème de Magnetohydrodynamique</u>, in M. HOLT [2].

K. BRYAN

[1] <u>A numerical investigation of a nonlinear model of a wind</u> <u>driven ocean</u>, J. Atmos. Sci. <u>20</u>, 1963, p. 594-606.
[2] <u>A numerical method for the study of circulation of the world</u> <u>ocean</u>, J. of Comp. Phys., <u>4</u>, 1969, p. 347-376.
[3] Editor, <u>Proceedings of a Conference on Oceanography</u>, held in

BUIANTON

1972. To appear.

[1] Fractional derivatives of solutions of nonlinear evolution equations in Banach spaces. To appear.

H. CABANNES and R. TEMAM [1] Editors, Proceedings of the Third International Conference on <u>Numerical Methods in Fluid Dynamics</u>, Paris, July 1972, Lecture Notes in Physics,\_\_\_\_, Springer-Verlag, 1973.

L. CATTABRIGA

[1] <u>Su un problema al contorno relativo al sistema di equazioni di Stokes</u>, Rend. Mat. Sem. Univ. Padova, <u>31</u>, 1961, p. 308-340.

- J. CEA
- [1] <u>Approximation variationelle des problèmes aux limites</u>, Ann. Inst. Fourier, <u>14</u>, 1964, p. 345-444.
- [2] Optimisation, Théorie et Algorithme Dunod, Paris, 1971.
- L. CEA and R. GLOWINSKY [1] <u>Méthode numérique pour l'écoulement laminaire d'un fluide</u> rigide visco-plastique, Rapport IRIA, Mars 1971.
- A.J. CHORIN
- [1] <u>A numerical method for solving incompressible viscous flow</u> problems, J. of Comp. Phys., <u>2</u>, 1967, p. 12-26.
- [2] <u>Numerical solution of the Navier-Stokes equations</u>, Math. Comp., 23, 1968, p. 341-354.
- [3] <u>Numerical solution of incompressible flow problems</u>, Studies in Num. Anal., <u>2</u>, 1968, p. 64-71.
- [4] Computational aspects of the turbulence problem, in M. HOLT [2].
- C.K. CHU and G. JOHANSSON
  [1] <u>Numerical studies of the heat conduction with highly</u> anisotropic tensor conductivity II. To appear.
- C.K. CHU and H.O. KREISS [1] Computational Fluid Dynamics, a book, to appear.
- C.K. CHU, K.W. MORTON and K.V. ROBERTS [1] <u>Numerical studies of the heat conduction equation with highly</u> anisotropic tensor conductivity; in CABANNES-TEMAM [1].
- P.G. CIARLET and P.A. RAVIART
  [1] <u>General Lagrange and Hermite interpolation in R<sup>n</sup> with</u>
  <u>applications to finite elements</u>, Arch.Rational Mech. Anal., <u>46</u>
  (No. 3), 1972, p. 177-199.

[2] Interpolation theory over curved elements, with applications to finite element methods, Computer Methods in Applied Mechanics and Engineering, <u>1</u>, 1972, p. 217-249.

[3] <u>The Combined Effect of Curved Boundaries and Numerical</u> Integration of Isoparametric Finite Element Methods, in I. BABUSHKA [1].

[4] Theory of Finite Element Methods, a book, to appear.

P.G. CIARLET and G. WAGSHALL

[1] <u>Multipoint Taylor formulas and applications to the finite</u> <u>element method</u>, Num. Math., <u>17</u>, 1971, p. 84-100.

#### COLLINS

[1] <u>Application de la mécanique des milieux continus à la</u> <u>Biomécanique</u>, Cours de 3<sup>ême</sup> cycle, Université de Paris VI, 1972.

CROUZEIX

[1]

CROUZEIX and P.A. RAVIART [1] To appear.

#### S.C.R. DENNIS

- [1] The numerical solution of the vorticity transport equation, Report CERN, Geneva, 1972.
- J. DENY and J.L. LIONS
  [1] Les espaces du type de BEPPO LEVI, Ann. Inst. Fourier, 5, 1954,
  p. 305-370.
- A. DOUGLIS

See S. AGMON, A. DOUGLIS and L. NIRENBERG.

- D. DUVAUT and J.L. LIONS [1] <u>Les inéquations en Mécanique et en Physique</u>, Dunod, Paris, 1971, English translation. To appear.
- D.G. EBIN and J. MARSDEN [1] <u>Groups of diffeomorphisms and the motion of an incompressible fluid</u>, Annals of Math., <u>92</u>, 1970, p. 102-163.

[2] <u>Convergence of Navier-Stokes fluid flow as viscosity goes to zero</u>. Initial boundary value problem. To appear.

S. FERNBACH

See B. ALDER and S. FERNBACH.

R. FINN

[1] On the exterior stationary problem for the Navier-Stokes equations and associated perturbation problems, Arch. Rational Mech. Anal., <u>19</u>, 1965, p. 363-406.

[2] On the steady state solution of the Navier-Stokes equations III, Acta Math., 105, 1961, p. 197-244.

[3] Estimates at infinity for stationary solutions of the Navier-Stokes equations, Bull. Math. Soc. Sci. Math. Phys. R.P. Roumanie, <u>3</u> (51), 1959, p. 387-418.

[4] On steady state solutions of the Navier-Stokes partial differential equations, Arch. Rational Mech. Anal., 3, 1959, p. 381-396.

[5] <u>Stationary solutions of the Navier-Stokes equations</u>, Proc. Symp. Appl. Math., <u>17</u>, 1965, p. 121-153.

R. FINN and D.R. SMITH

[1] On the linearized hydrodynamical equations in two dimensions, Arch. Rational Mech. Anal., 25, 1967, p.1-25.

[2] On the stationary solution of the Navier-Stokes equations in two dimensions, Arch. Rational Mech. Anal, <u>25</u>, 1967, p. 26-39.

E.B. FABES, B.F. JONES, and N. RIVIERE [1] <u>The initial boundary value problem for the Navier-Stokes equations with</u> <u>data in L<sup>P</sup></u>, Arch. Rational Mech. Anal., <u>45</u>, 1972, p. 222-240.

G. FIX

[1] To appear.

C. FOIAS

 Essais dans l'étude des solutions des équations de Navier-Stokes dans l'espace - L'unicité et la presque périodicité des solutions petites, Rend. Sem. Mat. Padova XXXII, 1962, p. 261-294.

[2] <u>Solutions statistiques des équations d'évolution non linéaires</u>, Lecture at the C.I.M.E. on non-linear problems, Varenna 1970, Ed. Gremoneze Publishers.

C. FOIAS and G. PRODI

[1] <u>Sur le comportement global des solutions non stationnaires des</u> équations de Navier-Stokes en dimension 2, Rend. Sem. Mat. Padova <u>XXXIX</u>, 1967, p. 1-34.

[2] Solutions statiques des équations de Navier-Stokes. To appear.

M. FORTIN

[1] <u>Approximation d'un opérateur de projection et application à un</u> schéma de résolution numérique des équations de Navier-Stokes, Thèse de 3 cycle, Université de Paris XI, 1970.

[2] <u>Calcul numérique des écoulements des fluides de Bingham et des fluides</u> newtoniens incompressibles par la méthode des éléments finis, Thèse, Université de Paris, 1972.

M. FORTIN and R. TEMAM

[1] <u>Numerical Approximation of Navier-Stokes Equations</u>, in O.M. BELOTSERKOVSKII [1].

M. FORTIN, R. PEYRET and R. TEMAM

[1] <u>Résolution numérique des equations de Navier-Stokes pour un fluide</u> <u>incompressible</u>, J. de Mécanique, 10, No. 3, 1971, p. 357-390, and an announcement of results in M. HOLT [2].

J.E. FROMM

[1] The time dependent flow of an incompressible viscous fluid, Methods of Comp. Phys. <u>3</u>, Academic Press, New York, 1964.

[2] <u>A method for computing nonsteady incompressible viscous fluid flows</u>, Los Alamos Scient. Lab. Report LA 2910, Los Alamos, 1963.

[3] <u>Numerical solutions of the nonlinear equations for heater fluid layer</u>, Phys. of Fluids, 8, 1965, p. 1757.

[4] <u>Numerical solution of the Navier-Stokes equations at high Reynolds</u> numbers and the problem of discretization of convective derivatives. Lecture in J.J. Smolderen [1].

H. FUJITA

[1] On the existence and regularity of the steady-state solutions of the Navier-Stokes Equations, J. Fac. Sci. Univ. Tokyo, Section 1, 9, 1961, p. 59-102.

H. FUJITA and T. KATO

[1] <u>On the Navier-Stokes initial value problem I</u>, Arch. Rational Mech. Anal., <u>16</u>, 1964, p. 269-315.

See also T. KATO and H. FUJITA.

- H. FUJITA and K. MASUDA [1] To appear.
- H. FUJITA and N. SAUER
  - [1] <u>Construction of weak solutions of the Navier-Stokes equation in a</u> <u>noncylindrical domain</u>, Bull. Amer. Math. Soc., <u>75</u>, 1969, p. 465-468.
- E. GAGLIARDO
  - [1] <u>Propriéta di al classi di funzioni in più variabili</u>, Ricerche di Mat., 7, 1958, p. 102-137.
- R. GLOWINSKY

See J. CEA and R. GLOWINSKY.

R. GLOWINSKY, J.L. LIONS, and R. TREMOLIERES [1] <u>Approximation des Inéquations de la Mécanique et de la Physique</u>, Dunod, Paris, 1973.

#### S.K. GODUNOV

[1] <u>Solution of one dimensional problems of gas dynamics with variable</u> meshes, NAUKA, Moscow, 1970.

## D. GREENSPAN

[1] <u>Numerical studies of prototype cavity flow problems</u>, Comp. Sci. Techn. Rept. No. 37, University of Wisconsin, July 1968.

See M. ISRAELI and H.P. GREENSPAN.

F.H. HARLOW and J.E. WELCH

[1] <u>Numerical calculation of time dependent viscous incompressible flow of fluid with a free surface</u>, Phys. Fluids, <u>8</u>, 1965, p. 2182-2189.

## J.E. HIRSH

[1] The finite element method applied to ocean circulation problems. To appear.

## M. HOLT

[1] Editor, <u>Basic Developments in Fluid Dynamics</u>, Vol. I, Academic Press, New York, 1965.

[2] Editor, Proceedings of the Second International Conference on Numerical Methods in Fluid Dynamics, Berkeley, Sept. 1970, Lecture Notes in Physics, 8, Springer-Verlag, 1971.

[3] <u>La Résolution Numérique de Quelques Problèmes de Dynamique des Fluide</u>s, Lecture Notes No. 25, Université of Paris-Sud, Orsay, France, 1972.

E. HOPF

 <u>Uber die Aufangswertaufgabe für die hydrodynamischen Grundgleichungen</u>, Math. Nachr., <u>4</u> (1951), p. 213-231.

[2] <u>On nonlinear partial differential equations</u>, Lecture Series of the Symposium on Partial Differential Equations, Berkeley, 1955. Ed. The Univ. of Kansas (1957), p. 1-29.

L. HURWICZ See K. ARROW, L. HURWICZ, and H. UZAWA.
M. ISRAELI and H.P. GREENSPAN

Nonlinear motions of a confined rotating fluid, Report, Department of Meteorology, M.I.T. To appear.

G. JOHANSSON

See C.K. CHU and G. JOHANSSON.

**B.F.** JONES

See E.B. FABES, B.F. JONES, and N. RIVIERE.

C. KAHANE

[1] <u>On the spatial analyticity of solutions of the Navier-Stokes equations</u>, Arch. Rational Mech. Anal., <u>33</u>, 1969, p. 386-405.

S. KANIEL and M. SHINBROT

[1] <u>Smcothness of weak solutions of the Navier-Stokes equations</u>, Arch. Rational Mech. Anal., <u>24</u>, 1967, p. 302-324.

[2] <u>A reproductive property of the Navier-Stokes equations</u>, Arch. Rational Mech. Anal., <u>24</u>, 1967, p. 363-369.

See also M. SHINBROT and S. KANIEL.

T. KATO

[1] <u>On classical solutions of two dimensional nonstationary Euler equation</u>, Arch. Rational Mech. Anal. <u>25</u>, 1967, p. 188-200.

[2] <u>Nonstationary flows of viscous and ideal fluids in R<sup>3</sup></u>, J. Functional Analysis, <u>9</u>, 1972, p. 296-305.

- T. KATO and H. FUJITA
  - [1] <u>On the stationary Navier-Stokes system</u>, Rend. Sem. Univ. Padova, <u>32</u>, 1962, p. 243-260.
- J.B. KELLER and S. ANTMAN

[1] Editors, <u>Bifurcation Theory and Nonlinear Eigenvalue Problems</u>, Benjamin, New York, 1969.

## H.O. KREISS

See C.K. CHU and H.O. KREISS.

A. KRZWICKI and O.A. LADYZHENSKAYA

[1] <u>A grid method for the Navier-Stokes equations</u>, Soviet Physics Dokl.,<u>11</u>, 1966, p. 212-216.

O.A. LADYZHENSKAYA

[1] <u>The Mathematical Theory of Viscous Incompressible Flow</u>, Gordon and Breach, New York, English translation, Second Edition, 1969.

[2] <u>Sur de nouvelles équations dans la dynamique des fluides visqueux et</u> <u>leur résolution</u>, Globale-Troudi Math. Inst. Stekloff, CII, 1967, p. 85-104 (in Russian).

[3] <u>Sur des modifications des équations de Navier-Stokes pour de grands</u> gradients de vitesses, Sem. Inst. Stekloff, Leningrad, <u>7</u>, 1968, p. 126-154 (in Russian). O.A. LADYZHENSKAYA (cont.)

[4] <u>Sur la solution globale unique du problème de Cauchy tridimensionnel</u> pour les équations de Navier-Stokes à symétrie axiale, Sem. Inst. Stekloff, Leningrad, <u>7</u>, 1968, p. 155-177 (in Russian).

[5] On convergent finite differences schemes for initial boundary value problems for Navier-Stokes equations, Fluid Dyn. Trans., 5, 1969, p. 125-134.

O.A. LADYZHENSKAYA See A. KRZYWICKI and O.A. LADYZHENSKAYA.

O.A. LADYZHENSKAYA and V.I. RIVKIND

[1] On the alternating direction method for the computation of a viscous incompressible fluid flow in cylindrical coordinates, Izv. Akad. Nauk, <u>35</u>, 1971, p. 259-268.

## P. LASCAUX

[1] <u>Application de la méthode des éléments finis en hydrodynamique</u> <u>bidimensionnelle utilisant les variables de Lagrange</u>, Technical Report, C.E.A. Limeil, 1972.

## P. LAX

J. LERAY

[1] <u>Etude de diverses équations intégrales nonlinéaires et de quelques</u> problèmes que pose l'hydrodynamique, J. Math. Pures Appl., <u>12</u>, 1933, p. 1-82.

[2] <u>Essai sur les mouvements plans d'un liquide visqueux que limitent des parois</u>, J. Math. Pures et Appl., <u>13</u>, 1934, p. 331-418.

[3] Sur le mouvement d'un liquide visqueux emplissant l'espace, Acta Math.
 63, 1934, p. 193-248.

J.L. LIONS

[1] <u>Problèmes aux limites dans les équations aux dérivées partielles</u>, Presses de l'Université de Montréal, 1962.

[2]

[3] <u>Quelques résultats d'existence dans les equations aux dérivées partielles</u> non <u>linéaires</u>, Bull. Soc. Math. France, <u>87</u>, 1959, p. 245-273.

[4] <u>On the numerical approximation of some equations arising in hydrodynamics</u>, A.M.S. Symposium, Durham, April 1968.

[5] <u>Sur la régularité et l'unicité des solutions turbulentes des équations</u> <u>de Navier-Stokes</u>, Rend. Sem. Mat. Padova, <u>30</u>, 1960, p. 16-23.

J.L. LIONS

See G. DUVAT and J.L. LIONS; J. DENY and J.L. LIONS; R. GLOWINSKY, J.L. LIONS and R. TREMOLIERES.

J.L. LIONS and E. MAGENES

[1] <u>Nonhomogeneous boundary value problems and applications</u>, Springer-Verlag, Berlin, New York, 1972.

- J.L. LIONS and G. PRODI
  - [1] <u>Un théorème d'existence et d'unicité dans les équations de Navier-Stokes</u> en dimension 2, C.R. Acad. Sci. Paris, <u>248</u>, 1959, p. 3519-3521.
- E. MAGENES and G. STAMPACCHIA
  - [1] <u>I Problemi al contorno per le equazioni differenziali di tipo ellittico</u>, Ann. Scuola. Norm. Sup. Pisa, <u>12</u>, 1958, p. 247-357, note (27), p. 320.
- G.I. MARCHUK

[1] In the Proceedings of the International Congress of Mathematicians, Nice, 1970, Gauthier-Villars, Paris, 1971.

J. MARSDEN

See D.G. EBIN and J. MARSDEN.

G. MARSHALL

[1] On the numerical treatment of viscous flow problems, Report NA-7, Delft, 1972.

[2] Numerical treatment of air pollution problems. To appear.

K. MASUDA

[1] On the analyticity and the unique continuation theorem for solutions of the Navier-Stokes equations, Proc. Japan Acad., <u>43</u> (No. 9), 1967, p. 827-832.

K. MASUDA

See H. FUJITA and K. MASUDA.

G. MEURANT

[1] <u>Quelques aspects théoriques et numériques de problèmes de valeurs</u> propres non linéaires, Thèse de 3<sup>eme</sup> cycle, Université de Paris-Sud, 1972.

G. MORETTI

[1] Lectures in J.J. SMOLDEREN [1].

K.W. MORTON

See C.K. CHU, K.W. MORTON, and K.V. ROBERTS; R.D. RICHTMYER and K.W. MORTON

J. NECAS

[1] <u>Les méthodes directes en théorie des équations elliptiques</u>, Masson, Paris, 1967.

L. NIRENBERG

See S. AGMON, A. DOUGLIS, and L. NIRENBERG.

- W.F. NOH
  - [1]
- M.C. PELISSIER [1] To appear.
- C.S. PESKIN [1] Flow patterns around heart values, J. of Comp. Phys.

[2]

R. PEYRET

See M. FORTIN, R. PEYRET and R. TEMAM.

G. PRODI

[1] <u>Un teorema di unicita per le equazioni di Navier-Stokes</u>, Annali di Mat., <u>48</u>, 1959, p. 173-182.

[2] <u>Qualche risultato riguardo alle equazioni di Navier-Stokes nel caso</u> bidimensionale, Rend. Sem. Mat. Padova, <u>30</u>, 1960, p. 1-15.

G. PRODT

See J.L. LIONS and G. PRODI; C. FOIAS and G. PRODI.

G. PROUSE

[1] <u>Soluzioni periodiche dell'equazione di Navier-Stokes</u>, Rend. Acad. Naz. Lincei, XXXV, 1963, p. 443-447.

[2] <u>Soluzioni quazi-periodiche dell'equazione di Navier-Stokes in due</u> <u>dimensioni</u>, Rend. Sem. Mat. Padova, <u>33</u>, 1963.

#### P.H. RABINOWITZ

[1] Periodic solutions of nonlinear hyperbolic partial differential equations (I) and (II), Comm. Pure Appl. Math., 20, 1967, p. 145-205, and 22, 1969, p. 25-39.

[2] <u>Existence and nonuniqueness of rectangular solutions of the Benard</u> problem, Arch. Rational Mech. Anal., <u>29</u>, 1968, p. 32-57.

[3] <u>Some aspects of nonlinear eigenvalue problems</u>, M.R.C. Technical Summary Report No. 1193, University of Wisconsin, 1972.

#### M.A. RAUPP

[1] <u>Galerkin mathods for two-dimensional unsteady flows of an ideal</u> incompressible fluid, Thesis, University of Chicago, 1971.

P.A. RAVIART

 [1] Sur l'approximation de certaines équations d'évolution linéaires et non linéaires, J. Math. Pures Appl., <u>46</u>, 1967, p. 11-107 et 109-183.
 [2] <u>Méthode des Elements Finis</u>, Cours de 3<sup>ème</sup> cycle, Université de Paris VI, 1972.

# P.A. RAVIART

See P. CIARLET and P.A. RAVIART; CROUZEIX and P.A. RAVIART.

G. de RHAM

[1] Variétés Différentiables, Hermann, Paris, 1960.

R.D. RICHTMYER and K.W. MORTON

[1] <u>Difference Methods for Initial Value Problems</u>, Wiley-Interscience, New York, 1957.

M. RIESZ

[1] <u>Sur les ensembles compacts de fonctions sommables</u>, Acta. Sci. Math. Szeged, <u>6</u>, 1933, p. 136-142.

#### N. RIVIERE

See E.B. FABES, B.F. JONES, and N. RIVIERE.

V.I. RIVKIND

See O.A. LADYZHENSKAYA and V.I. RIVKIND.

K.V. ROBERTS

See C.K. CHU, K.W. MORTON, and K.V. ROBERTS.

E. SANCHEZ-PALENCIA

[1] Existence des solutions de certains problèmes aux limites en magnétohydrodynamique, J. Mécanique, 7, 1968, p. 405-426.

[2] <u>Quelques résultats d'existence et d'unicité pour les écoulements</u> <u>magnétohydrodynamiques non stationnaires</u>, J. Mécanique, <u>8</u>, 1969, p. 509-541.

J. SATHER

[1] The initial boundary value problem for the Navier-Stokes equations in regions with moving boundaries, Thesis, University of Minnesota, 1963.

N. SAUER

See H. FUJITA and N. SAUER.\*

J.P. SAUSSAIS

[1] <u>Etude d'un problème de diffusion non linéaire lié à la physique des</u> plasmas, Thèse de 3<sup>eme</sup> cycle, Université de Paris-Sud, 1972.

L. SCHWARTZ

[1] Théorie des Distributions, Hermann, Paris, 1957.

[2] <u>Théorie des distributions à valeurs vectorielles</u>, <u>I</u>, <u>II</u>, Ann. Inst. Fourier, <u>7</u>, 1957, p. 1-139 and <u>8</u>, 1958, p. 1-209.

J. SERRIN

[1] <u>A note on the existence of periodic solutions of the Navier-Stokes</u> equations, Arch. Rational Mech. Anal., <u>3</u>, 1959, p. 120-122.

[2] <u>Mathematical Principles of Classical Fluid Dynamics</u>, Encyclopedia of Physics, vol. 13, 1, Springer-Verlag, 1959.

[3] <u>The initial value problem for the Navier-Stokes equations</u>, in "Nonlinear Problems," R.E. Langer editor, University of Wisconsin Press, 1963, p. 69-98.

[4] On the interior regularity of weak solutions of Navier-Stokes equations, Arch. Rational Mech. Anal., 9, 1962, p. 187-195.

M. SHINBROT and S. KANIEL

[1] <u>The initial value problem for the Navier-Stokes equations</u>, Arch. Rational Mech. Anal., <u>21</u>, 1966, p. 270-285.

D.R. SMITH

See R. FINN and D.R. SMITH.

J.J. SMOLDEREN

[1] Numerical Methods in Fluid Dynamics, AGARD Lecture Series, No. 48, 1972.

S.L. SOBOLEV

[1] <u>Applications de l'analyse fonctionnelle aux équations de la physique</u> <u>mathématique</u>, Leningrad, 1950, in Russian, and English translation by A.M.S., <u>19</u>.

# G. STAMPACCHIA

See E. MAGENES and G. STAMPACCHIA.

G. STRANG

[1] An Analysis of the Finite Elements Method, a book, to appear.

P. SZEPTYCKI

[1] The equations of Euler and Navier-Stokes on compact Riemannian manifolds, Technical Report, University of Kansas, 1972.

L. TARTAR

See C. BARDOS and L. TARTAR.

R. TEMAM

[1] <u>Sur la stabilité et la convergence de la méthode des pas fractionnaires</u>, Ann. Mat. Pura Appl., <u>LXXIV</u>, 1968, p. 191-380.

[2] <u>Une méthode d'approximation de la solution des équations de Navier-Stokes</u>, Bull. Soc. Math. France, 98, 1968, p. 115-152.

 [3] Sur l'approximation de la solution des équations de Navier-Stokes par la méthode des pas fractionnaires (I), Arch. Rational Mech. Anal., <u>32</u> (No. 2), 1969, p. 135-153.

[4] <u>Sur l'approximation des equations de Navier-Stokes par la méthode des pas</u> <u>fractionnaires (II)</u>, Arch. Rational Mech. Anal., <u>33</u> (No. 5), 1969, p. 377-385.

[5] <u>Approximation of Navier-Stokes Equations</u>, AGARD Lecture Series, No. <u>48</u>, 1970.

[6] <u>Sur la résolution exacte et approchée d'un problème hyperbolique non linéaire de T. CARLEMAN</u>, Arch. Rational Mech. Anal., <u>35</u> (No. 5), 1969, p. 351-362.

[7] <u>Méthodes de Décomposition en Analyse Numérique</u>, Proceedings of the International Conference of Mathematicians, Nice, 1970, Gauthier-Villars, Paris, 1971.
[8] <u>Numerical Analysis</u>, Reidel Publishing Company, Holland, 1973 (in English), and P.U.F. Paris, 1969 (in French).

[9] On the theory and numerical analysis of the Navier-Stokes equations. To appear.

R. TEMAM

See M. FORTIN and R. TEMAM; M. FORTIN, R. PEYRET, and R. TEMAM; J.P. BOUJOT, J.L. SOULE, and R. TEMAM; H. CABANNES and R. TEMAM.

J.P. THOMASSET

[1] Thèse de 3<sup>ème</sup> cycle Université de Paris-Sud, 1973.

R. TREMOLIERES

See R. GLOWINSKY, J.L. LIONS, and R. TREMOLIERES.

H. UZAWA

See K. ARROW, L. HURWICZ, and H. UZAWA.

W. VELTE

 [1] <u>Stabilitätsverhalten und Verzweigung Stationärer Lösungen der Navier-</u> <u>Stokesschen Gleichungen</u>, Arch. Rational Mech. Anal., <u>16</u>, 1964, p. 97-125.
 [2] <u>Stabilitäts und Verzweigung stationärer Lösungen der Navier-</u> <u>Stokesschen Gleichungen beim Taylor problem</u>, Arch. Rational Mech. Anal., <u>22</u>, 1966, p. 1-14.

### C. WAGSHALL

See P.G. CIARLET and C. WAGSHALL.

#### J.E. WELCHS

See F.H. HARLOW and J.E. WELCHS.

### WILKINS

[1] <u>Calculation of elastic plastic flow</u>, in Meth. of Comp. Phys., <u>3</u>, Academic Press, New York, 1964.

[2]

# N.N. YANENKO

[1] Fractional Step Methods, English translation, Springer-Verlag, 1971.

[2] In the <u>Proceedings of the International Conference of Mathematicians</u>, Nice, 1970, Gauthier-Villars, Paris, 1971.

# V.I. YUDOVICH

[1] <u>Nonstationnary flow of a perfect nonviscous fluid</u>, (in Russian), J. of Num. Math. and Math. Phys., <u>3</u>, 1963.

[2] <u>Periodic solutions of viscous incompressible fluids</u>, (in Russian), Dokl. Akad. Nauk, <u>130</u>, 1960, p. 1214-1217.

[3] <u>A two dimensional problem of unsteady flow of an ideal incompressible fluid across a given domain</u>, Amer. Math. Soc. Translation, <u>57</u>, 1966, p. 277-304.

## A. ZENISEK

[1] <u>Polynomial interpolations on the triangle</u>, Numer. Math., <u>15</u>, 1970, p. 283-296.

## M. ZLAMAL

[1] <u>On the finite element method</u>, Numer. Math., <u>12</u>, 1968, p. 394-409.
[2] <u>A finite element procedure of the second order of accuracy</u>, Numer. Math., <u>14</u>, 1970, p. 394-402.

#### M. ZLAMAL

See I. BABUSHKA and M. ZLAMAL.